MARGINAL COST PRICING FOR GAS DISTRIBUTION UTILITIES: PRELIMINARY ANALYSES AND MODELS

prepared by

Jean-Michel Guldmann

THE NATIONAL REGULATORY RESEARCH INSTITUTE 2130 Neil Avenue Columbus, Ohio 43210

November 1980

80-12

FOREWORD

This report was prepared by The National Regulatory Research Institute (NRRI) under Grant No. DE-FG-01-80RG10268 from the U.S. Department of Energy (DOE), Economic Regulatory Administration, Division of Regulatory Assistance. The opinions expressed herein are solely those of the authors and do not reflect the opinions nor the policies of either the NRRI or the DOE.

The NRRI is making this report available to those concerned with state utility regulatory issues since the subject matter presented here is believed to be of timely interest to regulatory agencies and to others concerned with utility regulation.

> Douglas N. Jones Director

EXECUTIVE SUMMARY

The effects of marginal cost pricing on the demand for natural gas and on changes in the capital and operating costs of gas distribution utilities are important issues in the Public Utility Regulatory Policies Act of 1978. However, the analysis of these effects has been, up to now, considerably inhibited because of lack of relevant data and methods on which to base the calculation of these marginal costs, in particular the marginal capacity costs. It is the purpose of this study to provide data and methods for the calculation of these costs and for the evaluation of the impacts of marginal cost pricing policies. These methods combine the use of econometric techniques and optimization/simulation algorithms.

Econometric models of distribution plant costs have been developed using community-level data for four U.S. gas distribution utilities: Long Island Lighting Company, Columbia Gas of Ohio, Inc., Pacific Gas and Electric Company, and National Fuel Gas Distribution Corporation. These cost models can be used for predicting future costs as well as for calculating marginal distribution capacity costs. Some major commonalities emerge from the comparison of the different models. Probably the most important one is the nonseparability of the distribution plant costs incurred to serve the different sectoral markets of the utility. Such a result is not surprising in view of the complex and nonseparable linkages that exist among the different customers served by the same pipeline The second most important commonality is related to the economies network. of scale achieved with respect to both residential and nonresidential gas sales. The two previous results imply that the sectoral sales marginal costs are (1) decreasing with the sector's size, and (2) depending upon the size of the other sector(s). Third, the population density variable turns out to be generally significant. Finally, the Pacific Gas and Electric Company analysis has demonstrated the usefulness of accounting for weather parameters when the utility's service territory is climatologically heterogeneous.

The exact calculation of the marginal supply, storage, and transmission costs implies the development of a complex gas network optimization model. In view of the problems involved in solving a complete network model, a simplified, aggregate, and nonspatialized model has been developed to calculate these marginal costs. This model, cast into a linear programming format, yields time-linked (monthly) marginal costs. In addition, it has been embedded into a larger simulation model designed to evaluate all the implications of marginal cost pricing under alternative assumptions (maximum supplies, demand elasticities, etc.). This general model has been applied to the East Ohio Gas Company. The major results of the optimization/simulation analysis are that (1) marginal costs highly depend upon supply conditions (maximum availability, charges, contracts, etc,) and upon various technological constraints; (2) peak-shifting problems are very likely to occur if distribution capacity marginal costs are wholly assigned to the peak period (month); (3) the excess revenue problem does not necessarily always occur, and its occurrence depends upon supply conditions, costs, technological constraints, financial parameters, and the price elasticities of the monthly demands. Although it would be highly premature to draw final conclusions from this partial analysis, it should be noted that the results do not clearly point out the superiority of a marginal cost pricing policy.

ACKNOWLEDGEMENTS

The author would like to express his sincerest gratitude to his research assistants, Mr. Kyubang Lee and Ms. Veena Khanna. Mr. Lee has been primarily responsible for carrying out the statistical computations reported in chapter 3 and has often contributed very useful and perceptive comments. Ms. Khanna has been very helpful in bibliographical searches as well as in data preparation and analysis.

The author would also like to express his sincerest thanks to the following persons whose help and collaboration have been essential to the completion of the study:

Mr. John P. Zekoll, Director - Gas Division, New York Public Service Commission;

Mr. Walter J. Cavagnaro, Energy Policy Consultant, California Public Utilities Commission;

Mr. Richard Hare, Jr., Vice-President, and Mr. Howard T. Rose, Manager -State Regulation, National Fuel Gas Distribution Corporation;

Mr. Alan W. Beringsmith, Coordinator, Corporate Planning Department, and Mr. Thomas C. Long, Supervisor - Gas Results of Operations, Pacific Gas and Electric Company;

as well as all the staff members of National Fuel Gas Corporation and Pacific Gas and Electric Company who prepared the data used in this study.

TABLE OF CONTENTS

Chapter

Page

| 1 | INTRODUCTION 1 |
|----------|---|
| 2 | NATURAL GAS DISTRIBUTION UTILITIES COSTS AND PRICING: GENERAL CONSIDERATIONS |
| | The Theoretical Rationale for Utility Marginal Cost Pricing |
| | Introductory Considerations |
| | Calculating Gas Distribution Marginal Costs: A Conceptual Approach |
| | Practical Approach |
| 3 | ECONOMETRIC MODELING OF DISTRIBUTION PLANT COSTS 15 |
| | General Considerations |
| | Extensions |
| 4 | UTILITY COST MINIMIZATION AND MARGINAL COST PRICING EVALUATION - APPLICATION TO THE EAST OHIO GAS COMPANY 89 |
| | Structure of the Gas Utility Marginal Cost Pricing Model (GUMCPM) |
| | Pricing Model (GUMCPM) |
| 5 | SUMMARY |
| Appendix | |
| А | LONG ISLAND LIGHT COMPANY DATA |
| В | COLUMBIA GAS OF OHIO COMPANY DATA |

| С | PACIFIC GAS AND ELECTRIC COMPANY DATA | 9 |
|---|---|---|
| D | NATIONAL FUEL GAS DISTRIBUTION CORPORATION DATA | 9 |
| E | COMPUTER PROGRAM OF THE GUMCP MODEL | 1 |
| F | SAMPLE OUTPUT OF THE GUMCP MODEL | 3 |

٦,

LIST OF FIGURES

Page

Figure

| 2.1 | A Hypothetical Gas Distribution System |
|-----|--|
| 3.1 | National Fuel Gas Company System 60 |
| 4.1 | Structure of the Gas Utility Marginal Cost Pricing Model (GUMCPM) |
| 4.2 | Maximum Gas Deliveries to Storage |
| 4.3 | Maximum Gas Withdrawals from Storage |
| 4.4 | Diagrammatic Representation of the EOGC System 108 |
| 4.5 | EOGC System Approximation |
| 4.6 | Typical Demand Curve and Consumers' Surplus |

LIST OF TABLES

Page

Table

| 3.1 | Neighborhood Prototypes Characteristics | 18 |
|------|---|----|
| 3.2 | Capital Costs of Providing Gas to Thousand Housing Units in Six Neighborhoods | 19 |
| 3.3 | Value of Gas Plant in Service at the End of 1978 and 1979 - LILCO | 25 |
| 3.4 | Volume of Gas Sales and Number of Customers in in 1978 and 1979 - LILCO | 26 |
| 3.5 | Definitions, Means, and Standard Deviations of the Variables - LILCO Static Analysis | 28 |
| 3.6 | Market Dynamics in LILCO Communities during the Period 1978–1979 | 34 |
| 3.7 | Definitions and Mean Values of the Dynamic Anlaysis Variables - LILCO | 34 |
| 3.8 | Gas Plant in Service at the End of 1976 and 1977 - CGOC (In Dollars) | 38 |
| 3.9 | Volume of Gas Sales and Number of Customers in 1976 and 1977 - CGOC | 39 |
| 3.10 | Means and Standard Deviations of the Variables - CGOC Static Analysis | 41 |
| 3.11 | Volume of Gas Sales and Number of Customers in 1978 and 1979 - PG&E | 46 |
| 3.12 | Value of Gas Plant in Service at the End of 1978 and 1979 — PG&E (In Dollars) | 47 |
| 3.13 | Definitions, Means, and Standard Deviations of the Variables - PG&E Static Analysis | 49 |

xi

| 3.14 | Definitions, Means, and Standard Deviations of the Dynamic Analysis Variables - PG&E |
|------|---|
| 3.15 | Volume of Gas Sales and Number of Customers in 1979 - NFGDC and Its New York Division 61 |
| 3.16 | Value of Gas Plant in Service at the End of 1979 - NFGDC and Its New York Division (In Dollars) 62 |
| 3.17 | Definitions and Means of the Variables for the Two Samples S ₁ and S ₂ - NFGDC Static Analysis |
| 3.18 | Comparison of Model - Calculated and Average Plant Values for the Average NFGDC Community (In Dollars) 85 |
| 3.19 | Summary of Marginal and Average Costs for the Average NFGDC Community (In Dollars) |
| 3.20 | Summary of Sales and Density Elasticities |
| 4.1 | Average Monthly Degree-Days and Sectoral and Total Loads (MMCF) |
| 4.2 | EOGC Own Gas Production |
| 4.3 | EOGC Distribution Plant |
| 4.4 | Sectoral Monthly Loads (MMCF) with Market Growth Rates Equal to 50% - Average Cost Pricing Policy |
| 4.5 | Costs Structure of the Optimum Solutions under Supply Cases S ₁ and S ₂ - Average Cost Pricing Policy (In Dollars) |
| 4.6 | Optimal Monthly Purchases from Consolidated and Panhandle and Storage Deliveries and Withdrawals (MMCF) - Average Cost Pricing Policy |
| 4.7 | Optimal Maximum Supplies from Consolidated and Panhandle, Wellhead and Field-Line Monthly Purchases, Incremental Production Capacity and Constant Monthly Production, Incremental Storage Capacity and Total Storage Deliveries, and Incremental Transmission Capacity - Average Cost Pricing Policy |
| 4.8 | Distribution Plant, Financial Variables, and Average Volumetric Rates - Average Cost Pricing Policy 128 |
| 4.9 | Evaluation Criteria for the Average Cost Pricing Policy 129 |
| 4.10 | Monthly Marginal Costs - Average Cost Pricing Policy Analysis (In Dollars) |

xii

| 4.11 | Price-Demand Patterns in Case $S_1 E_1 \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots $ |
|------|---|
| 4.12 | Price-Demand Patterns in Case $S_1 E_1 \cdots \cdots \cdots \cdots \cdots 132$ |
| 4.13 | Price-Demand Patterns in Case $S_2 E_1 \cdots \cdots$ |
| 4.14 | Price-Demand Patterns in Case $S_2 E_2 \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots $ |
| 4.15 | Solutions Characteristics and Evaluation in Case $S_1 E_1 \dots \dots$ |
| 4.16 | Solutions Characteristics and Evaluation in Case $S_1 E_2 \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots $ |
| 4.17 | Solutions Characteristics and Evaluation in Case $S_2 = E_1 \dots \dots$ |
| 4.18 | Solutions Characteristics and Evaluation in Case $S_2 E_2 \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots \cdots $ |
| 4.19 | Solutions Ranking |
| A.1 | Plant in Service and Gas Sales - Long Island Lighting Company |
| A.2 | Number of Customers, Population and Acreage - Long Island Lighting Company |
| B.1 | Gas Sales, Numbers of Customers, Net Plant in Service, Population and Acreage - Columbia Gas of Ohio Company 156 |
| C.1 | Average Number of Residential, Commercial, and Industrial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company |
| C•2 | Average Number of Residential Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company |
| C.3 | Average Number of Commercial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company |
| C.4 | Average Number of Industrial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company |
| C.5 | Total Residential, Commercial, and Industrial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company |

| C.6 | Total Residential Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company | 182 |
|------|---|--------------|
| C.7 | Total Commercial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company | 186 |
| C.8 | Total Industrial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company | 1 9 0 |
| C.9 | Population, Acreage, Distribution Plant, and Main Mileage - Pacific Gas and Electric Company | 194 |
| C.10 | Average Degree-Days for the Period 1941-1970 - Meteorological Stations in the Pacific Gas and Electric Company's Service Area | 198 |
| D.1 | Gas Sales and Average Numbers of Customers - 1979 - National Fuel Gas Distribution Corporation | 200 |
| D. 2 | Distribution Plant in Service, Total Plant in Service, Population and Land Area - 1979 - National Fuel Gas Distribution Corporation | 205 |

CHAPTER 1

INTRODUCTION

The theory and application of marginal cost pricing to electric utilities have been the subjects of much research and discussion during recent years, and following a tradition solidly established in Europe, various electricity marginal cost pricing experiments have been conducted in the U.S. There has been less discussion about applying marginal cost pricing principles to natural gas utilities, and such discussions have nearly always identified the relevant marginal cost as the commodity marginal cost;¹ that is, whenever an existing or new gas source is called upon to help fill the demand in a given system, the price for all gas sold in that system is set at the cost of this marginal supply. The marginal transmission and distribution capacity costs have often been dismissed as irrelevant because of an alleged excess capacity in those networks.

Nevertheless, it seems that marginal cost pricing for gas distribution utilities is slowly coming of age. For instance, the New York Public Service Commission issued on September 17, 1979, an opinion² stating that the marginal cost of gas is a relevant consideration in gas rate cases and requested explanations of calculations and estimates for the commodity and capacity marginal costs at different times, recognizing the effects of contract provisions with suppliers, storage costs, and plans for transmission,

²Opinion No. 79-19 - State of New York Public Service Commission.

¹R.A. Tybout, "Marginal Cost versus Rolled-in Pricing for Natural Gas," Public Utilities Fortnightly, March 31, 1977.

distribution, and storage. The commissioners also stated their awareness of the possibility that marginal cost based rates might provide excess revenues to the utility, and of the need to deal with this issue should it arise.

The effect of marginal cost pricing on the demand for natural gas and on changes in the capital and operating costs of gas distribution utilities is an important issue in the National Energy Act of 1978 (see PURPA: Section 306 - Gas Utility Rate Design Proposals). However, the analysis of these effects has been, up to now, considerably inhibited because of lack of relevant data on which to base the calculation of these marginal costs, in particular the marginal capacity costs. It is the purpose of this study to provide data and methods for the calculation of gas marginal costs, with a particular emphasis on capacity costs, and for the evaluation of the impacts of marginal cost pricing policies. The proposed methods are illustrated with data obtained from actual gas distribution utilities in the U.S. They combine the use of statistical/ econometric techniques and of optimization/simulation algorithms.

The remainder of the report is organized as follows: chapter 2 presents general considerations for the estimation of gas distribution utilities' marginal costs and outlines the approach selected in this study; chapter 3 describes the rationale for the econometric modeling of the distribution plant costs and the results obtained for four different distribution utilities; chapter 4 presents an optimization/simulation model designed to compute monthly marginal costs and to analyze the impacts of marginal cost rates in terms of economic efficiency, energy conservation, and utility revenue requirements. The applicability of this model is illustrated with data from the East Ohio Gas Company; chapter 5 concludes the study and outlines areas for further research.

CHAPTER 2

NATURAL GAS DISTRIBUTION UTILITIES COSTS AND PRICING: GENERAL CONSIDERATIONS

The purpose of this chapter is to analyze, in a general way, the problems involved in natural gas pricing at the distribution level, and to present the rationale for the methodology adopted in this study. In the first section, the theoretical underpinnings of marginal or peak-load pricing for public utilities are summarily presented. The next section reviews the principles of marginal cost pricing application to gas distribution. The third section describes the conceptually optimal approach to marginal costs calculation and the problems involved in its actual implementation. The final section outlines the practical methodology selected in this study.

The Theoretical Rationale for Utility Marginal Cost Pricing

The theory of marginal cost pricing and its application to public utility pricing have been discussed in numerous recent books and articles.³ Public utilities, in particular gas and electric distribution utilities, supply a commodity the demand for which is periodic and that is only partially, if at all, storable. What should then be the price charged to the users of this commodity?

To simplify the analysis, consider a commodity with two distinct demand periods: an off-peak period T_1 and a peak period T_2 , of durations

³See, for instance, the "Symposium on Peak Load Pricing", <u>The Bell</u> Journal of Economics 7, 1 (Spring 1976).

 τ_1 and τ_2 , respectively. Define the corresponding demands per unit of time for the commodity as Q_1 and Q_2 . These demands are charged at prices P_1 and P_2 , and the demand function $P_1(Q_1)$ and $P_2(Q_2)$ are assumed to be known. The operating costs for the utility per unit of commodity produced are C_1 and C_2 , and the unit capacity cost is noted as b_2 . The utility's capacity must be able to provide the peak demand Q_2 , and under the assumption that no reserve margins are necessary, this capacity is taken exactly equal to Q_2 . The total cost for the utility of producing (Q_1, Q_2) is

$$TC = \tau_1 C_1 Q_1 + \tau_2 C_2 Q_2 + b_2 Q_2$$
(2.1)

The net revenue for the utility - or the producer's surplus (PS) - is equal to the difference between gross sales revenue and costs, with

$$PS = \tau_1 P_1(Q_1)Q_1 + \tau_2 P_2(Q_2)Q_2 - TC$$
(2.2)

The net consumers' surplus (CS) is equal to the difference between their gross surplus and the cost of obtaining the commodity, with

$$CS = \tau_1 \int_{0}^{Q_1} P_1(Q) dQ + \tau_2 \int_{0}^{Q_2} P_2(Q) dQ - \tau_1 P_1(Q_1) Q_1 - \tau_2 P_2(Q_2) Q_2 \quad (2.3)$$

The total welfare function (W) for both the utility and its customers is the sum of the above defined producer's and consumers' surpluses, with

$$W = \tau_1 \int_{0}^{Q_1} P_1(Q) dQ + \tau_2 \int_{0}^{Q_2} P_2(Q) dQ - \tau_1 C_1 Q_1 - \tau_2 C_2 Q_2 - b_2 Q_2 \quad (2.4)$$

The above welfare W is a function of the commodity quantities $\ensuremath{\mathbb{Q}}_1$ and $\ensuremath{\mathbb{Q}}_2$ produced and consumed

$$W = W(Q_1, Q_2) \tag{2.5}$$

The optimal production/consumption situation is reached when W is maximized, i.e., when the partial derivatives of W with respect to Q_1 and Q_2 are equal to zero. Such conditions are restated as

$$\frac{\partial W}{\partial Q_1} = \tau_1 P_1(Q_1) - \tau_1 C_1 = 0$$
 (2.6)

$$\frac{\partial W}{\partial Q_2} = \tau_2 P_2(Q_2) - \tau_2 C_2 - b_2 = 0$$
(2.7)

or, after simplification

$$P_1(Q_1) = C_1$$
 (2.8)

$$P_2(Q_2) = C_2 + \frac{b_2}{\tau_2}$$
(2.9)

The interpretation of equations (2.8) and (2.9) is that

- the off-peak price should be set equal to the off-peak unit operating cost, which is also the marginal off-peak operating cost,
- (2) the peak price should be set equal to the sum of the marginal peak operating cost and of the marginal capacity cost.

In the above example, linear cost functions have been used for the sake of simplicity, and therefore average and marginal costs are equal. However, if nonlinear cost functions are used, then the results are valid only with the marginal costs, hence the "marginal cost pricing" term.

The above theoretical framework will be useful for understanding the optimization/simulation approach presented in chapter 4. However, it clearly fails to account for various important real-world features of public utilities. First, it is clear that no public utility is characterized by a homogeneous production capacity. Electricity can be produced by different types of generators (coal, nuclear, oil, gas) with different operating and capacity costs. A gas distribution utility can purchase its gas from many different suppliers with widely different prices and contractual requirements, as well as extract gas from the ground or manufacture it (propane plant). Also, storage may be technologically feasible. Second, the demand for such commodities as gas or electricity varies daily, weekly, and seasonally, and therefore the number of relevant demand periods is considerably larger than in the above example. Third,

this demand, even in a given period, is uncertain (it varies with weather and other random factors), and so is the supply because of equipment failures; therefore, the interactions between pricing and curtailment or rationing costs must be accounted for. Finally, it must be noted that in the above example, it was implicitly assumed that even with marginal cost pricing, the second period T_2 would remain the peak one. However, it is quite possible that the consumers, reacting to the new peak and off-peak prices, would shift their demand from the peak to the off-peak period, making the latter the new peak period. Then, the original prices would no longer be equal to the marginal costs corresponding to the new demand pattern. Of course, the magnitude of this shifting depends upon the own- and cross-price elasticity of the demands of the different periods. The analysis in chapter 4 will clearly demonstrate the importance of this shifting peak problem.

The above remarks do not negate the usefulness of marginal cost pricing principles but simply point out that their application is much more complicated than the simple prescriptions based on simple models. The purpose of the next section is to further the analysis of the applicability of marginal cost pricing principles in the case of gas distribution utilities.

Marginal Cost Pricing at the Gas Distribution Level: Introductory Considerations

The gas industry is made up of three major components: production, transmission, and distribution. Distributors may produce some of the gas they use, but generally they receive most of their gas from one or more interstate pipeline companies that in turn may purchase it from various producers or import it (Canada, Mexico, LNG). The relevant commodity costs, in the absence of any vertical integration of the gas industry, are, for the distributors, those they pay their suppliers. These costs are generally characterized by two-part rates: a commodity rate, related to the amount of gas actually purchased, and a demand rate, related to the contract demand, that is, the maximum daily deliveries that the supplier commits itself to deliver to the distributor. The demand rate provides for

payment of the capacity (pipeline, compressors, storage, etc.) that the supplier has to install to offer the required quality of service. Also, most of the long-term contracts between distributors and interstate pipeline companies involve take-or-pay clauses, that is, the distributor commits itself to purchase a minimum quantity of gas at the specified commodity rate or to pay for this minimum quantity if it has not been actually taken.

The importance of the above features is obviously related to the variability of gas requirements that highly depend upon weather. Gas requirements peak in the winter season (generally January) and are at a low point in the summer season (generally July and August). Of course, the magnitude of the seasonal swing depends upon the market mix of the distributor, i.e., the number and characteristics of its space-heating customers. One way to attenuate the impact of the requirements variability on the supply variability is for the distributor to install and operate a storage (generally underground) system or to rent the storage pools of other companies (very often its own suppliers), and to use peak-shaving SNG (synthetic natural gas) plants or other short-term peak supplies.

In addition to an eventual storage system, the gas distribution system is made of transmission and distribution lines that deliver gas to the ultimate users. Transmission lines, of larger diameters, convey gas at higher pressure from the takeoff points, where gas is purchased from the suppliers, to the load centers, generally communities and metropolitan areas, where gas is then injected into the local distribution networks. The capacity of the distribution system must be such that the firm requirements corresponding to the coldest weather experienced in the service territory (or peak-day requirements) can be met. This capacity is therefore going to be underutilized most of the time, and under marginal cost pricing principles, the marginal capacity costs should then be paid by those consumers responsible for the peak requirements. Of course, note that the required marginal capacity also depends upon the existing excess capacity of the system.

The previous discussion of supply, storage, and distribution capacity costs clearly indicates that marginal variations in gas requirements at different periods have highly different impacts on these costs, and therefore marginal cost time-variable rates are clearly justified. Also, it appears that supply, storage, and distribution capacity decisions are highly interrelated. For instance, the economic feasibility of storage depends upon the costs of storage and the demand charge of the supplier. If the latter is very high, then storage may become an attractive alternative for reducing demand costs. Thus, the relevant marginal costs are those corresponding to the least-cost trade-off among supply, storage, and distribution decisions. They also depend upon additional supplies availability as well as upon such constraints as maximum incremental storage capacity, SNG production capacity, etc., and upon the possibility for the distributor to renegotiate long-term contracts with particular suppliers.

In addition to their temporal variability, gas distribution marginal costs are also characterized by a spatial variability.⁴ Indeed, a gas distribution system is a spatialized system with complex technological interactions, and therefore increases in demands at different points of the network have different impacts in terms of the necessary additional capacity of the different pipeline links, storage pools, compressors, etc. To trace the impacts of increased gas requirements clearly implies the use of detailed gas distribution network models where the various flows are simulated and that account for the trade-offs between compressors size and pipeline diameters. The use of such models is discussed in the next section.

From the above discussion, it is clear that the approach envisioned here would encompass both short-term and long-term marginal costs. In

⁴The use of locational variations in rate design by distribution utilities has been minimal. However, the analysis in chapter 3 will demonstrate the importance of these locational variations.

other words, the capacity costs of distribution are not assumed to be sunk. A dynamic growing market is assumed to exist and to require capacity replacement and expansion.

Finally, it is necessary to conform revenues under marginal cost rates with the revenue requirements determined through the traditional rate base regulation. It has often been argued that setting prices equal to marginal costs would provide the utility revenues in excess of the authorized, regulated revenues. However, such a proposition has never been formally proven, and in fact, depends upon the specific characteristics of the utility, its suppliers, and its customers. These revenue considerations will be fully analyzed in chapter 4.

Calculating Gas Distribution Marginal Costs: A Conceptual Approach

Consider a hypothetical gas distribution utility as diagrammatically represented in figure 2.1.



Figure 2.1 A Hypothetical Cas Distribution System

The system presented in figure 2.1 represents both existing and potential (i.e., which may eventually be added) components. The end-use customers are grouped into load centers (communities, urban areas, etc.) in which the distribution lines (and the related equipment such as regulators, gas holders) are located. The pipeline links on figure 2.1 are therefore only the transmission ones that convey gas from the supply takeoff points to the seasonal storage fields and to the load centers. Assume that there are L load centers (l=1-L) and that the year can be subdivided into T periods (t=1-T), each characterized by given levels of gas requirements. The gas requirements of load center l during period t are then noted D_{lt} . The problem facing the utility planner is to determine the least-cost pattern of supply, operation, and capacity expansion decisions subject to various physical, technological, and other constraints, and to satisfying the gas requirements D_{lt} .

There are a large number of decision variables controlled by the uti _ty planner, such as the following

- the amounts of gas purchased from each supplier at each takeoff point during each period t
- the maximum daily deliverability from each supplier
- the amounts of gas conveyed in each selected pipe link during each period
- the diameters of these pipes
- the location and power of the compressors
- the storage fields' capacity and the corresponding periodic inflows and outflows, etc.

There are, of course, constraints bearing on the above variables, such as the following

- maximum available supplies
- maximum pipe and compressor capacities
- maximum storage capacities and deliverability, etc.

Assume that there are K decision variables $(k=1\rightarrow K)$ noted X_k and that the total cost associated with a given vector $\overline{X} = \{X_k\}$ is denoted $C(\overline{X})$. The constraints set is partitioned into two subsets:

- M_1 physical, technological, and resource availability constraints $(m=1 \rightarrow M_1)$
- M_2 (=LxT) constraints expressing the satisfaction of the requirements D $_{0+}$

The planning problem can then be expressed as

minimize
$$C(\bar{X})$$
 (2.10)

subject to the following constraints

$$F_{m}(\bar{X}) = 0 \qquad (m=1 \rightarrow M_{1})$$
 (2.11)

$$G_{lt}(\bar{X}) = D_{lt} \qquad (l=1 \rightarrow L; t=1 \rightarrow T)$$
(2.12)

The above model is a mathematical program that would turn out to be a linear program if the objective function $C(\overline{X})$ and the constraints were expressed linearly. In such a case, the marginal cost associated with a marginal variation of the requirements D_{lt} is exactly equal to the shadow price, or dual value, of the corresponding constraint (2.12). These shadow prices are a natural part of the solution of any linear program. Such an approach to the calculation of space-time marginal costs has been applied by Scherer⁵ in the case of electricity generation and distribution systems. When the system cannot be reduced to a linear format, a possible approach to the calculation of the marginal cost $MC(D_{lt})$ is to solve the above program while increasing the demand D_{lt} by an increment ΔD_{lt} and to compute the cost increment ΔC . The marginal cost is then approximated by

$$MC(D_{lt}) = \frac{\Delta C}{\Delta D_{lt}}$$
(2.13)

⁵C.R. Scherer, "Estimating Peak and Off-Peak Marginal Costs for an Electric Power System: An Ex Ante Approach," <u>The Bell Journal of Economics</u> 7, 2 (1976): 575-601.

Obviously, the above marginal cost would encompass supply, storage, and transmission marginal costs. However, providing for the increment ΔD_{lt} implies also additional distribution capacity costs within load center l. Conceptually, then, the internal structure of each load center should also be formalized as a network serving all the individual customers (residential, commercial, industrial), and the marginal distribution cost corresponding to the marginal variation of the demand of any customer should be computed through a procedure similar to the one discussed for the larger network. Through such a hierarchical analysis, the total marginal cost corresponding to any marginal variation in demand could be calculated.

What are the practical prospects for the previous approach? Various planning models have been developed for gas utilities, mostly at the ⁷ interstate transmission level,⁶ but also at the distribution level.⁷ The transmission models are all expressed as optimization models, whereas the distribution ones are cast into a simulation format. However, no model could be found that analyzes, comprehensively, the design and operation of a gas distribution network in an urban area (i.e., a load center). The review of the available literature shows that the design of a supply/storage/transmission optimization model is feasible, but that developing efficient solution algorithms may be quite difficult because of the highly nonlinear character of the model and the necessary inclusion

See, for instance: J.C. Heideman, "Optimal Development of a Natural Gas Transmission System," Society of Petroleum Engineers, SPE Preprint 3980, 1972; H.B. Martch and N.J. McCall, "Optimization of the Design and Operation of Natural Gas Pipeline Systems," SPE Preprint 4006, 1972; O. Flanigan, "Constrained Derivatives in Natural Gas Pipeline System Optimization," <u>Journal of Petroleum Technology</u> 24, 5 (1972); D.J. Fenton and J.H. Wilson, "Extending a Gas Pipeline Network," <u>Journal</u> of the Operational Research Society 29, 9 (1978).

⁷See, for instance: A.E. Yingling, D.L. Raphael, and G.E. Slater, "A Dynamic Linear Flow Model of a Gas Distribution System," SPE Preprint 4714, 1973; G.E. Slater, J.C. Erdle, D.L. Raphael, "Simulating the Operation of a Natural Gas Distribution System with Linear Flow Models," Journal of Canadian Petroleum Technology 16, 4 (1978).

of integer variables. The development of such models at the level of urban areas appears to be an even more difficult endeavor. The approach adopted in this study has been, therefore, to develop simplified models dealing with (a) supply, storage, and transmission costs on one side, and (b) distribution costs on the other side. The outline of this approach is presented in the next section.

Calculating Gas Distribution Marginal Costs: A Practical Approach

In view of the problems involved in calculating community-level distribution costs through a comprehensive network modeling approach, a statistical approach has been selected, wherein the actual distribution capacity costs of the various communities (or part of them) included in the utility's service territory are related to the size of their various submarkets, their population density, and their climatic characteristics, provided that the service territory is climatologically heterogeneous. The resulting econometric cost models can then be used to determine the marginal distribution plant costs incurred by a marginal increment of residential, commercial, or industrial demand (expressed in gas volume or number of customers). Clearly, these marginal costs represent average values for the whole community, encompassing higher or lower marginal costs for individual customers. This econometric approach is presented in chapter 3 and illustrated with the data provided by different gas distribution utilities in the U.S.

In view of the problems involved in solving a complete network model, a simplified, aggregate, and nonspatialized model has been developed to calculate the marginal supply, storage, and transmission costs. This model, cast into a linear programming format, yields time-linked marginal costs. In addition, it has been embedded into a larger simulation model designed to evaluate the implications of marginal cost pricing under alternative maximum supplies and demand elasticities assumptions. This general model - the Gas Utility Marginal Cost Pricing Model (GUMCPM) - has been applied to the East Ohio Gas Company, and the results of this application as well as the structure of the model are presented in chapter 4.

In order to establish a correspondence with the marginal capacity, energy, and customer costs customarily computed for electric utilities, the marginal costs computed in the above-mentioned approaches can be characterized as follows:

- The marginal distribution plant costs computed with the econometric models include both distribution and customer capacity costs
- (2) The marginal costs computed by the cost-minimization model include (a) energy (supply) costs, (b) capacity (production, storage, and transmission costs, and (c) operating (production and storage) costs, closely related to the energy costs. These marginal costs are complemented, in the simulation model, by the distribution capacity marginal costs and by the other operating marginal costs (transmission, distribution, customer, and administration)

The optimization/simulation approach demonstrates that the various marginal costs cannot be easily separated because of multiple and complex cost trade-offs taking place in a gas distribution system. The econometric approach emphasizes the impacts of market size and mix, and urban structure on local distribution marginal costs. These impacts are not considered in the current optimization/simulation approach because of its aggregated, nonspatialized character but could be so in an extended model.

CHAPTER 3

ECONOMETRIC MODELING OF DISTRIBUTION PLANT COSTS

The purpose of this chapter is to present the principles and results of an econometric analysis of distribution plant costs, based on communitylevel data obtained from different distribution utilities. The resulting distribution plant cost functions can then be used to predict future costs as well as used for the calculation of marginal costs. In the first section, the general characteristics of the distribution plant, a review of the available data, and the general structure of the econometric models are presented. The next section deals with an analysis of the results obtained for four particular companies: Long Island Lighting Company, Columbia Gas of Ohio, Inc., Pacific Gas and Electric Company, and National Fuel Gas Distribution Corporation. The last section consists of a comparative analysis and synthesis of the results and outlines possible extensions of the approach,

General Considerations

Gas Distribution Utilities Plant Structure

The capital equipment of gas distribution utilities is generally classified according to the following categories:

- (1) the intangible plant, generally very small, and including such items as "organization" and franchises and consents
- (2) the production plant, including both manufactured gas production plant and natural gas production and gathering plant
- (3) the natural gas storage plant, including both underground storage plant and other storage equipments, such as holders
- (4) the transmission plant, made up essentially of mains and compressor station equipment
- (5) the distribution plant, the major components of which are the mains, services, and meters

(6) the general plant, including transportation equipment, tools, shop and garage equipment, laboratory equipment, etc.

The distribution plant includes some or all of the following items:

- land and land rights
- structures and improvements
- mains
- compressor station equipment
- measuring and regulating station equipment
- services
- meters
- meter installations
- house regulators
- house regulating installations
- industrial measuring and regulating station equipment
- other property on customers premises
- other equipment

Mains, services, meters, and regulating equipment constitute most of the distribution plant. The mains comprise between 50% to 70% of the distribution plant value. They convey the gas taken from the transmission system to the final users and can be made of steel, cast iron, or plastic. Services comprise between 20% to 35% of the distribution plant. A gas service is the pipe between a distribution main and the customer's meter. Usually, it supplies a single building housing one or more customers. Both steel and plastic pipes are used for gas services. Meters are, of course, used to measure actual gas consumption by customers. Regulating equipment is used to control gas distribution pressures in both high-pressure and low-pressure systems. A gas pressure regulator automatically varies the rate of gas flow through a pipeline to maintain a preset outlet pressure.⁸

⁸For more technical details about the various components of the distribution network, see the <u>Gas Engineers Handbook</u>, sec. 9 (New York: Industrial Press, 1966).

Estimating Gas Distribution Plant Costs

The original (or historical) cost balances of the different components of a utility's plant in service, at the beginning and end of each year, are generally available in the annual reports that the utility submits to the regulatory authorities. The end-of-year value is equal to the beginningof-year value plus the value of the additions made during the year minus the original cost value of the plant retired during the same year. It is on the basis of these data that average plant costs per customer or per thousand cubic feet (MCF) delivered are estimated. Obviously, such an approach is ill fitted to deal with such considerations as joint, nonseparable costs, economies of scale, and population and land-use densities, inasmuch as they have an effect on plant costs.

The effects of market mix and density have been partially analyzed by some authors in the case of electrical distribution costs.⁹ In the case of gas distribution, the available data are even scarcer. One study reporting some relationships between gas distribution capital costs and density has been carried out by Real Estate Research Corporation for the Council on Environmental Quality and other government agencies, with the broader goal of assessing the environmental and economic costs of alternate housing types and development patterns at the urban fringe.¹⁰ Six neighborhood prototypes differing in housing type and density were analyzed. They are described in table 3.1.

⁹See, for instance: F. J. Wells, "The Effects of Customer Density on Electrical Distribution Costs," in P. B. Downing, ed., Local <u>Service Pricing Policies and Their Effect on Urban Spatial Structure</u> (Vancouver, University of British Columbia Press, 1977); M. L. Baughman and D. J. Bottaro, <u>Electric Power Transmission and Distribution Systems:</u> <u>Costs and Their Allocation</u>, National Science Foundation PB-247189 (Springfield, VA: National Technical Information Service, 1975).

Real Estate Research Corporation, <u>The Costs of Sprawl: Detailed Cost</u> <u>Analysis</u> (Washington, D.C.: U. S. Government Printing Office, 1974) Stock Number 4111-00023.

TABLE 3.1

| Neighborhood Prototype | | Population (Per 100 Acres) | Residential Density (Units per Acre) | | |
|------------------------|---------------------------------|-------------------------------|--|--|--|
| A. | Single Family Conventional | 3,520 | 2.0 | | |
| В. | Single Family Clustered | 3,520 | 2.5 | | |
| С. | Townhouse Clustered | 3,330 | 3.3 | | |
| D. | Walk Up Apartments | 3,330 | 5.0 | | |
| Ε. | High Rise Apartments | 2,825 | 10.0 | | |
| F. | Housing Mix 20% of A,B,C,D,E | 3,300 | 3.3 | | |

NEIGHBORHOOD PROTOTYPES CHARACTERISTICS

Source: The Costs of Sprawl: Detailed Cost Analysis, (Washington, D.C.: Government Printing Office, 1974).

The estimates for gas distribution capital costs are indicated in table 3.2 for each of the six neighborhood types. All figures are in 1973 dollars. It was assumed that all development prototypes would be typical of high-standard new suburban construction. Preexisting land uses and the relationships between the neighborhoods and the rest of the metropolitan area were not taken into account. Also, the study did not include the cost of debt servicing and replacement or upgrading costs for any facilities built within the development period. Thus, capital cost estimates are given only for pipelines and appurtenances within the neighborhood. The selected pipe materials were deemed to be typical of current practice in the U.S. (Use of other materials might alter cost estimates significantly.) Differences in costs due to differences in terrain, topography, and climate could not be considered. Also, 30% of the estimated costs of the pipelines were added to cover contractor's and subcontractor's profits and overhead plus engineering fees. The length of utility pipelines is close to street lengths. It was assumed that utility line length would be somewhat shorter than road length (10 to 15% less) to reflect sophisticated engineering and design practices.

TABLE 3.2

| | | | Neighborhood | | | an an Maria a Maria Manan a Anana an Anana an an an an an an Anana Maria Maria Manana Manana an Anana Manana Ma |
|---|--|--|--|------------|--|---|
| | A | В | C | D | E | F |
| Variable | Single | Single | - | 2 | | * |
| | Family | Family | Townhouse | Walk Up | High Rise | Housing Mix |
| | Conventional | Clustered | Clustered | Apartments | Apartments | (20% A,B,C,D,E) |
| Total | | | | | | |
| Pipeline | | | | | | |
| Length | 56,000' | 35,800' | 22,800' | 13,604' | 8,055' | 25,500' |
| Percentage of | | | | | | |
| Road Length | 90% | 80% | 80% | 80% | 90% | 85% |
| Cost per Linear Foot of Pipe | \$2.30 | \$2.30 | \$2.30 | \$3.00 | \$3.00 | In proportion to the Housing Mix |
| Total Pipeline Cost | \$124,200 | \$82 , 340 | \$52,440 | \$40,812 | \$24,165 | \$64,791 |
| Overhead and Profit | \$37,260 | \$24,702 | \$15,732 | \$12,244 | \$7 , 249 | \$19,437 |
| Total Capital Cost | \$161,460 | \$107,062 | \$68,172 | \$53,056 | \$31,414 | \$84,228 |
| alan di manana manana kana di manana manana aya mata mata mata ang mangana di kanang mangana di kana di kana di | an a | 188 - 1972 - 1979 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - 1970 - | a na | | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, | |

CAPITAL COSTS OF PROVIDING GAS TO THOUSAND HOUSING UNITS IN SIX NEIGHBORHOODS

Source: The Costs of Sprawl: Detailed Cost Analysis, (Washington, D.C.: U.S. Government Office, 1974).

Although the data in table 3.2 confirm, rather dramatically, the relationship between gas distribution capital costs and density, they remain too much prototype related to be of general use. Also, as noted earlier, they only refer to urban extensions and do not account for the cost implications of these extensions for the whole community or metropolitan area. For instance, such extensions may call for the reinforcement of the existing network to meet the increased loads through mains duplication or compressor stations installation, etc. Also, these new urban developments may be located at varying distances from the existing main lines, implying extensions of mains of varying lengths.

Next, it is important to remember that residential customers constitute only a part, important as it may be, of the gas market, and no specific cost data appear to be available for commercial and industrial customers that have consumption levels and load profiles significantly different from those of the residential customers. Hence, the distribution capital costs incurred to serve them can also be expected to be significantly different.

Finally, any given customer will consume more or less gas, depending upon the climate of the area where he is located, all other factors remaining the same, and it is necessary to account for the climatic factor in estimating and predicting distribution costs.

For all of the above-mentioned reasons, new approaches to the analysis of gas distribution capital costs are called for. The purpose of the next section is to outline the principles of such an approach.

An Econometric Approach to Distribution Plant Analysis

Most gas distribution utilities keep track of their capital investments at the community level. In some states, such as New York, they are required to do so for tax assessment purposes. They also keep track of their gas sales, numbers of customers, and revenues for market, revenue, and billing analyses. The communities located in the service territory of any utility display strong variations in terms of (1) the number of residential, commercial, and industrial customers and the

corresponding unit-average and total gas loads; (2) the amount of distribution plant in service within the community boundaries; (3) the population and land acreage, and hence the density of the community; and (4) climatic factors if the service territory is spread over a climatologically heterogeneous region. Once such data are gathered, the natural next step is to try to explain, through regression analysis, the variations of the distribution plant in service - the dependent variable - by the variations of such independent variables as market size and mix, population density, winter-cold severity, etc. Both additive (linear) and multiplicative (logarithmic) models should be tested. Examples of such models are¹¹

$$PS = a_0 + a_1 * RMCF + a_2 * CMCF + a_3 * IMCF + a_4 * TEDN$$
(3.1)

 $\ln(PS) = b_0 + b_1 * \ln(RMCF) + b_2 * \ln(CMCF) + b_3 * \ln(IMCF) + b_4 * \ln(TEDN)$ (3.2)

where

| | 'S is the amount of distribution plant in service (\$) | |
|-------|---|--|
| | MCF is the amount of annual residential gas sales (MCF) | |
| Galle | IMCF is the amount of annual commercial gas sales (MCF) | |
| ingel | IMCF is the amount of annual industrial gas sales (MCF) | |
| 1000 | EDN is the population density (population per acre) | |

In equations (3.1) and (3.2), the coefficients a_1 , a_2 , a_3 and/or b_1 , b_2 , b_3 are expected to be positive, and the coefficients a_4 and/or b_4 negative. If, for instance, the linear model prevails, then the coefficients a_1 , a_2 , and a_3 represent the marginal plant costs incurred by serving one additional MCF to the residential, commercial, and industrial sectors, respectively. If, on the other side, the logarithmic model prevails, then the costs of service to the three sectors are nonseparable, and the marginal cost of serving, say, one MCF to the residential sector depends upon the current levels of sales to the three markets. Also, the latter case implies the existence of economies or diseconomies of scale, whereas the linear model implies constant costs to scale.

One obvious problem with the above approach is related to the use of the original cost balance for measuring the value of the plant in service,

¹¹From here on, the text contains a combination of algebraic and FORTRAN notations; e.g., multiplication is sometimes designated by an asterisk(*).

instead of its replacement cost that should be the correct reference for measuring total and marginal costs. However, if the various communities of the service territory have plants in service made up, percentage-wise, of equipment of similar vintages, then it can reasonably be assumed that the ratio between historical cost and replacement cost is approximately constant. Such an assumption turned out to be verified for the distribution plant of Pacific Gas and Electric Company (in 1979 the replacement cost was equal to 2.79 times the historical cost) and will be retained for the other companies analyzed in this study.

Other functional forms can be tested. For instance, the sectoral numbers of customers instead of the sectoral sales can be used as independent variables. However, both types of variables should not be used simultaneously, for there may be a strong to very strong correlation between them (i.e., the number of residential customers and the MCF level of residential sales are generally very strongly correlated). Also, the independent variables may be aggregated in various ways: total sales or total number of customers, commercial and industrial customers or sales, etc.

Cost functions such as those illustrated by equations (3.1) and (3.2)are indicative of long-term total costs and marginal costs. Indeed, as the whole community plant is taken into consideration in the analysis, the resulting marginal cost of serving, say, one additional residential MCF includes both the marginal cost corresponding to the localized main extension and service and meter, and the marginal cost corresponding to the necessary adjustments in the whole community plant. The latter may be incurred much after the extension has been made, as a result of reaching some threshold point in the operation of the whole network. The former, however, may be termed a "short-term" marginal cost, directly incurred at the time of new installation and service. Such a short-term cost could, in principle, be analyzed with time-series data on plant in service and market size and structure. In this case, however, only the numbers of customers should be used as independent variables because gas sales may change significantly from one year to the next as a result of climatic changes, even while the numbers and characteristics of the customers do not change.
Examples of short-term cost models are

$$DPS = a_0 + a_1 * DRCUS + a_2 * DCCUS + a_3 * DICUS + a_4 * TEDN$$
(3.3)

 $\ln(DPS) = b_0 + b_1 * \ln(DRCUS) + b_2 * \ln(DCCUS) + b_3 * \ln(DICUS) + b_4 * \ln(TEDN)$ (3.4)

where

- DPS is the increase in the amount of distribution plant in service between two consecutive years, and
- DRCUS, DCCUS, and DICUS are the increases in the numbers of residential, commercial, and industrial customers during the same period, respectively

The feasibility of this short-term analysis depends upon (a) the availability of the corresponding data; and (b) the existence of actual market growth (i.e., DRCUS, DCCUS, DICUS > 0). As is well known, a ban on new customer hookups had been instituted in most states in the early 1970s because of steadily decreasing available gas supplies. However, because of wellhead gas pricing changes, supplies started to increase again in the late 1970s and the ban was removed. Utilities started to connect new customers, and the corresponding market growth has been particularly noticeable in 1978 and 1979. Some limited analyses of the "short-term" cost effect therefore turned out to be feasible within the framework of this study.

Based on the above principles and ideas, various analyses have been performed with data obtained from four different gas distribution utilities: Long Island Lighting Company, Columbia Gas of Ohio Company, Pacific Gas and Electric Company, and National Fuel Gas Distribution Corporation. The purpose of the next section is to describe the available data and the results of these analyses.

Applications of the Econometric Approach

Long Island Lighting Company (LILCO)

LILCO is a dual gas and electric, privately owned utility serving, in 1979, an estimated 2,884,601 people, including 97,343 persons residing on the Rockaway Peninsula of Queens, New York. Most of the served population is residing in communities located in the Long Island counties of Nassau and Suffolk.

The data in table 3.3 provide an overview of the gas plant in service at the end of 1978 and 1979, and those in table 3.4 present a summary of gas sales and average numbers of gas customers during the years 1978 and 1979.

Firm gas sales in 1979 totaled 39,400,000 MCF, down only 2.6% below 1978, despite much more moderate winter weather in 1979 (4,622 versus 5,441 annual heating degree-days; normal year average = 5,095 degree-days). Sales to interruptible commercial and industrial customers rose 170.5%. During 1979, the number of LILCO gas space-heating customers was increased by 5,600. However, the net balance of the average number of residential customers increased only by 297 because of a significant attrition of the existing market due to an overall population decline. Also, the existing firm customers were allowed to expand their firm gas requirements. Thus, LILCO was characterized by a dynamic market in 1978/1979 that will permit a limited "short-term" cost analysis.

The data in table 3.3 show that the distribution plant makes up for about 73% of the total plant. Mains and services, in turn, make up for about 59% and 29% of the distribution plant. In 1979, the changes in the distribution plant included (a) additions, valued at \$8,238,075; and (b) retirements, valued at \$609,177. The value of the additions corresponds to replacement costs, whereas the value of the retirements corresponds to historical (original) costs. A part of the additions is used to replace the retired plant, but most of it is likely to be related to new service, the exact amount depending upon the replacement costs.

LILCO

| | · · |
|--------------------|-------|
| 1 in 10 1 23 | ?C) |
| (TH DOTTOL | . ວ / |

| Plant Component | End of 1978 | End of 1979 |
|--|--|---------------|
| OVERVIEW | | |
| Manufactured Gas Production | \$ 5,833,120 | \$ 5,896,982 |
| Storage | 12,246,770 | 12,439,837 |
| Transmission | 48,127,301 | 48,506,446 |
| Distribution | 200,990,807 | 208,619,705 |
| General | 7,048,017 | 8,205,708 |
| | aaree aaree ay waa ah a | |
| Total | \$274,246,015 | \$283,668,678 |
| DISTRIBUTION PL4 | ANT | |
| Land and Land Rights | \$ 279,148 | \$ 280,014 |
| Structures and Improvements | 280,627 | 324,148 |
| Mains | 118,460,883 | 122,359,684 |
| Compressor Station Equipment | 9,585 | 9,585 |
| Measuring and Regulating Station Equipment | 2,294,563 | 2,453,773 |
| Services | 59,141,513 | 61,647,931 |
| Meters | 11,728,385 | 12,701,159 |
| Meter Installation | 6,791,098 | 6,826,330 |
| House Regulators | 2,005,005 | 2,017,081 |
| | | |

Sources: Annual Reports of LILCO to the State of New York Public Service Commission - 1978 and 1979.

VOLUME OF GAS SALES AND NUMBER OF CUSTOMERS IN 1978 AND 1979 - LILCO

| | | Year |
|-------------------------|---------------------|---------------------------------------|
| Sector | 1978 | 1979 |
| · · · | GAS SALES (MCF) | · · · · · · · · · · · · · · · · · · · |
| Residential | 27,470,883 | 26,369,644 |
| Commercial & Industrial | 14,549,342 | 17,241,741 |
| Public Authorities | 13,693 | 13,865 |
| Interdepartmental | 75,237 | 2,760,023 |
| Total | 42,109,155 | 46,385,272 |
| | NUMBER OF CUSTOMERS | · · · · · · · · · · · · · · · · · · · |
| Residential | 356,547 | 356,844 |
| Commercial & Industrial | 30,415 | 30,399 |
| Public Authorities | 60 | 58 |
| Total | 387,022 | 387,301 |
| | | |

Sources: Annual Reports of LILCO to the State of New York Public Service Commission - 1978 and 1979.

On the basis of the data in tables 3.3 and 3.4, the 1978 distribution historical unit costs per MCF and customer are the following:

- 4.73 \$/MCF, and

- 519 \$/customer¹²

Included in the Annual Reports submitted by LILCO to the State of New York Public Service Commission (NYPSC) are community-level data on annual gas sales and average number of customers for the residential sector and for the combined commercial and industrial sectors, as well as on the value of the total gas plant in service by the end of the year. Unfortunately, the latter data are not further disaggregated, and therefore the amount of distribution plant is not known, and the application of the econometric analysis to the total plant is likely to introduce some bias because such items as gas production plant and transmission plant are very unlikely to be related to the structure of the local markets. This is less so for the storage plant, which includes mostly short-term gas holders, and for the general plant, which can be both related to local variables to some extent. Thus, the best that can be done in this case is to specify econometric models with the total plant in service variable, and then adjust the resulting equations by the ratio of the distribution to total plants (0.7329 in 1978 and 0.7354 in 1979).

A complete set of plant and market data was prepared for 101 communities for both 1978 and 1979. These data are presented in appendix A. LILCO's estimates of the 1978 population of these communities were also included in the data set. Land area data were partly drawn from a 1970 Census of Population report, ¹³ and from census tract acreage data provided by R. J. Panzarella, Forecast Analyst at LILCO. Complete acreage data were gathered for 89 communities.

The long-term, or static, analysis has been performed with the 1978 data. At the end of 1978, the total plant in service in the 101

¹³1970 Census of Population - Population of Places of 2500 or more - 1950 and 1970 Supplementary Report PC(S1)-26 (August 1972.)

¹²LILCO's historical distribution unit costs are significantly larger than those of the other companies to be analyzed in this chapter. This is most likely due to the fact that LILCO's plant is made up of components of more recent vintages. However, this assumption could not be verified because the necessary data are lacking.

communities amounted to \$246,559,200, or 89.9% of the total LILCO plant in service. All the 387,022 LILCO 1978 customers were located in these 101 communities. The short-term, or dynamic, analysis has been performed by taking the difference between the 1979 and 1978 data on both plant and market variables.

a. The Static Analysis

The results presented in this section pertain to the 89 communities for which density figures could be prepared. The definitions and means and standard deviations of the various variables are presented in Table 3.5.

TABLE 3.5

| Variable | Definition | Mean | Standard Deviation |
|----------|---|-----------|-----------------------|
| DC | Plant in Service (ξ) - End of 1078 | 2 750 795 | 6 800 365 |
| TMCF | Total Cas Sales (MCF) $= 1978$ | 468,923 | 1,111,092 |
| RMCF | Residential Gas Sales (MCF) - 1978 | 305,941 | 722,895 |
| CIMCF | Commercial & Industrial Gas Sales (MCF) - 1978 | 162,982 | 406,177 |
| TCUS | Total Number of Customers - 1978 | 4,316 | 10,137 |
| RCUS | Number of Residential Customers - 1978 | 3,977 | 9,362 |
| CICUS | Number of Commercial & Industrial Customers - 1978 | 339 | 790 |
| TEDN | Population Density (people per acre) | 8.764 | 9.135 |

DEFINITIONS, MEANS, AND STANDARD DEVIATIONS OF THE VARIABLES LILCO STATIC ANALYSIS

Source: Author's calculations.

In a first stage, the plant in service (PS) was regressed on the aggregate sales or number of customers, and on the population density, for both the additive and multiplicative forms. The multiplicative model is expressed in final (nonlogarithmic) multiplicative form. The t-statistics of the coefficients are indicated in parenthesis at the appropriate places. The following four models were obtained.

$$PS = 238,641.4 + 6.03379 * TMCF - 36,196.96 * TEDN (R2 = 0.941) (3.5)(37.13)1.4 (1.83)PS = 6.11658 * TMCF1.0120 * TEDN-0.1514 (R2 = 0.927) (3.6)$$

(3.35)

(32.02)

$$\begin{cases} PS = 342,818.4 + 648.528 * TCUS - 44,667.47 * TEDN (R2 = 0.904) (3.7) \\ (28.42) (1.76) \end{cases}$$

$$PS = 1983.9955 * TCUS^{0.9141} * TEDN^{-0.3366} (R2 = 0.858) (3.8) \\ (22.04) (5.03) \end{cases}$$

The performances of the four above models, as measured by their R^2 , are overall quite good, with slightly higher R^2 in the linear case. However, the density coefficient is much more significant in the multiplicative case than in the linear one (where the confidence level is around 95% only). However, as is well known, R^2 for linear and log-linear models cannot be directly compared. For the same dependent variable and equivalent number of independent variables, the functional form that yields the minimum sum of squares of the residuals is generally to be selected.¹⁵ A transformation of PS that permits such a comparison is $PS_1 = C.PS$, where C is the inverse of the geometric mean of PS. The sum of squares of the residuals in the linear model must then be multiplied by C^2 , and the resulting value S_1 must be compared to the sum S_2 of squares of the residuals in the logarithmic case, the model with the smaller sum value being generally preferred. It is further possible to test whether the two functions are empirically equivalent by computing the d statistic

$$d = \frac{N}{2} \left| \ln \left(\frac{S_1}{S_2} \right) \right|$$
(3.9)

where N is the sample size. The larger of the two sums is placed in the numerator. If the two forms are equivalent, then d follows the chi-square

¹⁴The t-statistics, measuring the significance of the regression coefficients, are indicated in parenthesis below the corresponding coefficient. The level of significance, for a given t-value, depends upon the sample size and number of variables of the regression model.

¹⁵See: P. Rao, and R.L. Miller, <u>Applied Econometrics</u> (Belmont CA: Wadsworth Publishing Company, Inc.,1971),pp. 107-11.

distribution with one degree of freedom. (The critical value of the 90% level of confidence is 2.706).

On the basis of the previous criteria, the logarithmic models are clearly superior to the linear ones. For instance, the sum S_1 for equation (3.5) is equal to 138.83, whereas the sum S_2 for equation (3.6) is equal to 18.32. The multiplicative models are rewritten below to reflect only distribution costs by adjusting the total plant equations by the distribution to total plants ratio (0.7329)

$$PS = 4.48284 * TMCF^{1.0120} * TEDN^{-0.1514}$$
(3.10)

$$PS = 1,454.0703 * TCUS^{0.9141} * TEDN^{-0.3366}$$
(3.11)

The above cost functions imply nearly constant costs to scale (extremely slight diseconomies of scale) with respect to total sales and some economies of scale with respect to the total number of customers. The corresponding marginal distribution capacity cost functions are

$$MC(TMCF) = \frac{\partial PS}{\partial TMCF} = 4.53663 * TMCF^{0.0120} * TEDN^{-0.1514}$$
(3.12)

$$MC(TCUS) = \frac{\partial PS}{\partial TCUS} = 1,329.1657 * TCUS^{-0.0859} * TEDN^{-0.3366}$$
(3.13)

The marginal costs for a hypothetical average community characterized by the average figures in table 3.5 are

The next step of the analysis was to use sectoral sales and numbers of customers as independent variables. The results are

 $^{16}\ensuremath{\mathsf{Hereafter}}$ in this chapter referred to as the marginal cost.

$$\begin{cases} PS = 229,418.0 + 5.77470 * RMCF + 6.50241 * CIMCF - 34,814.58 * TEDN \\ (8.24) (5.23) (1.72) \\ (R^2 = 0.941) (3.14) \\ PS = 46.76794 * RMCF^{0.7370} * CIMCF^{0.1545} * TEDN^{-0.1765} \\ (15.00) (5.59) (3.03) \\ (R^2 = 0.883) (3.15) \\ (R^2 = 0.883) (3.15) \\ (R^2 = 0.883) (1.67) \\ (3.79) (1.88) (1.67) \\ (R^2 = 0.906) (3.16) \\ (R^2 = 0.906) (4.81) \\ \end{cases}$$

$$(R^2 = 0.874)$$
 (3.17)

The sums of the squares of the residuals are significantly smaller in the logarithmic cases, and therefore the multiplicative models are to be selected. It is also notable that the regression coefficients are much more significant in the multiplicative cases. Besides the previous statistical reasons for rejecting the linear models, there are also other substantive, logical reasons to do so. Indeed, the relative values of the coefficients of RMCF and CIMCF in equation (3.14) are highly questionable: it would appear that the distribution plant cost of one additional "commercial and industrial" MCF is higher than the corresponding residential one. This result is not consistent with the load profiles of these two sectors (the load factor of the residential sector being much lower) and the necessary relationship between distribution plant cost and peak load.

The multiplicative cost functions are characterized by significant economies of scale effects. The exponents can also be viewed as cost elasticities to sectoral sales or numbers of customers; that is, they indicate the percentage increase in plant cost due to 1% increase in the corresponding independent variables. As could be expected, the elasticities of the residential sector are larger than those of the

commercial-industrial one.

The multiplicative models are rewritten below to reflect only distribution costs by using the distribution to total plants ratio (0.7329)

$$PS = 34.27622 * RMCF^{0.7370} * CIMCF^{0.1545} * TEDN^{-0.1765}$$
(3.18)

 $PS = 3969.2046 * RCUS^{0.6057} * CICUS^{0.2768} * TEDN^{-0.3106}$ (3.19)

Focusing on the sales model (3.18), it is now possible to derive marginal cost functions with respect to residential and commercial-industrial sales, MC(RMCF) and MC(CIMCF), respectively, with

$$MC(RMCF) = \frac{\partial PS}{\partial RMCF} = 25.26055 * RMCF^{-0.2630} * CIMCF^{0.1545} * TEDN^{-0.1765}$$
(3.20)

$$MC(CIMCF) = \frac{\partial PS}{\partial CIMCF} = 5.29533 * RMCF^{0.7370} * CIMCF^{-0.8451} * TEDN^{-0.1765}$$
(3.21)

The marginal costs for the hypothetical average community characterized by the average sales and density figures presented in table 3.5 are then

 $\overline{MC}(RMCF) = 3.9661$ \$/MCF

 $\overline{MC}(CIMCF) = 1.5606$ \$/MCF

The above two values should be compared to the marginal cost of 3.82 \$/MCF when the total load is considered. (See equation 3.12.) Clearly the latter is not very helpful to discriminate between the two sectors and using it would heavily penalize the commercial-industrial sector while slightly advantaging the residential one. The customers-related marginal cost functions are derived similarly, with

$$MC(RCUS) = \frac{\partial PS}{\partial RCUS} = 2403.6471 * RCUS^{-0.3943} * CICUS^{0.2768} * TEDN^{-0.3106}$$
(3.22)

 $MC(CICUS) = \frac{\partial PS}{\partial CICUS} = 1098.8346 * RCUS^{0.6057} * CICUS^{-0.7232} * TEDN^{-0.3106}$ (3.23)

The marginal customer-related costs for the average community depicted by the data in table 3.5 are then

MC(RCUS) = 234.08 \$/residential customer

MC(CICUS) = 1255.38 \$/commercial-industrial customer

To illustrate the variations of these marginal costs with market size, consider a much smaller community with 500 residential customers, 20 commercial-industrial customers, and a density of 4 people per acre. The marginal costs are then

MC(RCUS) = 308.99 \$/residential customer

MC(CICUS) = 3,531.41 \$/commercial-industrial customer

The above values should be compared to the marginal cost of 311.897 \$/customer when the total number of customers is considered. (See equation 3.13.) Basing a pricing policy on the latter cost would considerably advantage (in fact subsidize) the commercial-industrial sector at the slight expense of the residential one. Note, however, that all the above cost figures are based on historical costs data and are therefore smaller than the corresponding replacement cost figures. Naturally, any pricing policy incorporating distribution capacity marginal costs should use replacement cost figures.

b. The Dynamic Analysis

The numbers of communities characterized by an increase, decrease, or no change in the numbers of their residential (DRCUS) and commercialindustrial (DCICUS) customers are indicated in table 3.6.

| Nuclear of Decidential | Number of Commercial-Industrial Customers (DCICUS) | | |
|------------------------|---|--------------------|------------------|
| Customers (DRCUS) | Decrease (< 0) | No Change (= 0) | Increase (>0) |
| Decrease (< 0) | 12 | 7 | 6 |
| No Change (= 0) | 2 | 12 | 3 |
| Increase (>0) | 22 | 19 | 18 |

MARKET DYNAMICS IN LILCO COMMUNITIES DURING THE PERIOD 1978-1979

Source: Author's calculations.

Three separate analyses were performed on the following groups of communities: (1) the 18 communities displaying growth in both sectors, (2) the 41 communities displaying growth in the residential sector only, and (3) the 9 communities displaying growth in the commercial-industrial sector only. A common feature of the three analyses is that the logarithmic model is, by far, superior to the linear one, and therefore only results pertaining to the former are presented. Also, the density variable turned out to be insignificant and was discarded. The definitions of the variables and their average values in the above three cases are presented in table 3.7.

TABLE 3.7

DEFINITIONS AND MEAN VALUES OF THE DYNAMIC ANALYSIS VARIABLES - LILCO

| | | Case 1 | Case 2 | Case 3 |
|----------|---|---------------------|---------------------------------|---------------------|
| Variable | | DRCUS>0 DCICUS>0 | DRCUS>0 DCICUS <u><</u> 0 | DRCUS<0 DCICUS>0 |
| DPS | Increase in Plant in Service (\$) | 198,842 | 39,540 | 180,622 |
| DRCUS | Increase in Residential Customers | 16.89 | 7.71 | -14.33 |
| DCICUS | Increase in Commercial- Industrial Customers | 1.72 | -2.73 | 4.78 |

Source: Author's calculations.

Case 1: DRCUS > 0; DCICUS > 0; 18 communities

The model, adjusted to reflect distribution costs only, is

DPS = 1,375.9258 * DRCUS^{0.7585} * DCICUS^{1.0904} (
$$R^2 = 0.486$$
) (3.24)
(1.42) (1.58)

The coefficients of DRCUS and DCICUS are significant at the 10% level. The corresponding marginal cost functions are

$$MC(DRCUS) = \frac{\partial DPS}{\partial DRCUS} = 1,043.5852 * DRCUS^{-0.2415} * DCICUS^{1.0904}$$
(3.25)

$$MC(DCICUS) = \frac{\partial DPS}{\partial DCICUS} = 1,500.2806 * DRCUS^{0.7585} * DCICUS^{0.0904}$$
(3.26)

The marginal costs for the hypothetical community depicted by Case 1 growth data in table 3.7 (DRCUS = 16.89, DCICUS = 1.72) are

MC(DCICUS) = 13,445.216 \$/new commercial-industrial customer

The ratio between the above residential "dynamic" marginal cost (\$952.405) and the residential "static" marginal cost computed for the average community in the previous section (\$234.076) is equal to 4.07. This ratio may be viewed as a first, rough estimate of the ratio between replacement and historical costs. It is larger than the one obtained for Pacific Gas and Electric Company (2.78), and this may be due to the older age of LILCO's plant. The corresponding ratio for commercial-industrial customers is much larger, equal to 10.71 (\$13,445.216/\$1,255.38). However, this ratio does change rapidly with market size, and it does not seem possible to specify the characteristics of two equivalent static and dynamic communities for which costs could be meaningfully compared. (For instance, if the community of 500 residential customers and 20 commercial-industrial customers is selected for the static case, the previous ratio becomes equal to 3.807).

Case 2: DRCUS > 0; DCICUS < 0; 42 communities

In this case, the plant increase is assumed to be solely related to residential growth. It is also assumed that there is no retirement of the plant in service related to the attrition of the commercial-industrial customers. The model, adjusted to reflect distribution costs only, is then

DPS = 1,748.723 * DRCUS^{0.9996} (
$$R^2 = 0.276$$
) (3.27)
(3.85)

Although the correlation coefficient is lower than in the previous case, it is significantly different from zero (at the 0.1% level), and the regression coefficient is also highly significant. The above model can be viewed as an almost constant-cost-to-scale one, with a constant distribution plant marginal cost equal to

MC(DRCUS) = 1,747 \$/new residential customer

The above marginal cost appears to be larger (by \$795) than the one obtained in case 1. A reasonable explanation for this difference is that in the present case the residential sector does not benefit from the positive technological externalities related to the addition, in the distribution system, of commercial and industrial customers. In other words, the joint-cost effect does not take place here, and the cost difference of \$795 is a measure of the economic benefit derived from this externality by the residential sector.

Case 3: DRCUS \leq 0; DCICUS > 0; 9 communities

In this case, it is assumed that the plant increase is solely related to commercial-industrial growth, and that there is no retirement of the plant in service related to the attrition of the residential customers. The model, adjusted to reflect distribution costs only, is then

DPS = 298.735 * DCICUS^{2.9557} (
$$R^2 = 0.5$$
) (3.28)
(2.64)

The above model has significant correlation and regression coefficients. The cost function is characterized by diseconomies of scale, and so is the corresponding marginal cost function

$$MC(DCICUS) = 882.9791 * DCICUS^{1.9557}$$
(3.29)

A comparison of the results obtained with equations (3.29) and (3.26) shows that for a <u>given</u> commercial-industrial growth the corresponding marginal customer cost is going to be lower in presence of residential growth, as compared to the no-residential-growth case, only below a given threshold of minimal residential growth. Assume, for instance, that DCICUS = 5 customers, then equation (3.29) would yield, in presence of no residential growth, a marginal cost of \$20,556.3. With reference to equation (3.26), the residential growth leading to the same commercial-industrial marginal cost is equal to 26 residential customers. If DRCUS = 5, then MC(DCICUS) = \$5,881, and if DRCUS = 40, then MC(DCICUS) = \$28,474. To determine the technological circumstances (if any) producing these cost effects would require much more in-depth analyses of local factors, a study that could not be performed in the framework of this research.

Columbia Gas of Ohio Company (CGOC)

Columbia Gas of Ohio is a privately owned distribution utility providing service to 360 communities in central, northern, and southern Ohio. Its major supplier is the Columbia Transmission Corporation that also owns the underground storage fields used, at a cost, by CGOC. Therefore, the major part of the CGOC plant is its distribution plant, as demonstrated by the plant in service data for 1976 and 1977 presented in table 3.8.

The data in table 3.8 show that the distribution plant makes up for about 97% of the total plant. Mains and services, in turn, make up for about 59% and 21% of the distribution plant. Meter-related equipment and house-regulators-related equipment make up for another 11% and 2% of this plant.

The data in table 3.9 provide a summary of gas sales and average numbers of gas customers during the years 1976 and 1977. On the basis of

VALUE OF GAS PLANT IN SERVICE AT THE END OF 1976 AND 1977 CGOC

| 7 - | The | ~ ~ | | · · |
|---------|-------|-----|------------|-----|
| () n | 100 | 11 | nr | C 1 |
| 1 1 1 1 | - 170 | 1 1 | <i>a</i> 1 | D / |
| < | | | ~ | ~ / |

| Plant Component | End of 1976 | End of 1977 | | | |
|---|--------------|--------------|--|--|--|
| DISTRIBUTION PLANT | | | | | |
| Land and Land Rights | \$ 3,384,835 | \$ 3,452,563 | | | |
| Structures and Improvements | 6,817,690 | 6,897,848 | | | |
| Mains | 238,835,096 | 245,473,197 | | | |
| Measuring and Regulating Station Equipment | 4,518,807 | 4,840,849 | | | |
| Measuring and Regulating Station Equipment - City Gate Check | 2,491,710 | 2,529,750 | | | |
| Services | 82,893,968 | 89,504,927 | | | |
| Meters | 33,883,409 | 34,763,419 | | | |
| Meter Installation | 10,794,359 | 11,126,744 | | | |
| House Regulators | 4,236,932 | 4,245,608 | | | |
| House Regulator Installation | 4,372,382 | 4,391,326 | | | |
| Industrial Measuring and Regulating | | | | | |
| Station Equipment | 5,395,295 | 5,392,429 | | | |
| Other Property on Customers' Premises | 877,333 | 877,333 | | | |
| Other Equipment | 1,814,603 | 1,679,221 | | | |
| | | • | | | |

Total Distribution Plant

\$400,316,419 \$415,175,214

| $T_{a} \leftarrow a 1$ | 114 2 1 2 4 | Dlamb | |
|------------------------|-------------|-------|--|
| TOLAT | ULIILV | Plant | |
| | | | |
| | | | |

\$412,424,960 \$428,715,064

Sources: Annual Reports of CGOC to the Public Utilities Commission of Ohio (PUCO).

| | Yea | ır |
|---|--|--|
| Sector | 1976 | 1977 |
| | GAS SALES (MCF) | |
| Residential Commercial Industrial | 158,014,211 64,601,949 130,904,784 | 151,145,232 56,232,729 102,155,711 |
| Total | 353,520,944 | 309,533,672 |
| | NUMBER OF CUSTOMERS | |
| Residential Commercial Industrial | 965,915 78,219 1,732 | 960,577 77,380 1,592 |
| Total | 1,045,866 | 1,039,549 |

VOLUME OF GAS SALES AND NUMBER OF CUSTOMERS IN 1976 AND 1977 - CGOC

Sources: Annual Reports of CGOC to the Public Utilities Commission of Ohio (PUCO).

these data, the 1976 distribution plant historical unit costs per MCF and customer are the following (using the 1977 sales figures could be misleading because of the heavy curtailments that took place in 1977)

- 1.132 \$/MCF, and

- 382.761 \$/customer

The home rule provision in Ohio's constitution and statutes permits a municipality to contract with a privately owned utility to obtain services by passage of a rate ordinance and its acceptance by the utility. On the basis of this provision, CGOC establishes gas rates separately with 360 Ohio communities. There are marked variations among these rates, related, according to CGOC, to variable costs of bringing gas to these communities. At the request of the PUCO, data were collected by The National Regulatory Research Institute (NRRI) to find out whether there could be significant improvements in the ratemaking procedures adopted by the PUCO and by the various communities. Various data have been gathered for a sample of 52 communities included in the set of the 291 municipalities that had rate changes through either PUCO rate orders or ordinance rate negotiations during the period 1976-1979. The data retained for the purpose of the present study are

- the net plant in service, or rate base: RB
- the residential, commercial, and industrial gas sales (MCF): RMCF, CMCF, IMCF
- the numbers of residential, commercial, and industrial customers: RCUS, CCUS, ICUS

This data set was complemented, for 42 communities, with population and acreage data. In addition, the combined commercial-industrial sector was also considered (as in the case of LILCO's analysis), with the corresponding sales and number of customers noted CIMCF and CICUS. The means and standard deviations of the above plant and market variables are presented in table 3.10, and the detailed community-level data in appendix The rate base, or net plant in service, is equal to the total plant in Β. service minus the accumulated provision for depreciation, amortization, and depletion. The latter was equal, at the end of 1977, to \$145,155,000, while the total plant in service was equal, at the same period, to \$428,715,064. (See table 3.8.) Thus, the ratio of total to net plants in service is equal to 1.512. It is assumed that this adjustment ratio can be uniformly applied to the rate bases of the 52 communities. Second, as was done in the case of LILCO, it is necessary to adjust the total plant figure to reflect only the distribution plant costs. The 1977 distribution to

| Variable | Mean | Standard Deviation |
|------------------------|-----------|-----------------------|
| RB (\$) | 2,352,877 | 6,690,295 |
| TMCF (total sales) | 1,978,637 | 6,083,876 |
| RMCF | 1,441,581 | 4,332,637 |
| CMCF | 496,148 | 1,649,132 |
| IMCF | 40,907 | 121,460 |
| CIMCF | 537,055 | 1,765,766 |
| TCUS (total customers) | 9,505 | 29,016 |
| RCUS | 8,871 | 27,123 |
| CCUS | 599 | 1,721 |
| ICUS | 35 | 215 |
| CICUS | 634 | 1,900 |

MEANS AND STANDARD DEVIATIONS OF THE VARIABLES CGOC STATIC ANALYSIS

Source: Author's calculations.

total plants ratio is selected, equal to 0.9684. Therefore, the rate base figures must be multiplied by 1.4642 to represent the distribution plant in service. Another problem is related to the fact that the data do not all pertain to the same year (28 communities refer to 1976 data, 12 to 1977 data, and the remainder equally to 1978 and 1979 data). Indeed, gas sales vary from one year to another because of weather changes, all other factors remaining equal (i.e., the numbers of customers). One way to eliminate this problem is to adjust gas sales with reference to an average-weather year (e.g., with an average number of degree-days). To perform this adjustment, the knowledge of the load equations of the different sectors is a prerequisite but could not be gathered in this study. Thus some bias is likely to exist in the resulting econometric models. However, in view of the excellent fits obtained, it is believed that this bias is probably negligible.

In a first stage, the distribution plant in service PS was regressed on the aggregate sales or number of customers, and on the population density, with both the additive and logarithmic forms. The density variable turned out to be highly insignificant and was therefore

discarded. Thus, the analyses were performed with the 52 communities' data. The following four models were obtained

$$PS = 385,565.5 + 1.60805 * TMCF (R2 = 0.997)$$
(3.30)
(138.53)

$$PS = 6.63155 * TMCF^{0.9141} \qquad (R^2 = 0.973) \qquad (3.31)$$

$$(42.54)$$

$$(PS = 248,023.6 + 438.848 * TCUS (R2 = 0.992)$$
(3.32)
(81.95)

$$(PS = 810.691 * TCUS^{0.9183} (R^2 = 0.963) (3.33)$$
(3.33)

The sums of the squares of the residuals are significantly smaller in the logarithmic cases, and therefore the multiplicative models are to be selected. They imply economies of scale of similar magnitude with respect to both sales and total number of customers. The corresponding marginal cost functions are

$$MC(TMCF) = \frac{\partial PS}{\partial TMCF} = 6.0617 * TMCF^{-0.0859}$$
(3.34)

$$MC(TCUS) = \frac{\partial PS}{\partial TCUS} = 744.482 * TCUS^{-0.0817}$$
(3.35)

The marginal costs for a hypothetical average community characterized by the average figures in table 3.10 are

$$\overline{MC}(TMCF) = 1.744$$
 \$/MCF

 $\overline{MC}(TCUS) = 352.35$ \$/customer

It is interesting to compare the above marginal cost functions with the corresponding ones obtained in the LILCO analysis (equations 3.12 and 3.13). If equation (3.13) is adjusted for an average density of 8.764

people per acre, the resulting equation

$$MC(TCUS) = 640.126 * TCUS^{-0.0859}$$
(3.36)

is very similar to equation (3.35), both with respect to the multiplicative constant and the exponent, hence the similar customer-related marginal costs. On the other side, the sales-related marginal cost of LILCO (3.82 \$/MCF) is about twice as large as the corresponding cost for CGOC (1.744 \$/MCF). This apparent contradiction is resolved when it is noted that the average CGOC customer annual load is about twice the corresponding LILCO load. Thus, there are significant economies of scale associated with customer's size, and this observation points out the need for further econometric analyses involving customers' sizes in addition to the presently used variables. These analyses could not be performed in the framework of this study.

The next step in the analysis was to use sectoral sales and numbers of customers as independent variables. The industrial sector-related variables turned out to be insignificant or having the wrong sign. This result is probably related to the fact that most of the industrial customers are located in the large cities of Columbus (1,555) and Toledo (107), while most of the communities have very few such customers or none at all (24 communities in the latter case). The commercial and industrial sectors have therefore been pooled together. The subsequent analyses are therefore strictly similar to those performed on LILCO's data. The results are

 $\begin{cases} PS = 204,983.7 + 2.1053 * RMCF + 0.3821 * CIMCF (R² = 0.998) (3.37) \\ (25.75) (1.90) \end{cases}$ $PS = 16.5992 * RMCF^{0.5835} * CIMCF^{0.3091} (R² = 0.974) (3.38) \\ (10.47) (5.93) \end{cases}$

$$\begin{cases} PS = 235414.1 + 278.039 * RCUS + 1,171.871 * CICUS (R2 = 0.993) (3.39) \\ (6.07) (1.79) \end{cases}$$

$$PS = 1691.0924 * RCUS^{0.5960} * CICUS^{0.3527} (R2 = 0.971) (3.40) \\ (8.32) (4.50) \end{cases}$$

The sums of the squares of the residuals are significantly smaller in the logarithmic cases. Note also that the "commercial-industrial" regression coefficients are much more significant in the latter cases. Therefore, the multiplicative models (3.38) and (3.40) are to be selected.

The CGOC cost functions are characterized by significant economies of scale effects. Such a result clearly confirms the company's contention that the cost of service varies from one community to the other. The comparison of equations (3. and (3.40) with LILCO's equations (3.18) and (3.19) reveals a significant similarity when the customers variables are concerned. With respect to gas sales, the exponent of RMCF is larger in LILCO's case, probably because of diseconomies of scale at the customer level (the average LILCO residential customer consumption is 76.93 MCF, while for CGOC it is equal to 162.50 MCF). Surprisingly, the exponent of CIMCF is smaller in LILCO's case, although the corresponding average customer consumption is about half the corresponding one for CGOC. An explanation of this apparent contradiction clearly requires further data analyses.

The sales and customers marginal cost functions are then

$$MC(RMCF) = \frac{\partial PS}{\partial RMCF} = 9.6856 * RMCF^{-0.4165} * CIMCF^{0.3091}$$
(3.41)

$$MC(CIMCF) = \frac{\partial PS}{\partial CIMCF} = 5.1315 * RMCF^{0.5835} * CIMCF^{-0.6908}$$
(3.42)

$$MC(RCUS) = \frac{\partial PS}{\partial RCUS} = 1,007.8742 * RCUS^{-0.4040} * CICUS^{0.3527}$$
(3.43)

$$MC(CICUS) = \frac{\partial PS}{\partial CICUS} = 596.4314 * RCUS^{0.5960} * CICUS^{-0.6473}$$
(3.44)

The marginal costs for the average community depicted by the data in table 3.10 are

MC(RMCF) = 1.557 \$/MCF MC(CIMCF) = 2.214 \$/MCF MC(RCUS) = 249.261 \$/residential customer MC(CICUS) = 2,063.917 \$/commercial-industrial customer

The striking feature in the above results is the fact that a marginal "commercial-industrial" MCF costs more than a marginal "residential" MCF. This counterintuitive result is here related to scale effects and to the relative sizes of the residential and commercial-industrial markets (the former is thrice as large as the latter). Consider now a community with equal-sized markets, each consuming 500,000 MCF (i.e., RMCF=CIMCF=500,000). In this case, the marginal sales costs are

MC(RMCF) = 2.367 \$/MCF MC(CIMCF) = 1.254 \$/MCF

and the traditionally expected cost ranking is observed.

Pacific Gas and Electric Company (PG&E)

PG&E is a dual gas and electric, privately owned utility providing service to the central and northern parts of California. The data in table 3.11 provide a summary of gas sales and average numbers of gas customers during the years 1978 and 1979, and those in table 3.12 present an overview of the gas plant in service at the end of these two years.

The residential and commercial markets have been characterized by a significant growth during the period 1978-1979 (2.36% for residential customers and 3.66% for commercial customers). The industrial market has experienced, during the same period, a slight decrease, due to industrial customers switching to other energy sources. The average consumptions per customer in 1979 are as follows: 90.409 MCF per residential customer, 849.339 MCF per commercial customer, and 66,654.11 MCF per industrial customer. The growth dynamics of the PG&E market will therefore permit to

| Үе | ar |
|-------------------|---|
| 1978 | 1979 |
| GAS SALES (MCF) | |
| 220,076,421 | 234,294,712 |
| 144,027,085 | 143,620,679 |
| 138,975,191 | 186,164,937 |
| 503,078,697 | 564,080,328 |
| 1,339 | 1,356 |
| 125,768,565 | 216,147,045 |
| 9,926,108 | 36,013,469 |
| 638,774,709 | 816,242,198 |
| MBER OF CUSTOMERS | |
| 2,531,755 | 2,591,507 |
| 163,117 | 169,097 |
| 2,853 | 2,793 |
| 2,697,725 | 2,763,397 |
| 1 | 1 |
| .5 | 5 |
| | Ye 1978 GAS SALES (MCF) 220,076,421 144,027,085 138,975,191 503,078,697 1,339 125,768,565 9,926,108 638,774,709 MBER OF CUSTOMERS 2,531,755 163,117 2,853 2,697,725 1 1 5 |

VOLUME OF GAS SALES AND NUMBER OF CUSTOMERS IN 1978 AND 1979 - PG&E

Sources: Annual Reports of PG&E to the California Public Utilities Commission (CPUC) - 1978 and 1979.

VALUE OF GAS PLANT IN SERVICE AT THE END OF 1978 AND 1979 PG&E

| (In Doll | ars) | | |
|--|-----------------------|-----------------|--|
| Plant Component | End of 1978 | End of 1979 | |
| OVERV | /IEW | | |
| Storage | \$ 103,974,935 | \$ 107,715,240 | |
| Transmission | 444,410,681 | 451,797,081 | |
| Distribution | 1,088,674,784 | 1,157,367,950 | |
| General | 12,736,349 | 13,241,884 | |
| Total (Excluding Production and Intangible Plant) | \$1,649,796,749 | \$1,730,122,155 | |
| DISTRIBUTI | ION PLANT | | |
| Land and Land Rights | \$ 3 ,5 51,839 | \$ 3,770,818 | |
| Structures and Improvements | 453,164 | 456,962 | |
| Mains | 522,350,801 | 551,964,541 | |
| Compressor Station Equipment | 68,185 | 65,030 | |
| Measuring and Regulating Station Equipment | 18,472,619 | 18,862,555 | |
| Services | 375,948,482 | 404,008,003 | |
| Meters | 129,106,410 | 136,917,955 | |
| House Regulators | 32,483,993 | 34,827,752 | |
| Industrial Measuring and Regulating Station Equipment | 5,071,754 | 5,330,965 | |
| Other Property on Customers' Premises | 49,637 | 45,468 | |
| Other Equipment | 1,117,900 | 1,117,901 | |
| • | | | |

Sources: Annual Reports of PG&E to the California Public Utilities Commission (CPUC) - 1978 and 1979, perform a "short-term" dynamic cost analysis.

The data in table 3.12 show that the distribution plant made up for about 66.89% of the total plant in 1979. Mains, services, and meters, in turn, made up for about 47.7%, 34.9%, and 11.8% of the distribution plant, respectively. In 1979, the changes in the distribution plant included (a) additions, valued at \$73,552,557; and (b) retirements, valued at \$4,859,391. The ratio of replacement to historical cost has been estimated by PG&E as equal to 2.79. Under the assumption that the whole retired plant is replaced, then the truly new distribution plant can be estimated at \$59,994,856 (= 73,552,557 - 2.79 * 4,859,391). The average cost of the new distribution plant per new customer (residential, commercial, and industrial sectors combined) would then be \$912.719. If the total 1979 distribution plant is considered, the historical unit costs per MCF and customer are

- 2.052 \$/MCF

- 418.82 \$/customer

If the replacement to historical costs ratio (= 2.79) is applied to the above customer cost, a figure of \$1,168.5 is obtained, higher than the "dynamic" cost of \$912.719. This result would confirm the hypothesis, presented in the first section of this chapter, that the "dynamic" costs are short-term, immediate costs (mains, services, meters) but do not include the longer term costs that the additions of new customers may call for later.

Included in the Annual Reports submitted by PG&E to the California Public Utilities Commission (CPUC) are community-level data on annual gas sales and average numbers of customers in the residential, commercial, and industrial sectors. These data are related to 94 communities with a population of 10,000 or more. These communities are regrouped into 13 geographical divisions. The gas sales and numbers of customers for those communities and for the years 1975 through 1979 are presented in appendix C. The Valuation Department of PG&E provided, for the same years, estimates of the historical and replacement costs of the distribution plant in these

communities. The historical costs, i.e., the plant in service, at the end of 1978 and 1979 are also presented in appendix C, together with the mileage of distribution mains at the same periods. A complete set of population and acreage data could be prepared on the basis of the 1970 Census of Population documentation and is also presented in appendix C. Completely new parameters considered in this analysis are the total annual and peak-month average number of heating degree-days. Indeed, the service territory of PG&E is climatologically heterogeneous, and the same customer is likely to consume more or less gas annually as well as during the peak month, depending upon where he is located. The data used to prepare 30year average figures for total annual and peak-month-heating degree-days are presented in appendix C. They refer to meteorological stations located in various divisions. When a division includes more than one station, the average value is selected. Then, the divisions total annual (DDT) and peak-month (DDM) figures are assigned to the communities located in the corresponding divisions.

The long-term, or static, econometric analysis has been performed with the 1979 data, and the short-term, dynamic analysis has been performed by taking the difference between the 1979 and 1978 data on both plant and market variables.

a. The Static Analysis

The definitions and means and standard deviations of the variables used in this analysis are presented in table 3.13.

In a first stage, the distribution plant in service was regressed on the aggregate sales or total number of customers, on the population density, and on the two degree-day measures alternatively. The variable DDT appeared with the wrong sign and also was highly insignificant. Therefore, only the specifications incorporating DDM were retained. In all cases, the multiplicative model appeared very superior to the additive one, and therefore only results pertaining to the former are presented. Also, in order to evaluate the impact of the introduction of the variable DDM on the

DEFINITIONS, MEANS, AND STANDARD DEVIATIONS OF THE VARIABLES - PG&E STATIC ANALYSIS

| Variable | Definition | Mean | Standard Deviation |
|----------|--|-----------|-----------------------|
| PS | Distribution Plant in Service - End of 1979 (\$) | 7,384,459 | 10,184,724 |
| TMCF | Total Gas Sales (MCF) - 1979 | 3,454,267 | 6,229,610 |
| RMCF | Residential Gas Sales (MCF) - 1979 | 1,742,436 | 2,979,162 |
| CMCF | Commercial Gas Sales (MCF) - 1979 | 969,256 | 2,586,112 |
| IMCF | Industrial Gas Sales (MCF) - 1979 | 742,575 | 1,738,055 |
| CIMCF | Commercial and Industrial Gas Sales (MCF) - 1979 | 1,711,831 | 4,205,098 |
| TCUS | Total Number of Customers - 1979 | 21,139 | 36,008 |
| RCUS | Number of Residential Customers - 1979 | 19,800 | 33,841 |
| CCUS | Number of Commercial Customers - 1979 | 1,321 | 2,188 |
| ICUS | Number of Industrial Customers - 1979 | 18 | 32 |
| CICUS | Number of Commercial and Industrial Customers - 1979 | 1,339 | 2,218 |
| TEDN | Population Density (people per acre) | 5.96 | 3.64 |
| DDM | Peak Month Average Number of Degree-Days | 534.81 | 56.30 |
| DDT | Annual Average Number of Degree-Days | 2796.96 | 374.22 |

Source: Author's calculations.

models specifications and to permit a comparison with the corresponding models derived for the other companies (for which no meteorological variability is considered), the results with an without DDM are presented. They are

$$PS = 229.1817 * TMCF^{0.7072} * TEDN^{-0.1328} (R^2 = 0.751)$$
(3.45)
(16.40) (1.79)

$$PS = 21.2111 * TMCF^{0.7082} * TEDN^{-0.1143} * DDM^{0.3720}$$
(R² = 0.753)
(16.40) (1.48) (0.89) (3.46)

$$PS = 1211.915 * TCUS^{0.9289} * TEDN^{-0.2879} (R^2 = 0.937) (3.47)$$

$$(3.47)$$

$$PS = 116.356 * TCUS^{0.9299} * TEDN^{-0.2697} * DDM^{0.3671} (R^{2} = 0.939)$$
(37.05) (6.87) (1.78) (3.48)

All the coefficients are significant at, at least, the 5% level, with the exception of the coefficient of DDM in equation (3.46), the significance of which is in the 15-20% range. It is notable that the exponents of the density variable in equations (3.45) and (3.47) are very close to those obtained in the LILCO analysis (see equations 3.10 and 3.11), where they are equal to - 0.1514 and -0.3366. Also remarkable is the similarity of the exponents of TCUS in the cases of LILCO, CGOC, and PG&E. (See equations 3.11 and 3.33.) If the PG&E and LILCO customer-related cost functions (equations 3.47 and 3.11) are adjusted for the average PG&E population density (= 5.96), they become

$$PG\&E: PS = 724.92 * TCUS^{0.9289}$$
 (3.49)

LILCO:
$$PS = 797.33 * TCUS^{0.9141}$$
 (3.50)

The above equations, when compared with the corresponding CGOC equation (3.33)

CGOC:
$$PS = 810.69 * TCUS^{0.9183}$$
 (3.51)

show a considerable degree of similarity. Therefore, it can be concluded that distribution plant costs are uniformly characterized by the same level of economies of scale when market size is measured by the total number of customers. When market size is measured by total gas sales, it appears that PG&E is characterized by larger economies of scale than CGOC and LILCO.

The marginal cost functions derived from equations (3.46) and (3.48) are

$$MC(TMCF) = \frac{\partial PS}{\partial TMCF} = 15.0217 * TMCF^{-0.2918} * TEDN^{-0.1143} * DDM^{0.3720}$$
(3.52)

$$MC(TCUS) = \frac{\partial PS}{\partial TCUS} = 108.2036 * TCUS^{-0.0701} * TEDN^{-0.2697} * DDM^{0.3671} (3.53)$$

The marginal costs for the hypothetical average community characterized by the average figures in table 3.13 are

MC(TMCF) = 1.567 \$/MCF MC(TCUS) = 333.97 \$/customer

The next step of the analysis was to use sectoral sales and numbers of customers as independent variables. In order to permit comparisons with the LILCO and CGOC models, the commercial and industrial variables were pooled together in a first stage. In the second stage, the disaggregated data were used. Again, the models specifications with and without the degree-day variable DDM are presented (the variable DDT turned out to be insignificant and was discarded). The first-stage results are

$$PS = 25.5817 * RMCF^{0.8402} * CIMCF^{0.0696} * TEDN^{-0.2514} (R^{2} = 0.925) (21.32) (2.96) (5.93) (3.54)$$

$$PS = 0.1926 * RMCF^{0.8632} * CIMCF^{0.0565} * TEDN^{-0.2192} * DDM^{0.7474} (22.72) (2.50) (5.30) (3.34) (R^{2} = 0.932) (3.55)$$

$$PS = 1700.1759 * RCUS^{0.8353} * CICUS^{0.0898} * TEDN^{-0.2844} (R^{2} = 0.937)$$
(3.56)

$$PS = 181.1541 * RCUS^{0.8492} * CICUS^{0.0772} * TEDN^{-0.2682} * DDM^{0.3451}$$

$$(15.27) (1.59) (6.80) (1.64)$$

$$(R^{2} = 0.939) (3.57)$$

All the regression coefficients in the above equations are significant at the 5% level. When the PG&E models are compared to the LILCO and CGOC models (see equations 3.18, 3.19, 3.38, and 3.40), it appears that they are characterized by lesser economies of scale with respect to the residential sales or number of customers, but by considerably larger economies of scale with respect to the commercial-industrial variables. The density elasticity is larger than LILCO's when sales are considered (-0.1765) and smaller when the variable is the number of customers (-0.3106). The significance of the density variable is here very high, and so is the significance of the degree-days variable DDM in the sales-related specification. (This variable is still significant at the 5% level in the customers-related specification.) The latter result clearly confirms the importance of the weather factor in the determination of the appropriate capacity of the distribution system.

The marginal cost functions derived from equations (3.55) and (3.57) are

$$MC(RMCF) = \frac{\partial PS}{\partial RMCF} = 0.1662 * RMCF^{-0.1368} * CIMCF^{0.0565} * TEDN^{-0.2192} *$$

$$DDM^{0.7474}$$
 (3.58)

$$MC(CIMCF) = \frac{\partial PS}{\partial CIMCF} = 0.0109 * RMCF^{0.8632} * CIMCF^{-0.9435} * TEDN^{-0.2192} * DDM^{0.7474}$$
(3.59)

 $MC(RCUS) = \frac{\partial PS}{\partial RCUS} = 153.837 * RCUS^{-0.1508} * CICUS^{0.0772} * TEDN^{-0.2682} * DDM^{0.3451}$ (3.60)

$$MC(CICUS) = \frac{\partial PS}{\partial CICUS} = 13.9906 * RCUS^{0.8492} * CICUS^{-0.9228} * TEDN^{-0.2682} * DDM^{0.3451}$$
(3.61)

The marginal costs for the hypothetical average community characterized by the average figures in table 3.13 are

MC(RMCF) = 3.874 \$/MCF
MC(CIMCF) = 0.258 \$/MCF
MC(RCUS) = 326.751 \$/residential customer
MC(CICUS) = 439.417 \$/commercial-industrial customer

The above sectoral sales marginal costs should be compared to the total sales marginal cost, $\overline{\text{MC}}(\text{TMCF}) = 1.567$ \$/MCF. Using the latter in a pricing policy would lead to a considerable subsidization of the residential customers by the commercial-industrial ones. Whereas the residential marginal cost is in the same value range as those estimated for the average LILCO and CGOC communities, it should be noted that the commercial-industrial PG&E marginal cost is much smaller than those of LILCO and CGOC. As there are no major interutility differences as far as customer size is concerned, such a difference is probably due to (a) a higher load factor for PG&E customers, and (b) local circumstances, such as the location of these customers within the community. Clearly, additional research is necessary to provide more definite explanations about these differences.

When using the disaggregated commercial and industrial variables, the results turned out to be acceptable only with the sales variables. Indeed, whenever used, the number of industrial customers turned out to be statistically insignificant and with the wrong sign. The sales-related models, with and without the peak-month degree-day variable DDM, are $PS = 30.1581 * RMCF^{0.8434} * CMCF^{0.0437} * IMCF^{0.0140} * TEDN^{-0.2494}$ (21.35) (1.37) (2.05) (5.96)

$$(R^2 = 0.926)$$
 (3.62)

 $PS = 0.2152 * RMCF^{0.8575} * CMCF^{0.0432} * IMCF^{0.0115} * TEDN^{-0.2160} * (22.61) (1.43) (1.76) (5.31)$

 $DDM^{0.7530} (R^2 = 0.935) (3.63) (3.45)$

All the regression coefficients are significant at the 5% level, with the exception of the commercial sales coefficient that is, nevertheless, significant at the 10% level. The exponents of RMCF, TEDN, and DDM are very close to those obtained when using the aggregate commercialindustrial sales variable. (See equation 3.55.) In the present model, the commercial sales elasticity (0.043) is about four times larger than the industrial one (0.011). This large difference can be explained by (a) the higher load factor of industrial customers, which are much less sensitive to weather than the commercial ones, and (b) customer-level economies of scale related to customer size. Indeed, the average commercial customer consumption, based on the data in table 3.13, is 734 MCF, whereas the corresponding industrial one is 41,254 MCF. Obviously, the above model might be further improved by introducing customer-size variables. Such an analysis is left for further research efforts.

The marginal cost functions derived from equation (3.63) are

 $MC(RMCF) = \frac{\partial PS}{\partial RMCF} = 0.1845 * RMCF^{-0.1425} * CMCF^{0.0432} * IMCF^{0.0115} *$

$$\text{TEDN}^{-0.2160} \text{* DDM}^{0.7530} \tag{3.64}$$

$$MC(CMCF) = \frac{\partial PS}{\partial CMCF} = 0.0093 * RMCF^{0.8575} * CMCF^{-0.9568} * IMCF^{0.0115} *$$

 $\text{TEDN}^{-0.2160} * \text{DDM}^{0.7530}$ (3.65)

$$MC(IMCF) = \frac{\partial PS}{\partial IMCF} = 0.0025 * RMCF^{0.8575} * CMCF^{0.0432} * IMCF^{-0.9885} *$$

 $\text{TEDN}^{-0.2160} * \text{DDM}^{0.7530}$ (3.66)

The marginal costs for the hypothetical average community characterized by the average figures in table 3.13 are then

 $\overline{MC}(RMCF) = 3.889 \text{ $/MCF}$ $\overline{MC}(CMCF) = 0.352 \text{ $/MCF}$ $\overline{MC}(IMCF) = 0.122 \text{ $/MCF}$

The above values should be compared to those obtained with the aggregate commercial-industrial sales ($\overline{MC}(RMCF) = 3.874$ \$/MCF, and $\overline{MC}(CIMCF) = 0.258$ \$/MCF). The residential marginal costs are nearly the same. However, the commercial marginal cost is about thrice the industrial one, and therefore basing a pricing policy on the aggregate commercial-industrial marginal cost would lead to a substantial subsidization of the commercial sector by the industrial one.

b. The Dynamic Analysis

Among the 94 communities analyzed in the previous section, 89 were characterized by a growth in both the residential sector and the combined commercial-industrial one, and 4 by a growth in the residential sector only. Because the sample in the latter case is too small, the following analysis only pertains to the 89 communities. The commercial and industrial sectors were combined because the results derived with the disaggregated data were not acceptable, basically because most of the growth has taken place in the commercial sector. This aggregation will also permit comparisons with the similar model derived for LILCO. Finally, note that the density and degree-day variables turned out to be insignificant and were discarded.

The definitions of the variables and their means and standard deviations are presented in table 3.14. These data imply an average new plant cost equal to \$864/customer.

DEFINITIONS, MEANS, AND STANDARD DEVIATIONS OF THE DYNAMIC ANALYSIS VARIABLES - PG&E

| Variable | Definition | Mean | Standard Deviation |
|----------|--|---------|--------------------|
| DPS | Increase in the Distribution Plant in Service (\$) | 491,501 | 632,069 |
| DTCUS | Increase in the Total Number of Customers | 569 | 703 |
| DRCUS | Increase in the Number of Residential Customers | 518 | 651 |
| DCICUS | Increase in the Number of Commercial-Industrial Customers | 51 | 64 |

Source: Author's calculations.

In a first stage, the increase in plant in service DPS was regressed on the increase in the total number of customers. The linear and logarithmic specifications are

DPS = 82,120.87 + 718.636 * DTCUS (
$$R^2 = 0.639$$
) (3.67)
(12.41)

$$DPS = 9,790.874 * DTCUS^{0.6004} (R^2 = 0.521)$$
(3.68)
(9.73)

The sums of the squares of the residuals are equal to 140.53 for equation (3.67) and to 44.44 for equation (3.68). The logarithmic model (3.68) is therefore to be selected. It is characterized by stronger economies of scale than in the case of the static approach. (See equation 3.47.) The corresponding marginal cost function is

$$MC(DTCUS) = \frac{\partial DPS}{\partial DTCUS} = 5879.0125 * DTCUS^{-0.3995}$$
(3.69)

The marginal cost for the average growth community characterized by the figures in table 3.14 (DTCUS = 569) is

$\overline{MC}(DTCUS) = 466.146$ \$/customer.

The static marginal cost computed in the static analysis was equal to 333.97 \$/customer. Using the replacement to historical costs ratio of 2.79, this static marginal cost is then equal to \$931.78 at current 1979 costs. As expected, the short-term, dynamic marginal cost is significantly smaller than the long-term one.

The next step in the analysis was to regress DPS on both residential and commercial-industrial customers increases (DRCUS and DCICUS). The linear and logarithmic specifications are

DPS = 69,756.53 + 614.579 * DRCUS + 2007.349 * DCICUS (R² = 0.629)(6.17) (1.99) (3.70)

$$DPS = 16,650.291 * DRCUS^{0.4088} * DCICUS^{0.1917} (R^2 = 0.495)$$
(3.71)
(5.91) (2.32)

The sums of the squares of the residuals are equal to 137.89 for equation (3.70) and to 46.79 for equation (3.71). Thus the multiplicative model is to be selected. It is characterized by stronger economies of scale effects in the residential sector as compared to the corresponding static model. (See equation 3.57.) However, the opposite feature characterizes the commercial-industrial sector. The marginal cost functions derived from equation (3.71) are

$$MC(DRCUS) = 6.806.334 * DRCUS^{-0.5912} * DCICUS^{0.1917}$$
(3.72)

$$MC(DCICUS) = 3,192.187 * DRCUS^{0.4088} * DCICUS^{-0.8083}$$
 (3.73)

The marginal costs for the average growth community characterized by the figures in table 3.14 (DRCUS = 518; DCICUS = 51) are then

MC(DRCUS) = 359.357 \$/residential customer

MC(DCICUS) = 1711.832 \$/commercial-industrial customer
National Fuel Gas Distribution Corporation (NFGDC)

National Fuel Gas Distribution Corporation is a privately owned gas distribution utility providing service to 471 communities in western New York, northwestern Pennsylvania, and a small portion of eastern Ohio. These communities have an aggregate population estimated, in 1979, at 2,400,000. The principal ones are Buffalo, Niagara Falls, and Jamestown, New York; and Erie and Sharon, Pennsylvania.

NFGDC is a subsidiary of the National Fuel Gas Company, a public holding company that owns 100% of NFGDC capital stock, as well as 100% of the stock of the National Fuel Gas Supply Corporation, which deals with storage and transmission, of the Seneca Resources Corporation, which deals with gas production and gasoline extraction, and of the National Gas Storage Corporation, which deals exclusively with storage. The National Fuel Gas Supply Corporation purchases about 81.5% of the gas requirements from five major interstate pipeline suppliers (see the system map in figure 3.1) and resells this gas to NFGDC. The supply balance is obtained from the purchase of synthetic gas, natural gas produced in the Appalachian area, and manufactured gas.

NFGDC is partitioned into three divisions: New York, Pennsylvania, and Ohio. The latter is extremely small. The New York division is about twice as large as the Pennsylvania one with respect to the residential and commercial markets, and slightly larger with respect to the industrial market. The New York division covers more than 5,100 square miles and has a population (1970 census) of over 1.6 million persons. As the communitylevel data used in the following analysis pertain to communities in the New York division, summary statistics are provided for both NFGDC and its New York division. The data in table 3.15 provide a summary of end-use gas sales and average numbers of customers in 1979, and those in table 3.16 present an overview of the gas plant in service at the end of 1979.

The data in table 3.16 show that the distribution plant makes up for about 85% of the total plant for both the total corporation and the New York division. Mains, services, and meter-related equipment, in turn, make up for about 65%, 21%, and 7.5% of the distribution plant. The

National Fuel Gas Company



Figure 3.1 National Fuel Gas Company System

Source: 1979 Financial and Statistical Report - National Fuel Gas Company

TABLE 3.15

VOLUME OF GAS SALES AND NUMBER OF CUSTOMERS IN 1979 - NFGDC AND ITS NEW YORK DIVISION

| | TOTAL CORPORATION | | NEW YORK DIVISION | |
|-------------|-------------------|------------------------|-------------------|------------------------|
| | Gas Sales (MCF) | Number of Customers | Gas Sales (MCF) | Number of Customers |
| Residential | 104,287,562 | 637,821 | 72,403,511 | 451,223 |
| Industrial | 67,187,351 | 1,470 | 34,286,874 | 805 |
| Total | 206,894,007 | 675,600 | 130,000,759 | 474,323 |

Sources: Annual Reports of NFGDC to the Federal Energy Regulatory Commission (FERC) and to the New York State Public Service Commission (NYPSC) - 1979.

New York division total plant represents about 72% of the total corporation plant.

The total average number of accounts has been virtually static in the past few years in the New York division, in part because of the highsaturation percentage in the residential market and the depressed economic climate of western New York. In 1979, installations of new main extensions and service lines were due principally to conversions of existing homes from oil to gas, but the largest segment of expenditures was for the replacement of mains and service lines because of obsolescence. Such a situation therefore precluded a dynamic analysis of the distribution plant.

On the basis of the data in tables 3.15 and 3.16, the distribution historical unit costs per MCF and customer for the New York division are

1.774 \$/MCF

- 486.23 \$/customer

As mentioned before, the community-level data used in the NFGDC static analysis pertain to communities located in the New York division. This is

TABLE 3.16

VALUE OF GAS PLANT IN SERVICE AT THE END OF 1979 - NFGDC AND ITS NEW YORK DIVISION

| (In | Dollars) |
|-----|-----------|
| (| v0114413/ |

| | Total Corporation | New York Division |
|--|-------------------|-------------------|
| | OVERVIEW | |
| Intangible | \$ 318,882 | \$ 195,236 |
| Production | 13,832,717 | 12,719,581 |
| Transmission | 27,268,184 | 16,511,380 |
| Distribution | 321,151,271 | 230,629,864 |
| General | 14,281,077 | 12,167,618 |
| Total | \$376,852,131 | \$272,223,679 |
| DIS | TRIBUTION PLANT | |
| Land and Land Rights | \$ 2,919,718 | \$ 1,831,204 |
| Structures and Improvements | 4,012,172 | 1,244,623 |
| Mains | 208,788,832 | 151,891,791 |
| Measuring and Regulating Station Equipment | 5,914,523 | 4,235,788 |
| Services | 69,216,169 | 51,300,074 |
| Meters | 19,870,029 | 13,842,716 |
| Meter Installation | 4,366,716 | 2,548,016 |
| House Regulators | 1,014,866 | 563,465 |
| House Regulators Installation | 1,024,251 | 727,054 |
| Industrial Measuring and Regulating Station Equipment | 3,213,289 | 2,329,333 |
| Other Equipment | 799,499 | 115,800 |

Sources: Annual Reports of NFGDC to the Federal Energy Regulatory Commission (FERC) and to the New York State Public Service Commission (NYPSC) - 1979. so because plant data are specifically prepared for these communities for submission to the New York State Board of Equalization and Assessment. Two different documents have been provided by NFGDC and used to prepare the data file: Form EA5.3 and Form EA4.3EG.

Form EA5.3 includes the following aggregate data for each community (or tax district):

- the total plant in service;
- the "personal" plant in service, including essentially meters, as well as other measuring devices and house regulators;
- the "highway" plant in service, which includes mains, regulator stations, and other equipment located on the street side of the curbs;
- the "private" plant in service, which includes mainly pipelines and regulators on the house side of the street.

Only the "personal" plant in service variable, PC, is used in the following analysis, and it is assumed that PC closely represents the investment in meters, meter installation, house regulators, and house regulator installation.

Form EA4.3EG includes, for each community, disaggregated account-level data for the production, transmission, distribution, and general plants. The distribution plant accounts do not refer to meters and house regulators, hence the use of the "personal" plant in service variable PC discussed above. Each account data are further disaggregated into "highway" and "private" plants. However, in the present analysis such differentiation has not been accounted for, and the total values only are considered. The following plant components have been included in the file:

- LAR: land and land rights;
- STI: structures and improvements;
- MAI: mains;
- MRS: measuring and regulating stations;
- SER: services.

A minor component, "Other Equipment," has been merged with the "personal" plant in service, and the resulting variable is defined as

MER: meters and house regulators

Finally, the total distribution plant PS has been defined as the sum of the above variables

$$PS = LAR + STI + MAI + MRS + SER + MER$$
(3.74)

Sales and numbers of customers data have been prepared by NFGDC staff for the specific purposes of this study. These data are related, for each community, to the residential, commercial, and industrial sectors and to a fourth category, Public Authorities, which refer to municipal, state, and federal buildings, as well as, in a very minor way, to some street lighting. In the data presented in table 3.15, the Public Authorities (P.A.) sector is combined with the commercial sector, as P.A.s display very much the same load characteristics as commercial customers (space heating is the dominant use of gas).

A complete set of plant and market data was prepared for 173 communities. These data are presented in appendix D. Population and land acreage data, however, could be prepared for 33 communities only (those with a population of 2,500 or more), on the basis of the 1970 Census of Population documentation. This smaller sample is referred to as S_2 , while the larger one is referred to as S_1 .

The means of the different variables to be used in the static NFGDC statistical analysis are presented in table 3.17 for the two samples S_1 and S_2 separately. The sample S_1 represents about 85% of the New York division distribution plant, and about 95% of the corresponding total sales and number of customers. The average distribution plant in sample S_2 is about 1.95 times larger than the corresponding plant in sample S_1 . The ratios of average sales and average numbers of customers between samples S_2 and S_1 are equal to 3.27 and 2.83, respectively. The latter figures indicate that the average gas consumption per customer is about 15% larger for sample S_2

TABLE 3.17

DEFINITIONS AND MEANS OF THE VARIABLES FOR THE TWO SAMPLES ${\rm S_1}$ AND ${\rm S_2}$ - NFGDC STATIC ANALYSIS

| Variable | Definition | Sample S ₁ (173) | Sample S ₂ (33) |
|----------|---|-----------------------------|---|
| | DISTRIBUTION PLANT (\$) - END OF 1979 | | |
| PS | Total Distribution Plant | 1,134,931 | 2,213,307 |
| LAR | Land and Land Rights | 3,267 | 7,963 |
| STI | Structures and Improvements | 5,836 | 18,811 |
| MAI | Mains | 815,216 | 1,597,039 |
| MRS | Measuring and Regulating Stations | 18,025 | 43,113 |
| SER | Services | 166,546 | 335,746 |
| MER | Meters and House Regulators | 126,041 | 210,634 |
| | MARKET DURING 1979 | | 80 9 - 1 () () () () () () () () () () () () () |
| TMCF | Total Gas Sales (MCF) | 714,926 | 2,338,234 |
| RMCF | Residential Gas Sales (MCF) | 396,903 | 1,177,212 |
| CMCF | Commercial Gas Sales (MCF) | 92,818 | 299,775 |
| IMCF | Industrial Gas Sales (MCF) | 190,636 | 755,969 |
| PMCF | Public Authorities Gas Sales (MCF) | 34,568 | 105,278 |
| CIPMCF | Total Nonresidential Gas Sales (MCF) | 318,023 | 1,161,022 |
| CPMCF | Total Commercial and Public Authorities Gas Sales (MCF) | 127,386 | 405,053 |
| TCIIS | Total Number of Customers | 2,618 | 7,419 |
| RCUS | Number of Residential Customers | 2,492 | 7.045 |
| CCUS | Number of Commercial Customers | 110 | 332 |
| ICUS | Number of Industrial Customers | 4 | 12 |
| PCUS | Number of Public Authorities Customers | 12 | 29 |
| CIPCUS | Number of Nonresidential Customers | 126 | 373 |
| CPCUS | Number of Commercial and Public Authorities Customers | 122 | 361 |
| TEDN | Population Density (people per acre) | | 6.053 |

Source: Author's calculations.

as compared to sample S1.

The major advantage of the NFGDC data is their disaggregated character. Indeed, it is now possible to develop econometric models not only for the total distribution plant, as was done for LILCO, CGOC, and PG&E, but also for the various components of this plant. The various plant variables have been regressed on aggregated and disaggregated sales and customers variables. The analyses have been carried out separately for samples S_1 and S_2 , with the intent to identify size effects eventually. (Sample S_2 refers to much larger communities.) This is, of course, only a first step in the segmentation of the market to get better models, and such a segmentation should be further considered in subsequent research efforts.

In the following, the total distribution plant and its various components are analyzed separately, and a synthesis of the results is then presented.

a. Total Distribution Plant

1. Sales-Related Analysis

Sample S₁

The multiplicative model turns out to be superior to the additive one in all cases. The results are acceptable only for (a) total sales, and (b) the two-sector market disaggregation (i.e., residential sales RMCF and nonresidential sales CIPMCF, with

 $PS = 30.3703 * TMCF^{0.7934} \qquad (R^2 = 0.819) \qquad (3.75)$ (27.83)

 $PS = 56.1046 * RMCF^{0.6920} * CIPMCF^{0.0883} (R^2 = 0.858) (3.76)$ (15.51) (2.82)

All the coefficients are highly significant (at least at the 1% level) and display the expected relative values. As in the other companies analyzed previously, economies of scale characterize distribution plant costs. The marginal cost functions derived from equations (3.75) and (3.76) are

$$MC(TMCF) = 24.0965 * TMCF^{-0.2066}$$
(3.77)

$$MC(RMCF) = 38.8227 * RMCF^{-0.3080} * CIPMCF^{0.0883}$$
(3.78)

$$MC(CIPMCF) = 4.9550 * RMCF^{0.6920} * CIPMCF^{-0.9117}$$
(3.79)

The marginal costs for the average community characterized by sample S_1 figures in table 3.17 are then

MC(TMCF) = 1.488 \$/MCF MC(RMCF) = 2.241 \$/MCF MC(CIPMCF) = 0.357 \$/MCF

The marginal total distribution cost per additional MCF is about six times larger in the residential sector, as compared to the nonresidential one, and using the total sales marginal cost would lead to a substantial subsidization of the residential customers by the commercial-industrial ones.

Sample S₂

The multiplicative model here also turns out to be the best one. The significance of the density variable is rather low. The results are therefore presented with and without this variable. As for sample S_1 , it is not possible to obtain satisfactory models beyond a two-sector market disaggregation. The results are

$$PS = 28.1441 * TMCF^{0.7775} (R^{2} = 0.784)$$
(3.80)
(10.61)
$$PS = 26.1733 * TMCF^{0.7865} * TEDN^{-0.0303} (R^{2} = 0.784)$$
(3.81)
(8.63) (0.17)

$$PS = 9.5978 * RMCF^{0.8028} * CIPMCF^{0.0965} (R^2 = 0.854)$$
(3.82)
(6.46) (1.08)

 $PS = 6.5054 * RMCF^{0.8495} * CIPMCF^{0.0969} * TEDN^{-0.1357} (R^2 = 0.858)$ (6.31) (1.08) (0.91) (3.83)

The density variable is totally insignificant in equation (3.81), and its significance level is about 20% in equation (3.83). Considering equations (3.75), (3.76), (3.80), and (3.82), one can notice that the rates of economies of scale are quite similar for both samples in the cases of TMCF and CIPMCF, while in the residential case, the rate of economies of scale is slightly smaller for S₂. Whether the previous result is an indication that lesser economies of scale are achieved in the larger communities is a somehow premature conclusion, and additional analyses are needed to confirm or invalidate this proposition.

Comparison with the Other Companies

NFGDC equations (3.75) and (3.76) should be compared with the corresponding equations for LILCO (3.10 and 3.15), for CGOC (3.31 and 3.38), and for PG&E (3.45 and 3.54). Although substantial differences exist, NFGDC equations compare best with PG&E equations, where the exponents of TMCF, RMCF, and CIMCF are equal to 0.707, 0.840, and 0.056. LILCO and CGOC display larger economies of scale in the residential sector and substantially smaller ones in the commercial-industrial sector.

2. Customers-Related Analysis

Sample S1

The multiplicative model turns out to be superior to the additive one in the case of the variable TCUS. No disaggregated model is acceptable (the sign of the variable CIPCUS is negative). The model is

$$PS = 1287.3305 * TCUS^{0.8771} (R^2 = 0.864)$$
(3.84)
(32.96)

The marginal cost function derived from equation (3.84) is

$$MC(TCUS) = 1129.1047 * TCUS^{-0.1229}$$
(3.85)

The marginal cost for the average community characterized by sample S_1 figures in table 3.17 is then

$\overline{MC}(TCUS) = 429.172$ \$/customer

Sample S₂

The multiplicative models, which again turn out to be the superior ones, are specified for both aggregated and disaggregated variables. In addition, the density variable appears to have much higher significance levels than when used with the sales variables. The results are

$$PS = 478.7315 * TCUS^{1.0080} * TEDN^{-0.2078} (R^2 = 0.871)$$
(3.86)
(12.05) (1.46)

 $PS = 767.1275 \& RCUS^{0.8092} * CIPCUS^{0.2183} * TEDN^{-0.1843}$ (R² = 0.873) (4.19) (1.05) (1.27) (3.87)

If equation (3.86) is compared to equation (3.84), it would appear that lesser economies of scale are achieved with the larger communities (sample S₂), which supports the preliminary conclusion derived in the sales models case. The density variable is significant at the 10% level, and its elasticity compares most closely with PG&E density elasticity. The marginal cost functions derived from equations (3.86) and (3.87) are

 $MC(TCUS) = 482.5829 * TCUS^{0.0080} * TEDN^{-0.2078}$ (3.88)

 $MC(RCUS) = 620.7375 * RCUS^{-0.1908} * CIPCUS^{0.2183} * TEDN^{-0.1844}$ (3.89)

 $MC(CIPCUS) = 167.4708 * RCUS^{0.8092} * CIPCUS^{-0.7817} * TEDN^{-0.1843}$ (3.90)

The above functions are then used to compute the marginal costs for the average community characterized by sample S_2 figures in table 3.17, with

MC(TCUS) = 356.64 \$/customer MC(RCUS) = 299.15 \$/residential customer MC(CIPCUS) = 1,524.37 \$/non-residential customer b. Land and Land Rights

1. Sales-Related Analysis

Sample S₁

The multiplicative models are again the superior ones. They are presented below for three cases: (1) total market, (2) two-sector market, and (3) three-sector market, with

LAR =
$$0.0000244 * \text{TMCF}^{1.2445}$$
 (R² = 0.408) (3.91)
(10.86)

LAR = $0.0001709 * \text{RMCF}^{0.9085} * \text{CIPMCF}^{0.2364}$ (R² = 0.393) (3.92) (4.43) (1.64)

LAR =
$$0.0002893 \times \text{RMCF}^{0.8219} \times \text{CPMCF}^{0.2602} \times \text{IMCF}^{0.0558}$$
 (R² = 0.400)
(3.83) (1.60) (1.11) (3.93)

All the coefficients are significant at the 5% level, but the exponent of IMCF in equation (3.93) with a significance level at about 15%. The land and land rights component is characterized by economies of scale effects with respect to sectoral sales, and the relative values of the different elasticities are as expected. However, LAR is characterized by diseconomies of scale when total sales are considered, and there is no clear explanation for this phenomenon. The marginal cost functions derived from equations (3.91) and (3.92) are

$$MC(TMCF) = 0.0000304 * TMCF^{0.2445}$$
(3.94)

$$MC(RMCF) = 0.0001553 * RMCF^{-0.0915} * CIPMCF^{0.2364}$$
(3.95)

$$MC(CIPMCF) = 0.0000404 * RMCF^{0.8219} * CIPMCF^{-0.7635}$$
(3.96)

The marginal land and land rights costs for the average community characterized by sample S_1 figures in table 3.17 are then

 $\overline{MC}(TMCF) = 0.00082 \ \text{$/MCF}$ $\overline{MC}(RMCF) = 0.00095 \ \text{$/MCF}$ $\overline{MC}(CIPMCF) = 0.00010 \ \text{$/MCF}$

Sample S₂

Interestingly, the additive models turn out to be superior in this case. However, the density variable is never significant. The acceptable models are

LAR = -513.6472 + 0.0036 * TMCF (R² = 0.934) (3.97) (20.96)

LAR = 443.4687 + 0.0061 * RMCF + 0.00028 * CIPMCF (R² = 0.985) (3.98) (23.64) (0.84)

The significance of CIPMCF is low, at the 20% level. However, the marginal costs, which are here read directly from the equations as the coefficients of the sales variables, display the expected relative values and are about four to six times larger than those obtained with sample S_1 . This would confirm the hypothesis that lesser economies of scale are achieved in the larger communities.

2. Customers-Related Analysis

The following multiplicative models are selected as both acceptable and superior to the corresponding additive ones

LAR =
$$0.0090414 * TCUS^{1.3696}$$
 (R² = 0.427) (3.99)
(11.29)

LAR =
$$0.0285463 * \text{RCUS}^{0.8532} * \text{CIPCUS}^{0.6113}$$
 (R² = 0.436) (3.100)
(2.78) (1.78)

LAR = $0.0379849 * \text{RCUS}^{0.7347} * \text{CPCUS}^{0.7392} * \text{ICUS}^{0.0068}$ (R² = 0.441) (2.34) (2.02) (0.13) (3.101) The significance of the exponent of ICUS in equation (3.101) is very low, which would imply that the number of industrial customers has very little impact on the necessary land and land rights. Equation (3.101) implies that residential and commercial customers have a similar impact on this component. Finally, note that the pattern of economies and diseconomies of scale when using aggregated and disaggregated variables is strictly similar to the pattern characterizing the sales-related models.

Sample S2

As for the sales-related models, the additive models are the superior ones, and the density variable is not significant. The acceptable models are

LAR = -137.4929 + 1.0919 * TCUS (R² = 0.991) (3.102) (57.55)

LAR = -357.2541 + 0.8796 * RCUS + 5.7123 * CIPCUS (R² = 0.991) (3.103) (3.70) (1.12)

LAR = -814.32124 + 0.6288 * RCUS + 4.9923 * CPCUS + 202.8004 * ICUS (3.05) (1.18) (3.99) (R² = 0.994) (3.104)

c. Structures and Improvements

1. Sales-Related Analysis

Sample S₁

The following multiplicative models are both acceptable and superior to the corresponding additive ones

STI = $0.0000564 * \text{TMCF}^{1.2214}$ (R² = 0.342) (3.105) (9.42) STI = $0.000321 * \text{RMCF}^{0.9217} * \text{CIPMCF}^{0.2159}$ (R² = 0.334) (3.106) (4.00) (1.33)

STI =
$$0.0003955 * \text{RMCF}^{0.8922}_{(3.68)} * \text{CPMCF}^{0.2207}_{(1.20)} * \text{IMCF}^{0.0293}_{(0.52)}$$
 (R² = 0.335)
(3.107)

All the coefficients are significant at the 10% level except the coefficient of IMCF in equation (3.107). The patterns of economies of scale with respect to sectoral sales and of diseconomies of scale with respect to total sales are strikingly similar to those pointed out for the land and land rights variable LAR. (See equations 3.91 through 3.93.) The marginal cost functions derived from equations (3.105) and (3.106) are

$$MC(TMCF) = 0.0000689 * TMCF^{0.2214}$$
(3.108)

$$MC(RMCF) = 0.0002959 * RMCF^{-0.0783} * CIPMCF^{0.2159}$$
(3.109)

$$MC(CIPMCF) = 0.0000854 * RMCF^{0.9217} * CIPMCF^{-0.7841}$$
(3.110)

The marginal structures and improvements costs for the average community characterized by $sample S_1$ figures in table 3.17 are then

 $\overline{MC}(TMCF) = 0.00136 \ \text{\$/MCF}$ $\overline{MC}(RMCF) = 0.00166 \ \text{\$/MCF}$ $\overline{MC}(CIPMCF) = 0.00060 \ \text{\$/MCF}$

Sample S₂

In this case, the additive specification turns out to be the superior one. The density variable is totally insignificant when the total sales are considered and weakly significant in the two-sector case. The results are

STI =
$$-1176.93 + 0.0085 * TMCF$$
 ($R^2 = 0.936$) (3.111)
(21.31)
STI = $2272.228 + 0.0148 * RMCF + 0.0003 * CIPMCF - 206.0199 * TEDN(31.89)$ (0.60) (0.84)

$$(R^2 = 0.992)$$
 (3.112)

The residential and nonresidential marginal costs, read directly from the equations, are about nine times larger and two times smaller than the corresponding costs in the case of sample S_1 , with the total marginal cost being about six times larger than in the case of sample S_1 . Overall, these results continue to support the hypothesis that lesser economies of scale are achieved in the larger communities.

2. Customers-Related Analysis

Sample S₁

The following multiplicative models are selected as both acceptable and superior to the corresponding additive ones

STI =
$$0.01713 * TCUS^{1.3585}$$
 (R² = 0.365) (3.113)
(9.91)
STI = $0.08913 * RCUS \frac{0.5854}{(1.70)} * CIPCUS^{0.9221}$ (R² = 0.383) (3.114)
(2.39)

It can be noted that the pattern of economies of scale with the sectoral variables and of diseconomies of scale with the aggregated variable is similar to those found out in the case of the sales models. (See equations 3.105 and 3.106.)

Sample S₂

As for the sales-related models, the additive specifications are the superior ones. In addition, the density variable turns out to be significant. The selected models are

STI = 2226.698 + 2.6239 * TCUS - 475.9217 * TEDN ($R^2 = 0.993$) (3.115) (60.47) (2.20) STI = 2074.321 + 2.5026 * RCUS + 5.2304 * CIPCUS - 470.554 * TEDN(5.26) (0.51) (2.13) ($R^2 = 0.994$) (3.116)

$$STI = 786.6148 + 2.0919 * RCUS + 4.6038 * CPCUS + 304.2455 * ICUS (4.54) (0.49) (2.67) - 362.1987 * TEDN (3.117) (1.76)$$

The significance of the commercial customers variable is very low, and it may be inferred from the above that in the nonresidential sector, the major impact on STI is related to industrial customers.

d. Mains

1. Sales-Related Analysis

Sample S1

The multiplicative specification is again the superior one. The selected models are

MAI =
$$1.0707 * \text{TMCF}^{1.0296}$$
 (R² = 0.615) (3.118)
(16.51)

MAI =
$$1.84172 * \text{RMCF}^{0.9211} * \text{CIPMCF}^{0.1175}$$
 (R² = 0.677) (3.119)
(9.14) (1.66)

All the coefficients are significant at the 5% level. The relative values of the residential and nonresidential sales elasticities are as expected and imply economies of scale with respect to sectoral sales. However, mains are characterized by very slight diseconomies of scale when total sales are considered, as was the case with the variables LAR and STI. The marginal cost functions derived from equations (3.118) and (3.119) are

$$MC(TMCF) = 1.1398 * TMCF^{0.0296}$$
(3.120)

$$MC(RMCF) = 1.6964 * RMCF^{-0.0789} * CIPMCF^{0.1175}$$
(3.121)

$$MC(CIPMCF) = 0.2165 * RMCF^{0.9211} * CIPMCF^{-0.8824}$$
(3.122)

The marginal mains costs for the average community characterized by sample S_1 figures in table 3.17 are then

 $\overline{MC}(TMCF) = 1.69868 \ \text{$/MCF}$ $\overline{MC}(RMCF) = 2.71985 \ \text{$/MCF}$ $\overline{MC}(CIPMCF) = 0.43319 \ \text{$/MCF}$

Sample S₂

The multiplicative specification is the superior one. The significance of the density variable is low in the case of the aggregate sales model, and therefore results are given with and without this variable. In the case of the two-sector model, the density variable has a significance level of 15%. The results are

MAI =
$$17.9452 * \text{TMCF}^{0.7836}$$
 (R² = 0.759) (3.123)
(9.88)
MAI = $15.1248 * \text{TMCF}^{0.8048} * \text{TEDN}^{-0.0714}$ (R² = 0.760) (3.124)
(8.17) (0.37)

MAI = $3.7767 * \text{RMCF}^{0.8613} * \text{CIPMCF}^{0.1041} * \text{TEDN}^{-0.1770}$ (R² = 0.831) (5.73) (1.04) (1.06) (3.125)

The comparison of equations (3.125) and (3.119) indicates that a higher rate of economies of scale with respect to residential sales characterizes the larger communities, while this feature is reversed in the case of nonresidential sales. With respect to total sales, the rate of economies of scale is significantly larger in the present case. Therefore, the present results would tend to support the hypothesis that larger economies of scale can be achieved in the larger communities.

2. Customers-Related Analysis

Sample S₁

The only acceptable model is the multiplicative one with the aggregate number of customers

$$MAI = 134.426 * TCUS^{1.1483} (R^2 = 0.660) (3.126)$$
(3.126)

The above model displays the same slight diseconomies of scale as the sales-related model. (See equation 3.118.)

Sample S₂

The acceptable models are

$$MAI = 298.3278 * TCUS^{1.0302} * TEDN^{-0.2516} (R^2 = 0.845)$$
(3.127)
(10.97) (1.57)

 $MAI = 456.8266 * RCUS^{0.8628} * CIPCUS^{0.1807} * TEDN^{-0.2322}$ (R² = 0.846) (3.95) (0.77) (1.42) (3.128)

The significance of the variable CIPCUS in equation (3.128) is rather low (about 20%). The significance of the density variable TEDN is higher (less than 10%). The sectoral elasticities are in the same ranges as those of the sales-related model. (See equation 3.125.)

e. Measuring and Regulating Stations

1. Sales-Related Analysis

Sample S₁

The multiplicative specification is the superior one, and the selected models are

$$MRS = 0.0000946 * TMCF^{1.3809} (R^2 = 0.422)$$
(3.129)
(11.17)

$$MRS = 0.0011062 * RMCF^{0.8799} * CIPMCF^{0.3745} (R^{2} = 0.419)$$
(3.130)
(4.02) (2.43)

 $MRS = 0.0019346 * RMCF^{0.8316} * CPMCF^{0.3435} * IMCF^{0.0789} (R^2 = 0.423)$ (3.62) (1.97) (1.47) (3.131)

All the coefficients are significant at, at least, the 10% level. The relative values of the sectoral elasticities are as expected, and all imply economies of scale with respect to sectoral sales. However, the pattern of diseconomies of scale with respect to total sales, noticed for the previous components, is also present here. The marginal cost functions derived from equations (3.129) and (3.130) are

$$MC(TMCF) = 0.0001306 * TMCF^{0.3809}$$
(3.132)

$$MC(RMCF) = 0.0009734 * RMCF^{-0.1201} * CIPMCF^{0.3745}$$
(3.133)

$$MC(CIPMCF) = 0.0004143 * RMCF^{0.8316} * CIPMCF^{-0.6255}$$
(3.134)

The marginal measuring and regulating stations costs for the average community characterized by sample S_1 figures in table 3.17 are then

 $MC(TMCF) = 0.02219 \ \text{$/MCF}$ $MC(RMCF) = 0.02381 \ \text{$/MCF}$ $MC(CIPMCF) = 0.00678 \ \text{$/MCF}$

Sample S₂

Again the multiplicative specification is the superior one. The density variable is significant at the 15% level in the total sales model, and at the 2.5% level in the two-sector sales model. The selected models are

 $MRS = 0.63398 * TMCF^{0.7880} * TEDN^{-0.1768} (R^2 = 0.765)$ (3.135) (8.62) (0.99)

 $MRS = 0.13818 * RMCF^{0.8696} * CIPMCF^{0.0895} * TEDN^{-0.2925} (R^2 = 0.858)$ (6.71) (1.04) (2.04)
(3.136)

The comparison of equations (3.136) and (3.130) points out a much higher rate of economies of scale with nonresidential sales in the larger communities, whereas this rate does not change significantly with respect to residential sales. These sectoral effects lead to overall higher economies of scale in the larger communities when total sales are considered. Similar to the main-related analysis in the previous section, these results tend to support the hypothesis that larger economies of scale can be achieved in the larger communities.

2. Customers-Related Analysis

Sample S1

The acceptable models are

$$MRS = 0.0597 * TCUS^{1.5384} (R^2 = 0.452)$$
(3.137)
(11.87)

 $MRS = 0.3644 * RCUS^{0.7032} * CIPCUS^{0.9899} (R^2 = 0.469)$ (3.138) (2.16) (2.72)

 $MRS = 0.3980 * RCUS^{0.7053} * CPCUS^{0.9411} * ICUS^{0.0210} (R^2 = 0.468)$ (2.11) (2.41) (0.39) (3.139)

The significance of the coefficient of ICUS in equation (3.139) is very low, and it therefore appears that in the nonresidential sector, the major impact on MRS is related to commercial customers.

Sample S₂

The acceptable models are

$$MRS = 11.8009 * TCUS^{1.0081} * TEDN^{-0.3526} (R^2 = 0.857)$$
(3.140)
(11.90) (2.45)

 $MRS = 21.4219 * RCUS^{0.7300} * CIPCUS^{0.3118} * TEDN^{-0.3190} (R^2 = 0.863)$ (3.79) (1.50) (2.20) (3.141) The significance of the density variable is quite high (2.5% level), as it was the case in the sales-related model (3.136). Larger economies of scale are achieved in the larger communities with respect to nonresidential customers, and this confirms the pattern noticed in the sales-related models.

f. Services

1. Sales-Related Analysis

Sample S1

The acceptable models are

SER =
$$0.4871 * \text{TMCF}^{0.9670}$$
 (R² = 0.813) (3.142)
(27.26)

SER =
$$1.6691 * \text{RMCF} 0.7195 * \text{CIPMCF} 0.1966$$
 (R² = 0.834) (3.143)
(12.20) (4.74)

All the coefficients are highly significant (at the 0.01% level) and imply economies of scale. The relative values of the residential and nonresidential elasticities are as expected. The comparison of equation (3.143) with the similar equation for mains (3.119) shows that the services components are characterized by higher economies of scale in the residential sector and lesser ones in the commercial-industrial sector. The marginal cost functions derived from equations (3.142) and (3.143) are

$$MC(TMCF) = 0.4710 * TMCF^{-0.0330}$$
(3.144)

$$MC(RMCF) = 1.2010 * RMCF^{-0.2805} * CIPMCF^{0.1966}$$
(3.145)

$$MC(CIPMCF) = 0.3281 * RMCF^{0.7195} * CIPMCF^{-0.8034}$$
 (3.146)

The marginal services costs for the average community characterized by sample S_1 figures in table 3.17 are then

MC(TMCF) = 0.30176 \$/MCF MC(RMCF) = 0.38985 \$/MCFMC(CIPMCF) = 0.13291 \$/MCF

Sample S2

The density variable turns out to be insignificant. The acceptable models are

SER =
$$6.1161 * \text{TMCF}^{0.7532}$$
 (R² = 0.690) (3.147)
(8.31)

SER =
$$2.0774 * \text{RMCF}^{0.7846} * \text{CIPMCF}^{0.0894}$$
 (R² = 0.755) (3.148)
(4.72) (0.74)

The significance of the coefficient of CIPMCF in equation (3.148) is rather low (at about the 20% level). The comparison of equations (3.147) and (3.142) indicates that a higher rate of economies of scale is achieved in the larger communities. However, the comparison of the sectoral models (3.148) and (3.143) shows that this is only true for the commercialindustrial sector, whereas slightly less economies of scale are achieved with respect to residential sales in the larger communities.

2. Customers-Related Analysis

Sample S1

The only acceptable model is the multiplicative one with the aggregate number of customers

SER =
$$49.3229 * TCUS^{1.0607}$$
 (R² = 0.844) (3.149)
(30.45)

Sample S2

Here also the only acceptable model is the aggregate multiplicative one

SER =
$$122.7594 * TCUS^{0.9038}$$
 (R² = 0.759) (3.150)
(9.89)

The comparison of equations (3.149) and (3.150) confirms the hypothesis that higher economies of scale for services are achieved in the larger communities.

g. Meters and House Regulators

1. Sales-Related Analysis

Sample S1

The acceptable models are

MER =
$$9.5988 * \text{TMCF}^{0.7079}$$
 (R² = 0.726) (3.151)
(21.31)

$$MER = 17.9205 * RMCF^{0.6138} * CIPMCF^{0.0751} (R^2 = 0.744)$$
(3.152)
(10.80) (1.88)

All the coefficients are significant at, at least, the 5% level, and imply significant economies of scale, in particular in the nonresidential sector. The resulting marginal cost functions are

$$MC(TMCF) = 6.7946 * TMCF^{-0.2921}$$
(3.153)

 $MC(RMCF) = 10.9994 * RMCF^{-0.3862} * CIPMCF^{0.0751}$ (3.154)

$$MC(CIPMCF) = 1.3460 * RMCF^{0.6138} * CIPMCF^{-0.9249}$$
(3.155)

The marginal meters and house regulators costs for the average community characterized by sample S_1 figures in table 3.17 are then

 $\overline{MC}(TMCF) = 0.13240 \text{ $/MCF}$ $\overline{MC}(RMCF) = 0.19606 \text{ $/MCF}$ $\overline{MC}(CIPMCF) = 0.02994 \text{ $/MCF}$

Sample S₂

The acceptable models are

 $MER = 2.0173 * TMCF^{0.8160} * TEDN^{-0.1370} (R^2 = 0.759)$ (3.156) (8.33) (0.72)

$$MER = 0.6852 * RMCF^{0.8053} * CIPMCF^{0.1471} * TEDN^{-0.2236} (R^2 = 0.814)$$
(5.14) (1.41) (1.29) (3.157)

The significance of the density variable is at the 20% level in equation (3.156) and at the 10% level in equation (3.157). The comparison of these equations with equations (3.151) and (3.152) indicates that lesser economies of scale are achieved in the larger communities.

2. Customers-Related Analysis

Sample S₁

The only acceptable model is the multiplicative one with the aggregate number of customers

MER =
$$275.0448 * TCUS^{0.7805}$$
 (R² = 0.762) (3.158)
(23.42)

Sample S2

In this case, both the aggregate and two-sector multiplicative models are acceptable

 $MER = 45.9453 * TCUS^{1.0282} * TEDN^{-0.3024} (R^2 = 0.828)$ (3.159) (10.45) (1.81)

 $MER = 113.1802 * RCUS^{0.5602} * CIPCUS^{0.5349} * TEDN^{-0.2440} \qquad (R^2 = 0.848)$ $(2.61) \qquad (2.31) \qquad (1.51) \qquad (3.160)$

The comparison of equations (3.158) and (3.159) confirms the hypothesis that as far as meters and regulators are concerned, lesser economies of

scale are achieved in the larger communities. Also, note that the density variable is significant at, at least, the 10% level.

h. Synthesis of the NFGDC Analysis

Some general conclusions emerge from the previous analysis.

- (1) The cost variations of the different components of the distribution plant are best explained by the multiplicative model, as was the case for the other companies. Note that the additive model, which implies constant marginal costs, is superior for two minor components (land and land rights, and structures and improvements) when sample S₂ (i.e., the larger communities) is used. It can therefore be concluded that in most instances, the cost effects of the different market sectors are nonseparable.
- (2) All the cost functions are characterized by economies of scale with respect to sectoral sales. The residential sales elasticity varies between 0.614 (meters and house regulators) and 0.922 (structures and improvements). The nonresidential sales elasticity varies between 0.075 (meters and house regulators) and 0.374 (measuring and regulating stations).
- (3) The significance of the density variable is generally low. It scores best in the cases of measuring and regulating stations, meters and house regulators, and mains. Such a disappointing result calls for further investigation. It is possible that sample S₂ is too small and not enough diversified as far as population density is concerned, and further data gathering may prove a beneficial investment.
- (4) The comparison of the results for samples S₁ and S₂ would imply that higher rates of economies of scale with respect to sectoral sales are achievable in the larger communities for mains and measuring and regulating stations. The opposite conclusion would be true for the

other distribution plant components. These results cannot be viewed as definitive but indicate that a segmentation of the communities by size and the specification of econometric models for different market segments might better account for size effects and lead to better, more accurate models.

In order to get an overview of the various models' performances the total and two-sector sales models derived with sample S_1 have been used to compute the plant components corresponding to the average figures in table 3.17 (TMCF = 714,926 MCF; RMCF = 396,903 MCF; CIPMCF = 318,023 MCF). The results are presented in Table 3.18.

TABLE 3.18

| | (In Do | ollars) | |
|--------------------|-----------------------|----------------------------|---|
| Plant Component | Total Sales Models | Two-Sector Sales Models | Sample S _l Actual Average Values |
| LAR | \$ 471 | \$ 417 | \$ 3,267 |
| STI | 797 | 715 | 5,836 |
| MAI | 1,179,518 | 1,171,998 | 815,216 |
| MRS | 11,488 | 10,742 | 18,025 |
| SER | 223,105 | 215,045 | 166,546 |
| MER | 133,727 | 126,781 | 126,041 |
| TOTAL | \$1,549,106 | \$1,525,698 | \$1,134,931 |
| PS | \$1,340,764 | \$1,285,537 | \$1,134,931 |
| Courses | Authorit colouist. | inna | |

COMPARISON OF MODEL-CALCULATED AND AVERAGE PLANT VALUES FOR THE AVERAGE NFGDC COMMUNITY

Source: Author's calculations.

The results obtained with the total and two-sector sales models are very similar, very slightly lower in the case of the two-sector sales models. The most significant differences between the model-calculated and the actual average values pertain to the mains and services costs that the models overestimate by 44% and 29%, respectively. Although the results only characterize the average community, they are indicative of the need to improve the analysis and obtain more accurate models. Another important comparison is between the sums of the values of the individual plant

components ("Total" in table 3.18) and the values obtained by using the models calibrated while using directly the total distribution plant variable PS. (See equations 3.75 and 3.76.) The sum of the individual components is about 15% to 18% larger than the value derived from the PS model. Such a result is not surprising. Indeed, a single-equation estimation procedure has been used for variables that are not independent, i.e., the sum of the individual plant components must be equal to PS. (See equation 3.74.) It is possible that the use of simultaneous-equations estimation procedures might reduce the observed discrepancies, although such a conclusion cannot be a priori taken for granted.¹⁷

Finally, the sectoral and total sales marginal costs computed in the previous sections for the average community of sample S_1 are presented in table 3.19, together with the average total sales costs computed on the basis of the data in table 3.17.

TABLE 3.19

| Plant Component | Sectoral Sales Residential Sector(1) | Marginal Costs Nonresidential Sector(2) | <u>(1)</u> (2) | Total Sales Marginal Costs | Total Sales Average Costs |
|--------------------|--|---|-------------------|----------------------------------|---------------------------------|
| LAR | 0.00095 | 0.00010 | 9.50 | 0.00082 | 0.00457 |
| STI | 0.00166 | 0.00060 | 2.77 | 0.00136 | 0.00816 |
| MAI | 2.71985 | 0.43319 | 6.28 | 1.69868 | 1.14028 |
| MRS | 0.02381 | 0.00678 | 3.51 | 0.02219 | 0.02521 |
| SER | 0.38985 | 0.13291 | 2.93 | 0.30176 | 0.23295 |
| MER | 0.19606 | 0.02994 | 6.55 | 0.13240 | 0.17630 |
| TOTAL | 3.33218 | 0.60352 | 5.52 | 2.15721 | 1.58747 |
| PS | 2.24123 | 0.37500 | 6.28 | 1.48798 | 1.58747 |

SUMMARY OF MARGINAL AND AVERAGE COSTS FOR THE AVERAGE NFGDC COMMUNITY

Source: Author's calculations.

¹⁷See, for instance: P. Rao and R.L. Miller, "Simultaneous Equations Model", in <u>Applied Econometrics</u> (Belmont CA: Wadsworth Publishing Company, Inc.), chap. 8, p. 185. As could be expected, the sums of the marginal costs computed for each plant component are larger than the marginal costs computed with the total plant (PS) equations. The ratio between residential and nonresidential marginal costs varies between 2.77 and 9.50. However, a ratio of 6 is probably best representative of this relationship when the total distribution plant is considered. When total sales are considered, marginal costs appear larger than average costs when the individual components are considered. However, this conclusion is reversed when using the total plant (PS) equation. Thus, additional analyses are called for to ascertain the relationship between marginal and average costs for the average (or any other) community.

Synthesis of the Econometric Analyses and Possible Extensions

Some major commonalities emerge from the previous analyses. Probably the most important one is the nonseparability of the distribution plant costs incurred to serve the different sectoral markets of the utility. Such a result is not surprising in view of the complex and nonseparable linkages that exist among the different customers served by the same pipeline network. The second most important commonality is related to the economies of scale achieved with respect to both residential and nonresidential gas sales. The two previous results imply that the sectoral sales marginal costs are (1) decreasing with the sector's size, and (2) depending upon the size of the other sector(s). Third, the density variable turns out to be highly significant for two companies (LILCO and PG&E), weakly so for NFGDC, and not at all for CGOC. The disappointing results for the two last companies call for additional investigations and may be related to poor quality data or to too small variations of the density variable. Finally, the PG&E analysis has demonstrated the usefulness of accounting for weather parameters when the utility's service territory is climatologically heterogeneous.

The elasticities of residential sales, nonresidential sales, and density for the four companies are presented in table 3.20. (The values for NFGDC correspond to sample S₂ analysis, where the density variable is considered.)

TABLE 3.20

| | | Non- | |
|---|-------------|-------------|------------|
| | Residential | Residential | |
| Utility | Sales | Sales | Density |
| - | Elasticity | Elasticity | Elasticity |
| Long Island Lighting Company | 0.737 | 0.154 | -0.176 |
| Columbia Gas of Ohio Company | 0.583 | 0.309 | NA |
| Pacific Gas & Electric Company | 0.840 | 0.069 | -0.251 |
| National Fuel Gas Distri- bution Corporation | 0.849 | 0.096 | -0.136 |
| | | | |

SUMMARY OF SALES AND DENSITY ELASTICITIES

Source: Author's calculations.

The data in table 3.20 point out a considerable similarity between PG&E and NFGDC with respect to the sales elasticities. LILCO and CGOC display higher economies of scale in the residential sector and lesser ones in the nonresidential sector. These interutility variations constitute an interesting and important area for further analysis and research. The variations of the sales elasticities are most likely due, in part, to variations in customers' sizes, and the introduction of these variables into the models should be tested. Another source of variations may be related to variable load characteristics (load factor, peak load) for a given sector among the different communities of the service area. These load characteristics can be determined on the basis of historical monthly consumptions and degree-days data. Such a determination for each community may be quite time consuming. However, this analysis is much more feasible at the level of company division and may permit a preliminary test of the importance of these variables.

CHAPTER 4

UTILITY COST MINIMIZATION AND MARGINAL COST PRICING EVALUATION -APPLICATION TO THE EAST OHIO GAS COMPANY

The purpose of this chapter is to present the structure of the <u>Gas</u> <u>Utility Marginal Cost Pricing Model</u> (GUMCPM) and the results of its application to the East Ohio Gas Company. In the first section, after an overview of the model's organization and logic, its different submodels are described in detail. In the next section, the assumptions used in applying the model to the East Ohio Gas Company (EOGC) are specified, the results thoroughly analyzed, and the feasibility and worth of a marginal cost pricing policy assessed. In the final section, some possible extensions of the model are outlined.

Structure of the Gas Utility Marginal Cost Pricing Model (GUMCPM)

An Overview of the Model

A general flow diagram of the model is presented in figure 4.1. The model consists of three major, interlinked blocks: (1) <u>Exogenous data and assumptions</u>, (2) <u>Average cost pricing policy</u>, and (3) <u>Marginal cost pricing policy</u>. The computer program of the model, a listing of which is presented in appendix E, is organized into a main program, where the basic data and assumptions are specified, and various subprograms - LOAD, MARCOS, DIST, REVREQ, EVAL1, EVAL2 - also indicated in figure 4.1.

The exogenous data and assumptions include (1) market-related parameters such as sectoral market growth, base and space-heating load coefficients, and price elasticities of monthly gas demands; (2) supplyrelated parameters such as maximum supplies and rates for the different available suppliers; and (3) utility-related parameters such as operating



Figure 4.1 Structure of the Gas Utility Marginal Cost Pricing Model (GUMCPM)

and capacity costs, maximum capacity expansions, the allowed rate of return, and other financial parameters (taxes, etc.).

The above data and assumptions are first used in the Average cost pricing policy block, where the monthly loads of the residential, commercial, and industrial sectors are calculated while using historically determined base and space-heating load coefficients and neglecting the price-effect component of the monthly load (demand) functions. These loads are then inputs to the utility supply, operating, and capacity costs minimization submodel that determines the optimal trade-off between supply mix and own production, storage and transmission operations, and capacity expansion decisions, subject to satisfying the above-mentioned loads and various utility-related technological constraints. The format of this cost minimization model is a linear program that yields automatically as an important by-product shadow prices for the monthly load constraints (satisfaction of demand), and these shadow prices are precisely the marginal costs incurred by marginal increases in demand. Note, however, that these marginal costs are defined only with respect to the costs considered in the linear program. Therefore, these marginal costs will have to be complemented by other marginal costs such as the distribution capacity marginal costs computed in the next step, together with the total new distribution plant. The total new plant (production, storage, transmission, distribution) is then calculated and serves as an input to the financial analysis submodel that very much replicates the financial analysis typically made in the context of rate cases. The utility's rate base is first calculated, and then so is the revenue from gas sales necessary to provide the allowed rate of return on this rate base. This revenue, divided by the total annual gas load, yields the appropriate average volumetric rate. This rate will be used as the reference rate when price effects are considered in load calculations within the Marginal cost pricing policy block. In other words, it is assumed that the price-effect components of the monthly load equations are equal to one when the monthly rates are set equal to the above average volumetric rate. (An alternative equilibrating procedure is outlined in the last section.) Finally, the average cost pricing policy is evaluated with respect to criteria such as

(1) total gas consumption, (2) peak monthly load, (3) load factor, and (4) consumers' and producer's surpluses (two measures of the overall economic efficiency of the pricing policy).

The Marginal cost pricing policy block consists of a repetition of a calculation cycle until an equilibrium between monthly gas supplies and demands is eventually reached, wherein the demands are determined by prices set equal to the total marginal costs of these demands. If this equilibrium is not reached, the calculations are terminated after a specified number of iterations. At the first iteration, the monthly sectoral loads are calculated while setting monthly rates equal to the total marginal costs derived from the Average cost pricing policy block. The resulting loads, necessarily different from those used in the Average cost pricing policy block, are then inputs to, and constraints for, the utility supply, operating, and capacity costs minimization submodel. The subsequent calculations are similar to those of the Average cost pricing policy block, up to the evaluation of the marginal cost pricing policy. The new total marginal costs corresponding to the sectoral loads are then computed and become inputs, in the next iteration, to the monthly load equations. If the new loads are equal to the loads computed in the previous iteration, it is then clear that the equilibrium has been reached. If this is not the case, the previous cycle of calculations is repeated.

The following subsections describe the structure of the different submodels and their adaptation to the specific features of the East Ohio Gas Company (EOGC) that serves the northeastern part of Ohio, including the cities of Cleveland, Akron, Canton, Warren, and Youngstown. It is the largest gas distribution utility in Ohio with respect to the number of customers: in 1977, the EOGC had 908,758 residential customers, 52,867 commercial customers, and 1,108 industrial customers. The data used have been drawn from the annual reports of the EOGC to the Federal Power Commission¹⁸ and to the Public Utilities Commission of Ohio for the period 1970-1977 or have been obtained directly from the EOGC.

¹⁸Now the Federal Energy Regulatory Commission.

The Load Analysis Submodel

The monthly gas requirements depend upon the specific characteristics and mix of the end-use customers, upon weather severity measured in degree-days, and in the longer term, upon gas prices. The EOGC's customers have been aggregated into three categories: residential, commercial, and industrial. Their observed monthly requirements (or loads) for the year 1972 have been regressed on the corresponding monthly degree-day values. This year has been selected because it was the most recent one (as from 1977) without curtailments, and therefore actual industrial usage closely approximated potential industrial requirements. The following regression models were obtained, with DGMRO_m, DGMCO_m, and DGMIO_m being defined as the residential, commercial, and industrial requirements during month m, and DD_m as the corresponding number of degree-days

$$DGMRO_{m} = 3,203.742 + 23.912 * DD_{m} (MMCF)$$
(4.1)
(R² = 0.989)

$$DGMCO_m = 1,516.625 + 9.104 * DD_m (MMCF)$$
 (4.2)
(R² = 0.989)

$$DGMIO_{m} = 10,179.264 + 3.567 * DD_{m} (MMCF)$$
(4.3)
(R² = 0.920)

The corresponding total gas requirements $DGMTO_m$ are then

$$DGMTO_m = DGMRO_m + DGMCO_m + DGMIO_m = 14,899.621 + 36.583 * DD_m$$
 (4.4)

Equations (4.1) through (4.4) are assumed to characterize the base or existing gas market throughout the following analysis. In equations (4.1) through (4.4), the first coefficient represents the base load, independent of weather, and the second one represents the space-heating load per degree-day. For an average annual number of degree-days equal to 6258, the

residential, commercial, and industrial base loads correspond to 20.5%, 24.2%, and 84.5% of the total sectoral load, respectively. For example, the residential base and space-heating loads represent 20.5% and 79.5% of the total residential load. Throughout the following analysis, 30-year average values of the monthly degree-days are used. These values, together with the corresponding base market sectoral and total loads, are presented in table 4.1.

| | | | | | · | |
|---------------------------|--|---------|-------------|--|------------|--|
| | | Average | Residential | Commercial | Industrial | Total |
| | | Degree- | Load | Load | Load | Load |
|] | Month | Days | DGMRO | DGMCO | DGMIO | DGMTO |
| | | | | | | |
| 1. | April | 506.6 | 15,317.56 | 6,128.71 | 11,986.31 | 33,432.59 |
| 2. | May | 248.2 | 9,138.70 | 3,776.24 | 11,064.60 | 23,979.53 |
| 3. | June | 50.5 | 4,411.30 | 1,976.38 | 10,359.40 | 16,747.07 |
| 4. | July | 11.0 | 3,466.77 | 1,616.77 | 10,218.50 | 15,302.05 |
| 5. | August | 18.9 | 3,655.68 | 1,688.69 | 10,246.68 | 15,591.05 |
| 6. | September | 120.5 | 6,085.14 | 2,613.66 | 10,609.09 | 19,307.88 |
| 7. | October | 371.6 | 12,089.44 | 4,899.67 | 11,504.76 | 28,493.88 |
| 8. | November | 712.6 | 20,243.44 | 8,004.14 | 12,721.11 | 40,968.68 |
| 9. | December | 1,071.6 | 28,827.85 | 11,272.47 | 14,001.66 | 54,101.98 |
| 10. | January | 1,207.7 | 32,082.27 | 12,511.53 | 14,487.13 | 59,080.92 |
| 11. | February | 1,046.3 | 28,222.87 | 11,042.14 | 13,911.42 | 53,176.43 |
| 12. | March | 892.5 | 24,545.20 | 9,641.95 | 13,362.81 | 47,549.96 |
| | and and a state of the state of | | | Bread States and a subscription of the subscri | | an selective production of the selection |
| ingen an Europerand State | Total | 6,258.0 | 188,086.22 | 75,172.34 | 144,473.48 | 407,732.04 |
| ~ | 4 . 1 | | | | | |

| TABLE | 4. | 1 |
|-------|----|---|
|-------|----|---|

AVERAGE MONTHLY DEGREE-DAYS AND SECTORAL AND TOTAL LOADS (MMCF)

Source: Author's calculations.

The next step is to introduce price effects in the monthly sectoral load functions. It is assumed that these effects interact multiplicatively with the weather-related components as described by equations (4.1) through (4.3) and are characterized by constant price elasticities. The constant elasticity assumption has been found to be appropriate at the annual level.
Nelson,¹⁹ for instance, estimated an elasticity of -0.280, and derived, through regression analysis, a predictive model for total annual energy demand of the multiplicative form, where the price and weather effects constitute two of the most significant factors. However, it seems that very little information is available on demand elasticities at the intraannual level, such as the month. In view of the resulting uncertainty, sensitivity analyses are necessary to assess the impacts of alternate elasticity values within reasonable ranges. If EL_m is the elasticity for month m, and P_m the corresponding price, the price effect PE_m is assumed to be measured by the quantity

$$PE_{m} = \left(\frac{P_{m}}{PAVG}\right)^{EL_{m}} (EL_{m} < o)$$
(4.5)

where PAVG is a reference price for which the price effect is equal to one. This reference price is, in the present approach, set equal to the average volumetric rate, enabling the utility to earn its allowed operating income exactly. (PAVG is determined in the financial analysis submodel of the <u>Average cost pricing policy</u> block.) The sectoral monthly elasticities are noted ELR_m, ELC_m, and ELI_m for the residential, commercial, and industrial sectors, respectively.

It is finally necessary to account for the change in demand due to market growth. If RMR, RMC, and RMI represent the residential, commercial, and industrial sectors growth rates, then the complete monthly sectoral load functions can be specified as

$$DGMR_{m} = DGMRO_{m} * (1 + RMR) * \left(\frac{P_{m}}{PAVG}\right)^{ELR_{m}}$$

$$DGMC_{m} = DGMCO_{m} * (1 + RMC) * \left(\frac{P_{m}}{PAVG}\right)^{ELC_{m}}$$

$$(4.6)$$

$$DGMI_{m} = DGMIO_{m} * (1 + RMI) * \left(\frac{P_{m}}{PAVG}\right)^{ELI_{m}}$$
(4.8)

¹⁹J. P. Nelson, "The Demand for Space Heating Energy," <u>The Review of</u> <u>Economics and Statistics</u>, 1975, pp. 508-12.

In the application of the model, the three market growth rates are taken equal to 0.5 (50%). As mentioned in the overview of the model, the price effects are neglected in the <u>Average cost pricing policy</u> block. This is simply done by setting initially the P_m 's (m = 1-12) and PAVG equal to 1. In the <u>Marginal cost pricing policy</u> block, the P_m 's are set equal to the corresponding monthly total marginal costs, and PAVG is taken as the average volumetric rate determined by the financial analysis submodel at the end of the Average cost pricing policy block.

The Utility Supply, Operating, and Capacity Costs Minimization Submodel

There are noticeable variations in the structure of gas distribution utilities in terms of the characteristics of their suppliers (number, maximum supplies, rate structure, take-or-pay clauses, etc.), of their own gas production, of their own storage system or of the storage they rent, and of the importance of their transmission system. It is therefore necessary to adapt the costs minimization submodel to the specific features of the utility dealt with. In the following subsections, the supply, production, storage, and transmission components of the EOGC are described, and the corresponding mathematical relationships and constraints are formulated. In the final subsection, the submodel is summarized and the objective function - the total system cost - formulated.

a. EOGC Gas Supply

From 1970 to 1977, the EOGC has purchased between 87% and 91% of its annual supply from two interstate pipeline companies: The Consolidated Gas Supply Corporation (71% to 74%) and Panhandle Eastern Pipeline Company (13.5% to 18.4%). The remainder was obtained from wellhead and field-line purchases in Ohio, as well as, to a small extent, from production by EOGC itself. The latter will be discussed in the next subsection.

In the model, the monthly purchases from Consolidated and Panhandle are noted $SUP1_m$ and $SUP2_m$ for month m, respectively. In order to keep

up with seasonal definitions and constraints, the year is defined as the period extending from April 1 to March 31 (with months numbered accordingly). It is assumed that there are limits to the total annual supplies purchasable from these two companies. These limits are noted SUP1T and SUP2T for Consolidated and Panhandle, respectively. These constraints are expressed mathematically as

 $\sum_{m=1}^{12} SUP1_m \leq SUP1T: Consolidated's annual supply (4.9)$

12 $\Sigma \text{SUP2}_m \leq \text{SUP2T: Panhandle's annual supply}$ (4.10) m=1

The rate structure of Consolidated includes a commodity charge, CCl, related to the amount actually purchased, a demand charge, DCl, related to the maximum allowable daily purchase DAYMX1, and a winter requirement charge, WRC, related to the total winter gas purchases (from November 1 to March 31). The rate structure of Panhandle, in addition to a commodity charge, CC2, and a demand charge, DC2, includes a take-or-pay clause stating that the minimum monthly bill must include a minimum commodity charge based upon 75% use of the demand contract DAYMX2. The demand contracts DAYMX1 and DAYMX2 are taken, in this analysis, as decision variables of the model. Assuming that the monthly purchases SUP1_m and SUP2_m are uniformly spread over the month, the following maximum monthly purchases constraints must hold for each month m (where N_m is the number of days in month m)

 $SUPl_m - N_m DAYMXI < 0 (m = 1 \rightarrow 12)$: Consolidated (4.11)

$$SUP2_m - N_m DAYMX2 < 0 (m = 1 \rightarrow 12)$$
: Panhandle (4.12)

The take-or-pay clause of Panhandle makes it necessary to introduce into the model a new monthly variable, $SUPV_m$, equal to the highest of (1) the actual monthly supply $SUP2_m$ and (2) 75% of the monthly equivalent of

the daily demand contract. The following constraints ensure the endogenous determination of $\ensuremath{\text{SUPV}}_m$

$$SUPV_m - SUP2_m > 0 \ (m = 1 \rightarrow 12)$$
 (4.13)

$$SUPV_m - 0.75 * N_m * DAYMX2 \ge 0 \ (m = 1 \rightarrow 12)$$
 (4.14)

The total annual cost of supply from Consolidated, CTS1, includes commodity, winter requirement, and demand costs, with

$$CTS1 = \begin{bmatrix} \sum CC1 * SUP1_m \end{bmatrix} + 12 * \begin{bmatrix} \sum WRC * SUP1_m \end{bmatrix} + 12 * DC1 * DAYMX1 \quad (4.15)$$

m=1 m=8

The total annual cost of supply from Panhandle, CTS2, includes commodity and demand costs, with

$$CTS2 = \left[\sum_{m=1}^{12} CC2 * SUPV_{m}\right] + 12 * DC2 * DAYMX2$$
(4.16)

The intrastate wellhead and field-line purchases are assumed constant over time. The monthly wellhead and field-line purchases, SUPWH and SUPFL, are limited by maximum production capacities SUPWHT and SUPFLT, hence the constraints

SUPWH < SUPWHT: maximum wellhead purchases (4.17)

The above purchases are assumed subject to commodity charges only, noted CWH and CFL. The total annual cost of wellhead and field-line purchases, CTWF, is then

$$CTWF = 12 * [CWH * SUPWH + CFL * SUPFL]$$

$$(4.19)$$

The actual rate values used in the model are those which were in effect in 1977, with

CC1 = 1,202.4 \$/MMCF DC1 = 980.0 \$/MMCF WRC = 8.075 \$/MMCF CC2 = 1,009.2 \$/MMCF DC2 = 1,860.0 \$/MMCF

The wellhead and field-line purchase costs were taken equal to the average 1977 corresponding costs, with

CWH = 787.0 \$/MMCF CFL = 1,481.0 \$/MMCF

Alternate assumptions with respect to Consolidated's and Panhandle's maximum annual supplies, SUPIT and SUP2T, will be considered in the application of the model in the next section. The assumptions with respect to maximum wellhead and field-line purchases are

> SUPWHT = 2000 MMCF/month SUPFLT = 2500 MMCF/month

b. EOGC Gas Production

The actual annual amounts of gas produced by the EOGC for the period 1972-1977 are indicated in table 4.2.

Taken on a monthly basis, this capacity is then

$$PROC = \frac{11,372}{12} = 947.667 \text{ MMCF/month}$$

| Year | Gas Production | (MMCF) |
|------|----------------|--------|
| | | |
| 1972 | 3,740 | |
| 1973 | 11,163 | |
| 1974 | 9,486 | |
| 1975 | 11,372 | |
| 1976 | 6,785 | |
| 1977 | 6,200 | |

TABLE 4.2 EOGC OWN GAS PRODUCTION

Source: EOGC's Annual Reports to the Public Utilities Commission of Ohio.

The total operating cost for production in 1977 amounted to \$5,711,000. The average 1977 unit operating cost was selected in this model, with

 $COMP = \frac{5,711,000}{6,200} = 921.129$ \$/MMCF

The 1977 historical (or book) value of the production plant amounted to \$73,299,000. In view of the fact that the production plant has been constructed quite recently, it was assumed that its 1977 replacement value would be equal to 1.5 times its historical value, or \$109,948,500. The replacement cost per unit of monthly production capacity is then equal to 116,020.22 \$/(MMCF/month). It is, however, necessary to use annualized investment cost figures in the model. These annualized figures were computed while assuming (1) an investment lifetime of 30 years, and (2) an interest rate of 12%. The ratio of the annualized cost to the total present cost is then equal to

CRF =
$$1/\left\{\frac{1}{0.12}\left[1 - \frac{1}{(1+0.12)^{-30}}\right]\right\} = 0.1241$$

The annualized production capacity investment cost is then

$$CIP = 14,398.11 \ \text{(MMCF/month)}$$

The production-related decision variables are

- the monthly production levels PR_m , and
- the additional monthly production capacity DPRO

The total annual production cost is then

$$CTP = CIP * DPRO + \sum_{m=1}^{12} COMP * PR_m$$
(4.20)

It is assumed that there is an upper limit, DPROM, to the incremental production capacity, with

DPROM = 3,000 MMCF/month

In addition, it is assumed that the utility is constrained by the regulatory authorities to supply a certain share, SHP, of new customers with its own gas. Such a constraint was actually imposed on the EOGC by the Public Utilities Commission of Ohio (PUCO) in 1978, when the EOGC applied for a relief order from the then existing moratorium on new hookups. In the present analysis, this share is taken equal to 10% of the new total annual load DDGT. The production-related constraints are then written as

DPRO
$$\leq$$
 DPROM: maximum incremental production capacity (4.21)
12
 $\sum PR_m \geq$ SHP * DDGT: minimum total production (4.22)
m=1
PR_m - DPRO \leq PROC (m = 1 \rightarrow 12): the monthly production levels are limited
by the production capacity (4.23)

c. EOGC Gas Storage

The EOGC storage system includes five storage fields, with a total certified storage capacity of 147,594.1 MMCF. From 1970 to 1977, the average annual total deliveries and withdrawals were equal to 54,125.348 MMCF and 53,614.929 MMCF, respectively, with an average gas loss of 306.947 MMCF. The total amount of cushion gas was equal to 90,937.838 MMCF at the end of 1977 (i.e., 61.6% of the certified capacity). This cushion gas must be maintained in storage at all times to ensure effective utilization of the reservoir. Additional volumes must also be injected and maintained above the cushion gas volume to establish reservoir volume and pressure conditions necessary to provide minimum withdrawal rates: in the case of the EOGC, these additional volumes correspond to 15.4% of the certified capacity.

Although the different storage fields have different capacities and porosities, only the aggregate storage capacity is considered. This simplification is acceptable inasmuch as one field makes up for 88.3% of the total capacity. The major difference between a gas storage system and a water storage system is that both injection and withdrawal maximal rates depend, at any time, upon the amount of gas stored, i.e., the reservoir pressure. The main technological difference between gas withdrawal and delivery is that compressors are required for delivery, whereas natural storage pressure is used to transfer gas out of storage into the mains. Monthly constraints on storage deliveries and withdrawals have been determined on the basis of historical data. Using monthly storage flows, the level of gas in storage at the beginning of each month m, GSTOR_m, has been determined, and a storage saturation rate, RSTOR_m, taken as a proxy for storage pressure, has been defined as

$$RSTOR_m = GSTOR_m / STCO$$
 (4.24)

where STCO is the existing certified storage capacity. The monthly deliveries and withdrawals for the period 1971-1976 were plotted against the corresponding saturation rates, and the maximum flows were approximated

by straight-line segments, as presented in figures 4.2 and 4.3. It appears that storage was active when the saturation rate was comprised between a minimum and a maximum saturation rate, $R_{min}(=.77)$ and $R_{max}(=1.18)$, respectively. The maximum deliveries and withdrawals during any month m are noted by MAXINS_m and MAXOUS_m and are defined functionally as

$$MAXINS_{m} = A_{1} * RSTOR_{m} + B_{1}$$
(MMCF) (4.25)

$$MAXOUS_{m} = A_{2} * RSTOR_{m} + B_{2}$$
(MMCF) (4.26)

with $A_1 = -11,463.415$, $B_1 = 20,726.83$, $A_2 = 22,500$, $B_2 = -9,825$. If GINST_m and GOUST_m are the actual deliveries and withdrawals during month m, if GSTOR₀ is the cushion and operational (nonwithdrawable) gas, and if it is assumed that there is no usable gas in storage at the beginning of the first month, it follows that

$$RSTOR_{m} = GSTOR_{m}/STCO = [GSTOR_{o} + \sum_{\mu=1}^{m-1} (GINST_{\mu} - GOUST_{\mu})]/STCO \qquad (4.27)$$

The above coefficients A_1 , B_1 , A_2 , B_2 characterize the existing storage system (STCO = 147,594.1 MMCF). To extend the applicability of equations (4.25) through (4.27), it is assumed that these coefficients are linear functions of the total storage capacity, STCAP, whatever the incremental storage capacity DSTC added to STCO. The coefficients A_{10} , B_{10} , A_{20} , B_{20} of these functions are specified on the basis of the existing storage characteristics. For instance

$$A_{10} = \frac{A_1}{STCO} = - \frac{11,463.415}{147,594.1} = -0.07766852$$

and the coefficient A1 is redefined as

$$A_1 = A_{10} * (STCO + DSTC)$$

(4.28)







Figure 4.3 Maximum Gas Withdrawals from Storage

The other coefficients are then

 $B_{10} = 0.14043129$ $A_{20} = 0.15244512$ $B_{20} = -0.06656770$

In addition, it is assumed that the minimum and maximum storage saturation rates, R_{min} and R_{max} , remain constant whatever the storage capacity. The maximum monthly storage deliveries and withdrawals constraints, and the maximum and minimum storage saturation constraints

$$GINST_m \leq MAXINS_m$$
 (m = 1 \rightarrow 12) (4.29)

$$GOUS T_m \leq MAXOUS_m \qquad (m = 1 \rightarrow 12) \tag{4.30}$$

$$R_{\min} \leq RSTOR_{m} \leq R_{\max} \quad (m = 1 \rightarrow 12)$$
(4.31)

are then rewritten as follows

m

$$\begin{array}{c} m-1 \\ \text{GINST}_{m} - A_{10} * \sum (\text{GINST}_{\mu} - \text{GOUST}_{\mu}) - (A_{10} * \text{Rmin} + B_{10}) * \text{DSTC} \leq \\ \mu=1 \end{array}$$

$$\begin{array}{c} (A_{10} * \text{Rmin} + B_{10}) * \text{STCO} (\text{m} = 1 \rightarrow 12): \text{ maximum delivery} \\ \text{m-1} \end{array}$$

$$\begin{array}{c} (A_{10} * \text{Rmin} + B_{10}) * \text{STCO} (\text{m} = 1 \rightarrow 12): \text{ maximum delivery} \\ \text{m-1} \end{array}$$

$$\begin{array}{c} (A_{20} * \text{Rmin} + B_{20}) * \text{STCO} (\text{m} = 1 \rightarrow 12): \text{ maximum withdrawal} \end{aligned}$$

$$\begin{array}{c} (A_{20} * \text{Rmin} + B_{20}) * \text{STCO} (\text{m} = 1 \rightarrow 12): \text{ maximum withdrawal} \end{aligned}$$

$$\begin{array}{c} (A_{20} * \text{Rmin} + B_{20}) * \text{STCO} (\text{m} = 1 \rightarrow 12): \text{ maximum withdrawal} \end{aligned}$$

$$\begin{array}{c} (A_{33}) \\ (\text{GINST}_{\mu} - \text{GOUST}_{\mu}) - (\text{Rmax} - \text{Rmin}) * \text{DSTC} \leq (\text{Rmax} - \text{Rmin}) * \text{STCO} \\ (\text{m} = 1 \rightarrow 12): \text{ maximum saturation rate} \end{aligned}$$

$$\sum_{\mu=1}^{\sum} (\text{GINST}_{\mu} - \text{GOUST}_{\mu}) \ge 0 \quad (m = 1 \rightarrow 12): \text{ minimum saturation rate} \quad (4.35)$$

Constraints (4.32) through (4.35) are derived from constraints (4.29) through (4.31) while accounting for the fact that in equation (4.27)

$$GSTOR_{o} = R_{min} * (STCO + DSTC)$$
(4.36)

In addition to the previous storage operations constraints, it is assumed that there is a limit, DSTCM, to the incremental storage capacity, hence the constraint

DSTC \langle DSTCM (4.37)

In the present analysis, DSTCM is taken equal to 100,000 MMCF.

Gas storage costs include initial capital costs, mostly related to wells, gathering lines, compressors, regulating equipment, land, etc., and annual operation and maintenance costs, mostly related to wells maintenance, compressor fuel, gas losses, storage well royalties, supervision, etc. There is much uncertainty in the estimation of new storage capital costs, which depend on an annualized basis, upon the project lifetime, discount rate, and amount of cushion and operational gas necessary to maintain adequate pressure conditions. On the basis of various data, an annual capital cost range of 32.0 \$/MMCF to 77.1 \$/MMCF was obtained, consistent with the Federal Power Commission National Gas Survey 1975 average estimate of 57.0 \$/MMCF of storage capacity. The figure selected in this study is

CIST = 50 \$/MMCF

The EOGC average annual operation and maintenance expense per MMCF of gas delivered to storage is equal to \$66.46. This cost figure, apportioned equally among deliveries and withdrawals, is used in the present study, with

CS = 33.23 \$/MMCF

The total storage investment and operation and maintenance cost, CTS, is finally

$$12 CTS = CIST * DSTC + CS * \sum_{m=1}^{12} (GINST_m + GOUST_m)$$
(4.38)

d. EOGC Gas Transmission

The EOGC tranmission mains convey gas from the points of connection with its suppliers to the distribution networks of the various communities and metropolitan areas served by the company. Many important transmission mains do so while passing through the EOGC storage system, as illustrated in figure 4.4. Abstracting from the spatial complexities of the system's network, it is assumed that the transmission system may be decomposed into two components: (1) T1, conveying gas from the suppliers to storage as well as directly to the end-use customers; and (2) T_2 , conveying gas from storage and from the suppliers to the end-use customers. This simplification of the system is illustrated in figure 4.5. Clearly, then, the capacity of T_1 is determined by the peak purchases, while the capacity of T_2 is determined by the peak sales to the end-use customers. The peak sales are, on a monthly basis, exogenously specified for the utility costs minimization model and only vary in the iterative simulation of the Marginal cost pricing policy block, where rates are iteratively readjusted equal to the total marginal costs. On the other side, the peak monthly purchases are endogenously determined by the costs minimization model and may be reduced by increasing the available storage capacity. Obviously, there is a cost trade-off between the incremental transmission and storage capacities that must be accounted for in the model.

By the end of 1977, the maximum daily sales had taken place on January 8, 1970, with an amount of 2,853.1 MMCF. At such a constant daily rate, the monthly sales would have been equal to 88,446.1 MMCF. In the following, the existing monthly capacity of T₂ is assumed to be

 $PT_{20} = 88,500 \text{ MMCF}$





109

Figure 4.5 EOGC System Approximation

The peak daily purchases have taken place on February 1, 1971, when the balance between sales and storage withdrawal/delivery was at a maximum. The sales were equal, on that day, to 2,796.0 MMCF and the withdrawal from storage to 1,025.9 MMCF. The corresponding daily purchase rate of 1,770.1 MMCF would lead to a monthly purchase rate of 54,873.1 MMCF. In the following, the existing monthly capacity of T_1 is assumed to be

 $PT_{10} = 55,000 \text{ MMCF}$

The 1977 historical (or book) value of the transmission plant amounted to \$102,837,912. In view of the age of the system, it was assumed that the 1977 replacement value of this plant would be equal to 2.5 times its historical value, or \$257,094,785. In addition, it was assumed that (1) component T_1 represents 40% of this investment and component T_2 the remainder, (2) the lifetime of a transmission investment is 30 years, and (3) the discount rate is 12%. The annualized unit expansion costs of the transmission components T_1 and T_2 are then computed as

 $CIPT_1 = \frac{0.4 * 0.1241 * 257,094,785}{55,000} = 232.0397 \ \text{\$/(MMCF/month)}$

 $CIPT_2 = \frac{0.6 * 0.1241 * 257,094,785}{88,500} = 216.30822 \ \text{\$/(MMCF/month)}$

Calculations related to component T_2 are described in the distribution plant analysis section. With respect to component T_1 , the decision variable is the incremental monthly transmission capacity DPT₁. The augmented capacity is the upper limit to monthly transmission flows, hence the constraint

$$SUPl_m + SUP2_m + SUPWH + SUPFL + PR_m - DPT_1 < PT_{10} (m = 1 \rightarrow 12)$$
 (4.39)

Finally, the annualized transmission capacity expansion cost is defined as

$$CTPT_1 = CIPT_1 * DPT_1 \tag{4.40}$$

The transmission operating costs are considered later, together with the distribution and other operating costs, and are taken as proportional to the end-use sales.

e. EOGC Gas Requirements

The monthly total loads computed in the load analysis submodel, DGMT_m , must always be satisfied, hence the supply-demand equality constraints

 $SUP1_m + SUP2_m + SUPWH + SUPFL + PR_m - GINST_m + GOUST_m = DGMT_m$ (4.41) (m = 1 \rightarrow 12)

The shadow prices of these constraints are noted MC_m . They are precisely equal to the marginal costs incurred by an increase of one unit of demand during any month m. Note, however, that these marginal costs refer only to the costs considered in this linear program (supply, production operations and investment, storage operations and investment, and transmission investment). Therefore, they do not constitute the total marginal costs relevant to marginal cost pricing policy and will be complemented by other investment and operations marginal costs later.

f. Summary of the Utility Costs Minimization Submodel

The linear program described in the previous subsections is made of

- 79 decision variables: $SUPl_m$ (m = 1 \rightarrow 12); $SUP2_m$ (m = 1 \rightarrow 12); $SUPV_m$ (m = 1 \rightarrow 12); DAYMX1; DAYMX2; SUPWH; SUPFL; DPRO; PR_m (m = 1 \rightarrow 12); GINST_m (m = 1 \rightarrow 12); GOUST_m (m = 1 \rightarrow 12); DSTC; DPT₁

- 139 constraints: maximum monthly and annual purchases from Consolidated and Panhandle; maximum wellhead and field-line purchases; maximum production capacity expansion; minimum total annual production; maximum monthly production rate; maximum storage capacity expansion; maximum monthly storage deliveries and withdrawals; maximum monthly transmission flows; and monthly gas loads provision.

The objective function of the program is the sum of all the costs considered, i.e., purchases, production, storage, and transmission costs, with

$$CT = CTS1 + CTS2 + CTWF + CTP + CTS + CTPT_1$$
(4.42)

The linear program has been solved by the code LPCODE developed by Professor C. H. Martin, Department of Industrial and Systems Engineering, The Ohio State University.

The Distribution Plant Capacity Submodel

This submodel determines the incremental capacities necessary to accommodate the peak sales to the end-use customers. As such, it deals with (1) the component T₂ of the transmission system, as described in the costs minimization submodel; and (2) the distribution system. For both cases, it is first necessary to determine the peak sales month m_p . (This peak month is initially January ($m_p = 10$) but is likely to change with changes in the rate structure.)

a. Case of the Transmission Component T₂

If the peak sales $DGMT_{m_p}$ are smaller than the existing transmission capacity PT_{20} (= 88,500 MMCF/month), then there is no need for expanding component T₂, and the corresponding marginal capacity cost, $CMPT_2$, and present value of the incremental plant, NPT₂, are both equal to zero. In the other case, it follows that

$$CMPT_{2} = \begin{cases} CIPT_{2} = 216.30822 \text{ $/MMCF during month } m_{p} \\ 0 \text{ during all the other months} \end{cases}$$
(4.43)

and

$$NPT_2 = CIPT_2 * (DGMT_{m_p} - PT_{20})/CRF$$
 (4.44)

b. Case of the Distribution Plant

The existing capacity of the distribution network has been assumed equal to the January peak sales of the base gas market (see table 4.1), with

 $PD_0 = 59,081 \text{ MMCF/month}$

The above figure is significantly lower than the capacity of transmission component T_2 (PT₂₀ = 88,500 MMCF). This is so because it can be reasonably assumed that with the increased use of compressors, the capacity of the transmission system can be easily increased, and this is much less the case for smaller distribution mains. In addition, note that, even if there was some excess capacity in some distribution mains, the taking of new customers necessarily requires such new investments as meters and services. Obviously, the above assumption is an approximation and should be submitted to a sensitivity analysis.

The 1977 historical (or book) value of the EOGC's distribution plant amounted to \$372,284,403. The breakdown of this plant into various components is indicated in table 4.3. In view of the age of the system, it was assumed that the 1977 replacement value of this plant is equal to 2.5 times

| Plant Category | Historical Value S | |
|----------------------|-----------------------|--|
| fiant Gategory | Varue 9 | |
| Land and Land Rights | 2,105,000 | |
| Structures | 11,755,000 | |
| Mains | 264,543,000 | |
| Regulators | 9,935,000 | |
| Services | 50,723,000 | |
| Meters | 28,618,000 | |
| Other | 4,604,000 | |
| | | |

EOGC DISTRIBUTION PLANT

Source: EOGC's 1977 Annual Report to the PUCO.

its historical value, or \$930,711,000. Under the assumptions that (1) the lifetime of a distribution investment is 30 years and (2) the discount rate is 12%, the annualized unit expansion cost of the distribution system is then computed as

$$CIPD = \frac{0.1241 * 930,711,000}{59,081} = 1954.964 \ \text{(MMCF/month)}$$

If the peak sales $DGMT_{m_p}$ are smaller than the existing distribution capacity PD₀ (= 59,081 MMCF/month), then there is no need for expanding the distribution system, and the corresponding marginal capacity cost, CMPD, and the present value of the incremental plant, NPD, are both equal to zero. In the other case, it follows that

$$CMPD = \begin{cases} CIPD = 1954.964 \text{ $/MMCF during month } m_p \\ 0 \text{ during all the other months} \end{cases}$$
(4.45)

and

$$NPD = CIPD * (DGMT_{m_p} - PD_0)/CRF$$
(4.46)

Total New Plant and Marginal Costs Calculations

The total cost CT (see equation 4.42) minimized in the utility supply, operating, and capacity costs minimization submodel includes both operating and annualized investment costs, noted OMC_1 and PIS_1 , respectively. The investment costs PIS_1 include the production, storage, and transmission capacity costs, with

 $PIS_1 = CIP * DPRO + CIST * DSTC + CIPT_1 * DPT_1$ (4.47)

The present value of this plant is then

$$NEWPIS_1 = PIS_1/CRF$$
(4.48)

The operating costs OMC1 include the supply and the production and storage operating costs, with

$$OMC_1 = CT - PIS_1 \tag{4.49}$$

The next step is to compute the present value of the total new plant, NEWPIS, including the transmission component T_2 and the distribution system, with

 $NEWPIS = NEWPIS_1 + NPT_2 + NPD$ (4.50)

The total new plant NEWPIS constitutes then a major input to the financial analysis submodel described in the next section.

The total marginal costs include (1) the marginal costs MC_m corresponding to the costs considered in the costs minimization submodel, (2) the transmission component T₂ and the distribution system capacity marginal costs, CMPT₂ and CMPD, which must be exclusively assigned to the

peak sales month m_p , and (3) other operating marginal costs, COM_2 , corresponding to costs proportional to sales and not considered previously (i.e. transmission and distribution operating costs, customers services, administration, etc.). The value of COM_2 was estimated as

 $COM_2 = 209.48495$ \$/MMCF

The total monthly marginal costs,
$$TMC_m$$
, are then computed as

$$TMC_m = \begin{cases} MC_m + COM_2 + CMPT_2 + CMPD, \text{ if } m = m_p, \\ MC_m + COM_2, \text{ if } m \neq m_p \end{cases}$$
(4.51)

The above monthly marginal costs then serve as monthly prices in the load analysis submodel within the Marginal cost pricing policy block.

The Financial Analysis Submodel

This submodel very much replicates the main calculations typically performed prior to rate case proceedings that take place when the utility requests a change in its retail prices in order to be able to achieve the allowed rate of return on the net value of the utility's plant in service (or rate base), as determined by the state regulatory authorities. The equations and procedures used in this analysis are based on a former study applied to the East Ohio Gas Company, and are not, therefore, discussed in great detail here. For more information, the reader is referred to the corresponding published materials.²⁰

The first part of the analysis consists in determining the net plant in service (or rate base) and the depreciation expense. It is assumed that the whole new plant is put in service in the same single period (i.e., within a year's time), and that the market growth takes place in a similar way. Of course, this is an approximation of reality, wherein the growth in

²⁰See for instance: J.M. Guldmann and D.Z. Czamanski, "A Simulation Model of Market Expansion Policies for Natural Gas Distribution Utilities," <u>Energy</u> 5, 10 (1980): 1013-43; D.Z. Czamanski and J.M. Guldmann, <u>The</u> <u>Allocation of Increasing Gas Supplies in Ohio</u> (Columbus, Ohio: The National Regulatory Research Institute, 1978).

both plant and market takes place progressively. However, such an approximation should be acceptable in view of the stated purpose of the model, i.e., a general evaluation of marginal cost pricing policy. The total plant in service, TOTPIS, is equal to the sum of

(1) the initial total plant in service, PISBEG (= \$617,338,511)

(2) the replacement plant, REPPIS, which is to replace those parts of PISBEG normally retired, with

REPPIS = 0.03625 * PISBEG (4.52)

(3) the new plant in service, NEWPIS, the calculation of which has been described in previous sections

It follows that

TOTPIS = PISBEG + REPPIS + NEWPIS(4.53)

A single average depreciation rate, estimated with historical data, is used to compute the depreciation expense DEPEXP for the three plant types, with

DEPEXP = 0.02939 * TOTPIS (4.54)

The total accumulated provision for depreciation, TAPD, is credited for amounts recovered during the year, such as insurance and salvage value of plant, by adjusting the depreciation expenses with an accumulated provision factor also estimated with historical data, so that

TAPD = TAPDO + 0.82528 * DEPEXP (4.55)

where TAPDO (= \$224,690,519) is the initial accumulated provision for depreciation before the replacement and new plants are put in service. The net plant in service (or rate base), NETPIS, is finally calculated as the difference between the total plant in service and the total accumulated provision for depreciation, with

The second part of the analysis consists in determining the revenue from gas sales, X, that enables the utility to earn the allowed rate of return, ALLROR, on its rate base. It is assumed that this rate of return is equal to 12.06% (1978 value prescribed by the Public Utilities Commission of Ohio). The allowed operating income, AOPINC, is then

AOPINC = 0.1206 * NETPIS (4.57)

The actual operating expenses of the utility, ACOPEX, are the sum of the operating and depreciation expenses. The operating expenses include

(a) the operating costs OMC_1 , determined in the costs minimization submodel (see equation 4.49), and

(b) the other operating costs OMC₂ include the transmission, distribution, customer, and administrative operating costs; they are assumed proportional to the total gas sales DGT, with unit cost COM₂ (= 209.48495 \$/MMCF), so that

$$OMC_2 = 209.48495 * DGT$$
 (4.58)

It follows that

$$ACOPEX = OMC_1 + OMC_2 + DEPEXP$$
(4.59)

The total operating revenues, OPREVS, are the sum of the revenues from gas sales, X, and of other revenues derived from the transportation of gas of others and from nonutility operations such as building rentals. These other revenues are empirically related to the total plant in service, TOTPIS. The total operating revenues are then

$$OPREVS = X + 0.005263 * TOTPIS$$
 (4.60)

In order to determine the net operating income, it is necessary to account for various taxes and deductions. The income before federal income taxes, INCBFT, is calculated while accounting for revenues taxes, REVTAX, property taxes, PRPTAX, and payroll taxes, PAYTAX, with

| REVTAX = 0.041454 * OPREVS | (4.61) |
|----------------------------|--------|
| PRPTAX = 0.021 * NETPIS | (4.62) |
| $PAYTAX = 0.03 * OMC_2$ | (4.63) |

The income before federal income taxes is then

$$INCBFT = OPREVS - ACOPEX - REVTAX - PRPTAX - PAYTAX$$
 (4.64)

or

INCBFT = 0.958546 * X + 0.00504483 * TOPIS - ACOPEX - PRPTAX - PAYTAX (4.65)

In order to simplify later calculations, the following notations are used

$$x_1 = 0.00504483 * TOTPIS - ACOPEX - PRPTAX - PAYTAX (4.66)$$

and

$$INCBFT = 0.958546 * X + X_1$$
 (4.67)

The federally taxable income, TAXINC, and the federal income tax, INCTAX, are then computed while accounting for liberalized depreciation, LIBDEP, interest charges, INTCHG, and investment tax credits, INVTXC, with

| LIBDEP = 0.3 * DEPEXP | (4.68) |
|----------------------------------|--------|
| INTCHG = 0.01759 * TOTPIS | (4.69) |
| INVTXC = 0.1 * (NEWPIS + REPPIS) | (4.70) |

and

$$TAXINC = INCBFT - LIBDEP - INTCHG$$
(4.71)
INCTAX = 0.46 * TAXINC - INVTXC (4.72)

The net operating income, NOPINC, is finally determined as

and must be equal to the allowed operating income, hence

$$NOPINC = AOPINC$$
(4.74)

Equation 4.74 includes, implicitly, the unknown X - the revenues from gas sales. In order to simplify the resolution of this equation, the following quantity is defined

$$X_2 = 0.46 * LIBDEP + 0.46 * INTCHG + INVTXC$$
 (4.75)

The necessary revenues from gas sales are then

$$X = (AOPINC - X_2 - 0.54 * X_1)/(0.51761484)$$
(4.76)

and the corresponding average volumetric rate is

$$PAVG = X/DGT$$
(4.77)

The Pricing Policy Evaluation Submodel

This submodel is implemented through two computer subprograms - EVAL1 and EVAL2 - that are essentially identical with the exception of the way they deal with gas sales revenues surpluses or deficits. Indeed, in the <u>Average cost pricing policy</u> block (case of EVAL1), there are neither deficits nor surpluses, for gas revenues are determined to yield exactly the allowed operating income, whereas in the <u>Marginal cost pricing policy</u> block these revenues depend upon (1) the monthly rates taken equal to the total marginal costs, and (2) the corresponding monthly gas demands.

The submodel first computes the load factor of the end-use customers sales. If m_p is the peak sales month, this load factor is defined as

$$LF = DGT/(12 * DGMT_{m_p})$$
(4.78)

(4.73)

The next step consists in estimating the sectoral consumers' suppluses, taken as consumers' measures of the economic efficiency of the pricing policy. Such surpluses are computed for each month separately. Consider the typical demand curve in figure 4.6. The consumers' surplus at price P_0 is measured by the shaded area S_0 . If the functional relationship between price P and demand D is known, this area can be estimated as



Figure 4.6 Typical Demand Curve and Consumers' Surplus

In the present study, the monthly demand functions are of the constant price elasticity type. (See equations 4.6 through 4.8.) In such a case, it is impossible to integrate the demand function up to an infinite price, and for practical purposes, the upper bound of the integral (4.79) has been set equal to 10,000 \$/MMCF. The residential consumers' surplus for month m is then expressed as

$$CSR_{m} = DGMRO_{m} * \left[\frac{PAVG^{-ELR}_{m}}{1 + ELR_{m}}\right] * \left[10,000^{(1+ELR_{m})} - P_{m}^{(1+ELR_{m})}\right]$$
(4.80)

Similar functional relationships are derived for the commercial and industrial sectors by replacing the weather component $DGMRO_m$ and the elasticity ELR_m by the corresponding sectors' values. The total annual sectoral surpluses are obtained as the sum of the corresponding monthly surpluses.

The production efficiency of the utility can be measured by its net income, NETINC. In the <u>Average cost pricing policy</u> block, this net income is, by definition, equal to the allowed operating income

$$NETINC = AOPINC$$
(4.81)

When the marginal cost pricing policy is applied, the net income may be higher or lower than the allowed operating income. The difference can be found by comparing the gas sales revenues XE necessary to earn the allowed operating income and computed in the financial analysis submodel at the end of the computations cycle, with the actual gas sales XA computed as

$$XA = \sum_{m=1}^{12} P_m * DGMT_m$$
(4.82)

The revenue deficit (DF < 0) or surplus (DF > 0) is then

$$DF = XA - XE \tag{4.83}$$

and the net income is adjusted for DF while accounting for tax effects

$$NETINC = AOPINC + 0.5176 * DF$$
(4.84)

The aggregate efficiency of the pricing policy is finally measured by the sum of (1) all the consumers' surpluses, and (2) the utility's net income.

Application of the Gas Utility Marginal Cost Pricing Model (GUMCPM)

The Assumptions

As stated several times in the description of the model, there is some uncertainty about the exact value of different parameters, either because of forecasting difficulties or because of approximations made while formulating the model. Therefore, sensitivity analyses are called for on such items as (1) demand functions parameters, (2) market growth rates, (3) operating and capacity unit costs, (4) operating and capacity expansion technological constraints, (5) supply costs and constraints, and (6) financial parameters and allowed rate of return.

Obviously, all the above sensitivity analyses could not be performed in the framework of this study. In order to illustrate the potentialities and usefulness of the model, a limited sensitivity analysis was conducted, focusing on the maximum annual supplies available from Consolidated (SUPIT) and Panhandle (SUP2T), and on the monthly price elasticities of demand that were assumed to be equal for all the months and sectors. More specifically, the following values were considered

1. Supply cases

S₁: SUP1T = 500,000 MMCF; SUP2T = 200,000 MMCF S₂: SUP1T = 200,000 MMCF; SUP2T = 500,000 MMCF

2. Elasticity cases

 E_1 : $ELR_m = ELC_m = ELI_m = -0.1$ (m = 1 \rightarrow 12) E_2 : $ELR_m = ELC_m = ELI_m = -0.5$ (m = 1 \rightarrow 12)

The model was then applied under the following four combinations of assumptions: S_1E_1 , S_1E_2 , S_2E_1 , S_2E_2 . The results of these applications are described and discussed in the following section.

The Results

a. Case of the Average Cost Pricing Policy

As the price elasticities of demand are not accounted for in the <u>Average cost pricing policy</u> block, the monthly load patterns are the same for all the four combinations $(S_1E_1, S_1E_2, S_2E_1, S_2E_2)$, and therefore the results differ only with respect to the supply assumptions S_1 and S_2 .

The sectoral monthly loads, corresponding to residential, commercial, and industrial rates of growth all equal to 50%, are indicated in table 4.4. These loads are part of the constraints of the costs minimization submodel run under both supply assumptions. The corresponding minimum

TABLE 4.4

| | | | | | ويستعين المنطقين سيرك معزون ويتكاف الاستبار معارك معوود والقصاري ويسترك متوجه |
|--|--|-------------|------------|---|---|
| | | Residential | Commercial | Industrial | Total |
| | | Load | Load | Load | Load |
| | Month | DGMR | DGMC | DGMI | DGMT |
| | | | | | |
| 1. | April | 22,976.35 | 9,193.07 | 17,979.46 | 50,148.88 |
| 2. | May | 13,708.05 | 5,664.36 | 16,596.89 | 35,969.30 |
| 3. | June | 6,616.95 | 2,964.57 | 15,539.10 | 25,120.61 |
| 4. | July | 5,200.16 | 2,425.15 | 15,327.75 | 22,953.07 |
| 5. | August | 5,483.52 | 2,533.04 | 15,370.02 | 23,386.58 |
| 6. | September | 9,127.71 | 3,920.49 | 15,913.63 | 28,961.83 |
| 7. | October | 18,134.17 | 7,349.51 | 17,257.14 | 42,740.82 |
| 8. | November | 30,365.16 | 12,006.20 | 19,081.67 | 61,453.03 |
| 9. | December | 43,241.77 | 16,908.71 | 21,002.50 | 81,152.97 |
| 10. | January | 48,123.40 | 18,767.29 | 21,730.70 | 88,621.39 |
| 11. | February | 42,334.31 | 16,563.21 | 20,867.13 | 79,764.65 |
| 12. | March | 36,817.81 | 14,462.92 | 20,044.22 | 71,324.94 |
| 1000-10 ⁻⁰ 0-000-000 | n an | | ##### | ĸĸĊŢŴŦĊĊŢĸĸĊĬĬŔĬŦŊĸŊĸŦĬĬĬĬĬĬĸŎŢĸĸŔĬĬŔŊġŢĸŦŦŎŢĸĸĿĸĿŔĸŎŎĸŢŎŢĹĊĸŢĬĬĸ | n n fan skriet fan en sjonen sjonen sjon fan skriet oan fan skriet skriet fan skriet fan skriet skriet skriet s |
| | Total | 282,129.34 | 112,758.50 | 216,710.22 | 611,598.06 |
| 11 de la 11 | | | | | |

SECTORAL MONTHLY LOADS (MMCF) WITH MARKET GROWTH RATES EQUAL TO 50% AVERAGE COST PRICING POLICY

Source: Author's calculations.

costs CT and their breakdown into various components are presented in table 4.5. The lower cost of case S_2 is related to the much higher availability of gas from Panhandle (500,000 MMCF) in case S_2 than in case S_1 (200,000 MMCF), and to the significantly lower commodity cost for

Panhandle's gas (1,009.2 \MMCF versus 1,202.4 \MMCF for Consolidated's gas), hence the difference of 57,959,986 in total commodity charge. This decrease is compensated, to a smaller extent, by an increase of 10,274,363 in the total demand charge, as Panhandle's charge (1,860 \MMCF) is about double that of Consolidated's charge (980 \MMCF). When shifting from case S_1 to case S_2 , the total winter requirement charge decreases by 12,683,008 because of considerably lower winter gas purchases from Consolidated. All the other cost components have the same values under both cases.

TABLE 4.5

COSTS STRUCTURE OF THE OPTIMUM SOLUTIONS UNDER SUPPLY CASES S₁ and S₂ AVERAGE COST PRICING POLICY (In Dollars)

| Cost Component | Supply Case S ₁ | Supply Case S ₂ | |
|---|--|--|--|
| Total Cost CT | \$ 754,632,290 | \$ 693,263,101 | |
| Total Commodity Charge Total Demand Charge Total Winter Requirement Charge | \$ 643,375,002 32,057,349 15,964,072 | \$ 585,415,016 42,331,712 3,281,064 | |
| Wellhead Purchases Field-line Purchases Production Operations Storage Operations | 18,888,000 0 18,778,652 6,746,598 | 18,888,000 0 18,778,652 5,746,038 | |
| Total of Above Operating Costs OMC ₁ | \$ 735,809,718 | \$ 674,440,529 | |
| Total Annualized Investment Costs PIS | \$ 18,822,572 | \$ 18,822,572 | |
| Total Discounted Investment Costs NEWPIS | \$ 151,672,576 | \$ 151,672,576 | |

Source: Author's calculations.

The previous results are further illustrated and clarified by the optimal values of the model's decision variables, as presented in tables

OPTIMAL MONTHLY PURCHASES FROM CONSOLIDATED AND PANHANDLE

TABLE 4.6

AND STORAGE DELIVERIES AND WITHDRAWALS (MMCF)

AVERAGE COST PRICING POLICY

| | Supply Case Sl | | Supply Case S2 | | | |
|----------------------------------|--|--|--|---|---|--|
| Consolidated SUP ₁ | Panhandle SUP2 | Storage Deliveries (-) and Withdrawals (+) | Consolidated SUP ₁ | Panhandle SUP ₂ | Storage Deliveries (-) and Withdrawals (+) | |
| 44,745,61 | 14,634,15 | -12,929,76 | 12,542.19 | 36.525.55 | - 2.617.75 | |
| 28,747,76 | 14,634,15 | -11.111.49 | 0.00 | 36,525,55 | - 4,255.13 | |
| 24,882,98 | 14,634,15 | -18,095.40 | 0.00 | 36,525.55 | -15,103.82 | |
| 21,309.99 | 14,634.15 | -16,689.96 | 984.39 | 36 525.55 | -18,255,75 | |
| 20,447.22 | 14,634.15 | -15,393.67 | 0.00 | 36,525.55 | -16,837.85 | |
| 24,826.86 | 14,634.15 | -14,198.07 | 4,267.48 | 36,525.55 | -15,530.08 | |
| 37,503.11 | 14,634.15 | -13,095.32 | 15,557.07 | 37,343.08 | -13,858.22 | |
| 10,185.18 | 19,512.20 | +28,056.77 | 0.00 | 48,700.73 | + 9,053.41 | |
| 37,508.63 | 19,512,20 | +20,433.26 | 4,371.81 | 48,700.73 | +24,381.55 | |
| 44,745.61 | 19,512.20 | +20,664.70 | 15,557.07 | 48,700.73 | +20,664.70 | |
| 39,039.10 | 19,512.20 | +17,514.47 | 9,850.57 | 48,700.73 | +17,514.47 | |
| 33,269.39 | 19,512.20 | +14,844.47 | 4,080.86 | 48,700.73 | +14,844.47 | |
| 367,211.44 | 200,000.00 | 0.00 | 67,211.44 | 500,000.00 | 0.00 | |
| - | Consolidated SUP1 44,745.61 28,747.76 24,882.98 21,309.99 20,447.22 24,826.86 37,503.11 10,185.18 37,508.63 44,745.61 39,039.10 33,269.39 367,211.44 | Supply Case Consolidated SUP1 Panhandle SUP2 44,745.61 14,634.15 28,747.76 14,634.15 24,882.98 14,634.15 21,309.99 14,634.15 20,447.22 14,634.15 24,826.86 14,634.15 37,503.11 14,634.15 10,185.18 19,512.20 37,508.63 19,512.20 39,039.10 19,512.20 33,269.39 19,512.20 367,211.44 200,000.00 | Supply Case S_1 Supply Case S_1 Consolidated SUP1Panhandle SUP2Storage Deliveries (-) and Withdrawals (+)44,745.61 28,747.76 28,747.76 14,634.15 24,882.98 20,447.22 21,309.99 20,447.22 14,634.15 24,826.86 20,447.22 14,634.15 24,826.86 14,634.15 24,826.86 14,634.15 24,826.86 14,634.15 24,826.86 14,634.15 24,826.86 20,447.501 20,185.18 19,512.20 19,512.20 20,433.26 44,745.61 19,512.20 420,664.70 39,039.10 39,039.10 19,512.20 414,844.47367,211.44 367,211.44200,000.000.00 | Supply Case S_1 Storage Deliveries (-)Consolidated SUP1Panhandle SUP2and Withdrawals (+)Consolidated SUP144,745.6114,634.15-12,929.7612,542.1928,747.7614,634.15-11,111.490.0024,882.9814,634.15-16,689.96984.3920,447.2214,634.15-15,393.670.0024,826.8614,634.15-14,198.074,267.4837,503.1114,634.15-13,095.3215,557.0710,185.1819,512.20+28,056.770.0037,508.6319,512.20+20,433.264,371.8144,745.6119,512.20+17,514.479,850.5733,269.3919,512.20+14,844.474,080.86367,211.44200,000.000.0067,211.44 | Supply Case S_1 Supply Case S_2 Supply Case S_1 Supply Case S_1 Supply Case S_1 Supply Case S_1 Consolidated SUP1Panhandle SUP2A4,745.6114,634.15-12,929.7612,542.1936,525.5528,747.7614,634.15-12,929.7612,542.1936,525.5528,747.7614,634.15-11,111.490.0036,525.5528,747.7614,634.15-11,111.490.0036,525.5528,482.9814,634.15-16,689.96984.3936,525.5520,447.2214,634.15-14,198.074,267.4836,525.5524,826.8614,634.15-13,095.3215,557.0737,343.0810,185.1819,512.20+28,056.770.0048,700.7337,508.6319,512.20+28,056.770.0048,700.7339,039.1019,512.20+14,844.474,080.8648,700.73367,211.44200,000.000.00 <td cols<="" td=""></td> | |

4.6 and 4.7. As previously noted, all the available supplies from Panhandle are purchased, and in such a way that the take-or-pay clause (75% of the contract demand) is never implemented. All the available wellhead gas is purchased because of its low cost (787 %/MMCF), whereas field-line gas is never purchased because of its high cost (1481 %/MMCF). Production is not a cost-attractive alternative, and the production capacity is expanded by 751.22 MMCF/month, just enough to provide for a constant monthly production of 1698.88 MMCF, or 20,386.56 MMCF for the whole year. This amount simply covers 10% of the total demand increment of 203,866 MMCF, as stipulated in the constraint set. In both cases, the maximum incremental storage capacity (100,000 MMCF) is developed. However, it is fully used only in case S₁ (total storage deliveries equal to 101,513.67 MMCF). Finally, the peak purchases in month 10 (January) determine the level of incremental transmission capacity (12,956.69 MMCF/month).

The results of the analyses performed in the distribution and financial submodels are presented in table 4.8. The lower revenue requirement and average volumetric rate in case S_2 are attributable to the corresponding lower operating expenses (\$975,488,359 versus \$1,039,511,566, and 1,594.98 \$/MMCF versus 1,699.66 \$/MMCF).

The evaluation criteria are presented in table 4.9. The gas consumption/conservation criteria have the same values under both cases, simply because the sales patterns are the same. The economic efficiency criteria with respect to consumers depend both upon the reference average volumetric rate (1,699.66 %/MMCF for case S₁ and 1,594.98 %/MMCF for case S₂) and upon the assumed price elasticities. Four different cases must therefore be considered, and the results pertaining to each of them are also indicated in table 4.9. Note that the highest aggregate efficiency is obtained under case S₂E₁ (\$4,692,805,148), because of both the lowest reference volumetric rate and the highest demand curve (see figure 4.6). However, it is important to remember that the four sets of values are not comparable among themselves because they each refer to different demand curves but will constitute benchmarks for the evaluation of the marginal cost pricing policy as described in the next section.

OPTIMAL MAXIMUM SUPPLIES FROM CONSOLIDATED AND PANHANDLE, WELLHEAD AND FIELD-LINE MONTHLY PURCHASES, INCREMENTAL PRODUCTION CAPACITY AND CONSTANT MONTHLY PRODUCTION, INCREMENTAL STORAGE CAPACITY AND TOTAL STORAGE DELIVERIES, AND INCREMENTAL TRANSMISSION CAPACITY AVERAGE COST PRICING POLICY

| Variable | Supply Case S1 | Supply Case S2 |
|--|--|--|
| | diter of a state of the state o | nig 2002 With Cold and the Dirichler Cold and the Cold and |
| Consolidated's Maximum Supply: | | |
| - Daily (MMCF) | 1,491.52 | 518.57 |
| - Monthly (MMCF) | 44,745.61 | 15,557.07 |
| Panhandle's Maximum Supply: | | |
| - Daily (MMCF) | 650-41 | 1,623,36 |
| - Monthly (MMCF) | 19 512 20 | 48 700 73 |
| nonenity (mor) | 179712020 | 40,700:75 |
| Monthly Wellhead Purchases (MMCF) | 2,000.00 | 2,000.00 |
| Monthly Field-line Purchases (MMCF) | 0.00 | 0.00 |
| Incremental Production Capacity | | |
| (MMCF/month) | 751.22 | 751,22 |
| Monthly Production (MMCF) | 1,698.88 | 1,698.88 |
| ՠՠֈֈֈֈֈ֎ՠֈ֎ֈՠֈ֎ՠՠֈֈֈֈֈֈֈֈֈֈֈֈֈֈֈֈֈֈֈֈֈ | ĸĸŧġĸĸĸĸĔĊĸŔĬĊĔŔĸĸŗġĬĸĸĸĸĔĬĸŴĊĸŎĸĸĸĸĸĸĬĬĬŔĸĸĸĸĸĬĬĬĬŔĸĸŢĸĔĹĸŔĸſġĸĸŔĊŎĸĸĬĬĬĊŔĸĸĸĔŢĸĸĸŔĬŢ | |
| Incremental Storage Capacity (MMCF) | 100,000.00 | 100,000.00 |
| Total Storage Deliveries (MMCF) | 101,513.67 | 86,458.60 |
| Transmission Component T ₁ Incre- mental Capacity (MMCF/month) | 12,956.69 | 12,956.69 |

Source: Author's calculations.

TABLE 4.8

DISTRIBUTION PLANT, FINANCIAL VARIABLES, AND AVERAGE VOLUMETRIC RATES AVERAGE COST PRICING POLICY

| Variable | Supply Case S ₁ | Supply Case S ₂ |
|---|--|--|
| New Transmission Plant T ₂ (\$) New Distribution Plant (\$) Total New Plant (\$) Net Plant in Service (\$) Allowed Operating Income (\$) Actual Operating Expenses (\$) | 211,576 465,353,472 617,237,504 1,001,776,640 120,814,248 900,872,207 | 211,576 465,353,472 617,237,504 1,001,776,640 120,814,248 839,503,017 |
| Revenue Requirement (\$) Average Volumetric Rate (\$/MMCF) Source: Author's calculations. | 1,039,511,566 1,699.66 | 975,488,359 1,594.98 |

| Variable | Supply Case S_1 | Supply Case S ₂ |
|---|--|--|
| Gas Consumption/Conservation | | |
| Peak Sales Month Peak Sales (MMCF) Sales Load Factor Total Gas Consumption (MMCF) | January 88,621.39 0.5751 611,598.06 | January 88,621.39 0.5751 611,598.06 |
| Economic Efficiency | | |
| E_{1} . Price Elasticity = -0.1 | _ | |
| Total Residential Surplus (\$) Total Commercial Surplus (\$) Total Industrial Surplus (\$) Total Consumers' Surplus (\$) Net Utility Income (\$) Aggregate Efficiency (\$) | 2,092,875,719 836,458,689 1,607,587,336 4,536,921,745 120,814,248 4,657,735,992 | 2,109,053,072 842,924,285 1,620,013,544 4,571,990,900 120,814,248 4,692,805,148 |
| E_2 . Price Elasticity = -0.5 | | |
| Total Residential Surplus (\$) Total Commercial Surplus (\$) Total Industrial Surplus (\$) Total Consumers' Surplus (\$) Net Utility Income (\$) Aggregate Efficiency (\$) | 1,367,218,105 546,435,440 1,050,192,561 2,963,846,106 120,814,248 3,084,660,354 | 1,353,510,283 540,956,841 1,039,663,259 2,934,130,383 120,814,248 3,054,944,631 |

EVALUATION CRITERIA FOR THE AVERAGE COST PRICING POLICY

Source: Author's calculations.

Finally, it is interesting to analyze the implications of a marginal cost pricing policy with monthly demands such as those indicated in table 4.4, that is, totally price-inelastic demands. The marginal costs produced by the cost minimization model, MC_m , and the total marginal costs, TMC_m , are presented in table 4.10 for the two supply cases.

Under a marginal cost pricing policy, the utility's revenues from gas sales, RGS, are

$$RGS = \sum_{m=1}^{1.2} TCM_m * DGMT_m$$

(4.85)

with

MONTHLY MARGINAL COSTS AVERAGE COST PRICING POLICY ANALYSIS (In Dollars)

| | Supply Case Sl | | Supply Case S ₂ | |
|-----------|---|--|---|--|
| Month | Cost Minimization Model Marginal Costs MC _m | Total Marginal Costs TMC _m | Cost Minimization Model Marginal Costs MC _m | Total Marginal Costs TMC _m |
| April | 1202.40 | 1411.88 | 1202.40 | 1411.88 |
| May | 1202.40 | 1411.88 | 1202.40 | 1411.88 |
| June | 1202.40 | 1411.88 | 1202.40 | 1411.88 |
| July | 1202.40 | 1411.88 | 1202.40 | 1411.88 |
| August | 1202.40 | 1411.88 | 1139.30 | 1348.78 |
| September | 1202.40 | 1411.88 | 1202.40 | 1411.88 |
| October | 1202.40 | 1411.88 | 1208.16 | 1417.65 |
| November | 1299.30 | 1508.78 | 1274.62 | 1484.11 |
| December | 1299.30 | 1508.78 | 1299.30 | 1508.78 |
| January | 1923.34 | 4304.10 | 1917.58 | 4298.34 |
| February | 1299.30 | 1508.78 | 1299.30 | 1508.78 |
| March | 1299.30 | 1508.78 | 1299.30 | 1508.78 |

Source: Author's calculations.

- Case S1: RGS = \$1,148,274,700 or a revenue above the "normal" revenue by \$108,763,200

- Case S₂: RGS = \$1,145,019,200 or a revenue above the "normal" revenue by \$169,530,850

The above results would therefore confirm the widespread view that marginal cost pricing is likely to bring to the utility revenues higher than those it is entitled to by regulation. However, these results constitute extreme cases because it is assumed that there is no change in the demand pattern as a consequence of the marginal cost pricing pattern. It is the purpose of the next section to analyze the implications for marginal cost pricing policy of price-dependent demands for gas.

b. Case of the Marginal Cost Pricing Policy

The major feature of the four applications (S_1E_1 , S_1E_2 , S_2E_1 , S_2E_2) is that no equilibrium of supply and demand can be reached, at least through
the iterative procedure implemented here, because of constantly shifting peaks. This "negative" result is in itself significant, for it gives additional strength to the widespread assertion that time-differentiated marginal cost based rates are likely to induce customers to shift their peak consumption, with a new peak load eventually higher than the one under the existing rate structure, and therefore implying more peaking capacity for the system.

In each application, the disequilibrium is characterized by a shift between two different demand-supply patterns, each providing price inputs to the other; that is, the total marginal costs of one solution constitute the price pattern used in the determination of the other solution. The two alternating price-demand patterns are noted A and B for each of the four cases and are presented in tables 4.11 through 4.14. In all cases, the peak-rate month is alternatively January or December, and the peak-sales month alternatively December or January, respectively. One pattern (A for cases S_1E_1 and S_1E_2 , and B for cases S_2E_1 and S_2E_2) is characterized by a unique major peak rate in January, while the other pattern includes a major and a minor peak rate (3,463.75 \$/MMCF in December and 2,132.82 \$/MMCF in February). The latter rate pattern always implies higher peak sales, but not higher annual sales, with the exception of case S₂E₂ (679,557 MMCF versus 601,327 MMCF). The characteristics of the optimal solutions for each pattern and case, as well as their evaluations, are presented in tables 4.15 through 4.18. As indicated earlier, the four cases - S1E1, S1E2, S_2E_1 , S_2E_2 , - cannot be compared among themselves, because they refer to different demand functions assumptions, and instead must be compared to the corresponding average cost pricing solutions described in tables 4.5 through 4.9. For instance, the outputs of the cost minimization, distribution, and financial submodels for patterns A and B in case S_1E_1 must be compared to the corresponding variables for supply case S1 in tables 4.5 through 4.9, whereas the economic efficiency criteria (surpluses, aggregate efficiency) must be compared with those of the supply case S1 and the price elasticity case E_1 (E = -0.1) in table 4.9. To illustrate the previous remark, the aggregate efficiency of pattern A/S_1E_1 , equal to \$4,593,440,450, must be compared to the reference aggregate efficiency of

| Month | Pattern A | | Pattern B | | |
|--|---|---|---|---|--|
| nonen | Price | Total Demand | Price (¢/MMCE) | Total Demand | |
| | (\$/FINGE) | (PITIOF) | (3) HHOF) | (FINOF) | |
| April | 1411.88 | 51,087.85 | 1411.88 | 51,087.86 | |
| May | 1411.88 | 36,642.78 | 1411.88 | 36,642.78 | |
| June | 1411.88 | 25,590.96 | 1411.88 | 25,590.96 | |
| July | 1411.88 | 23,382.83 | 1411.88 | 23,382.83 | |
| August | 1411.88 | 23,824.46 | 1411.88 | 23,824.46 | |
| September | 1411.88 | 29,504.10 | 1411.88 | 29,504.10 | |
| October | 1411.88 | 43,541.09 | 1411.88 | 43,541.09 | |
| November | 1508.78 | 62,189.47 | 1508.78 | 62,189.47 | |
| December | 1508.78 | 82,125.50 | 3463.75 | 75,576.38 | |
| January | 4304.10 | 80,758.20 | 1508.78 | 89,683.41 | |
| February | 1508.78 | 80,720.54 | 2132.82 | 77,974.26 | |
| March | 1508.78 | 72,179.69 | 1508.78 | 72,179.69 | |
| analaikaanaa mata-faranakikati kiin parayondari kiinaadikkii | nin engen andere et kommen in Miller en | n or an | en nederse ziele de General de Gen | ŢġĸĸĸĸġĊĸĸĸĸŦĔĊĸĸŔĊŢġĸĸĬĸŢŎĊĸŔŔŢĊŢġĸĸĊĸŎĊĊĬĊŎĬŎĸĸĸŎĬĊĊŎĬŔŢĸĊĬĸĬĬĬŎĬĬŎŎŎŎ | |
| Total | | 611,547.47 | 10523-62629 | 611,177.29 | |
| Source: Au | thor's calc | ulations. | | ALLONG STOCK ST | |

PRICE-DEMAND PATTERNS IN CASE S_1E_1

TABLE 4.12

|--|

| Month | Pat | Pattern A | | Pattern B | | |
|------------|--------------------|---|--------------------|------------------------|--|--|
| Honen | Price (\$/MMCF) | Total Demand (MMCF) | Price (\$/MMCF) | Total Demand (MMCF) | | |
| | | Ennaminen an fallen fan wegen gener gener fan en fan en fan de | * | | | |
| April | 1411.88 | 55,022.86 | 1411.88 | 55,022.86 | | |
| May | 1411.88 | 39,465.17 | 1411.88 | 39,465.17 | | |
| June | 1411.88 | 27,562.09 | 1411.88 | 27,562.09 | | |
| July | 1411.88 | 25,183.88 | 1411.88 | 25,183.88 | | |
| August | 1411.88 | 25,659.53 | 1411.88 | 25,659.53 | | |
| September | 1411.88 | 31,776.64 | 1411.88 | 31,776.64 | | |
| October | 1411.88 | 46,894.81 | 1411.88 | 46,894.81 | | |
| November | 1508.78 | 65,224.57 | 1508.78 | 65,224.57 | | |
| December | 1508.78 | 86,133.56 | 3463.75 | 56,847.69 | | |
| January | 4304.10 | 55,690.25 | 1508.78 | 94,060.33 | | |
| February | 1508.78 | 84,660.03 | 2132.82 | 71,205.66 | | |
| March | 1508.78 | 75,702.36 | 1508.78 | 75,702.36 | | |
| Total | casa gay | 618,975.75 | 15200-62029 | 614,605.59 | | |
| Source: Au | thor's calc | ulations. | | | | |

| Month | Pat | Pattern A | | Pattern B | | |
|-----------|--------------------|--------------|--------------------|--------------|--|--|
| nonen | Price (\$/MMCF) | Total Demand | Price (\$/MMCF) | Total Demand | | |
| | (0/11101) | (11101) | (9/10101) | | | |
| April | 1411.88 | 50,764.12 | 1411.88 | 50,764.12 | | |
| May | 1240.52 | 36,884.79 | 1321.04 | 36,653.54 | | |
| June | 1218.68 | 25,805.76 | 311.84 | 29,574.16 | | |
| July | 218.91 | 27,995.63 | 1321.04 | 23,389.70 | | |
| August | 209.48 | 28,650.16 | 1321.04 | 23,831.46 | | |
| September | 1218.68 | 29,751.74 | 1321.04 | 29,512.76 | | |
| October | 1411.88 | 43,265.18 | 1411.88 | 43,265.18 | | |
| November | 1508.78 | 61,795.40 | 1497.28 | 61,842.70 | | |
| December | 3463.75 | 75,097.48 | 1508.78 | 81,605.10 | | |
| January | 1508.78 | 89,115.12 | 4304.10 | 80,246.47 | | |
| February | 2132.82 | 77,480.16 | 1508.78 | 80,209.04 | | |
| March | 1508.78 | 71,722.32 | 1508.78 | 71,722.32 | | |
| Total | | 618,327.86 | | 612,616.55 | | |
| Source: A | Author's calc | ulations. | | | | |

PRICE-DEMAND PATTERNS IN CASE S_{2E_1}

TABLE 4.14

| Month | Pattern A | | Pattern B | | |
|------------|--------------------|---|--------------------|---|--|
| | Price (\$/MMCF) | Total Demand (MMCF) | Price (\$/MMCF) | Total Demand (MMCF) | |
| Annil | 1/11 00 | 52 201 52 | 1/11 00 | 53 301 52 | |
| APLIL | 1411.00 | 10,001,02 10,705,75 | 1411.00 | JJ,JUI®JZ | |
| May | 1240.52 | 40,/85./5 | 1411.88 | 38,230.33 | |
| June | 1218.68 | 28,738.39 | 1362.74 | 27,177.03 | |
| July | 218.91 | 61,956.67 | 1416.02 | 24,360.35 | |
| August | 209.48 | 64,530.98 | 1416.02 | 24,820.44 | |
| September | 1218.68 | 33,132.81 | 1411.88 | 30,782.53 | |
| October | 1411.88 | 45,427.75 | 1411.88 | 45,427.75 | |
| November | 1508.78 | 63,184.07 | 1483.21 | 63,726.49 | |
| December | 3463.75 | 55,069.26 | 1483.21 | 84,155.24 | |
| January | 1508.78 | 91,117.73 | 4295.82 | 53,999.97 | |
| February | 2132.82 | 68,978.04 | 1508.78 | 82,011.50 | |
| March | 1508.78 | 73,334.07 | 1508.78 | 73,334.07 | |
| | | errenrene an an bielen af Calastan bielen bielen staten span igen opgeneen. | | 94 m 93 m 97 4 4 68 4 7 68 6 4 68 6 4 7 6 7 7 6 7 7 7 6 7 7 7 6 7 7 7 7 7 | |
| Total | E019-4180 | 679,557.04 | Kridb ettips | 601,327.42 | |
| Source: Au | thor's calc | ulations. | | 0- 1000 1-100 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 | |

PRICE-DEMAND PATTERNS IN CASE S2E2

| Variable | Pattern A | Pattern B |
|---|---|--|
| OUTPUT OF THE COST N | INIMIZATION SUBMODE | an and an and an and an all and an and an all and all and all and a I have a set of the |
| | | |
| Total Cost CT (\$) | /51,846,983 | /55,//6,881 |
| Total Commodity Charge (\$) | 643,316,202 | 64Z,885,930 |
| Total Demand Charge (\$) | 32,968,116 | 35,745,832 |
| Total Winter Requirement Charge (\$) | 19,516,151 | 19,480,778 |
| Total Operating Costs OMC1 (\$) | /3/,48/,349 | /39,/88,055 |
| Annualized Investment Costs (§) | 14,359,634 | 15,988,826 |
| Discounted Investment Costs (§) | 115,/10,144 | 128,838,208 |
| Maximum Monthly Supply: Consolidated | 1 47 060 | 54 155 |
| (THOF) Mawinum Manthlu Supplus Dephandla | 47,005 | |
| (MCE) | 10 512 | 10 510 |
| (MMOF) | 29,212 | |
| (MCE) | 367 163 | 266 905 |
| (MMOF) | 307,103 | 200 _e 005 |
| (MCE) | 200,000 | 200,000 |
| (MMOF) Monthly Mallhard Duraharaa (MMOF) | 200,000 | 200,000 |
| Monthly wellhead Purchases (MMCF) | 2,000 | 2,000 |
| (MCE) | 7 = 1 | 750 |
| (MMCF) Monthly Droduction Data (MCCE) | 1 600 | 1 607 |
| Monthly Production Rate (MMCF) | 1,090 | 1,097 |
| Storage Incremental Capacity (MMCF) | 60 512 | 60 512 |
| Annual Storage Deliveries (MMCF) | 00,212 | 00,010 |
| Consolity (MCE) | 15 070 | 22 265 |
| capacity (MMCF) | 1,2/9 | 22,303 |
| OUTPUTS OF THE DISTRIBUT | ION AND FINANCIAL SU | BMODELS |
| | | |
| New Transmission Plant T_2 (\$) | 0 | 2,062,706 |
| New Distribution Plant T_2 (\$) | 363,022,848 | 482,083,840 |
| Total New Plant (\$) | 478,732,800 | 612,984,576 |
| Net Plant in Service (\$) | 866,631,424 | 997,626,880 |
| Actual Operating Expenses (\$) | 898,468,587 | 904,637,404 |
| Equilibrium Revenue (\$) | 1,033,292,770 | 1,043,322,900 |
| Equilibrium Volumetric Rate | | |
| (\$/MMCF) | 1,689.64 | 1,707.07 |
| OUTPUT OF THE EV | ALUATION SUBMODEL | ĸĸĸĸijĸĸĿġĸĸġġĸĸġġĸĸġġĸĸġĸĬĸŎĸĬŎĸĬĬĬĬĸĸġĸĸġŧĸĊĬĬĬĸŎĸĸġĸĸġĿĸŎĿĸĬŎĸĸġĸĸġĸĸġĸĸġĸĿĬĬĬĬĬĬĬ |
| ՠՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ֎ՠ | and and a second sec | ŊĸĸĸŦŎĸŦĸġţĸĸĸġĸĸĸġĸĸĸĔĸĊĸĿġġĸĸġġĸĸĸġġĸĸġġĸĸċġġĸĸŎġŎĸŎŎŎĸŎŎĸĸŢĊĸŎŎĸŎŎŎĸĸŎŎĸĸŎŎġĸĊŎġĸŎŎŎĸ |
| Peak Sales (MMCF) - Month 82, | ,125.60 - December | 89,683.41 - January |
| Sales Load Factor | 0.6205 | 0.5679 |
| Total Gas Consumption (MMCF) | 611,547.47 | 611,177.29 |
| Total Residential Surplus (\$) | 2,026,957,733 | 2,047,161,873 |
| Total Commercial Surplus (\$) | 811,509,256 | 819,270,035 |
| Total Industrial Surplus (\$) | 1,602,573,597 | 1,607,825,070 |
| Total Consumers' Surplus (\$) | 4,441,040,587 | 4,474,256,978 |
| Revenue Surplus (+) or Deficit (-) | | |
| (\$) | + 92,511,836 | + 52,586,954 |
| Aggregate Efficiency (\$) | 4,593,440,450 | 4,621,789,773 |
| | | |

SOLUTIONS CHARACTERISTICS AND EVALUATION IN CASE ${\rm S}_1{\rm E}_1$

Source: Author's calculations.

1000000

ġ.

| Variable | Pattern A | Pattern B |
|---|---|---------------------------------------|
| OUTPUT OF THE COST | MINIMIZATION SUBMODE | EL |
| ₩₩₩₽₩₽₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩ | ĸŦŧġŊŎŦĸŢŎĊŎĹŎĹŦŦŦŎŊĸĊĊŢŎŎŦŎŎŎĸŎŎŎŎŎŎŎŎĸĸŎĊŎĸĸŎŎĸĸŎŎŎĸĸŎŎŢĸĸŎŊĸŎŎŎŎŎŎŎŎŎŎ | ₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩ |
| Total Cost CT (\$) | 762,422,982 | 761,315,954 |
| Total Commodity Charge (\$) | 651,950,236 | 646,870,720 |
| Total Demand Charge (\$) | 34,504,308 | 37,457,850 |
| Total Winter Requirement Charge (\$) | 18,482,633 | 18,065,047 |
| Total Operating Costs OMC $_{ m l}$ (\$) | 746,852,138 | 744,174,395 |
| Annualized Investment Costs (\$) | 15,570,844 | 17,141,558 |
| Discounted Investment Costs (\$) | 125,470,096 | 138,126,944 |
| Maximum Monthly Supply: Consolidate | d | |
| (MMCF) | 50,988 | 58,522 |
| Maximum Monthly Supply: Panhandle | | |
| (MMCF) | 19,512 | 19,512 |
| Annual Purchases from Consolidated | 07/0/0 | 270 110 |
| (MMCF) | 3/4,343 | 3/0,119 |
| Annual Furchases from Pannandle | 200 000 | 200 000 |
| (MMCE) | 200,000 | 200,000 |
| Incremental Production Consolty | 2,000 | 2,000 |
| (MMCF) | 772 | 759 |
| | 112 | 155 |
| Storage Incremental Capacity (MMCF) | 0 | 0 |
| Annual Storage Deliveries (MMCF) | 60.513 | 60,513 |
| Transmission Plant T ₁ Incremental | 00,010 | |
| Capacity (MMCF) | 19,219 | 26,742 |
| | | - |
| OUTPUTS OF THE DISTRIBUT | ION AND FINANCIAL S | UBMODELS |
| | | |
| New Transmission Plant T ₂ (\$) | 0 | 9,691,736 |
| New Distribution Plant T_2 (\$) | 426, 162, 432 | 551,033,856 |
| Total New Plant (\$) | 551,632,384 | 698,852,352 |
| Net Plant in Service (\$) | 937,762,816 | 1,081,411,840 |
| Actual Operating Expenses (\$) | 911,532,006 | 912,265,578 |
| Equilibrium Revenue (Ş) | 1,048,923,004 | 1,053,604,046 |
| Equilibrium Volumetric Rate | 1 (0/ (1 | 1 71 / 28 |
| (S/MMCF) | 1,094.01 | 1,/14.28 |
| OUTDUT OF THE | FUALLATION CURMODEL | |
| USITOT OF THE | TANTOVITON PODUODED | |
| Peak Sales (MMCF) - Month 86 | ,133.56 - December | 94,060.33 - January |
| Sales Load Factor | 0,5989 | 0.5445 |
| Total Gas Consumption (MMCF) | 618,975,75 | 614.605.59 |
| Total Residential Surplus (\$) | 1,324,976,057 | 1,334,202,912 |
| Total Commercial Surplus (\$) | 530,746,277 | 534,237,643 |
| Total Industrial Surplus (\$) | 1,056,754,515 | 1,057,426,049 |
| Total Consumers' Surplus (\$) | 2,912,476,849 | 2,925,866,603 |
| Revenue Surplus (+) or Deficit (-) | | |
| (\$) | + 16,273,191 | + 4,897,271 |
| Aggregate Efficiency (\$) | 3,033,994,034 | 3,058,819,683 |

solutions characteristics and evaluation in case $\ensuremath{s_{1\text{E}_2}}$

Source: Author's calculations.

| Variable | Pattern A | Pattern B |
|---|--|----------------------|
| OUTPUT OF THE COS | ST MINIMIZATION SUBMOD | DEL. |
| | 700 710 000 | 701 / 00 070 |
| Total Cost CT (\$) | /03,/13,329 | 701,439,979 |
| Total Commodity Charge (\$) | 594, 313, 620 | 595,979,109 |
| Total Demand Charge (\$) | 44,077,088 | 40,040,190 |
| lotal Winter Requirement Charge (| (\$) 6,806,471 | 9,720,772 |
| Total Operating Costs OMC_1 (S) | 687,570,384 | 087,100,270 |
| Annualized Investment Costs (§) | 16,142,945 | 14,283,704 |
| Discounted investment Costs (\$) | 130,080,096 | 115,098,304 |
| Maximum Monthly Supply: Consolida | 17 E 20 | 25 072 |
| (MMCF) Maninum Monthla Gunnlas Danhandl | 27,539 | 23,073 |
| maximum Montilly Supply: Pannanole | | 60 102 |
| (MMCF) | 43,040 | 40,193 |
| Annual Purchases from Consolidate | | 11/ 012 |
| (MMCF) | /3,/1/ | 114,015 |
| Annual Purchases from Pannandle | 500 000 | 453 003 |
| (MMCF) Monthly Hollbood Durchasse (MMCF) | 500,000 | 4,000 |
| Monthly wellnead Purchases (MMCF, | 2,000 | 2,000 |
| (MCR) | 770 | フェル |
| (MMCF) | 1 710 | 1 702 |
| Storage Increased Consister (100 | 1,/10 | 1,702 |
| Appuel Storage Deliveries (MCE) | | 55 791 |
| Annual Storage Deliveries (MMGF) | 10,000 | 77,101 |
| Consolty (MMCE) | L 21 707 | 14 768 |
| Capacity (MMCF) | 2.1,177 | 14,700 |
| OUTPUTS OF THE DISTRI | BUTION AND FINANCIAL S | SUBMODELS |
| | ۲۵۵ – ۱۹۵۵ ۲۵ پر ۲۵۵ میلید اور در ۲۵۵ میلید دارد اور ۲۵ میلید میلید اور ۲۵ میلید در ۲۵ میلید و ۲۵ میلید در ۲۵ م ۱۹۵۵ – ۱۹۵۵ ۲۵ پر ۲۵ میلید در ۲۵ میلید در ۲۵ میلید میلید اور ۲۰ میلید در ۲۵ میلید و ۲۵ میلید و ۲۵ میلید در ۲۵ م | n y |
| New Transmission Plant T ₂ (\$) | 1,072,165 | 0 |
| New Distribution Plant T ₂ (\$) | 473,131,264 | 354,824,960 |
| Total New Plant (\$) | 604,283,392 | 469,923,072 |
| Net Plant in Service (\$) | 989,136,640 | 858,035,200 |
| Actual Operating Expenses (\$) | 853,661,944 | 848,102,551 |
| Equilibrium Revenue (\$) | 989,956,661 | 980,519,500 |
| Equilibrium Volumetric Rate | | |
| (\$/MMCF) | 1,601.02 | 1,600.54 |
| OUTPUT OF THE | EVALUATION SUBMODEL | |
| | | |
| reak Sales (MMCF) - Month | 89,115.12 - January | 81,605.10 - December |
| Sales Load Factor | 0.5782 | 0.6256 |
| Total Gas Consumption (MMCF) | 618,327.86 | 612,616.55 |
| Total Residential Surplus (\$) | 2,053,543,610 | 2,025,366,022 |
| Total Commercial Surplus (\$) | 822,855,564 | 811,346,108 |
| Total Industrial Surplus (\$) | 1,646,548,121 | 1,616,808,550 |
| iotal Consumers' Surplus (\$) | 4,522,947,295 | 4,453,520,680 |
| kevenue Surplus (+) or Deficit (| -) | |
| | + 29,670,941 | +101,592,054 |
| Aggregate Efficiency (\$) | 4,657,594,838 | 4,609,583,759 |

SOLUTIONS CHARACTERISTICS AND EVALUATION IN CASE ${\rm S_{2}E_{1}}$

i,

Source: Author's calculations.

| SOLUTIONS CHARACTERISTICS | AND | EVALUATION | IN | CASE | S ₂ E ₂ |
|---------------------------|-----|------------|----|------|-------------------------------|
|---------------------------|-----|------------|----|------|-------------------------------|

| Variable | Pattern A | Pattern B |
|--|--|--|
| OUTPUT OF THE COST | MINIMIZATION SUBMODE | EL |
| | | |
| Total Cost CT (\$) | //6,9/6,123 | 684,397,599 |
| Total Commodity Charge (\$) | 664,405,084 | 580,540,481 |
| Total Demand Charge (\$) | 46,437,113 | 41,243,689 |
| Total Winter Requirement Charge (\$) | 4,327,615 | 7,315,443 |
| Total Operating Costs OMC ₁ (\$) | 757,919,648 | 670,147,158 |
| Annualized Investment Costs (\$) | 19,056,475 | 14,250,441 |
| Discounted Investment Costs (\$) | 153,557,360 | 114,830,272 |
| Maximum Monthly Supply: Consolidate | d | |
| (MMCF) | 26,412 | 26,347 |
| Maximum Monthly Supply: Panhandle | | |
| (MMCF) | 48,499 | 41,553 |
| Annual Purchases from Consolidated | | |
| (MMCF) | 132,905 | 91,996 |
| Annual Purchases from Panhandle | | |
| (MMCF) | 500,000 | 465,287 |
| Monthly Wellhead Purchases (MMCF) | 2,000 | 2,000 |
| Incremental Production Capacity | | |
| (MMCF) | 940 | 723 |
| Monthly Production Rate (MMCF) | 1,888 | 1,670 |
| Storage Incremental Capacity (MMCF) | 0 | 0 |
| Annual Storage Deliveries (MMCF) | 45,087 | 55,615 |
| Transmission Plant T _l Incremental | | |
| Capacity (MMCF) | 23,799 | 16,571 |
| | TON AND PINANCIAL O | IDMODEL C |
| UUIPUIS OF THE DISIKIBUI | ION AND FINANCIAL S | DPMODET2 |
| New Transmission Plant T ₂ (\$) | 4,562,736 | 0 |
| New Distribution Plant T ₂ (\$) | 504,678,656 | 394,997,504 |
| Total New Plant (\$) | 662,798,592 | 509,827,584 |
| Net Plant in Service (\$) | 1,046,232,580 | 896,972,032 |
| Actual Operating Expenses (\$) | 938,557,560 | 829,901,323 |
| Equilibrium Revenue (\$) | 1,080,492,967 | 962,526,324 |
| Equilibrium Volumetric Rate | | , , , , , , , , , , , , , , , , , , , |
| (\$/MMCF) | 1,589.99 | 1,600.67 |
| | ۲ ۵۰ - ۱۹۹۵ - ۱۹۹۵ - ۱۹۹۵ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱۹۹۹ - ۱ | ۲ |
| OUTPUT OF THE EV | ALUATION SUBMODEL | ND-+231-434405-4324432-424434-424-434-424-434-434-434- |
| Peak Sales (MMCF) - Month 01 | 117 73 - Tenurozati | 9/ 155 9/ December |
| Sales Load Factor | 111.73 - January 0.6215 | 04,102,24 - Decemper 0 5955 |
| Total Cas Consumption (MMCF) | 670 557 02 | 601 207 11 |
| Total Residential Surplus (C) | 1 317 070 020 | 1 286 012 282 |
| Total Commercial Surplus (3) | 1, JI/, J/2, JJU 520 1/1 000 | 1,200,013,302 515 12/ 020 |
| Total Industrial Surplus (\$) | JZ7,141,007 1 NON 619 110 | JIJ,IJ4,727 1 005 516 000 |
| Total Concumeral Sumplus (3) | 1,070,412,110 2 027 526 027 | 1,02,040,007 2,026 605 150 |
| $\frac{1}{2} = \frac{1}{2} = \frac{1}$ | 2,737,320,737 | 2,020,093,130 |
| (¢) | - 106 702 006 | + 66 670 651 |
| VY/ Accrecate Efficiency (c) | - 100,/02,000 2 000 /72 157 | |
| ABRIESALE DILICIENCY (5) | ఎ,000,473,157 | ∠, ४७४, ३४३, ३১I |

Source: Author's calculations.

\$4,657,735,992. An ordinal comparison of the values of a selected number of variables, for the two patterns A and B and the reference one R, is presented in table 4.19 for the four cases.

Case S_1E_1 - The average cost pricing policy (R) dominates the two marginal cost related pricing patterns A and B with respect to the economic efficiency criteria. The consumers would be the losers in patterns A and B while the utility would, in both cases, earn high excess revenues (\$92 million and \$52 million). This is so although pattern R implies very slightly higher annual gas consumption, and a new plant significantly higher than in case A (\$478,732,800), and is second ranked with respect to peak sales, load factor, and the equilibrium volumetric rate.

Case S_1E_2 - From the economic efficiency viewpoint, the reference average cost pricing pattern R dominates the two others. Note that the utility achieves excess revenues much smaller than in case S_1E_1 . Pattern R is also the optimal one with respect to total annual gas consumption and remains second ranked for the other variables.

Case $S_{2E_{1}}$ - From the economic efficiency viewpoint, pattern R again dominates the two others. Note that the utility achieves its overall highest excess revenues in case B (\$101,592,054). Pattern R is also the optimum one with respect to the total annual gas consumption and the equilibrium volumetric rate but turns out to be the least desirable with respect to load factor and amount of new plant in service.

Case S_2E_2 - Although here again pattern R dominates the two others with respect to aggregate efficiency, this is no longer so when the consumers and the utility are considered separately. Indeed, the total consumers' surplus in pattern A is, by a small margin, the highest one, mainly because of its industrial component. On the other side, it is under pattern A that the utility, for the first time, has a revenue deficit of quite a substantial magnitude (\$106,702,886). Pattern R achieves the lowest load factor and is second ranked with respect to the other variables.

138

| SOLULIONS KANKING | S ₀ | LUT | IONS | RANKING |
|-------------------|----------------|-----|------|---------|
|-------------------|----------------|-----|------|---------|

| | CAS | SE | |
|--|---|---|---|
| S ₁ E ₁ | S ₁ E ₂ | S ₂ E ₁ | S ₂ E ₂ |
| OMIC EFFIC | CIENCY | ali ang ang ang akarat ng ang akarat ng ang akarat ng ang akarat ng akarat ng akarat ng akarat ng akarat ng ak Mga mga mga mga ng akarat ng aka | |
| A <b<r A<b<r A<b<r A<r<b A<r<b R<b<a< td=""><td>A<b<r A<b<r R<a<b R<a<b A<b<r R<b<a< td=""><td>B<a<r B<a<r B<a<r B<r<a B<r<a R<a<b< td=""><td>B<a<r B<a<r B<a<r B<r<a B<r<a A<r<b< td=""></r<b<></r<a </r<a </a<r </a<r </a<r </td></a<b<></r<a </r<a </a<r </a<r </a<r </td></b<a<></b<r </a<b </a<b </b<r </b<r </td></b<a<></r<b </r<b </b<r </b<r </b<r | A <b<r A<b<r R<a<b R<a<b A<b<r R<b<a< td=""><td>B<a<r B<a<r B<a<r B<r<a B<r<a R<a<b< td=""><td>B<a<r B<a<r B<a<r B<r<a B<r<a A<r<b< td=""></r<b<></r<a </r<a </a<r </a<r </a<r </td></a<b<></r<a </r<a </a<r </a<r </a<r </td></b<a<></b<r </a<b </a<b </b<r </b<r | B <a<r B<a<r B<a<r B<r<a B<r<a R<a<b< td=""><td>B<a<r B<a<r B<a<r B<r<a B<r<a A<r<b< td=""></r<b<></r<a </r<a </a<r </a<r </a<r </td></a<b<></r<a </r<a </a<r </a<r </a<r | B <a<r B<a<r B<a<r B<r<a B<r<a A<r<b< td=""></r<b<></r<a </r<a </a<r </a<r </a<r |
| UMPTION/CO | DNSERVATIO | N | |
| B≺A≺R A≺R≺B B≺R≺A | R <b<a A<r<b B<r<a< td=""><td>R<b<a B<r<a R<a<b< td=""><td>B<r<a B<r<a R<b<a< td=""></b<a<></r<a </r<a </td></a<b<></r<a </b<a </td></r<a<></r<b </b<a | R <b<a B<r<a R<a<b< td=""><td>B<r<a B<r<a R<b<a< td=""></b<a<></r<a </r<a </td></a<b<></r<a </b<a | B <r<a B<r<a R<b<a< td=""></b<a<></r<a </r<a |
| HER VARIA | BLES | | |
| A <b<r< td=""><td>A<r<b A<r<b< td=""><td>B<a<r R≤B≤A</a<r </td><td>B<r<a A<r<b< td=""></r<b<></r<a </td></r<b<></r<b </td></b<r<> | A <r<b A<r<b< td=""><td>B<a<r R≤B≤A</a<r </td><td>B<r<a A<r<b< td=""></r<b<></r<a </td></r<b<></r<b | B <a<r R≤B≤A</a<r | B <r<a A<r<b< td=""></r<b<></r<a |
| | S ₁ E ₁ OMIC EFFIC A <b<r A<b<r A<b<r A<r<b A<r<c B A<r<c B A<r<c B C MPTION/CC B<a C A<r C C C C C C C C C C C C C C C C C C C</r </a </r<c </r<c </r<c </r<b </b<r </b<r </b<r | S1E1 S1E2 OMIC EFFICIENCY A <b<r< td=""> A<b<r< td=""> A<b<r< td=""> A<s<r< td=""> A<s< td=""> A<</s<></s<></s<></s<></s<></s<></s<></s<></s<></s<></s<></s<></s<></s<r<></b<r<></b<r<></b<r<> | S1E1S1E2S2E1OMIC EFFICIENCY $A < B < R$ $A < R < R$ $A < B < R$ $A < B < R$ $B < A < R$ $A < B < R$ $A < B < R$ $B < A < R$ $A < B < R$ $A < B < R$ $B < A < R$ $A < B < R$ $A < B < R$ $B < A < R$ $A < B < R$ $A < B < R$ $B < R < A < R$ $A < B < R$ $A < B < R$ $B < R < A < R$ $A < B < R$ $A < B < R$ $B < R < A < R$ $A < B < R$ $A < B < R < R < B < R < A < R$ $A < R < B$ $A < R < B$ $B < R < A < R < B$ $M > T < O < S < R < R < R < R < R < R < R < R < R$ |

Source: Author's calculations.

It would be unwise to draw from the previous analysis a definite conclusion about the feasibility and economic efficiency of marginal cost pricing for gas distribution utilities because of both the uncertainties bearing upon the values of various parameters and the failure to achieve an equilibrium. To reach such a conclusion (or its opposite), additional research and analyses are necessary. They are summarily outlined in the next section.

Possible Extensions of the Modeling Approach

The model presented in this chapter could be improved in at least two ways. First, the average cost pricing policy analysis could include a supply-demand equilibrium procedure similar to that used in the marginal

139

cost pricing policy analysis. Under such an approach, it could be possible to analyze the implications of various assumptions with a given set of demand functions. Second, <u>marginal cost based</u> pricing policies should be tested. Such pricing patterns should imply a lesser difference between peak and nonpeak prices, for instance through spreading the distribution capacity marginal cost over the winter months. In such a case, an equilibrium between supply and demand is more likely to be reached. Also, some theoretical research is called for with respect to the existence or conditions for the existence of a supply-demand equilibrium when prices are set equal to the marginal costs as computed in the present model.

Finally, it is obvious that the implications of marginal cost pricing for gas utilities should be analyzed while applying the model to other utilities. The data base for such applications is currently being prepared for the National Fuel Gas Company, which serves western parts of the states of New York and Pennsylvania, and for Pacific Gas and Electric Company, which serves northern and central California, and it is expected that the results of these applications will be reported in a future research report.

CHAPTER 5

SUMMARY

The purpose of this research effort was to develop methods for the calculation of the marginal costs of gas distribution utilities, and for the evaluation of gas pricing policies based on marginal costs. Two different approaches have been followed: (1) the distribution plant costs have been analyzed statistically with community-level data, and econometric models predicting these costs on the basis of such variables as market size and mix, population density, and weather have been specified; (2) an aggregate, nonspatialized optimization model has been developed to calculate monthly supply, storage, and transmission marginal costs, and this model has been embedded into a larger simulation model analyzing the implications of marginal cost pricing policies.

The major results of the econometric analysis are that (1) the total distribution plant costs incurred to serve different sectoral markets are nonseparable; (2) economies of scale are achieved with respect to both residential and nonresidential gas sales; (3) the community's population density is generally an important determinant of distribution plant costs, and so is the weather pattern if the service territory is climatologically heterogeneous. The above results imply that the marginal plant costs with respect to sectoral sales decrease with the sector's size and depend upon the size of the other sector(s).

The major results of the optimization/simulation analysis are that (1) marginal costs highly depend upon supply conditions (maximum availability, charges, contracts, etc.) and upon various technological

141

constraints; (2) peak-shifting problems are very likely to occur if distribution capacity marginal costs are wholly assigned to the peak period (month); (3) the excess revenue problem does not necessarily always occur, and its occurrence depends upon supply conditions, costs, technological constraints, financial parameters, and the price elasticities of the monthly demands. It would be premature to draw final conclusions from this partial analysis, but it should be noted that the results do not clearly demonstrate the superiority of this marginal cost pricing method.

The previous analyses can of course be improved and further developed in a number of ways. The econometric models could probably be improved by including such variables as sectoral load factors and average customer sizes. An important endeavor would be to explain fully the interutility differences in the estimated parameters. The optimization/simulation model should be used to test marginal cost based pricing policies, wherein peakcapacity marginal costs could be spread over longer periods. Such policies might help to avoid the peak-shifting problem and turn out to be preferable to a pure marginal cost pricing policy. Finally, this optimization/simulation model might be spatialized through a complete network representation of the system (as outlined in chapter 2) in order to compute timeand space-related marginal costs. This spatialization would be the appropriate way to integrate the econometric and optimization/simulation approaches presented in this study.

142

APPENDIX A

LONG ISLAND LIGHTING COMPANY DATA

The purpose of this appendix is to present the community-level data used in the econometric analysis of the Long Island Lighting Company, as reported in chapter 3. The data are indicated for 101 different villages, towns, and cities, and for two years - 1978 and 1979. The plant in service and the residential and commercial/-industrial gas sales are presented in table A-1. The residential and commercial/industrial numbers of customers, the 1978 population, and the corresponding total and residential acreage are presented in table A-2.

PLANT IN SERVICE AND GAS SALES - LONG ISLAND LIGHTING COMPANY

| | | Plent In | Service (\$) | Gas S | Sales (MCF) - 19 | 179 | Gas S | Sales (MCF) - 19 | 78 |
|------|--------------------|-------------|--------------|-------------|------------------|-------------|-------------|------------------|-------------|
| Case | Municipality | End of 1979 | End of 1978 | Total | Residential | Comm./Ind. | Total | Residential | Comm./Ind. |
| 1 | Town of Hempstead | 46,980,146 | 45,435,491 | 7,382,642.5 | 4,475,445.1 | 2,907,197,4 | 6,962,310,4 | 4,598,579.0 | 2,363,731.4 |
| | Village of: | | | | | | | | |
| 2 | Atlantic Beach | 264,698 | 262,219 | 87,032.4 | 76,045.9 | 10,986.5 | 91,436.4 | 77,904.7 | 13,531.7 |
| 3 | Bellerose | 111,262 | 110,668 | 27,914.6 | 25,421.4 | 2,493.2 | 28,375.1 | 26,027.2 | 2,347,9 |
| 4 | Cederhurst | 528,631 | 508,407 | 146,444.0 | 98,210.6 | 48,233.4 | 149,163.4 | 100,397.4 | 48,766.0 |
| 5 | East Rockway | 754,654 | 746, 384 | 193,355.6 | 124,645.2 | 68,710.4 | 180,735.6 | 128,915.6 | 51,820.0 |
| 6 | Floral Park | 1,045,572 | 987,715 | 270,438.8 | 232,954.2 | 37,484.6 | 268,441.2 | 236,401.0 | 32,040.2 |
| 7 | Freeport | 3,491,923 | 3,287,353 | 639,616.6 | 340,004.7 | 299,611.9 | 615,635.2 | 345,217.0 | 270,418.2 |
| 8 | Garden City | 4,066,477 | 4,043,941 | 581,456.3 | 397,691.2 | 183,765.1 | 557,572.6 | 406,813.7 | 150,758.9 |
| 9 | Hempstead | 3,932,371 | 3,881,974 | 761,152.0 | 462,577.5 | 298, 574. 5 | 775,942.2 | 477,430.7 | 298,511.5 |
| 10 | Hewlett Bay Park | 145,727 | 139,947 | 28,272.8 | 27,666.9 | 605.9 | 29,077.4 | 28,337.6 | 739.8 |
| 11 | Hewlett Harbor | 315,672 | 305,003 | 52,566.1 | 51,354.3 | 1,211.8 | 53,414.0 | 52,138.5 | 1,275.5 |
| 12 | Hewlett Neck | 80, 833 | 78,969 | 17,252.6 | 17,252.6 | | 17,838.5 | 17,838,5 | |
| 13 | Island Park | 1,914,524 | 1,911,004 | 115,263.1 | 86,472.3 | 28,790.8 | 116, 323.8 | 87,440.8 | 28,883.0 |
| 14 | Lawrence | 809,159 | 779,088 | 239,459.9 | 206,015.8 | 33, 444.1 | 253,406.2 | 214,263.8 | 39, 142. 4 |
| 15 | City of Long Beach | 7,268,240 | 7,250,461 | 612,912.6 | 346,826.2 | 266,086.4 | 626,477.7 | 354, 588. 8 | 271,888.9 |
| | Village of: | | | | | | | | |
| 16 | Lynbrook | 1,785,069 | 1,756,340 | 384,302.3 | 241,902.9 | 142,399.4 | 392,465.0 | 243,012.6 | 149,452.4 |
| 17 | Malverne | 578,674 | 568,026 | 153,273.0 | 141,859.8 | 11,413.5 | 155,982.1 | 145,032.9 | 10,949.2 |
| 18 | New Hyde Park | 239,645 | 224,053 | 73,344.9 | 50,416.3 | 22,928.6 | 69,043.2 | 50,249.1 | 18,794.1 |
| 19 | Rockville Centre | 3,225,988 | 3,177,206 | 479,375.0 | 239,917.2 | 239,457.8 | 434,456.0 | 240,669.9 | 193,786.1 |
| 20 | South Floral Park | 181,429 | 169,697 | 19,442.6 | 19,439.3 | 3.3 | 19,567.3 | 19,563.8 | 3.5 |

| TABLE A-1 |
|-----------|
|-----------|

PLANT IN SERVICE AND GAS SALES - LONG ISLAND LIGHTING COMPANY (Continued)

| | | Plant In | Service (\$) | Gas S | Sales (MCF) - 19 | MCF) - 1979 Gas Sales (MCF) - 19 | | | |
|------|----------------------|-------------|--------------|-------------|------------------|----------------------------------|-------------|---------------------------------------|------------|
| Саве | Municipality | End of 1979 | End of 1978 | Total | Residential | Comm./Ind. | Total | Residential | Comm./Ind. |
| | Village of: | ······ | | | | | | · · · · · · · · · · · · · · · · · · · | |
| 21 | Stewart Manor | 173,930 | 168,883 | 41.510.9 | 37.677.3 | 3,833.6 | 42,483.3 | 38, 520, 3 | 3,963,0 |
| 22 | Valley Stream | 2,865,374 | 2,742,910 | 715,290,5 | 543,374.6 | 171,915.9 | 732,533.6 | 555, 356, 7 | 177,176.9 |
| 23 | Woodsburgh | 102,715 | 93,256 | 27,082.6 | 25, 536, 5 | 1,546.1 | 29,498.1 | 27,705.0 | 1,793.1 |
| 24 | Town of N. Hempstead | 8,812,002 | 8,616,141 | 2,344,402.8 | 1,381,263.7 | 963,139.1 | 2,340,527.3 | 1,425,974.1 | 914, 553.2 |
| | Village of | | | | | | | | r |
| 25 | Baxter Estates | 78,486 | 76,358 | 16,356.3 | 10,032.4 | 6,323.9 | 16,241.4 | 10,164.0 | 6,077.4 |
| 26 | East Hills | 409,224 | 396, 526 | 81,588.3 | 51,488.7 | 30,099.6 | 87,066.5 | 55, 503, 5 | 31,563.0 |
| 27 | East Williston | 275,521 | 253,448 | 36,676.4 | 35,546.7 | 1,129.7 | 37,626.4 | 36,357.2 | 1,269.2 |
| 28 | Floral Park | 183,221 | 178,534 | 40,180.3 | 30,451.5 | 9,728.8 | 41,949.3 | 31,539.8 | 10,409.5 |
| 29 | Flower Hill | 500,309 | 495,373 | 76,690.8 | 59,994.7 | 16,696.1 | 80,791.6 | 62,599.5 | 18,192.1 |
| 30 | Great Neck | 872,134 | 839,705 | 174,773.7 | 142,964.9 | 31,808.8 | 179,139.9 | 146,833.0 | 32,306.9 |
| 31 | Great Neck Estates | 263, 374 | 253,618 | 76,096.2 | 65,627.9 | 10,468.3 | 80,265.6 | 69,559.6 | 10,706.0 |
| 32 | Great Neck Plaza | 263,421 | 251,086 | 133,723.2 | 14,515.6 | 119,207.6 | 119,686.6 | 15,745.1 | 103,941.5 |
| 33 | Kensington | 81,153 | 81,692 | 34,851.0 | 26,057.9 | 8,793.1 | 34,376.0 | 26,671.2 | 7,704.8 |
| 34 | Kings Point | 697,278 | 645,583 | 139,764.5 | 114,665.5 | 25,099.0 | 209,176.2 | 123, 469. 5 | 85,706.7 |
| 35 | Lake Success | 691,283 | 709, 362 | 234,894.6 | 49,237.2 | 185,657.4 | 114,397.5 | 51,500.4 | 62,897.1 |
| 36 | Manorhaven | 361,401 | 351,420 | 67,230,1 | 42,143.7 | 25,086.4 | 68,562.8 | 43, 164. 4 | 25,398.4 |
| 37 | Mineola | 1,306,279 | 1,254,710 | 499,668.7 | 209,265.6 | 290,403.1 | 420,881.6 | 211,540.5 | 209,341.1 |
| 38 | Munsey Park | 195,292 | 189,123 | 42,737.2 | 40,947.5 | 1,789.7 | 44,708.4 | 42,734.4 | 1,974.0 |
| 39 | New Hyde Park | 388,062 | 376,891 | 93,095.3 | 75,494.2 | 17,601.1 | 93,532.9 | 75,832.5 | 17,700.4 |
| 40 | North Hills | 429,374 | 413,061 | 18,260.2 | 1,313.2 | 16,947.0 | 28,173.3 | 9,149.6 | 19,023.7 |

PLANT IN SERVICE AND GAS SALES - LONG ISLAND LIGHTING COMPANY (Continued)

| | | Plant In | Service (\$) | Gas S | Sales (MCF) - 19 | 79 | Gas S | ales (MCF) - 19 | 78 |
|------|------------------------|-------------|--------------|-------------|------------------|-------------|--|---|-------------|
| Case | Municipality | End of 1979 | End of 1978 | Total | Residential | Comm./Ind. | Total | Residential | Comm./Ind. |
| | Village of: | | | | | | an a | an na hafan da na an | |
| 41 | Old Westbury | 529,566 | 534,343 | 86,303.7 | 22,640.6 | 63,663.1 | 47,707.7 | 21,563.8 | 26,143.9 |
| 42 | Plandome | 100,872 | 84,284 | 27,976.1 | 27,832.9 | 143.2 | 29,006.7 | 28,845.8 | 160.9 |
| 43 | Plandome Heights | 74,672 | 69,212 | 18,903.6 | 18,903.6 | | 19,541.1 | 19,541.1 | 03103 |
| 44 | Plandome Manor | 79,228 | 77, 386 | 16,982.6 | 15,201.0 | 1,781.6 | 18,190.1 | 16,572.5 | 1,617.6 |
| 45 | Port Washington N. | 329,446 | 328,929 | 137,033.6 | 88,857.6 | 48,176.0 | 134,191.4 | 89, 500. 9 | 44,690.5 |
| 46 | Roslyn | 554,355 | 547,619 | 77,281,1 | 12,626.5 | 64,654.6 | 75,508.7 | 13,783.3 | 61,725.4 |
| 47 | Roslyn Estates | 122,099 | 119,570 | 29,141.4 | 24,713.7 | 4,427.7 | 29,527.6 | 26,048.4 | 3,479.2 |
| 48 | Roslyn Harbor | 362,748 | 361,265 | 17,503,3 | 15,367.9 | 2,135.4 | 19,628.8 | 17,577.4 | 2,051.4 |
| 49 | Russell Gardens | 95,248 | 95,506 | 25,828.7 | 14,919.1 | 10,909.6 | 25,493.5 | 16,705.5 | 8,788.0 |
| 50 | Saddle Rock | 40,495 | 46,608 | 5,724.6 | 4,989.6 | 735.0 | 6,410.1 | 5,618.4 | 791.7 |
| 51 | Sands Point | 88,223 | 89,678 | 15,457.0 | 13,466.2 | 1,990.8 | 15,436.9 | 13,220.2 | 2,216.7 |
| 52 | Thomaston | 541,583 | 541,527 | 54,723.2 | 43,608.1 | 11,115.1 | 57,377.3 | 44,879.4 | 12,497.9 |
| 53 | Westbury | 1,559,449 | 1,347,889 | 232,105.5 | 163,936.6 | 68,168.9 | 236, 219. 5 | 165,912.2 | 70,307.3 |
| 54 | Williston Park | 431,152 | 408,462 | 146,347.4 | 123,381.5 | 22,965.9 | 149,867.8 | 124,141.1 | 25,726.7 |
| 55 | Town of Oyster Bay | 28,140,961 | 26,964,148 | 4,537,342.2 | 1,903,364.6 | 2,633,977.6 | 4,095,979.7 | 1,978,531.9 | 2,117,447.8 |
| | Village of: | | | | | | | | |
| 56 | Bayville | 478,867 | 458,814 | 131,058.3 | 113,755.3 | 17,303.0 | 130,945.7 | 114,756.9 | 16,188.8 |
| 57 | Brookville | 371,003 | 358,567 | 94,237.6 | 13,144.7 | 81,092.9 | 46,440.8 | 14,361.9 | 32,078,9 |
| 58 | Farmingdale | 936, 281 | 734,474 | 78,736.7 | 35, 594. 5 | 43,142.2 | 80, 398. 1 | 35,840.4 | 44,557.7 |
| 59 | City of Glen Cove | 2,501,348 | 2,474,959 | 564,490.3 | 253,457.4 | 311,032.9 | 460,020.6 | 265,413.7 | 194,606.9 |
| 60 | Village of Lattingtown | 85,175 | 73,634 | 11,604.5 | 10,147.6 | 1,456.9 | 12,754.1 | 10,900.3 | 1,853.8 |

PLANT IN SERVICE AND GAS SALES - LONG ISLAND LIGHTING COMPANY (Continued)

| icipality | | | Plant In Service (\$) Gas Sales (MCF) - 1979 | | | | Gas Sales (MCF) - 1978 | | | |
|-----------------------------------|-------------|---|--|---|--|---|--|---|--|--|
| Municipality | End of 1979 | End of 1978 | Total | Residential | Comm./Ind. | Total | Residential | Comm./Ind. | | |
| of: | | | | | | | | | | |
| l Hollow | 8,972 | 8,972 | 1,556,0 | (4.7) | 1,560,7 | 1.504.8 | 24.4 | 1,480,4 | | |
| negua Park | 1.060.091 | 947, 593 | 137,403,5 | 107.509.6 | 29,893,9 | 133 262.0 | 109,216,9 | 24,045,1 | | |
| ecock | 93,122 | 96,166 | 18,983,5 | 13.019.7 | 5,963,8 | 17,882,4 | 13, 336, 4 | 4,546.0 | | |
| Neck | 108,276 | 106,827 | 6,242,7 | 6.242.7 | ., | 5,853,3 | 5,853,3 | 24,761.3 | | |
| ດໂດຟກ | 414,143 | 409,128 | 71,201,8 | 48,688,9 | 22,512,9 | 77.390.0 | 52,628,7 | 11,107,9 | | |
| rookville | 87,380 | 87.768 | 25,194,2 | 13.643.4 | 11.550.8 | 25,501,4 | 14, 393, 5 | | | |
| r Bay Cove | 143,116 | 143,660 | 22,038,4 | 22.038.4 | ,, | 23,100.0 | 23,100.0 | | | |
| a Harbor | 167,088 | 163,564 | 13.056.5 | 13,056,5 | | 13,369,4 | 13,369,4 | | | |
| liff | 969,885 | 934,454 | 104,494.8 | 94,553,4 | 9,941.4 | 108,268.3 | 98,178,6 | 10,089.7 | | |
| Brookville | 50,742 | 49,401 | 5,872.0 | 4,478.7 | 1,393.3 | 5, 593.4 | 4,287,2 | 1,306.2 | | |
| Babylon | 13,634,907 | 13,140,246 | 2,511,961,4 | 1,302,246,4 | 1,209,715.0 | 2,456,395.8 | 1,373,302,1 | 1,083,093.7 | | |
| of: | | | | | | | | | | |
| ville | 1,164,722 | 1,096,064 | 186,306.1 | 101,035.2 | 85,270,9 | 172,461.6 | 103,706.0 | 68,755.6 | | |
| on | 1,735,463 | 1,609,857 | 130,913.0 | 102,836.7 | 28,076.3 | 134,137.6 | 105,758.0 | 28,379.6 | | |
| nhurst | 2,615,850 | 2,559,887 | 288, 314.8 | 196,766.3 | 91, 548, 5 | 298, 229, 6 | 202,744.1 | 95,485.5 | | |
| Brookhaven | 28,662,610 | 28,474,846 | 3,057,570.4 | 2,074,118.1 | 983,452.3 | 3,063,332.5 | 2,221,412.1 | 841,920.4 | | |
| of: | | | | | | | | | | |
| Terre | 30,644 | 30,644 | 2,443.9 | 2,410.2 | 33.7 | 2,699.7 | 2,633.4 | 66.3 | | |
| ort | 93, 327 | 91,212 | 10,832.2 | 8,500.8 | 2,331.4 | 11,868.7 | 9,167.2 | 2,701.5 | | |
| leld | 41,568 | 37,201 | 5,596.6 | 5,596.6 | | 6,165.6 | 6,165.6 | | | |
| ogue | 926, 307 | 875,718 | 177,986.5 | 62,782.4 | 115,204.1 | 170,100.7 | 66,374.8 | 103,725.9 | | |
| ŧ | 5,093 | 5,093 | 1,958.1 | 1,958.1 | | 1,902,1 | 1,902.1 | | | |
| of: Ter ort leld ogue | re | re 30,644 93,327 41,568 926,307 5,093 | re 30,644 30,644 93,327 91,212 41,568 37,201 926,307 875,718 5,093 5,093 | re 30,644 30,644 2,443.9 93,327 91,212 10,832.2 41,568 37,201 5,596.6 926,307 875,718 177,986.5 5,093 5,093 1,958.1 | re 30,644 30,644 2,443.9 2,410.2 93,327 91,212 10,832.2 8,500.8 41,568 37,201 5,596.6 5,596.6 926,307 875,718 177,986.5 62,782.4 5,093 5,093 1,958.1 1,958.1 | re 30,644 30,644 2,443.9 2,410.2 33.7 93,327 91,212 10,832.2 8,500.8 2,331.4 41,568 37,201 5,596.6 5,596.6 926,307 875,718 177,986.5 62,782.4 115,204.1 5,093 5,093 1,958.1 1,958.1 | re 30,644 30,644 2,443.9 2,410.2 33.7 2,699.7 93,327 91,212 10,832.2 8,500.8 2,331.4 11,868.7 41,568 37,201 5,596.6 5,596.6 6,165.6 926,307 875,718 177,986.5 62,782.4 115,204.1 170,100.7 5,093 5,093 1,958.1 1,958.1 1,958.1 | re 30,644 30,644 2,443.9 2,410.2 33.7 2,699.7 2,633.4 93,327 91,212 10,832.2 8,500.8 2,331.4 11,868.7 9,167.2 41,568 37,201 5,596.6 5,596.6 6,165.6 6,165.6 926,307 875,718 177,986.5 62,782.4 115,204.1 170,100.7 66,374.8 5,093 5,093 1,958.1 1,958.1 1,958.1 1,902.1 1,902.1 | | |

PLANT IN SERVICE AND GAS SALES - LONG ISLAND LIGHTING COMPANY (Continued)

| 81 82 83 84 85 86 87 | | Plant In | Service (\$) | Gas S | ales (MCF) - 19 | 179 | Gas S | ales (MCF) - 19 | 978 |
|--|----------------------|-------------|--------------|--|--|--|-------------|--|-------------|
| Сазе | Municipality | End of 1979 | End of 1978 | Total | Residential | Comm./Ind. | Total | Residential | Comm./Ind. |
| | Village of: | | | angeneration and the Station of the second | 9007 - 80 MW 400,000 - 10 - 20 - 20 - 20 MW - 10 M | 999-99 49 49 49 49 49 49 49 49 49 49 49 49 4 | <u></u> | φ. «δαναθετά βαντία «Ολιντία» «Αγγοργγαριατική στη πογο φ | |
| 81 | Port Jefferson | 482.526 | 459,112 | 98, 583, 4 | 53.174.4 | 45,409,0 | 105.849.5 | 56,282,6 | 49,566,9 |
| 82 | Lake Grove | 469,750 | 470,550 | 127,548.1 | 75,285,6 | 52,262,5 | 123,666.8 | 81,427,9 | 42,238,9 |
| 83 | Town of East Hampton | 349,463 | 332,952 | 31,440,7 | 21.398.4 | 10,042.3 | 33,483.9 | 22.874.7 | 10,609.2 |
| | Village of: | - • • | | · - | • | • | • | | |
| 84 | East Hampton | 731,615 | 714,225 | 127,958.6 | 104,792,3 | 23,166,3 | 134,256,5 | 108,521.2 | 25,735,3 |
| 85 | Sag Habor | 149, 591 | 148,192 | 15,497,9 | 14,726.7 | 771.2 | 14,540.6 | 13,895.0 | 645.6 |
| 86 | Town of Huntington | 17,232,837 | 16,763,582 | 3,601,048.3 | 2,505,424.0 | 1,095,624.3 | 3,592,536.5 | 2,662,009.1 | 930,527,4 |
| | Village of: | | | | | | | | |
| 87 | Asharoken | 18,235 | 16,912 | 9,346.8 | 9,274.0 | 72.8 | 9,909.5 | 9,836.7 | 72.8 |
| 88 | Huntington Bay | 137,151 | 130,715 | 37,850.5 | 36,355.9 | 1,494.6 | 39,970.5 | 38,483.8 | 1,486.7 |
| 89 | Lloyd Harbor | 26,053 | 26,053 | 5,935.9 | 4,590.2 | 1,345.7 | 6,287.0 | 4,766.5 | 1,520.5 |
| 90 | Northport | 562,781 | 562,932 | 136,929.5 | 88,167.4 | 48,762.1 | 119,277.7 | 91,288.0 | 27,989.7 |
| 91 | Town of Islip | 21,774,198 | 20,776,994 | 4,643,850.1 | 2,696,224.0 | 1,947,626.1 | 4,302,760.5 | 2,850,006.0 | 1,452,754.5 |
| | Village of: | | | | | | | | |
| 92 | Brightwaters | 565,747 | 532,958 | 60,906.5 | 50,858.1 | 10,048.4 | 60,941.2 | 50,812.5 | 10,128.7 |
| 93 | Town of Riverhead | 3,599,077 | 3,474,546 | 203,746.7 | 87,292.4 | 116,454.3 | 200,198.6 | 91,125.6 | 109,073.0 |
| 94 | Town of Smithtown | 8,716,555 | 8,598,667 | 1,917,720.1 | 1,451,195.8 | 466,524.3 | 1,985,300.1 | 1,559,929.4 | 425,370.7 |
| | Village of: | | | | | | | | |
| 95 | Head of the Harbor | 21,219 | 21,219 | 4,360.1 | 3,128.2 | 1,231.9 | 4,084.3 | 3,195.5 | 888.8 |
| 96 | The Branch | 79,985 | 75,028 | 27,525.6 | 3,265.4 | 24,260.2 | 26,441.6 | 3,393.9 | 23,047.7 |
| 97 | Town of Southampton | 3,356,573 | 3,343,291 | 194,196.9 | 120,865.4 | 73,331.5 | 200,232.6 | 126,802.1 | 73,430.5 |
| | Village of: | | | | | | | | |
| 98 | Sag Harbor | 323,867 | 319,300 | 38,262.8 | 28,432.3 | 9,830.5 | 39,062.9 | 29,303.7 | 9,759.2 |
| 99 | Southampton | 1,062,846 | 1,042,636 | 182,061.5 | 137,019.9 | 45,041.6 | 181,930.4 | 135,111.9 | 46,818.5 |
| 100 | Town of Southold | 1,092,078 | 1,070,275 | 155,251.6 | 115,662.2 | 39,589.4 | 161,359.1 | 119,108.7 | 42,250.4 |
| 101 | City of New York | 6,946,156 | 7,082,285 | 1,729,420.9 | 1,018,890.4 | 696,665.7 | 1,681,734.3 | 1,059,414.9 | 608,626.4 |
| | | | | | | | | | |

Source: Annual Reports fo the New Public Service Commission, 1978-1979, Long Island Lighting Company

NUMBER OF CUSTOMERS, POPULATION AND ACREAGE - LONG ISLAND LIGHTING COMPANY

| | | Numb | er of Customers | - 1979 | Number | r of Customers | - 1978 | | | |
|------|-----------------------------------|--------|-----------------|------------|--|--|--------|--------------------|------------------|------------------------|
| Case | Municipality | Total | Residential | Comm./Ind. | n a se a s | 98-28-48-48-48-48-48-48-48-48-48-48-48-48-48 | 4: | Population 1978 | Total Acreage | Residential Acreage |
| 1 | Town of Hempstead | 76,334 | 70,886 | 5,448 | 76,218 | 70,795 | 5,423 | 542,402 | 83,200 | 0 |
| • | Village of: | 3-1 | 71.6 | 20 | 764 | | 20 | 1 (0) | 200 | 270 |
| 2 | Atlantic Beach | /54 | /15 | 39 | /50 | /1/ | 39 | 1,024 | 388 | 270 |
| 3 | Bellerose | 362 | 345 | 1/ | 360 | 345 | 15 | 1,090 | 114 | 91 |
| 4 | Cedarhurst | 2,108 | 1,8// | 231 | 2,105 | 1,8// | 229 | 6,928 | 436 | 383 |
| 5 | East Rockway | 2,528 | 2,363 | 165 | 2,520 | 2,358 | 162 | 11,401 | 831 | 496 |
| 6 | Floral Park | 4,577 | 4,405 | 172 | 4,545 | 4,3/5 | 170 | 16,10/ | 896 | 0 |
| 7 | Freeport | 7,449 | 6,872 | 577 | 7,443 | 6,866 | 577 | 40,997 | 3,219 | 2,544 |
| 8 | Garden City | 5,319 | 5,129 | 190 | 5,294 | 5,096 | 198 | 24,914 | 3,505 | 2,404 |
| 9 | Hempstead | 9,635 | 8,878 | 757 | 9,649 | 8,864 | 785 | 40,365 | 2,360 | 1,773 |
| 10 | Hewlett Bay Park | 134 | 130 | 4 | 133 | 129 | 4 | 601 | 213 | 174 |
| 11 | Hewlett Harbor | 231 | 229 | 2 | 229 | 227 | 2 | 1,501 | 534 | 436 |
| 12 | Hewlett Neck | 120 | 120 | | 119 | 119 | | 541 | 192 | 157 |
| 13 | Island Park | 1,283 | 1,163 | 120 | 1,285 | 1,161 | 124 | 5,578 | 269 | 234 |
| 14 | Lawrence | 1,941 | 1,854 | 87 | 1,954 | 1,861 | 93 | 6,425 | 3,007 | 1,991 |
| 15 | City of Long Beach Village of: | 8,837 | 8,022 | 81 5 | 8,795 | 7,975 | 820 | 34,546 | 1,564 | 1,364 |
| 16 | Lyphrook | 5.700 | 5, 193 | 507 | 5,693 | 5,187 | 506 | 22.853 | 1,280 | 0 |
| 17 | Malverne | 2,602 | 2 509 | 93 | 2,595 | 2,500 | 95 | 10,024 | 308 | 273 |
| 18 | New Hyde Park | 1 126 | 1,088 | 38 | 1 129 | 1,089 | 40 | 4,263 | 299 | 198 |
| 19 | Rockyille Contro | 5,637 | 5,222 | 415 | 5 644 | 5,233 | 411 | 28,535 | 2.148 | 1.607 |
| 2.0 | South Floral Park | 296 | 295 | 1 | 293 | 292 | 1 | 1,105 | 83 | 58 |

NUMBER OF CUSTOMERS, POPULATION AND ACREAGE - LONG ISLAND LIGHTING COMPANY (Continued)

| | | Numb | er of Customers | a — 1979 | Numbe | r of Customer | s — 1978 | | 9999 - P Anno 1999 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 19 | Residential Acreage |
|------|----------------------|--------|-----------------|------------|---|--|--|--------------------|---|------------------------|
| Саве | Municipality | Total | Residential | Comm./Ind. | ილესიტიატიკელტიაფეონებიტი იღეთანიადიადი | 993 * EL-9777 9 (9) * 7) * 83 * 93 * 93 * 93 * 93 * 93 * 93 * 93 | an a | Population 1978 | Total Acreage | |
| | Village of: | | | | | | | | | |
| 21 | Stewart Manor | 607 | 580 | 27 | 609 | 582 | 27 | 2,147 | 127 | 105 |
| 22 | Valley Stream | 10,462 | 9,833 | 629 | 10,473 | 9,842 | 631 | 40,200 | 4,215 | 2,079 |
| 23 | Woodsburgh | 248 | 243 | 5 | 247 | 242 | 5 | 806 | 284 | 232 |
| 24 | Town of N. Hempstead | 22,154 | 20,510 | 1,644 | 22,158 | 20,494 | 1,664 | 103,742 | 35,840 | 0 |
| | Village of | - | • | - | | | | - | · | |
| 25 | Baxter Estates | 231 | 198 | 33 | 227 | 196 | 31 | 1,036 | 0 | 0 |
| 26 | East Hills | 437 | 367 | 70 | 439 | 366 | 73 | 8,708 | 1,408 | 0 |
| 27 | East Williston | 514 | 509 | 5 | 509 | 504 | 5 | 2,805 | 370 | 292 |
| 28 | Floral Park | 734 | 693 | 41 | 732 | 689 | 43 | 1,936 | 118 | 118 |
| 29 | Flower H111 | 563 | 528 | 35 | 561 | 525 | 36 | 4,610 | 1,055 | 885 |
| 30 | Great Neck | 2,569 | 2,425 | 144 | 2,565 | 2,421 | 144 | 10,661 | 904 | 852 |
| 31 | Great Neck Estates | 803 | 763 | 40 | 800 | 761 | 39 | 3,082 | 327 | 278 |
| 32 | Great Neck Plaza | 1,987 | 1,765 | 222 | 1,984 | 1,759 | 225 | 6,113 | 203 | 95 |
| 33 | Kensington | 244 | 241 | 3 | 244 | 241 | 3 | 1,605 | 168 | 147 |
| 34 | Kings Point | 648 | 628 | 20 | 640 | 623 | 17 | 5,799 | 2,559 | 2,483 |
| 35 | Lake Success | 383 | 336 | 47 | 385 | 337 | 48 | 3,434 | 1,045 | 692 |
| 36 | Manorhaven | 514 | 442 | 72 | 510 | 440 | 70 | 5,911 | 256 | 0 |
| 37 | Mineola | 5,351 | 4,909 | 442 | 5,342 | 4,894 | 448 | 20,497 | 1,190 | 821 |
| 38 | Munsey Park | 530 | 519 | 11 | 528 | 518 | 10 | 3,004 | 333 | 333 |
| 39 | New Hyde Park | 1,700 | 1,596 | 104 | 1,696 | 1,587 | 109 | 5,907 | 299 | 198 |
| 40 | North Hills | 38 | 25 | 13 | 40 | 26 | 14 | 1,329 | 563 | 248 |

NUMBER OF CUSTOMERS, POPULATION AND ACREAGE - LONG ISLAND LIGHTING COMPANY (Continued)

| | | Numb | er of Customers | - 1979 | Number | r of Customers | - 1978 | | | Residential Acreage |
|------|------------------------|--------|-----------------|------------|--|--|--|--------------------|------------------|------------------------|
| Case | Municipality | Total | Residential | Comm./Ind. | ₩Ĵ~ĠĸĠĸġĸġĊĸġŦĸġĸġĊĸġĊĸġĊĸġĸĸţŦĸġĿĸġ ĸĸġ~ĠĸġĸĸġĸĸġŦĸġŦĸġċĸġĊĸġŎĸġĿĸġĸĸġ | alpendentingundung begindentingendentingen | ෮෩෯෩෯෮෭෫෮෭෨෫෯෯෯෯෯෯෯෯෯෯෯෯෯෯ ෮෪෪෫෪෪෯෨෩෩෫෪෪෪෯෯෨෯෪෯෨෯෪෯ | Population 1978 | Total Acreage | |
| | Village of: | | | | | | | | | |
| 41 | Old Westbury | 101 | 88 | 13 | 99 | 87 | 12 | 2,389 | 3,637 | 3,407 |
| 42 | Plandome | 287 | 283 | 4 | 286 | 282 | 4 | 1,604 | 375 | 292 |
| 43 | Plandome Heights | 246 | 246 | | 244 | 133 | | 1,056 | 247 | 192 |
| 44 | Plandome Manor | 111 | 107 | 4 | 110 | 106 | 4 | 820 | 192 | 149 |
| 45 | Port Washington N. | 606 | 538 | 68 | 608 | 540 | 68 | 3,009 | 330 | 211 |
| 46 | Roslyn | 631 | 537 | 94 | 628 | 532 | 96 | 2,621 | 717 | 503 |
| 47 | Roslyn Estates | 202 | 195 | 7 | 203 | 195 | 8 | 1,438 | 394 | 276 |
| 48 | Roslyn Harbor | 93 | 89 | 4 | 92 | 88 | 4 | 848 | 232 | 163 |
| 49 | Russell Gardens | 315 | 302 | 13 | 314 | 301 | 13 | 1,103 | 117 | 98 |
| 50 | Saddle Rock | 35 | 30 | 5 | 35 | 30 | 5 | 885 | 330 | 290 |
| 51 | Sands Point | 76 | 74 | 2 | 77 | 75 | 2 | 3,112 | 2,302 | 1,429 |
| 52 | Thomaston | 704 | 679 | 25 | 700 | 675 | 25 | 2,648 | 281 | 244 |
| 53 | Westbury | 3,106 | 2,872 | 234 | 3,087 | 2,853 | 234 | 15,924 | 1,525 | 1,525 |
| 54 | Williston Park | 2,190 | 2,056 | 134 | 2,183 | 2,043 | 140 | 8,923 | 751 | 497 |
| 55 | Town of Oyster Bay | 33,723 | 30,258 | 3,465 | 33,642 | 30,219 | 3,423 | 282,159 | 72,960 | 0 |
| | Village of: | · | • | - | | - | | | - | |
| 56 | Bayville | 1,459 | 1,396 | 63 | 1,452 | 1,392 | 60 | 6,981 | 896 | 0 |
| 57 | Brookville | 67 | 51 | 16 | 64 | 49 | 15 | 3,435 | 4,266 | 2,683 |
| 58 | Farmingdale | 1,310 | 1,092 | 218 | 1,307 | 1,091 | 216 | 9,568 | 72.2 | 624 |
| 59 | City of Glen Cove | 4,920 | 4,577 | 343 | 4,931 | 4,589 | 342 | 27,684 | 4,460 | 3,572 |
| 60 | Village of Lattingtown | 88 | 83 | 5 | 88 | 82 | 6 | 1,912 | 3,076 | 1,177 |

NUMBER OF CUSTOMERS, POPULATION AND ACREAGE - LONG ISLAND LIGHTING COMPANY (Continued)

| Designation of the second | | Numbo | er of Customers | 3 - 1979 | Numbe | r of Customers | s - 1978 | - | | Residential Acreage |
|---------------------------|--------------------|--------|-----------------|------------|--|---|---|--------------------|------------------|------------------------|
| Case | Municipality | Total | Residential | Comm./Ind. | terdendenstationspreizenspreizenstations | na na si na ang mangangang mangang mang | 56477142841667184294126412641264126412641264126412641264126 | Population 1978 | Total Acreage | |
| | Village of: | | | | | | | | | |
| 61 | Laurel Hollow | 10 | 1 | 9 | 10 | 2 | 8 | 1,560 | 2,897 | 3,188 |
| 62 | Massapequa Park | 2,419 | 2,314 | 105 | 2,398 | 2,303 | 95 | 22,200 | 1,393 | 1,341 |
| 63 | Matinecock | 80 | 63 | 17 | 80 | 63 | 17 | 886 | 1,425 | 544 |
| 64 | Mill Neck | 34 | 34 | | 32 | 32 | 15 | 1,039 | 1,671 | 640 |
| 65 | Muttontown | 200 | 185 | 15 | 200 | 185 | 13 | 2,753 | 485 | 386 |
| 66 | Old Brookville | 84 | 71 | 13 | 84 | 71 | | 2,084 | 367 | 292 |
| 67 | Oyster Bay Cove | 76 | 76 | | 76 | 76 | -0-m | 1,717 | 2,982 | 2,823 |
| 68 | Roslyn Harbor | 75 | 75 | | 73 | 73 | 940-946 | 305 | 0 | 0 |
| 69 | Sea Cliff | 1,605 | 1,511 | 94 | 1,594 | 1,504 | 90 | 6,123 | 704 | 0 |
| 70 | Upper Brookville | 33 | 27 | 6 | 33 | 27 | 6 | 1,331 | 235 | 186 |
| 71 | Town of Babylon | 17,559 | 15,457 | 2,102 | 17,547 | 15,461 | 2,086 | 165,651 | 33,920 | 0 |
| | Village of: | • | | | | | - | | | |
| 72 | Amityville | 1,892 | 1,684 | 208 | 1,894 | 1,683 | 211 | 10,776 | 1,344 | 0 |
| 73 | Babylon | 2,012 | 1,873 | 139 | 2,019 | 1,877 | 142 | 13,499 | 1,600 | 0 |
| 74 | Lindenhurst | 4,082 | 3,791 | 291 | 4,090 | 3,796 | 294 | 30,457 | 2,368 | 0 |
| 75 | Town of Brookhaven | 19,075 | 17,845 | 1,230 | 18,980 | 17,754 | 1,226 | 321, 322 | 166,400 | 0 |
| | Village of: | - | | | | | | 877 | 0 | 0 |
| 76 | Belle Terre | 15 | 14 | 1 | 15 | 14 | 1 | | | |
| 77 | Bellport | 252 | 232 | 20 | 254 | 234 | 20 | 2,978 | 896 | 0 |
| 78 | Old Field | 43 | 43 | | 44 | 44 | | 872 | 0 | 0 |
| 79 | Patchogue | 1,825 | 1,618 | 207 | 1,842 | 1,632 | 210 | 11,299 | 1,472 | 0 |
| 80 | Poquott | 31 | 31 | | 31 | 31 | | 521 | 0 | 0 |

NUMBER OF CUSTOMERS, POPULATION AND ACREAGE - LONG ISLAND LIGHTING COMPANY (Continued)

| | | Numb | er of Customers | - 1979 | Numbe | r of Customers | s — 1978 | | | |
|------|----------------------|--------|-----------------|------------|--|---|---|--------------------|-------------------------|------------------------|
| Case | Municipality | Total | Residential | Comm./Ind. | පත සංකාශය පත සංකාශය පතමාධපතා ක්ෂවා කරගත් පරිකාශකයෙක් පරිකාශකයෙක් පරිකාශකයෙක් පරිකාශකයෙක් පරිකාශකයෙක් පරිකාශකයෙක් පරිකාශක | n⊴318351.cl.adjetdensjen⊴2r#364c23r#35nobeng3n⊴5r n⊴318351.cl.adjetdensjen⊴3r#364c23r#35nobeng3ng5ng5ng5ng5ng5ng n⊴318535.cl.adjetdensjen#364c23nobeng3ng5ng5ng5ng5ng5ng5ng5ng5ng5ng5ng5ng5ng5n | ෯෪෯෬෯෦෪෮෦෭෩෫෯෪෯෪෯෪෯෪෯෪෯෩෯෭෩෯ ෯෪෯෬෯෪෯෪෯෪෯෪෯෪෯෪෯෪෯෩෯෭෯෩෯ | Population 1978 | Total <u>Acreage</u> | Residential Acreage |
| | Village of: | | | | | | | | | |
| 81 | Port Jefferson | 997 | 836 | 161 | 993 | 829 | 164 | 6,315 | 1,280 | 0 |
| 82 | Lake Grove | 630 | 557 | 73 | 632 | 558 | 74 | 9,445 | 1,856 | 0 |
| 83 | Town of East Hampton | 410 | 371 | 39 | 410 | 371 | 39 | 12,013 | 44,800 | 0 |
| | Village of: | | | | | | | • | • | |
| 84 | East Hampton | 1,025 | 932 | 93 | 1,023 | 928 | 95 | 2,044 | 0 | 0 |
| 85 | Sag Habor | 311 | 298 | 13 | 311 | 298 | 13 | 946 | 0 | 0 |
| 86 | Town of Huntington | 23,336 | 21,474 | 1,862 | 23,299 | 21,434 | 1,865 | 203,028 | 60,160 | 0 |
| | Village of: | | , | , | | • | | • | | |
| 87 | Asharoken | 65 | 64 | - 1 | 64 | 63 | 1 | 644 | 0 | 0 |
| 88 | Huntington Bay | 265 | 261 | 4 | 262 | 258 | 4 | 1,925 | 0 | 0 |
| 89 | Llovd Harbor | 33 | . 31 | 2 | 33 | 31 | 2 | 3,930 | 5,888 | 0 |
| 90 | Northport | 1,451 | 1,331 | 120 | 1,462 | 1,338 | 124 | 8,212 | 1,600 | 0 |
| 91 | Town of Islip | 31,772 | 29,438 | 2,334 | 31,858 | 29,536 | 2,322 | 309,016 | 65,280 | 0 |
| | Village of: | • | • | • | - | • | | | • | |
| 92 | Brightwaters | 672 | 628 | 44 | 670 | 628 | 42 | 3,808 | 576 | 0 |
| 93 | Town of Riverhead | 1,851 | 1,590 | 261 | 1,869 | 1,603 | 266 | 23,921 | 49,920 | 0 |
| 94 | Town of Smithtown | 12,197 | 11,376 | 821 | 12,181 | 11,372 | 809 | 121,723 | 1,952 | 0 |
| | Village of: | - | • | | | - | | | - | |
| 95 | Head of the Harbor | 68 | 62 | 6 | 69 | 63 | 6 | 1,093 | 0 | 0 |
| 96 | The Branch | 148 | 80 | 68 | 148 | 83 | 65 | 1,856 | 0 | 0 |
| 97 | Town of Southampton | 1,909 | 1,693 | 216 | 1,899 | 1,682 | 217 | 38,355 | 92,800 | 0 |
| | Village of: | | | | | | | | | |
| 98 | Sag Harbor | 576 | 507 | 69 | 577 | 507 | 70 | 1,860 | 0 | 0 |
| 99 | Southampton | 1,579 | 1,418 | 161 | 1,575 | 1,411 | 164 | 5,541 | 4,096 | 0 |
| 100 | Town of Southold | 2,109 | 1,941 | 168 | 2,102 | 1,933 | 169 | 17,067 | 33,920 | 0 |
| 101 | City of New York | 16,535 | 15,053 | 1,482 | 16,690 | 15,173 | 1,457 | 97,343 | 3,455 | 0 |

Source: Annual Reports to the New York Public Service Commission, 1978-1979, Long Island Lighting Company



APPENDIX B

COLUMBIA GAS OF OHIO COMPANY DATA

The purpose of this appendix is to present the community-level data used in the econometric analysis of the Columbia Gas of Ohio Company, as reported in chapter 3. The residential, commercial, and industrial gas sales and numbers of customers, the net plant in service, and the population and acreage of 52 communities included in the company's service territory are indicated in Table B-1.

| IABLE B-I |
|-----------|
|-----------|

GAS SALES, NUMBERS OF CUSICMERS, NET PLANT IN SERVICE, POPULATION AND ACREAGE - COLUMBIA GAS OF OHIO CONTAINT

| | | | Gas Sales (M | CF) | | · · · · · · · · · · · · · · · · · · · | Number of Cus | tomers | | Net Plant | | |
|-----|--|-------------|--------------|------------|----------|---------------------------------------|---------------|------------|---------|------------|------------|---------|
| Cas | e Community | Residential | Commercial | Industrial | A11 | Residential | Commercial | Industrial | A11 | In Service | Population | Acreage |
| | | Sector | Sector | Sector | Sectors | Sector | Sector | Sector | Sectors | (\$) | | • |
| | ······································ | | | | | | | | | | · · · · · | |
| 1 | Toledo | 18921600 | 5834240 | 595056 | 25350896 | 105789 | 7398 | 107 | 113294 | 29121200 | 383818 | 51968 |
| 2 | Lorain | 3678430 | 1201380 | 86090 | 4965900 | 20851 | 1469 | 5 | 22325 | 6493670 | 78185 | 14272 |
| 3 | New Riegel | 16557 | 13220 | 7483 | 37260 | 103 | 26 | ì | 130 | 47796 | | |
| 4 | Mansfield | 3034030 | 1253110 | 156290 | 4443430 | 16891 | 1791 | 22 | 18704 | 5039310 | 55047 | 15424 |
| 5 | Parma | 4801800 | 1079030 | 17740 | 5898570 | 29119 | 1350 | 3 | 30472 | 7182480 | 100216 | 13312 |
| 6 | Westlake | 876060 | 384577 | 29313 | 1289950 | 4298 | 272 | 4 | 4574 | 1648710 | 15689 | 9984 |
| 7 | Dublin | 21767 | 109011 | 0 | 130778 | 143 | 33 | 0 | 176 | 257064 | 5000 | 11520 |
| 8 | Bexley | 913233 | 117658 | 0 | 1030890 | 4606 | 151 | 0 | 4757 | 919970 | 14888 | 1536. |
| 9 | Brice | 8109 | 6001 | . 0 | 14109 | 48 | 8 | 0 | 56 | 5616C | 250 | 64 |
| 10 | Canal Winchester | 99570 | 51673 | 1363 | 152605 | 628 | 73 | 1 | 702 | 173420 | 3200 | 6400 |
| 11 | Columbus | 25089392 | 10502700 | 643108 | 36235200 | 166263 | 9999 | 1555 | 177817 | 38948896 | 539677 | 86144 |
| 12 | Gahanna | 573314 | 156232 | 0 | 729546 | 3938 | 174 | 0 | 4112 | 1138880 | 12400 | 4288 |
| 13 | Grove City | 643216 | 143203 | 14310 | 800729 | 4669 | 197 | 2 | 4868 | 1321340. | 13911 | 2880 |
| 14 | Groveport | 135525 | 40600 | 0 | 176124 | 1012 | 68 | 0 | 1080 | 319436 | 4000 | 4480 |
| 15 | Hilliard | 327003 | 83408 | 44317 | 454733 | 2435 | 123 | 4 | 2562 | 747516 | 8369 | 2752 |
| 16 | Marble Cliff | 40101 | 40053 | 0 | 80154 | 267 | 25 | 0 | 292 | 102601 | 680 | 192 |
| 17 | Minerva | 84134 | 8272 | Ō | 92406 | 463 | 28 | 0 | 491 | 132142 | 1600 | 1920 |
| 18 | New Albany | 22698 | 18114 | 0 | 40812 | 149 | 31 | 0 | 180 | 70959 | 530 | 4480. |
| 19 | New Rome | 5465 | 6297 | 0 | 11762 | 38 | 16 | 0 | 54 | 33023 | 110 | 640 |
| 20 | Obetz | 104418 | 41444 | 30245 | 176107 | 709 | 31 | 2 | 742 | 238919 | 3500 | 1920 |
| 21 | Revnoldsburg | 600557 | 153428 | 0 | 753980 | 4803 | 216 | 0 | 5019 | 1249880 | 13921 | 3008 |
| 22 | Upper Arlington | 2249180 | 385746 | õ | 2634930 | 12013 | 357 | Ċ | 12370 | 3234450 | 38630. | 6144 |
| 23 | Urbancrest | 28271 | 3764 | õ | 32035 | 173 | 7 | 0 | 180 | 76474 | | |
| 24 | Vallevview | 40632 | 860 | Ō | 41492 | 265 | 3 | 0 | 268 | 29922 | 1000 | 3200 |
| 25 | Westerville | 610072 | 192861 | 56299 | 859232 | 4282 | 309 | 5 | 4596 | 1762330 | 12530 | 5440. |
| 26 | Whitehall | 923728 | 345809 | 13123 | 1282660 | 8196 | 523 | 1 | 8720 | 1815520 | 25263 | 3712 |
| 27 | Worthington | 717015 | 253311 | 30254 | 1000580 | 4565 | 309 | - 4 | 4878 | 1426700 | 15326 | 3264 |
| 28 | Ashville | 85484 | 23294 | 4882 | 113660 | 573 | 64 | 1 | 638 | 256988 | 2309 | 640 |
| 29 | Mount Sterling | 86954 | 30670 | 4002 | 117623 | 528 | 77 | ō | 605 | 217049 | | |
| 30 | Waldo | 19027 | 9644 | 9126 | 37797 | 126 | 31 | 1 | 158 | 44432 | 437 | 256 |
| 31 | Baltimore | 117863 | 26736 | 176 | 144774 | 822 | 78 | 1 | 901 | 397805 | 3150 | 2560 |
| 32 | Centerburg | 72004 | 23868 | 10523 | 106394 | 397 | 65 | 2 | 464 | 121475 | | |
| 33 | Granvilla | 141405 | 50180 | 4673 | 205059 | 713 | 122 | 3 | 838 | 229058 | 3963 | 448 |
| 34 | Magnetic Spring | 17466 | 3713 | 4475 | 203038 | 108 | 11 | õ | 119 | 25393 | | |
| 35 | Springfield | 4023530 | 1212120 | 147110 | 5382760 | 24709 | 2120 | 26 | 26855 | 5811170 | 81926 | 10688 |

| ****** | | ······ | Gas Sales (M | CF) | | | Number of Cus | tomers | | | | |
|--------|----------------|-------------|--------------|------------|---------|-------------|---------------|------------|---------|------------|------------|---------|
| | | | (| , | | | Number of 000 | comere | | Net Plant | | |
| | | Residential | Commercial | Industrial | A11 | Residential | Commercial | Industrial | A11 | In Service | | |
| Cas | e Community | Sector | Sector | Sector | Sectors | Sector | Sector | Sector | Sectors | (\$) | Population | Acreage |
| | | | | | | | | | | | | |
| 36 | Tremont City | 17912 | 2617 | 0 | 20528 | 117 | 11 | 0 | 128 | 19693 | | |
| 37 | Columbiana | 240345 | 106271 | 59393 | 406009 | 1598 | 162 | 6 | 1766 | 486780 | 4959 | 1920 |
| 38 | Martins Ferry | 534975 | 179553 | 31953 | 746481 | 3171 | 267 | 3. | 3441 | 998015 | 10757 | 1344 |
| 39 | Shadyside | 212723 | 33110 | 868 | 246701 | 1528 | 78 | 1 | 1607 | 367350 | 5070 | 512 |
| 40 | Mingo Junction | 203308 | 30532 | 6837 | 240677 | 1318 | 88 | 3 | 1409 | 407877 | 5278 | 1408 |
| 41 | Steubenville | 1539250 | 675665 | 30525 | 2245440 | 8641 | 776 | 14 | 9431 | 2480390 | 27105 | 4800 |
| 42 | Jewett | 47076 | 9983 | 0 | 57059 | 255 | 27 | 0 | 282 | 86611 | • | |
| 43 | Quaker City | 33585 | 10493 | 0 | 44078 | 206 | 29 | 0 | 235 | 88118 | | |
| 44 | Frazeysburg | 53826 | 12582 | 0 | 66408 | 342 | 39 | 0 | 381 | 94825 | | |
| 45 | Lower Salem | 5913 | 1111 | 0 | 7024 | 46 | 8. | 0 | 54 | 27526 | 102 | 320 |
| 46 | Hemlock | 9462 | 390 | 0 | 9852 | 63 | 2 | 0 | 65 | 25232 | 199 | 256 |
| 47 | Shawnee | 40183 | 9137 | 0 | 49319 | 267 | 35 | 0 | 302 | 129723 | 914 | 448 |
| 48 | Chillicothe | 1206150 | 315401 | 35569 | 1557120 | 8082 | 770 | . 9 | 8861 | 2532240 | 24842 | 5312 |
| 49 | Cheshire | 9348 | 1986 | 0 | 11334 | 74 | 12 | 0 | 86 | 30480 | | |
| 50 | Middleport | 143105 | 36443 | 10515 | 190063 | 988 | 125 | 4 | 1117 | 369244 | 2784 | 704 |
| 51 | New Boston | 152239 | 38217 | 0 | 190456 | 1023 | 129 | 0 | 1152 | 354208 | 3325 | 640 |
| 52 | Portsmouth | 1383200 | 520979 | 50311 | 1954490 | 8871 | 1064 | 13 | 9948 | 3159150 | 27633 | 7808 |

TABLE B-1

GAS SALES, NUMBERS OF CUSTOMERS, NET FLANT IN SERVICE, FOPULATION AND ACREAGE - COLUMBIA GAS OF CHIO COMPANY (Continued)

Source: Public Utilities Commission of Ohio

157

l

APPENDIX C

PACIFIC GAS AND ELECTRIC COMPANY DATA

The purpose of this appendix is to present the community-level data used in the econometric analysis of the Pacific Gas and Electric Company, as reported in chapter 3. The numbers of residential, commercial, and industrial gas customers and the corresponding sales for the years 1975 through 1979 and for 94 communities of 10,000 population or more are indicated in tables C-1 through C-8. The 1970 population and acreage, and the 1978 and 1979 distribution plant and main mileage of these communities are indicated in table C-9. Finally, average degree-day data are indicated in table C-10.

| Average | Number | of | Residential, | Commercial, | and | Industrial | Gas | Customers | |
|---------|--------|-----|---------------|--------------|------|------------|-----|-----------|--|
| | Co | mmu | nities of 10, | 000 Populati | on o | r More | | | |
| | | | Pacific Gas a | nd Electric | Comp | anv | | | |

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------|---------|---------|---------|---------|---------|
| Community | | | | | |
| Coast Valleys | | | | | |
| Monterey | 9,288 | 9,455 | 9,545 | 9,583 | 9,644 |
| Pacific Grove | 6,666 | 6,705 | 6,717 | 6,766 | 6,893 |
| Salinas | 22,609 | 23,255 | 23,975 | 24,445 | 25,038 |
| Seaside | 6,855 | 6,905+ | 6,955 | 6,931 | 6,963 |
| Colgate | | | | | |
| Yuba City | 5,785 | 5,986 | 6,320 | 7,160 | 7,558 |
| De Sabla | | | | | |
| Chico | 8,278 | 8,576 | 9,075 | 9,378 | 9,770 |
| Drum | | | | | |
| Roseville | 7,465 | 7,693 | 7,994 | 8,280 | 8,552 |
| East Bay | | | | | |
| Alameda | 20,554 | 20,678 | 20,680 | 20,675 | 20,827 |
| Albany | 5,757 | 5,764† | 5,772 | 5,781 | 5,780 |
| Antioch | 11,712 | 12,093 | 12,774 | 13,742 | 14,609 |
| Berkeley | 41,788 | 41,913 | 41,912 | 41,958 | 42,199 |
| Concord | 30,817 | 31,747 | 32,962 | 33,943 | 35,027 |
| El Cerrito | 9,631 | 9,675 | 9,707 | 9,768 | 9,807 |
| Fremont | 35,507 | 36,441 | 37,605 | 38,835 | 40,645 |
| Hayward | 29,665 | 30,186 | 30,556 | 30,984 | 31,462 |
| Lafayette | 7,182 | 7,276 | 7,349 | 7,430 | 7,535 |
| Livermore | 15,409 | 15,699 | 15,879 | 16,033 | 16,140 |
| Martinez | 6,770 | 7,038 | 7,444 | 7,849 | 8,297 |
| Moraga | | | 4,103 | 4,188 | 4,317 |
| Newark | 8,105 | 8,455+ | 8,806 | 9,012 | 9,198 |
| Oakland | 126,639 | 126,913 | 127,219 | 127,223 | 128,497 |

160

TABLE C-1 Average Number of Residential, Commercial, and Industrial Gas Customers Communities of 10,000 Population or More

Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|------------------|--------|----------|--------|--------|---------|
| Community | | | | | |
| East Bay (cont.) | | | | | |
| Piedmont | 3,797 | 3,808 | 3,822 | 3,828 | 3,848 |
| Pinhole | 4,739 | 4,836 | 4,858 | 4,868 | 4,910 |
| Pittsburg | 8,673 | 9,111 | 9,529 | 9,938 | 10,433 |
| Pleasant Hill | 7,868 | 7,944 | 7,991 | 8,117 | 8,560 |
| Pleasanton | 9,477 | 10,086 | 10,502 | 10,734 | 10,999 |
| Richmond | 26,543 | 26,776 | 26,962 | 27,054 | 27,289 |
| San Leandro | 23,696 | 23,893 | 23,937 | 24,027 | 24,312 |
| San Pablo | 7,229 | 7,221 | 7,241 | 7,243 | 7,258 |
| Union City | 9,473 | 10,220 | 10,743 | 11,102 | 11,515 |
| Walnut Creek | 16,634 | 17,032 | 17,533 | 17,986 | 18,800 |
| Humboldt | | | | | |
| Arcata | 3,762 | 3,842 | 4,037 | 4,238 | 4,385 |
| Eureka | 10,393 | 10,464 | 10,589 | 10,702 | 10,810 |
| North Bay | | | | | |
| Benicia | | | 4,372 | 4,634 | 5,058 |
| Larkspur | 4,265 | 4,313 | 4,329 | 4,375 | 4,418 |
| Mill Valley | 4,915 | 5,005 | 5,107 | 5,119 | 5,133 |
| Napa | 17,187 | 17,660 | 18,051 | 18,463 | 18,973 |
| Novato | 10,020 | 10,548 | 11,061 | 11,455 | 12,330 |
| Petaluma | 10,577 | 10,857 | 10,995 | 11,266 | 11,551 |
| Rohnert Park | | sum dime | 4,296 | 4,895 | 5,468 |
| San Anselmo | 4,978 | 4,988 | 4,985 | 5,006 | 5,024 |
| San Rafael | 15,913 | 16,597 | 16,853 | 16,958 | 17,024 |
| Santa Rosa | 25,982 | 27,247 | 28,739 | 30,008 | 30,966 |
| Ukiah | 3,228 | 3,496 | 3,718 | 3,881 | 4,156 |
| Vallejo | 23,602 | 24,184 | 24,472 | 25,247 | 26,228 |
| Sacramento | | | | | |
| Davis | 10,038 | 10,459 | 10,942 | 11,527 | 12,052 |
| Fairfield | 12,466 | 12,979 | 13,704 | 14,198 | 14,911 |
| Sacramento | 96,944 | 97,796 | 98,123 | 98,711 | 100,155 |

| , . | Pacific | Gas and Electr | ic Company (Com | ntinued) | |
|---------------------|---------|----------------|-----------------|----------|---------|
| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 9,358 | 10,288 | 11,295 | 12,360 | 13,275 |
| Woodland | 8,933 | 9,201 | 9,557 | 10,005 | 10,311 |
| San Francisco | | | | | |
| Daly City | 23,945 | 24,157 | 24,403 | 24,539 | 24,702 |
| Millbrae | 7,126 | 7,157 | 7,220 | 7,280 | 7,301 |
| Pacifica | 11,344 | 11,421 | 11,424 | 11,452 | 11,525 |
| San Bruno | 11,790 | 11,793 | 11,825 | 11,835 | 11,863 |
| San Francisco | 256,230 | 256,674 | 257,136 | 259,013 | 260,507 |
| South San Francisco | 15,946 | 16,201 | 16,277 | 16,384 | 16,624 |
| San Joaquin | | | | | |
| Atwater | 3,568 | 3,706 | 3,854 | 4,052 | 4,390 |
| Bakersfield | 28,698 | 29,828 | 31,412 | 33,224 | 34,826 |
| Clovis | 7,311 | 8,355 | 9,484 | 10,273 | 11,254 |
| Fresno | 67,238 | 70,025 | 73,825 | 76,852 | 79,905 |
| Los Banos | 3,238 | 3,275 | 3,296 | 3,343 | 3,427 |
| Madera | 6,344 | 6,477 | 6,681 | 6,932 | 7,323 |
| Merced | 10,413 | 10,875 | 11,396 | 11,928 | 12,298 |
| Ridgecrest | 4,273 | 4,062 | 4,170 | 4,404 | 4,686 |
| Sanger | 3,462 | 3,566 | 3,607 | 3,710 | 3,896 |
| San Jose | | | | | |
| Belmont | 8,460 | 8,601 | 8,802 | 8,939 | 9,054 |
| Burlingame | 11,151 | 11,158 | 11,136 | 11,172 | 11,190 |
| Campbell | 9,189 | 9,360 | 9,335 | 9,530 | 9,792 |
| Cupertino | 7,336 | 7,629 | 7,851 | 8,098 | 9,917 |
| Foster City | 4,943 | 5,272 | 5,616 | 6,150 | 6,369 |
| Gilroy | 4,826 | 5,071 | 5,387 | 5,814 | 6,286 |
| Los Altos | 9,130 | 9,265 | 9,436† | 9,608 | 9,650 |
| Los Gatos | 8,665 | 8,910 | 9,217 | 9,584 | 9,811 |
| Menlo Park | 11,472 | 11,603 | 11,683 | 11,783 | 11,819 |
| Milpitas | 8,858 | 8,951 | 9,258 | 9,738 | 10,391 |

| Average Number of Residential, Commercial, and Industrial Gas Customers Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued) | | | | | | | | |
|---|-----------|-----------|-----------|-----------|-----------|--|--|--|
| Year | 1975 | | 1977 | 1978 | 1979 | | | |
| Community | | | | | | | | |
| San Jose (cont.) | | | | | | | | |
| Mountain View | 19,236 | 19,596 | 20,246 | 20,615 | 21,092 | | | |
| Morgan Hill | - | | 3,364 | 3,904 | 4,306 | | | |
| Redwood City | 20,030 | 20,147 | 20,208 | 20,429 | 20,708 | | | |
| San Carlos | 10,273 | 10,382 | 10,453 | 10,540 | 10,598 | | | |
| San Jose | 167,500 | 172,938 | 179,362 | 184,728 | 188,598 | | | |
| San Mateo | 27,871 | 28,034 | 28,106 | 28,382 | 28,704 | | | |
| Santa Clara | 27,321 | 27,642 | 28,228 | 28,473 | 29,250 | | | |
| Santa Cruz | 13,428 | 13,677 | 13,965 | 14,335 | 14,641 | | | |
| Saratoga | 8,465 | 8,651 | 8,799 | 8,832 | 9,142 | | | |
| Sunnyvale | 32,083 | 32,708 | 33,207 | 34,169 | 34,439 | | | |
| Watsonville | 5,998 | 6,417 | 6,702 | 6,938 | 7,104 | | | |
| Shasta | | | | | | | | |
| Redding | 3,714 | 3,811 | 3,950 | 7,419 | 9,893 | | | |
| Stockton | | | | | | | | |
| Ceres | | | 3,682 | 3,947 | 4,232 | | | |
| Lodi | 12,455 | 12,768 | 13,040 | 13,421 | 13,797 | | | |
| Manteca | 6,147 | 6,470 | 6,993 | 7,796 | 8,550 | | | |
| Modesto | 30,125 | 31,551 | 33,191 | 34,963 | 36,721 | | | |
| Stockton | 42,463 | 44,401 | 46,723 | 49,369 | 52,266 | | | |
| Tracy | 5,888 | 5,979 | 6,130 | 6,325 | 6,499 | | | |
| Turlock | 6,538 | 6,802 | 7,500 | 8,012 | 8,805 | | | |
| Customers in Communities of 10,000 Popula- tion and Over | 1,795,687 | 2,112,814 | 1,883,132 | 1,936,143 | 1,990,815 | | | |
| Customers in Commu- nities of Less Than 10,000 Population | 724,092 | 462,057 | 752,108 | 761,582 | 772,582 | | | |

† See Table C−2

163

Average Number of Residential, Commercial, and Industrial Gas Customers Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|----------------------------|-----------|-----------|-----------|-----------|-----------|
| Other Sales to | | | | | |
| Public Authorities | 1 | 1 | 1 | 1 | 1 |
| Sales for Resale | 4 | 4 | 4 | 6 | 8 |
| Interdepartmental Sales | | | | | |
| Total | 2,519,784 | 2,574,876 | 2,635,245 | 2,697,732 | 2,763,406 |

Source: Annual Reports to the California Public Utilities Commission, 1975-1979, Pacific Gas and Electric Company. Average Number of Residential Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------|---------|-----------|---------|---------|---------|
| Community | | | | | |
| Coast Valleys | | | | | |
| Monterey | 8,293 | 8,435 | 8,524 | 8,557 | 8,589 |
| Pacific Grove | 6,347 | 6,368 | 6,383 | 6,424 | 6,539 |
| Salinas | 20,924 | 21,531 | 22,233 | 22,634 | 23,161 |
| Seaside | 6,426 | 6,475† | 6,524 | 6,502 | 6,532 |
| Colgate | | | | | |
| Yuba City | 5,187 | 5,355 | 5,688 | 6,428 | 6,767 |
| De Sabla | | | | | |
| Chico | 7,352 | 7,645 | 8,144 | 8,439 | 8,763 |
| Drum | | | | | |
| Roseville | 6,925 | 7,140 | 7,429 | 7,722 | 7,959 |
| East Bay | | | | | |
| Alameda | 19,723 | 19,805 | 19,786 | 19,776 | 19,915 |
| Albany | 5,449 | 5,444+ | 5,439 | 5,443 | 5,441 |
| Antioch | 11,181 | 11,551 | 12,213 | 13,159 | 14,002 |
| Berkeley | 39,093 | 39,154 | 39,183 | 39,205 | 39,439 |
| Concord | 29,315 | 30,209 | 31,344 | 32,233 | 33,223 |
| El Cerrito | 9,233 | 9,254 | 9,295 | 9,351 | 9,387 |
| Fremont | 34,101 | 34,969 | 36,067 | 37,257 | 38,959 |
| Hayward | 27,395 | 27,856 | 28,153 | 28,506 | 28,919 |
| Lafayette | 6,701 | 6,781 | 6,862 | 6,935 | 7,003 |
| Livermore | 14,798 | 15,070 | 15,241 | 15,340 | 15,418 |
| Martinez | 6,417 | 6,682 | 7,084 | 7,492 | 7,928 |
| Moraga | ++ | 1000 BEEN | 3,976 | 4,063 | 4,189 |
| Newark | 7,787 | 8,124† | 8,461 | 8,644 | 8,814 |
| Oakland | 118,503 | 118,633 | 119,169 | 119,229 | 120,421 |

Average Number of Residential Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|------------------|--------|------------|--------|--------|--------|
| Community | | | | | |
| East Bay (cont.) | | | | | |
| Piedmont | 3,760 | 3,768 | 3,785 | 3,794 | 3,815 |
| Pinhole | 4,573 | 4,663 | 4,685 | 4,688 | 4,721 |
| Pittsburg | 8,219 | 8,651 | 9,072 | 9,461 | 9,939 |
| Pleasant Hill | 7,472 | 7,537 | 7,575 | 7,684 | 8,094 |
| Pleasanton | 9,121 | 9,702 | 10,080 | 10,278 | 10,509 |
| Richmond | 25,042 | 25,259 | 25,451 | 25,550 | 25,750 |
| San Leandro | 21,801 | 21,966 | 21,997 | 22,089 | 22,358 |
| San Pablo | 6,791 | 6,782 | 6,826 | 6,838 | 6,859 |
| Union City | 9,118 | 9,838 | 10,356 | 10,698 | 11,104 |
| Walnut Creek | 15,650 | 16,022 | 16,505 | 16,927 | 17,638 |
| Humboldt | | | | | • |
| Arcata | 3,340 | 3,404 | 3,600 | 3,797 | 3,928 |
| Eureka | 9,264 | 9,280 | 9,390 | 9,475 | 9,570 |
| North Bay | | | | | |
| Benicia | - | | 4,101 | 4,350 | 4,755 |
| Larkspur | 4,054 | 4,103 | 4,113 | 4,145 | 4,165 |
| Mill Valley | 4,630 | 4,699 | 4,787 | 4,794 | 4,797 |
| Napa | 16,124 | 16,564 | 16,936 | 17,290 | 17,750 |
| Novato | 9,586 | 10,098 | 10,595 | 10,978 | 11,758 |
| Petaluma | 9,897 | 10,145 | 10,269 | 10,529 | 10,777 |
| Rohnert Park | | seen other | 4,137 | 4,698 | 5,254 |
| San Anselmo | 4,683 | 4,687 | 4,684 | 4,704 | 4,713 |
| San Rafael | 14,383 | 15,041 | 15,269 | 15,333 | 15,345 |
| Santa Rosa | 24,063 | 25,303 | 26,803 | 28,042 | 28,887 |
| Ukiah | 2,811 | 3,056 | 3,269 | 3,402 | 3,623 |
| Vallejo | 22,171 | 22,783 | 23,071 | 23,827 | 24,792 |
| Sacramento | | | | | |
| Davis | 9,652 | 10,069 | 10,559 | 11,142 | 11,616 |
| Fairfield | 11,852 | 12,314 | 13,005 | 13,473 | 14,129 |
| Sacramento | 91,255 | 92,059 | 92,374 | 92,885 | 94,209 |
Average Number of Residential Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------------|---------|---------|---------|---------|---------|
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 8,949 | 9,864 | 10,857 | 11,899 | 12,772 |
| Woodland | 8,330 | 8,584 | 8,939 | 9,343 | 9,591 |
| San Francisco | . · · | | | | |
| Daly City | 23,247 | 23,474 | 23,720 | 23,860 | 24,027 |
| Millbrae | 6,822 | 6,856 | 6,919 | 6,976 | 6,998 |
| Pacifica | 11,062 | 11,113 | 11,124 | 11,162 | 11,219 |
| San Bruno | 11,212 | 11,201 | 11,230 | 11,238 | 11,262 |
| San Francisco | 239,190 | 239,021 | 239,965 | 241,808 | 243,298 |
| South San Francisco | 14,747 | 14,987 | 15,052 | 15,139 | 15,342 |
| San Joaquin | | | | | |
| Atwater | 3,328 | 3,468 | 3,619 | 3,803 | 4,126 |
| Bakersfield | 25,873 | 26,971 | 28,537 | 30,221 | 31,727 |
| Clovis | 6,934 | 7,939 | 9,036 | 9,768 | 10,658 |
| Fresno | 62,054 | 64,641 | 68,299 | 71,157 | 73,879 |
| Los Banos | 2,924 | 2,956 | 2,982 | 3,026 | 3,108 |
| Madera | 5,824 | 5,952 | 6,167 | 6,428 | 6,755 |
| Merced | 9,480 | 9,924 | 10,438 | 10,964 | 11,287 |
| Ridgecrest | 3,970 | 3,740 | 3,851 | 4,069 | 4,328 |
| Sanger | 3,172 | 3,266 | 3,312 | 3,420 | 3,596 |
| San Jose | | | | | |
| Belmont | 8,065 | 8,192 | 8,396 | 8,525 | 8,632 |
| Burlingame | 10,153 | 10,148 | 10,129 | 10,149 | 10,159 |
| Campbell | 8,215 | 8,325 | 8,258 | 8,429 | 8,624 |
| Cupertino | 6,923 | 7,201 | 7,399 | 7,617 | 9,387 |
| Foster City | 4,832 | 5,156 | 5,486 | 6,001 | 6,185 |
| Gilroy | 4,327 | 4,574 | 4,885 | 5,295 | 5,709 |
| Los Altos | 8,579 | 8,699 | 8,867† | 9,036 | 9,064 |
| Los Gatus | 8,000 | 8,230 | 8,532 | 8,870 | 9,059 |
| Menlo Park | 10,758 | 10,876 | 10,962 | 11,046 | 11,077 |
| Milpitas | 8,529 | 8,622 | 8,927 | 9,394 | 10,005 |

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|--|-----------|---|---------------|--|----------------|
| | | | | ala - manakar - maanifi na - maaka 1990 me | |
| Community | | | | | |
| San Jose (cont.) | | | | | |
| Mountain View | 17,538 | 17,870 | 18,481 | 18,760 | 19,156 |
| Morgan Hill | | | 3,103 | 3,633 | 3,949 |
| Redwood City | 18,692 | 18,792 | 18,859 | 19,062 | 19,338 |
| San Carlos | 9,346 | 9,446 | 9,511 | 9,575 | 9,627 |
| San Jose | 159,884 | 165,080 | 171,269 | 176,404 | 179,886 |
| San Mateo | 26,128 | 26,252 | 26,328 | 26,574 | 26,887 |
| Santa Clara | 25,167 | 25,433 | 25,884 | 26,002 | 26,616 |
| Santa Cruz | 12,178 | 12,390 | 12,666 | 13,015 | 13,314 |
| Saratoga | 8,139 | 8,309 | 8,451 | 8,476 | 8,769 |
| Sunnyvale | 30,413 | 30,956 | 31,385 | 32,263 | 32,327 |
| Watsonville | 5,305 | 5,703 | 5,969 | 6,193 | 6,347 |
| Shasta | | | | | • |
| Redding | 2,902 | 2,945 | 3,079 | 6,316 | 8,596 |
| Stockton | | | | | |
| Ceres | | - | 3,437 | 3,700 | 3,972 |
| Lodi | 11,497 | 11,806 | 12,078 | 12,434 | 12,776 |
| Manteca | 5,729 | 6,041 | 6,542 | 7,303 | 8,030 |
| Modesto | 28,246 | 29,631 | 31,220 | 32,895 | 34,493 |
| Stockton | 39,533 | 41,403 | 43,666 | 46,195 | 48,938 |
| Tracy | 5,487 | 5,551 | 5,710 | 5,903 | 6,052 |
| Turlock | 5,882 | 6,147 | 6,851 | 7,338 | 8,075 |
| Customers in Communities of 10,COO Popula- tion and Over | 1,681,474 | 1,977,114 | 1,764,986 | 1,814,894 | 1,864,84 |
| Customers in Commu- nities of Less Than 10,000 Population | 680,737 | 436,479 | 707,916 | 716,861 | 726,66 |
| Total | 2 362 211 | 2 /13 503 | 2 / 72 002 | 2 531 755 | 2 501 50 |
| | ±±2و202و2 | ۵, ۲, ۲, ۲, ۲, ۶, ۶, ۶, ۶, ۶, ۶, ۶, ۶, ۶, ۶, ۶, ۶, ۶, | 2016 و2117 و2 | رر، ۱ و ـ د ر و <u>ـ</u> | JU e ± 7 C e 2 |

168

+ : value linearly interpolated (original data inconsistent)
++ : See Table C-3 168

TABLE C-2

Average Number of Commercial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|-----------------|-------|-------|-------|-------|-------|
| Community | | | | | |
| | | | | | |
| Coast Valleys | | | | | |
| Monterey | 983 | 1,009 | 1,012 | 1,017 | 1,046 |
| Pacific Grove | 316 | 334 | 331 | 339 | 351 |
| Salinas | 1,665 | 1,705 | 1,727 | 1,796 | 1,861 |
| Seaside | 428 | 429† | 430 | 428 | 430 |
| Colgate | | | | | |
| Yuba City | 590 | 622 | 623 | 724 | 783 |
| <u>De Sabla</u> | | | | | |
| Chico | 914 | 919 | 920 | 928 | 996 |
| Drum | | | | | |
| Roseville | 528 | 542 | 557 | 550 | 586 |
| East Bay | | | | | |
| Alameda | 811 | 855 | 876 | 882 | 895 |
| Albany | 303 | 315+ | 328 | 333 | 334 |
| Antioch | 522 | 533 | 554 | 578 | 602 |
| Berkeley | 2,623 | 2,690 | 2,675 | 2,734 | 2,741 |
| Concord | 1,494 | 1,530 | 1,609 | 1,701 | 1,795 |
| El Cerrito | 397 | 420 | 411 | 416 | 419 |
| Fremont | 1,381 | 1,448 | 1,518 | 1,558 | 1,665 |
| Hayward | 2,221 | 2,282 | 2,359 | 2,433 | 2,498 |
| Lafayette | 481 | 495 | 487 | 495 | 532 |
| Livermore | 605 | 623 | 634 | 689 | 719 |
| Martinez | 345 | 348 | 352 | 349 | 361 |
| Moraga | ++ | | 124 | 124 | 127 |
| Newark | 290 | 306† | 322 | 346 | 362 |
| Oakland | 7,911 | 8,076 | 7,869 | 7,819 | 7,904 |

| Year 1975 1976 1977 Community East Bay (cont.) Piedmont 37 40 37 Piadmont 37 40 37 173 Pitedmont 37 40 37 Pinhole 166 173 173 Pittsburg 445 451 449 Pleasant Hill 396 407 416 Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt Arcata 414 433 435 Eureka 1,108 1,166 1,184 North Bay 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,08 | Average Number of Commercial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued) | | | | |
|---|---|-------|--|--|--|
| Lommunity East Bay (cont.) Piedmont 37 40 37 Pinhole 166 173 173 Pittsburg 445 451 449 Pleasant Hill 396 407 416 Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 1,108 1,166 1,184 North Bay 1,108 1,166 1,184 North Eay 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park | 1978 | 1979 | | | |
| East Bay (cont.) Piedmont 37 40 37 Pinhole 166 173 173 Pittsburg 445 451 449 Pleasant Hill 396 407 416 Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 1,108 1,184 1,184 North Bay 1,108 1,166 1,184 North Bay 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 301 | | | | | |
| Piedmont 37 40 37 Pinhole 166 173 173 Pittsburg 445 451 449 Pleasant Hill 396 407 416 Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 1,108 1,166 1,184 North Bay 1,108 1,166 1,184 North Bay 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park - 159 San Anselmo 29 | | | | | |
| Pinhole 166 173 173 Pittsburg 445 451 449 Pleasant Hill 396 407 416 Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 1,108 1,166 1,184 North Bay 1 1 1 1 Benicia 267 2 Larkspur 206 206 213 3 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park - 159 San Kafael 1,520 1,545 1,573 Santa Rosa | 34 | 33 | | | |
| Pittsburg 445 451 449 Pleasant Hill 396 407 416 Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 433 435 Eureka 1,108 1,166 1,184 North Bay 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park - 159 San Auselmo 295 301 301 Santa Rosa 1,908 1,933 1,926 Ukiah 415 <td< td=""><td>180</td><td>189</td></td<> | 180 | 189 | | | |
| Pleasant Hill 396 407 416 Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 433 435 Eureka 1,108 1,166 1,184 North Bay 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park - 159 San Auselmo 295 301 301 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 469 | 486 | | | |
| Pleasanton 348 378 417 Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 267 Arcata 414 433 435 Eureka 1,108 1,166 1,184 North Bay 267 Larkspur 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 </td <td>433</td> <td>466</td> | 433 | 466 | | | |
| Richmond 1,425 1,443 1,444 San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt Arcata 414 433 435 Eureka 1,108 1,166 1,184 North Bay 267 Larkspur 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 451 | 485 | | | |
| San Leandro 1,801 1,837 1,860 San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 433 435 Arcata 414 433 435 Eureka 1,108 1,166 1,184 North Bay 206 206 Larkspur 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 1,439 | 1,475 | | | |
| San Pablo 435 436 412 Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt 413 433 435 Arcata 414 433 435 5 Eureka 1,108 1,166 1,184 North Bay 206 206 213 Mill Valley 285 306 320 320 Napa 1,054 1,088 1,110 301 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park - 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 1,858 | 1,878 | | | |
| Union City 317 344 352 Walnut Creek 981 1,007 1,026 Humboldt Xarcata 414 433 435 Arcata 414 433 435 Eureka 1,108 1,166 1,184 North Bay North Bay Senicia 267 Larkspur 206 206 213 306 320 Napa 1,054 1,088 1,110 301 301 Novato 433 450 466 | 402 | 396 | | | |
| Walnut Creek 981 1,007 1,026 Humboldt Arcata 414 433 435 Eureka 1,108 1,166 1,184 North Bay Benicia 267 Larkspur 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 370 | 378 | | | |
| HumboldtArcata414433435Eureka1,1081,1661,184North BayBenicia267Larkspur206206213Mill Valley285306320Napa1,0541,0881,110Novato433450466Petaluma665697712Rohnert Park159San Anselmo295301301San Rafael1,5201,5451,573Santa Rosa1,9081,9331,926Ukiah415438447Vallejo1,4191,3901,395 | 1,057 | 1,160 | | | |
| Arcata414433435Eureka1,1081,1661,184North BayBenicia267Larkspur206206213Mill Valley285306320Napa1,0541,0881,110Novato433450466Petaluma665697712Rohnert Park159San Anselmo295301301Santa Rosa1,9081,9331,926Ukiah415438447Vallejo1,4191,3901,395 | | | | | |
| Eureka 1,108 1,166 1,184 North Bay Benicia 267 Larkspur 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 439 | 455 | | | |
| North Bay Benicia 267 Larkspur 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 1,212 | 1,227 | | | |
| Benicia 267 Larkspur 206 206 213 Mill Valley 285 306 320 Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | | | | | |
| Larkspur206206213Mill Valley285306320Napa1,0541,0881,110Novato433450466Petaluma665697712Rohnert Park159San Anselmo295301301San Rafael1,5201,5451,573Santa Rosa1,9081,9331,926Ukiah415438447Vallejo1,4191,3901,395 | 280 | 299 | | | |
| Mill Valley285306320Napa1,0541,0881,110Novato433450466Petaluma665697712Rohnert Park159San Anselmo295301301San Rafael1,5201,5451,573Santa Rosa1,9081,9331,926Ukiah415438447Vallejo1,4191,3901,395 | 227 | 250 | | | |
| Napa 1,054 1,088 1,110 Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 325 | 336 | | | |
| Novato 433 450 466 Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 1,168 | 1,218 | | | |
| Petaluma 665 697 712 Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 477 | 572 | | | |
| Rohnert Park 159 San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 723 | 761 | | | |
| San Anselmo 295 301 301 San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 | 197 | 214 | | | |
| San Rafael 1,520 1,545 1,573 Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 Sacramento 5 5 5 | 302 | 311 | | | |
| Santa Rosa 1,908 1,933 1,926 Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 Sacramento Sacramento Sacramento Sacramento Sacramento | 1,615 | 1,669 | | | |
| Ukiah 415 438 447 Vallejo 1,419 1,390 1,395 Sacramento | 1,956 | 2,069 | | | |
| Vallejo 1,419 1,390 1,395 Sacramento Image: Constraint of the second se | 477 | 531 | | | |
| Sacramento | 1,414 | 1,430 | | | |
| | | | | | |
| Davis 383 387 380 | 382 | 433 | | | |
| Fairfield 611 661 697 | 723 | 780 | | | |
| Sacramento 5,585 5,642 5,666 | 5,743 | 5,864 | | | |

Average Number of Commercial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------------|--------|--------|--------|--------|--------|
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 404 | 419 | 434 | 457 | 500 |
| Woodland | 594 | 608 | 611 | 655 | 713 |
| San Francisco | | | | | |
| Daly City | 690 | 676 | 676 | 673 | 669 |
| Millbrae | 303 | 300 | 301 | 304 | 303 |
| Pacifica | 279 | 306 | 299 | 289 | 305 |
| San Bruno | 573 | 587 | 592 | 594 | 598 |
| San Francisco | 16,739 | 17,415 | 16,971 | 17,014 | 17,026 |
| South San Francisco | 1,121 | 1,141 | 1,161 | 1,183 | 1,222 |
| San Joaquin | | | | • | |
| Atwater | 239 | 237 | 234 | 248 | 263 |
| Bakersfield | 2,790 | 2,824 | 2,844 | 2,972 | 3,068 |
| Clovis | 376 | 415 | 447 | 504 | 595 |
| Fresno | 5,082 | 5,291 | 5,443 | 5,612 | 5,944 |
| Los Banos | 308 | 313 | 309 | 314 | 317 |
| Madera | 514 | 518 | 509 | 499 | 563 |
| Merced | 921 | 939 | 948 | 954 | 1,001 |
| Ridgecrest | 302 | 321 | 318 | 334 | 357 |
| Sanger | 279 | 289 | 285 | 280 | 290 |
| San Jose | | | | | |
| Belmont | 394 | 408 | 405 | 413 | 422 |
| Burlingame | 992 | 1,004 | 1,001 | 1,017 | 1,029 |
| Campbell | 971 | 1,032 | 1,074 | 1,099 | 1,166 |
| Cupertino | 405 | 419 | 442 | 469 | 518 |
| Foster City | 111 | 116 | 130 | 149 | 184 |
| Gilroy | 491 | 489 | 495 | 511 | 569 |
| Los Altos | 550 | 565 | 568† | 571 | 585 |
| Los Gatos | 660 | 675 | 681 | 710 | 748 |
| Menlo Park | 704 | 718 | 712 | 728 | 734 |
| Milpitas | 307 | 308 | 312 | 326 | 367 |

| А | erage Number of Commercial Gas Customers - Communities of 10,000 opulation or More - Pacific Gas and Electric Company (Continued) | | | | | |
|---|--|------------|---------|---------|---------|--|
| Year | 1975 | 1976 | 1977 | 1978 | 1979 | |
| Community | | | | | | |
| | | | | | | |
| San Jose (cont.) | | | | | | |
| Mountain View | 1,660 | 1,688 | 1,727 | 1,817 | 1,900 | |
| Morgan Hill | | | 257 | 268 | 354 | |
| Redwood City | 1,309 | 1,328 | 1,324 | 1,345 | 1,349 | |
| San Carlos | 909 | 918 | 925 | 948 | 955 | |
| San Jose | 7,476 | 7,723 | 7,963 | 8,190 | 8,576 | |
| San Mateo | 1,728 | 1,769 | 1,767 | 1,798 | 1,808 | |
| Santa Clara | 2,108 | 2,165 | 2,303 | 2,434 | 2,598 | |
| Santa Cruz | 1,227 | 1,265 | 1,280 | 1,301 | 1,308 | |
| Saratoga | 324 | 340 | 346 | 354 | 371 | |
| Sunnyvale | 1,631 | 1,713 | 1,789 | 1,875 | 2,082 | |
| Watsonville | 674 | 697 | 716 | 728 | 740 | |
| Shasta | | | | | | |
| Redding | 799 | 854 | 864 | 1,091 | 1,284 | |
| Stockton | | | • | | | |
| Ceres | · | novik asat | 244 | 246 | 259 | |
| Lodi | 937 | 941 | 941 | 966 | 1,001 | |
| Manteca | 415 | 426 | 448 | 490 | 517 | |
| Modesto | 1,842 | 1,885 | 1,938 | 2,035 | 2,197 | |
| Stockton | 2,868 | 2,939 | 2,999 | 3,116 | 3,270 | |
| Tracy | 394 | 422 | 415 | 417 | 442 | |
| Turlock | 633 | 631 | 627 | 654 | 711 | |
| Customers in Communities of 10,000 Popula- tion and Over | 112,066 | 133,292 | 116,371 | 119,544 | 124,303 | |
| Customers in Comm nities of Less Than 10,000 Population | 41,939 | 24,638 | 43,030 | 43,573 | 44,794 | |
| | | | | | | |
| Total | 154,005 | 157,930 | 159,401 | 163,117 | 169,097 | |

† See Table C-2

 †† data not indicated (community with less than 10,000 population at that time) 172

TABLE C-3

Average Number of Industrial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------|------|------|------|------|------|
| Community | | | | | |
| Coast Valleys | | | | | |
| Monterey | 12 | 11 | 9 | 9 | 9 |
| Pacific Grove | 3 | 3 | 3 | 3 | 3 |
| Salinas | 20 | 19 | 15 | 15 | 16 |
| Seaside | 1 | 1† | 1 | 1 | 1 |
| Colgate | | | | | |
| Yuba City | 8 | 9 | 9 | 8 | 8 |
| De Sabla | | | | • | |
| Chico | 12 | 12 | 11 | 11 | 11 |
| Drum | | | | | |
| Roseville | 12 | 11 | 8 | 8 | 7 |
| East Bay | | | | | |
| Alameda | 20 | 18 | 18 | 17 | 17 |
| Albany | 5 | 5† | 5 | 5 | 5 |
| Antioch | 9 | 9 | 7 | 5 | 5 |
| Berkeley | 72 | 69 | 54 | 19 | 19 |
| Concord | 8 | 8 | 9 | 9 | 9 |
| El Cerrito | 1 | 1 | 1 | 1 | 1 |
| Fremont | 25 | 24 | 20 | 20 | 21 |
| Hayward | 49 | 48 | 44 | 45 | 45 |
| Lafayette | 0 | 0 | 0 | 0 | 0 |
| Livermore | 6 | 6 | 4 | 4 | 3 |
| Martinez | 8 | 8 | 8 | 8 | 8 |
| Moraga | 0 | 0 | 1 | 1 | 1 |
| Newark | 28 | 25† | 23 | 22 | 22 |
| Oakland | 225 | 204 | 181 | 175 | 172 |

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|------------------|------|------|------|------|------|
| Community | | | | | |
| East Bay (cont.) | | | | | |
| Piedmont | 0 | ð | Q | 0 | 0 |
| Pinhole | 0 | 0 | 0 | 0 | 0 |
| Pittsburg | 9 | 9 | 8 | 8 | 8 |
| Pleasant Hill | 0 | 0 | 0 | 0 | 0 |
| Pleasanton | 8 | 6 | 5 | 5 | 5 |
| Richmond | 76 | 74 | 67 | 65 | 64 |
| San Leandro | 94 | 90 | 80 | 80 | 76 |
| San Pablo | 3 | 3 | 3 | 3 | 3 |
| Union City | 38 | 38 | 35 | 34 | 33 |
| Walnut Creek | 3 | 3 | 2 | 2 | 2 |
| Humboldt | | | | | |
| Arcata | 8 | 5 | 2 | 2 | 2 |
| Eureka | 21 | 18 | 15 | 15 | 13 |
| North Bay | | | | | |
| Benicia | 0 | 0 | 4 | 4 | 4 |
| Larkspur | 5 | 4 | 3 | 3 | 3 |
| Mill Valley | 0 | 0 | 0 | 0 | 0 |
| Napa | 9 | 8 | 5 | 5 | 5 |
| Novato | 1 | 0 | 0 | 0 | 0 |
| Petaluma | 15 | 15 | 14 | 14 | 13 |
| Rohnert Park | 0 | 0 | 0 | 0 | 0 |
| San Anselmo | 0 | 0 | 0 | 0 | 0 |
| San Rafael | 10 | 11 | 11 | 10 | 10 |
| Santa Rosa | 11 | 11 | 10 | 10 | 10 |
| Ukiah | 2 | 2 | 2 | 2 | 2 |
| Vallejo | 12 | 11 | 6 | 6 | 6 |
| Sacramento | | | | | |
| Davis | 3 | . 3 | 3 | 3 | 3 |
| Fairfield | 3 | 4 | 2 | 2 | 2 |
| Sacramento | 104 | 95 | 83 | 83 | 82 |

Average Number of Industrial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

Average Number of Industrial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------------|------|------|------|------|------|
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 5 | 5 | 4 | 4 | 3 |
| Woodland | 9 | 9 | 7 | 7 | 7 |
| San Francisco | | | | | |
| Daly City | 8 | 7 | 7 | 6 | 6 |
| Millbrae | 1 | 1 | 0 | 0 | 0 |
| Pacifica | 3 | 2 | 1 | 1 | 1 |
| San Bruno | 5 | 5 | 3 | 3 | 3 |
| San Francisco | 301 | 238 | 200 | 191 | 183 |
| South San Francisco | 78 | 73 | 64 | 62 | 60 |
| San Joaquin | | | | | |
| Atwater | 1 | 1 | 1 | 1 | 1 |
| Bakersfield | 35 | 33 | 31 | 31 | 31 |
| Clovis | 1 | 1 | 1 | 1 | 1 |
| Fresno | 102 | 93 | 83 | 83 | 82 |
| Los Banos | 6 | 6 | 5 | 3 | 2 |
| Madera | 6 | 7 | 5 | 5 | 5 |
| Merced | 12 | 12 | 10 | 10 | 10 |
| Ridgecrest | 1 | 1 | 1 | 1 | 1 |
| Sanger | 11 | 11 | 10 | 10 | 10 |
| San Jose | | | | | |
| Belmont | 1 | 1 | 1 | 1 | 0 |
| Burlingame | 6 | 6 | 6 | 6 | 2 |
| Campbell | 3 | 3 | 3 | 2 | 2 |
| Cupertino | 8 | 9 | 10 | 12 | 12 |
| Foster City | 0 | 0 | 0 | 0 | 0 |
| Gilrov | 8 | 8 | 7 | 8 | 8 |
| Los Altos | 1 | 1 | 1† | 1 | -1 |
| Los Gatos | 5 | 5 | 4 | 4 | 4 |
| Menlo Park | 10 | 9 | 9 | 9 | 8 |
| Milpitas | 22 | 21 | 19 | 18 | 19 |

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---|-------|-------|-------|-------|-------|
| Community | | | | | |
| <u>San Jose</u> (cont.) | | | | | |
| Mountain View | 38 | 38 | 38 | 38 | 36 |
| Morgan Hill | 0 | 0 | 4 | 3 | 3 |
| Redwood City | 29 | 27 | 25 | 22 | 21 |
| San Carlos | 18 | 18 | 17 | 17 | 16 |
| San Jose | 140 | 135 | 130 | 134 | 136 |
| San Mateo | 15 | 13 | 11 | 10 | 9 |
| Santa Clara | 46 | 44 | 41 | 37 | 36 |
| Santa Cruz | 23 | 22 | 19 | 19 | 19 |
| Saratoga | 2 | 2 | 2 | 2 | 2 |
| Sunnyvale | -39 | 39 | 33 | 31 | 30 |
| Watsonville | 19 | 18 | 17 | 17 | 17 |
| Shasta | | | | | |
| Redding | 13 | 12 | 7 | 12 | 13 |
| Stockton | • | | | | |
| Ceres | 0 | 0 | 1 | 1 | 1 |
| Lodi | 21 | 21 | 21 | 21 | 20 |
| Manteca | 3 | 3 | 3 | 3 | 3 |
| Modesto | 37 | 35 | 33 | 33 | 31 |
| Stockton | 62 | 59 | 58 | 58 | 58 |
| Tracy | 7 | 6 | 5 | 5 | 5 |
| Turlock | 23 | 24 | 22 | 20 | 19 |
| Customers in Communities of 10,000 Popula- tion and Over | 2,147 | 2,408 | 1.775 | 1.705 | 1,667 |
| Customers in Commu- nities of Less Than 10,000 | 2,14/ | 2,400 | 1,1/0 | 1,100 | 1,007 |
| ropulation | 1,416 | 940 | 1,162 | 1,148 | 1,126 |
| Total | 3,563 | 3,348 | 2,937 | 2,853 | 2,793 |

Average Number of Industrial Gas Customers - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

Source: Annual Reports to the California Public Utilities Commission, 1975-1979, Pacific Gas and Electric Company.

. .

| ructife out and ficetife company | | | | | | |
|----------------------------------|------------|------------|------------|---|------------|--|
| Year | 1975 | 1976 | 1977 | 1978 | 1979 | |
| Community | | | | anaganggang ata ang ang ata ang ang ata ang ang ata ang | | |
| Gommentey | | | | | | |
| Coast Valleys | | | | | | |
| Monterey | 1,877,721 | 1,702,422 | 1,502,737 | 1,546,457 | 1,611,966 | |
| Pacific Grove | 833,733 | 752,351 | 665,553 | 659,489 | 682,675 | |
| Salinas | 3,655,150 | 3,366,499 | 3,175,063 | 2,992,291 | 3,051,111 | |
| Seaside | 1,003,854 | 921,335 | 836,335 | 797,061 | 816,355 | |
| Colgate | | | | | | |
| Yuba City | 965,774 | 981,723 | 979,611 | 1,020,505 | 1,074,526 | |
| De Sabla | | | | | | |
| Chico | 1,396,994 | 1,243,349 | 1,173,591 | 1,172,709 | 1,219,075 | |
| Drum | | | | | | |
| Roseville | 1,333,668 | 1,197,468 | 1,096,588 | 1,119,684 | 1,192,469 | |
| East Bay | | | | | | |
| Alameda | 3,869,983 | 8,539,378 | 3,016,016 | 3,240,332 | 3,373,209 | |
| Albany | 916,673 | 783,969 | 740,174 | 724,132 | 746,765 | |
| Antioch | 1,803,985 | 1,540,515 | 1,208,426 | 1,164,128 | 1,310,911 | |
| Berkelev | 8,081,846 | 7,096,082 | 6,371,396 | 6,263,691 | 6,774,647 | |
| Concord | 3,889,735 | 3,630,623 | 3,224,044 | 3,182,167 | 3,474,423 | |
| El Cerríto | 1,118,488 | 967,266 | 860,443 | 844,402 | 896,743 | |
| Fremont | 6,367,067 | 6,640,732 | 6,369,574 | 6,360,047 | 6,385,544 | |
| Havward | 6,411,453 | 5,834,443 | 5,441,285 | 5,085,216 | 5,314,905 | |
| Lafavette | 1,200,410 | 1,087,509 | 884,165 | 923,342 | 966,592 | |
| Livermore | 2,015,434 | 1,932,895 | 1,767,623 | 1,738,383 | 1,855,264 | |
| Martinez | 1,253,438 | 1,196,100 | 1,064,888 | 1,092,961 | 1,156,164 | |
| Moraga | † | 4000 8000 | 627,179 | 651,437 | 675,885 | |
| Newark | 3,265,622 | 3,512,503 | 3,371,592 | 3,433,326 | 3,597,987 | |
| Oakland | 24,669,933 | 22,976,464 | 21,198,287 | 20,626,399 | 20,560,562 | |

Total Residential, Commercial, and Industrial Gas Sales (MCF) Communities of 10,000 Population or More Pacific Gas and Electric Company

TABLE C-5

| Year | 1975 | 1976 | 1977 | 1978 | 1979 | | |
|------------------|-------------|-------------------|------------|------------|------------|--|--|
| Community | | | | | | | |
| East Bay (cont.) | | | | | | | |
| Piedmont | 689,097 | 603,652 | 517,789 | 525,085 | 541,126 | | |
| Pinhole | 660,494 | 577,306 | 486,309 | 474,587 | 500,696 | | |
| Pittsburg | 1,580,584 | 1,504,286 | 1,393,551 | 1,181,382 | 1,265,576 | | |
| Pleasant Hill | 1,100,694 | 1,002,124 | 855,757 | 833,715 | 958,614 | | |
| Pleasanton | 1,608,356 | 1,532,728 | 1,342,160 | 1,320,816 | 1,405,069 | | |
| Richmond | .37,779,623 | 39,288,828 | 38,157,081 | 33,725,841 | 37,749,652 | | |
| San Leandro | 6,432,310 | 6,142,142 | 5,371,415 | 5,129,881 | 5,308,937 | | |
| San Pablo | 920,150 | 852,023 | 789,955 | 762,108 | 818,421 | | |
| Union City | 3,780,590 | 3,862,223 | 3,847,246 | 3,359,126 | 1,825,373 | | |
| Walnut Creek | 2,510,115 | 2,323,741 | 1,977,903 | 2,045,057 | 2,230,412 | | |
| Humboldt | | | | | | | |
| Arcata | 669,559 | 684,280 | 611,187 | 611,893 | 640,147 | | |
| Eureka | 1,753,302 | 1,689,825 | 1,488,633 | 1,392,447 | 1,335,709 | | |
| North Bay | | | | | | | |
| Benicia | | | 6,968,118 | 6,241,088 | 8,691,701 | | |
| Larkspur | 636,197 | 547,419 | 455,417 | 471,369 | 495,107 | | |
| Mill Valley | 721,607 | 636,107 | 553,225 | 554,845 | 594,768 | | |
| Napa | 2,293,803 | 2,144,129 | 1,944,668 | 1,854,164 | 2,011,234 | | |
| Novato | 1,781,294 | 1,647,691 | 1,440,789 | 1,427,362 | 1,574,491 | | |
| Petaluma | 1,754,898 | 1,608,681 | 1,446,915 | 1,397,985 | 1,552,879 | | |
| Rohnert Park | | | 546,918 | 581,688 | 675,710 | | |
| San Anselmo | 647,176 | 578,999 | 489,621 | 485,137 | 527,377 | | |
| San Rafael | 2,638,724 | 2,394,386 | 2,011,923 | 2,004,087 | 2,177,882 | | |
| Santa Rosa | 3,643,899 | 3,435,754 | 3,162,562 | 3,191,788 | 3,466,752 | | |
| Ukiah | 379,809 | 373,551 | 363,863 | 367,414 | 409,608 | | |
| Vallejo | 4,828,587 | 4,296,100 | 3,721,535 | 3,590,773 | 3,824,548 | | |
| Sacramento | | | | | | | |
| Davis | 1,890,382 | 1,563,556 | 1,467,487 | 1,373,552 | 1,741,012 | | |
| Fairfield | 2,515,316 | 2,434,110 | 2,866,868 | 2,759,021 | 3,147,235 | | |
| Sacramento | 17,797,055 | 15,643,733 178 | 14,281,043 | 13,727,446 | 14,675,218 | | |

Total Residential, Commercial, and Industrial Gas Sales (MCF) Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

| Tota | l Residential, Communitie Pacifi | Commercial, as s of 10,000 Po c Gas and Elec | nd Industrial (pulation or Mon tric Company ((| Gas Sales (MCF) re Continued) |) |
|---------------------|--|--|---|-------------------------------------|------------|
| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 2,010,436 | 1,663,972 | 1,813,798 | 1,644,492 | 1,855,923 |
| Woodland | 2,113,720 | 1,733,024 | 1,564,876 | 1,481,938 | 1,686,809 |
| San Francisco | | | | | |
| Daly City | 3,340,203 | 2,941,924 | 2,724,055 | 2,568,475 | 2,659,802 |
| Millbrae | 1,041,351 | 965,827 | 847,290 | 816,937 | 846,997 |
| Pacifica | 1,520,756 | 1,432,371 | 1,267,778 | 1,189,224 | 1,222,686 |
| San Bruno | 1,538,000 | 1,408,365 | 1,273,442 | 1,197,805 | 1,256,700 |
| San Francisco | 43,373,592 | 39,401,383 | 36,050,171 | 34,172,294 | 34,949,780 |
| South San Francisco | 4,435,984 | 4,113,417 | 3,871,814 | 3,511,151 | 3,687,949 |
| San Joaquin | | | | | |
| Atwater | 838,152 | 702,575 | 690,103 | 657,298 | 645,048 |
| Bakersfield | 4,106,024 | 3,706,383 | 3,589,258 | 3,456,051 | 3,755,702 |
| Clovis | 769,461 | 766,624 | 782,166 | 792,844 | 901,363 |
| Fresno | 9,470,263 | 9,094,627 | 8,538,188 | 8,311,411 | 8,746,587 |
| Los Banos | 625,509 | 619,310 | 599,945 | 540,861 | 543,387 |
| Madera | 752,412 | 712,931 | 673,759 | 652,027 | 715,232 |
| Merced | 1,988,054 | 1,663,913 | 1,739,652 | 1,556,981 | 1,775,263 |
| Ridgecrest | 892,707 | 799,415 | 750,341 | 668,690 | 775,121 |
| Sanger | 508,364 | 487,210 | 462,342 | 419,229 | 460,342 |
| San Jose | | | | | |
| Belmont | 1,151,063 | 1,057,767 | 939,031 | 913,834 | 979,115 |
| Burlingame | 1,879,993 | 1,744,457 | 1,535,851 | 1,439,529 | 1,531,932 |
| Campbell | 1,203,173 | 1,135,268 | 997,953 | 981,730 | 1,059,984 |
| Cupertino | 1,288,263 | 1,231,831 | 1,165,809 | 1,203,947 | 1,397,050 |
| Foster City | 822,187 | 780,809 | 692,296 | 706,754 | 782,274 |
| Gilroy | 2,431,266 | 1,866,065 | 1,963,178 | 1,952,207 | 2,226,136 |
| Los Altos | 1,473,978 | 1,385,311 | 1,390,082† | 1,199,359 | 1,199,693 |
| Los Gatos | 1,344,856 | 1,311,744 | 1,180,082 | 1,213,727 | 1,250,052 |
| Menlo Park | 2,031,234 | 1,905,437 | 1,615,186 | 1,577,077 | 1,603,887 |
| Milpitas | 1,884,370 | 1,817,967 | 1,763,434 | 1,718,793 | 1,933,144 |

| | 1075 | 1070 | 1077 | 1070 | 1070 |
|---|-------------|-------------|-------------|-------------|-------------|
| iear | 1975 | 1976 | 1977 | 1978 | 1979 |
| Community | | | | | |
| San Jose (cont.) | | | | | |
| Mountain View | 3,253,993 | 3,088,847 | 2,778,691 | 2,737,996 | 2,865,935 |
| Morgan Hill | | | 433,545 | 465,214 | 529,074 |
| Redwood City | 3,075,480 | 2,935,215 | 2,609,250 | 2,527,263 | 2,664,320 |
| San Carlos | 1,486,902 | 1,393,617 | 1,215,296 | 1,153,331 | 1,251,155 |
| San Jose | 28,807,313 | 26,942,260 | 25,745,168 | 24,887,438 | 26,208,402 |
| San Mateo | 4,013,489 | 3,762,436 | 3,287,607 | 3,134,361 | 3,329,022 |
| Santa Clara | 7,388,178 | 7,468,244 | 7,214,952 | 7,246,007 | 7,842,281 |
| Santa Cruz | 2,326,062 | 2,106,514 | 1,877,738 | 1,844,627 | 1,918,862 |
| Saratoga | 1,705,397 | 1,664,581 | 1,414,374 | 1,399,227 | 1,397,975 |
| Sunnvvale | 7,070,918 | 6,412,670 | 5,984,184 | 5,897,853 | 6,141,255 |
| Watsonville | 1,106,149 | 1,093,686 | 1,120,089 | 1,073,951 | 1,087,247 |
| Shasta | | | | | |
| Redding | 739,644 | 661,856 | 653,788 | 874,899 | 1,209,403 |
| Stockton | | | | | |
| Ceres | | | 383,537 | 384,216 | 416,545 |
| Lodi | 2,263,772 | 1,928,928 | 1,978,887 | 1,840,563 | 2,071,888 |
| Manteca | 687,505 | 653,325 | 661,578 | 708,207 | 779,622 |
| Modesto | 5,481,992 | 5,140,000 | 5,281,631 | 5,228,792 | 5,680,889 |
| Stockton | 10,140,602 | 9,383,709 | 8,370,091 | 7,493,400 | 8,971,866 |
| Tracy | 1,661,645 | 1,598,753 | 1,627,626 | 1,683,575 | 1,789,874 |
| Turlock | 1,264,154 | 1,052,911 | 1,034,665 | 1,113,997 | 1,214,500 |
| Total Sales in Communities of 10,000 Popula- tion and Over | 354,975,928 | 333,006,041 | 317,498,993 | 304,267,398 | 325,120,401 |
| Sales in Communi- ties of Less Than 10,000 Population | 306,435,048 | 269,114,358 | 232,492,042 | 198,811,299 | 238,959,927 |

Total Residential, Commercial, and Industrial Gas Sales (MCF) Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

Total Residential, Commercial, and Industrial Gas Sales (MCF) Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|--------------------------------------|-------------|-------------|-------------|-------------|-------------|
| Other Sales to Public Authorities | 6,447 | 3,056 | 1,683 | 1,339 | 1,356 |
| Sales for Resale | 9,458,672 | 8,715,861 | 7,810,276 | 9,926,108 | 36,013,469 |
| Interdepartmental Sales | 159,223,561 | 195,063,458 | 217,368,151 | 125,768,565 | 216,147,045 |
| Total | 829,999,656 | 805,902,774 | 775,171,145 | 638,774,709 | 816,242,198 |

Source: Annual Reports to the California Public Utilities Commission, 1975-1979, Pacific Gas and Electric Company.

+ : See Table C-3
++ : See Table C-2

Total Residential Gas Sales (MCF) - Communities of 10,000 Population or More Pacific Gas and Electric Company

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------|------------|------------|------------|-----------|------------|
| Community | | | | | |
| Coast Valleys | | | | | |
| Monterey | 869,015 | 840,933 | 795,610 | 797,806 | 823,245 |
| Pacific Grove | 626,673 | 582,432 | 523,835 | 526,698 | 545,668 |
| Salinas | 2,157,332 | 2,028,124 | 1,915,264 | 1,815,034 | 1,861,519 |
| Seaside | 700,579 | 684,080 | 692,595 | 667,698 | 674,521 |
| Colgate | | | | | |
| Yuba City | 553,569 | 510,970 | 478,916 | 498,731 | 562,436 |
| De Sabla | | | | | |
| Chico | 763,992 | 680,163 | 652,586 | 662,895 | 682,021 |
| Drum | | | | | |
| Roseville | 778,268 | 682,128 | 637,052 | 658,093 | 714,503 |
| East Bay | | | | | • |
| Alameda | 2,139,111 | 1,924,907 | 1,691,719 | 1,634,630 | 1,748,534 |
| Albany | 479,356 | 469,035 | 430,336 | 407,733 | 432,689 |
| Antioch | 1,093,148 | 989,945 | 905,287 | 962,414 | 1,075,233 |
| Berkeley | 3,890,451 | 3,416,774 | 3,073,915 | 2,966,999 | 3,139,785 |
| Concord | 3,137,965 | 2,960,868 | 2,615,114 | 2,607,092 | 2,832,474 |
| El Cerrito | 960,094 | 826,827 | 736,244 | 733,550 | 768,599 |
| Fremont | 3,907,581 | 3,672,330 | 3,221,078 | 3,198,767 | 3,411,058 |
| Hayward | 2,865,249 | 2,773,671 | 2,474,486 | 2,435,219 | 2,570,195 |
| Lafayette | 1,079,792 | 972,254 | 782,289 | 827,391 | 862,246 |
| Livermore | 1,694,658 | 1,580,801 | 1,403,516 | 1,356,636 | 1,438,202 |
| Martinez | 680,858 | 623,427 | 570,377 | 584,446 | 655,058 |
| Moraga | ++ | | 516,811 | 546,203 | 571,066 |
| Newark | 905,260 | 859,813 | 757,160 | 732,533 | 777,332 |
| Oakland | 12,214,946 | 11,009,179 | 10,245,149 | 9,920,488 | 10,567,025 |

Total Residential Gas Sales (MCF) - Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|------------------|------------|-----------|-----------|-----------|-----------|
| Community | | | | | |
| East Bay (cont.) | | | | | |
| Piedmont | 664,901 | 582,156 | 499,219 | 505,645 | 522,121 |
| Pinhole | 584,500 | 514,713 | 431,091 | 422,662 | 446,883 |
| Pittsburg | 852,725 | 796,673 | 749,397 | 760,778 | 842,130 |
| Pleasant Hill | 871,185 | 788,272 | 664,258 | 656,726 | 721,599 |
| Pleasanton | 1,204,412 | 1,167,978 | 1,051,496 | 1,034,885 | 1,079,611 |
| Richmond | 2,239,908 | 2,394,774 | 2,213,632 | 2,128,694 | 2,279,539 |
| San Leandro | 2,137,116 | 2,012,360 | 1,758,923 | 1,697,389 | 1,810,176 |
| San Pablo | 594,407 | 539,587 | 532,279 | 515,315 | 545,571 |
| Union City | 924,764 | 947,526 | 885,450 | 845,953 | 935,110 |
| Walnut Creek | 1,868,032 | 1,728,199 | 1,451,322 | 1,509,420 | 1,658,778 |
| Humboldt | | | | | |
| Arcata | 348,414 | 347,827 | 312,570 | 301,987 | 317,660 |
| Eureka | 1,040,022 | 978,177 | 858,231 | 821,016 | 835,895 |
| North Bay | | | | | |
| Benicia | പതം എം | | 361,226 | 366,847 | 425,006 |
| Larkspur | 451,516 | 424,057 | 362,739 | 379,782 | 396,620 |
| Mill Valley | 625,768 | 539,353 | 466,094 | 467,779 | 501,216 |
| Napa | 1,677,803 | 1,568,430 | 1,424,567 | 1,385,821 | 1,522,013 |
| Novato | 1,279,345 | 1,288,921 | 1,147,650 | 1,145,392 | 1,252,352 |
| Petaluma | 1,184,837 | 1,101,651 | 976,989 | 949,145 | 1,049,115 |
| Rohnert Park | | | 475,756 | 512,744 | 589,954 |
| San Anselmo | 554,921 | 489,363 | 412,265 | 412,179 | 449,332 |
| San Rafael | 1,952,415 | 1,745,383 | 1,433,583 | 1,466,119 | 1,575,828 |
| Santa Rosa | 2,544,234 | 2,461,734 | 2,281,636 | 2,320,576 | 2,529,231 |
| Ukiah | 245,764 | 241,228 | 233,327 | 236,893 | 261,776 |
| Vallejo | 2,276,122 | 2,115,647 | 1,976,475 | 1,887,515 | 2,143,014 |
| Sacramento | | | | | |
| Davis | 1,017,941 | 934,608 | 870,328 | 869,833 | 929,105 |
| Fairfield | 1,306,828 | 1,256,033 | 1,269,744 | 1,262,660 | 1,407,056 |
| Sacramento | 10,484,470 | 9,185,083 | 8,411,887 | 8,070,455 | 8,666,892 |

Total Residential Gas Sales (MCF) - Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1.979 |
|---------------------|------------|------------|------------|------------|------------|
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 931,703 | 921,945 | 921,198 | 984,500 | 1,101,254 |
| Woodland | 951,786 | 853,670 | 797,249 | 785,214 | 839,853 |
| San Francisco | | | | | |
| Daly City | 2,722,094 | 2,437,040 | 2,241,245 | 2,124,042 | 2,189,948 |
| Millbrae | 864,312 | 799,388 | 692,931 | 673,155 | 701,509 |
| Pacifica | 1,394,441 | 1,269,338 | 1,127,636 | 1,075,577 | 1,100,742 |
| San Bruno | 1,295,462 | 1,184,002 | 1,057,249 | 1,001,764 | 1,049,695 |
| San Francisco | 25,369,811 | 22,955,461 | 21,504,945 | 20,481,213 | 21,337,976 |
| South San Francisco | 1,630,653 | 1,499,912 | 1,350,052 | 1,284,290 | 1,346,685 |
| San Joaquin | | | | | |
| Atwater | 326,179 | 305,922 | 370,829 | 369,617 | 388,047 |
| Bakersfield | 2,504,914 | 2,302,950 | 2,295,435 | 2,277,852 | 2,440,822 |
| Clovis | 624,194 | 653,898 | 664,766 | 672,836 | 754,886 |
| Fresno | 5,859,503 | 5,483,399 | 5,263,631 | 5,160,075 | 5,439,752 |
| Los Banos | 289,846 | 270,159 | 250,890 | 243,255 | 252,148 |
| Madera | 547,578 | 513,224 | 488,127 | 476,645 | 517,574 |
| Merced | 949,117 | 885,722 | 850,599 | 829,667 | 880,058 |
| Ridgecrest | 335,777 | 281,952 | 282,850 | 294,158 | 337,048 |
| Sanger | 277,036 | 262,503 | 246,597 | 232,268 | 256,500 |
| San Jose | | | | | |
| Belmont | 946,193 | 874,747 | 775,625 | 772,957 | 834,182 |
| Burlingame | 1,164,578 | 1,075,634 | 933,876 | 897,354 | 945,835 |
| Campbell | 840,288 | 800,860 | 702,752 | 693,726 | 739,657 |
| Cupertino | 865,120 | 812,687 | 731,187 | 726,081 | 865,854 |
| Foster City | 760,501 | 728,938 | 644,332 | 660,750 | 725,218 |
| Gilroy | 441,289 | 430,826 | 433,608 | 432,124 | 474,617 |
| Los Altos | 1,274,907 | 1,194,131 | 1,113,660+ | 1,033,190 | 1,031,288 |
| Los Gatos | 965,928 | 932,450 | 838,888 | 872,344 | 896,707 |
| Menlo Park | 1,145,507 | 1,080,680 | 932,235 | 919,820 | 956,357 |
| Milpitas | 965,493 | 934,851 | 798,132 | 785,280 | 903,247 |

Total Residential Gas Sales (MCF) - Communities of 10,000 Population or More Pacific Gas and Electric Company (Continued)

| <u>Community</u> <u>San Jose</u> (cont.) Mountain View Morgan Hill | | | | | |
|---|-------------|-------------|-------------|-------------|-------------|
| <u>San Jose</u> (cont.) Mountain View Morgan Hill | | | | | |
| Mountain View Morgan Hill | | | | | |
| Morgan Hill | 1,818,780 | 1,793,185 | 1,592,456 | 1,583,405 | 1,682,799 |
| | 10.00 CT20 | 4200 gass | 330,610 | 370,877 | 410,844 |
| Redwood City | 1,855,800 | 1,763,346 | 1,526,452 | 1,499,345 | 1,584,665 |
| San Carlos | 1,071,892 | 990,452 | 848,075 | 824,824 | 887,614 |
| San Jose | 18,082,088 | 17,296,897 | 16,107,682 | 15,679,755 | 16,298,261 |
| San Mateo | 2,964,846 | 2,746,856 | 2,401,025 | 2,328,236 | 2,494,957 |
| Santa Clara | 2,642,167 | 2,487,194 | 2,187,779 | 2,156,501 | 2,324,099 |
| Santa Cruz | 1,246,024 | 1,121,192 | 994,168 | 1,005,762 | 1,061,224 |
| Saratoga | 1,487,346 | 1,419,818 | 1,212,817 | 1,209,108 | 1,200,158 |
| Sunnyvale | 3,477,768 | 3,381,453 | 3,085,050 | 3,018,703 | 3,116,413 |
| Watsonville | 473,266 | 458,089 | 444,009 | 448,604 | 471,708 |
| Shasta | | | | | |
| Redding | 277,479 | 233,935 | 233,396 | 391,695 | 653,724 |
| Stockton | | | | | |
| Ceres | igaa vaa | - | 288,716 | 297,837 | 324,118 |
| Lodi | 1,136,378 | 1,019,757 | 979,966 | 966,244 | 1,038,718 |
| Manteca | 535,253 | 509,091 | 507,381 | 545,383 | 605,905 |
| Modesto | 2,938,952 | 2,782,117 | 2,750,027 | 2,761,110 | 2,987,979 |
| Stockton | 3,826,435 | 3,480,984 | 3,421,042 | 3,530,681 | 3,874,219 |
| Tracy | 488,430 | 438,835 | 419,955 | 411,883 | 445,499 |
| Turlock | 553,167 | 525,610 | 514,877 | 543,625 | 611,629 |
| Total Sales in Communities of 10,000 Popula- tion and Over | 181,936,348 | 169,163,622 | 155,924,710 | 153,808,598 | 164,149,998 |
| Sales in Communi- ties of Less Than 10,000 Population | 80,426,464 | 74,094,258 | 67,806,827 | 66,267,823 | 70,144,714 |
| Total 2 | 262,362,812 | 243,257,880 | 233,731,537 | 220,076,421 | 234,294,712 |

: See Table C-3 ++

Total Commercial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company

| Year | 1975 | 1.976 | 1977 | 1978 | 1979 |
|---------------|-----------|-----------|-----------|-----------|-----------|
| Community | | | | | |
| Coast Valleys | | | | | |
| Monterey | 725,093 | 608,906 | 507,310 | 508,617 | 549,736 |
| Pacific Grove | 196,138 | 158,445 | 130,429 | 123,005 | 129,118 |
| Salinas | 995,535 | 899,140 | 927,824 | 864,641 | 899,595 |
| Seaside | 300,099 | 233,851 | 140,397 | 126,095 | 138,742 |
| Colgate | | | | | |
| Yuba City | 222,403 | 189,503 | 171,428 | 182,435 | 204,659 |
| De Sabla | | | | | |
| Chico | 462,481 | 410,243 | 391,815 | 387,868 | 408,897 |
| Drum | | | | | |
| Roseville | 220,434 | 205,786 | 253,902 | 266,315 | 286,145 |
| East Bay | | | | | |
| Alameda | 503,549 | 472,678 | 426,556 | 408,297 | 438,808 |
| Albany | 186,535 | 121,322 | 152,341 | 160,117 | 156,199 |
| Antioch | 202,917 | 183,875 | 183,675 | 179,453 | 205,392 |
| Berkeley | 1,231,508 | 1,163,789 | 1,149,592 | 1,267,498 | 1,387,022 |
| Concord | 698,318 | 578,116 | 520,657 | 497,199 | 563,140 |
| El Cerrito | 155,891 | 137,988 | 122,121 | 109,237 | 126,116 |
| Fremont | 656,150 | 643,658 | 1,833,520 | 2,232,282 | 2,110,599 |
| Hayward | 1,202,075 | 1,075,203 | 1,350,338 | 1,200,090 | 1,282,788 |
| Lafayette | 120,618 | 115,255 | 101,876 | 95,951 | 104,346 |
| Livermore | 217,112 | 205,138 | 231,905 | 217,319 | 244,649 |
| Martinez | 222,155 | 203,618 | 197,286 | 174,283 | 187,573 |
| Moraga | + | | 54,076 | 48,438 | 50,060 |
| Newark | 120,821 | 128,955 | 1,156,361 | 1,084,094 | 1,131,966 |
| Oakland | 4,152,076 | 3,875,599 | 5,636,380 | 4,438,228 | 4,513,713 |

Total Commercial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|------------------|-----------|-----------|------------|------------|------------|
| Community | | | | | |
| East Bay (cont.) | | | | | |
| Piedmont | 24,196 | 21,496 | 18,570 | 19,440 | 19,005 |
| Pinhole | 75,994 | 62,593 | 55,218 | 51,925 | 53,813 |
| Pittsburg | 224,925 | 191,089 | 206,064 | 220,724 | 237,366 |
| Pleasant Hill | 229,509 | 213,852 | 191,499 | 176,989 | 237,015 |
| Pleasanton | 194,154 | 180,343 | 218,428 | 183,810 | 183,427 |
| Richmond | 670,657 | 617,815 | 30,850,461 | 29,743,789 | 22,345,971 |
| San Leandro | 906,860 | 802,506 | 1,101,693 | 1,032,320 | 1,103,443 |
| San Pablo | 258,264 | 243,197 | 189,638 | 170,741 | 176,506 |
| Union City | 148,737 | 140,482 | 180,907 | 194,468 | 179,863 |
| Walnut Creek | 548,079 | 512,682 | 451,890 | 456,867 | 489,488 |
| Humboldt | | | | | |
| Arcata | 254,811 | 286,087 | 294,860 | 306,967 | 319,797 |
| Eureka | 490,923 | 497,228 | 485,950 | 451,244 | 453,568 |
| North Bay | | | | | |
| Benicia | | | 4,681,235 | 5,239,132 | 7,414,686 |
| Larkspur | 103,233 | 83,724 | 57,566 | 56,883 | 66,632 |
| Mill Valley | 95,839 | 96,754 | 87,131 | 87,066 | 93,552 |
| Napa | 442,076 | 388,825 | 398,695 | 361,691 | 387,947 |
| Novato | 499,517 | 358,359 | 293,139 | 281,970 | 322,139 |
| Petaluma | 301,768 | 270,071 | 239,204 | 220,785 | 252,265 |
| Rohnert Park | 5000 ACC | 8097 enn; | 71,162 | 68,944 | 85,756 |
| San Anselmo | 92,255 | 89,636 | 77,356 | 72,958 | 78,045 |
| San Rafael | 575,646 | 538,658 | 484,178 | 454,552 | 512,091 |
| Santa Rosa | 882,493 | 749,054 | 691,086 | 674,827 | 746,264 |
| Ukiah | 124,249 | 127,208 | 126,130 | 126,975 | 144,395 |
| Vallejo | 725,729 | 657,772 | 644,762 | 620,315 | 663,204 |
| Sacramento | | | | | |
| Davis | 246,813 | 189,539 | 159,670 | 145,422 | 154,669 |
| Fairfield | 1,037,248 | 903,367 | 819,964 | 801,896 | 856,771 |
| Sacramento | 3,285,563 | 2,787,686 | 3,059,851 | 2,706,905 | 2,912,056 |

Total Commercial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---|------------|-----------|-----------|-----------|-----------|
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 238,985 | 175,389 | 705,690 | 502,864 | 585,617 |
| Woodland | 355,938 | 316,152 | 293,179 | 353,017 | 397,267 |
| San Francisco | | | | | |
| Daly City | 539,614 | 437,901 | 416,749 | 383,098 | 403,210 |
| Millbrae | 157,696 | 151,606 | 154,359 | 143,782 | 145,488 |
| Pacifica | 105,180 | 155,018 | 136,211 | 109,760 | 117,896 |
| San Bruno | 179,868 | 169,425 | 189,324 | 172,325 | 172,233 |
| San Francisco | 10,245,135 | 9,688,528 | 8,892,647 | 8,370,113 | 8,529,606 |
| South San Francisco | 731,512 | 804,739 | 1,408,751 | 1,417,045 | 1,469,596 |
| San Joaquin | | | | | |
| Atwater | 216,664 | 186,855 | 90,355 | 73,834 | 75,219 |
| Bakersfield | 1,199,707 | 1,072,653 | 1,028,108 | 936,678 | 1,043,138 |
| Clovis | 142,070 | 109,067 | 113,437 | 115,670 | 141,986 |
| Fresno | 2,097,452 | 2,011,785 | 2,050,016 | 1,997,497 | 2,117,089 |
| Los Banos | 116,302 | 100,082 | 274,729 | 253,447 | 202,526 |
| Madera | 185,820 | 167,420 | 172,947 | 163,225 | 179,451 |
| Merced | 419,410 | 406,961 | 419,647 | 403,575 | 468,404 |
| Ridgecrest | 422,395 | 380,119 | 373,167 | 346,455 | 326,948 |
| Sanger | 81,790 | 80,988 | 98,343 | 93,211 | 104,006 |
| San Jose | | | | | |
| Belmont | 183,865 | 166,174 | 148,751 | 130,904 | 144,933 |
| Burlingame | 617,196 | 574,604 | 515,872 | 457,393 | 506,741 |
| Campbell | 344,916 | 307,208 | 271,712 | 268,381 | 301,845 |
| Cupertino | 285,372 | 284,776 | 286,575 | 290,885 | 324,834 |
| Foster City | 61,686 | 51,871 | 47,964 | 46,004 | 57,056 |
| Gilrov | 198,683 | 161,297 | 1,150,644 | 1,139,327 | 1,259,816 |
| Los Altos | 196,901 | 189,237 | 274,729 | 164,135 | 166,542 |
| Los Catos | 308,956 | 306,134 | 265,910 | 270,624 | 288,813 |
| Menlo Park | 378,254 | 358,142 | 329,458 | 300,547 | 327,248 |
| Milnitas | 234,988 | 186,374 | 542,014 | 725,399 | 802,304 |
| TTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTT | - | | | | |

Total Commercial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---|------------|------------|-------------|-------------|-------------|
| Community | | | | | |
| San Jose (cont.) | | | | | |
| Mountain View | 739,153 | 663,867 | 597,911 | 596,972 | 654,421 |
| Morgan Hill | | | 84,071 | 86,675 | 106,819 |
| Redwood City | 616,008 | 560,377 | 493,067 | 464,420 | 508,477 |
| San Carlos | 236,538 | 228,959 | 212,786 | 183,177 | 215,939 |
| San Jose | 4,022,086 | 3,412,825 | 3,515,566 | 3,337,699 | 3,522,211 |
| San Mateo | 777,294 | 753,954 | 693,647 | 638,407 | 662,486 |
| Santa Clara | 1,161,076 | 1,106,621 | 2,228,929 | 2,120,901 | 2,444,283 |
| Santa Cruz | 516,166 | 477,970 | 435,348 | 442,449 | 487,037 |
| Saratoga | 180,132 | 209,400 | 164,561 | 156,964 | 162,538 |
| Sunnyvale | 1,250,392 | 1,064,827 | 1,451,337 | 1,443,682 | 1,559,127 |
| Watsonville | 235,511 | 245,323 | 252,047 | 243,166 | 234,651 |
| Shasta | | | | | |
| Redding | 350,138 | 332,757 | 354,573 | 384,135 | 448,523 |
| Stockton | | | | | |
| Ceres | | | 83,690 | 79,297 | 83,053 |
| Lodi | 364,540 | 316,836 | 421,820 | 350,463 | 367,407 |
| Manteca | 137,086 | 123,654 | 132,451 | 130,638 | 135,239 |
| lodesto | 907,538 | 855,492 | 1,041,505 | 1,127,016 | 1,224,623 |
| Stockton | 1,634,999 | 1,476,281 | 1,440,761 | 1,472,378 | 1,567,339 |
| Tracy | 145,464 | 165,720 | 182,442 | 187,863 | 200,457 |
| Turlock | 267,734 | 246,181 | 312,165 | 264,259 | 256,648 |
| Total Sales in Communities of 10,000 Popula- tion and Over | 59,237,512 | 54,101,832 | 96,526,061 | 92,840,823 | 91,157,216 |
| Sales in Communi- ties of Less Than 10,000 Population | 23,911,228 | 20,499,556 | 67,204,285 | 51,186,262 | 52,463,463 |
| · ····· | 83 148 740 | 74 601 388 | 163 730 346 | 144 027 085 | 143 620 679 |

Source: Annual Reports to the California Public Utilities Commission, 1975-1979, Pacific Gas and Electric Company.

† : See Table C-3

Total Industrial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------|-----------|-----------|-----------|-----------|-----------|
| Community | | | | | |
| Coast Valleys | | | | | |
| Monterey | 283,613 | 252,583 | 199,817 | 240,034 | 238,985 |
| Pacific Grove | 10,922 | 11,474 | 11,289 | 9,786 | 7,889 |
| Salinas | 502,283 | 439,235 | 331,975 | 312,616 | 289,997 |
| Seaside | 3,176 | 3,404 | 3,343 | 3,268 | 3,092 |
| Colgate | | | | | |
| Yuba City | 189,802 | 281,250 | 329,267 | 339,339 | 307,431 |
| De Sabla | | | | | |
| Chico | 170,521 | 152,943 | 129,190 | 121,946 | 128,157 |
| Drum | | | | | |
| Roseville | 334,966 | 309,554 | 205,634 | 195,276 | 191,821 |
| East Bay | | | | | |
| Alameda | 1,227,323 | 1,141,793 | 897,786 | 1,197,405 | 1,185,867 |
| Albany | 250,782 | 193,612 | 157,497 | 156,282 | 157,877 |
| Antioch | 507,920 | 366,695 | 119,464 | 22,261 | 30,286 |
| Berkeley | 2,959,887 | 2,515,519 | 2,147,889 | 2,029,194 | 2,247,840 |
| Concord | 53,452 | 91,639 | 88,273 | 77,876 | 78,809 |
| El Cerrito | 2,503 | 2,451 | 2,078 | 1,615 | 2,028 |
| Fremont | 1,803,336 | 2,324,744 | 1,314,976 | 928,998 | 863,887 |
| Hayward | 2,344,129 | 1,985,569 | 1,616,461 | 1,449,907 | 1,461,922 |
| Lafayette | 0 | 0 | 0 | 0 | 0 |
| Livermore | 103,664 | 146,956 | 132,202 | 164,428 | 172,413 |
| Martinez | 350,425 | 369,055 | 297,225 | 334,232 | 313,533 |
| Moraga | 0 | 0 | 56,292 | 56,796 | 54,759 |
| Newark | 2,239,541 | 2,523,735 | 1,458,071 | 1,616,699 | 1,688,689 |
| Oakland | 8,302,911 | 8,091,686 | 5,316,758 | 6,267,683 | 5,479,824 |

Total Industrial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|------------------|------------|------------------|------------|-----------|------------|
| Community | | | | | |
| East Bay (cont.) | | | | | |
| Piedmont | 0 | 0 | 0 | 0 | 0 |
| Pinhole | 0 | 0 | 0 | 0 | 0 |
| Pitcsburg | 502,934 | 516,524 | 438,090 | 199,880 | 186,080 |
| Pleasant Hill | 0 | 0 | 0 | 0 | 0 |
| Pleasanton | 209,790 | 184,407 | 72,236 | 102,121 | 142,031 |
| Richmond | 34,869,058 | 36,276,240 | 5,092,988† | 1,853,358 | 13,124,142 |
| San Leandro | 3,388,334 | 3,327,276 | 2,510,799 | 2,400,172 | 2,395,318 |
| San Pablo | 67,479 | 69,239 | 68,038 | 76,052 | 96,344 |
| Union City | 2,707,089 | 2,774,215 | 2,780,889 | 2,318,705 | 710,400 |
| Walnut Creek | 94,004 | 82,860 | 74,691 | 78,770 | 82,146 |
| Humboldt | | | | | |
| Arcata | 66,334 | 50,366 | 3,757 | 2,939 | 2,690 |
| Eureka | 222,357 | 214,420 | 144,452 | 120,187 | 46,246 |
| North Bay | | | | | |
| Benicia | 0 | 0 | 1,925,657 | 635,109 | 852,009 |
| Larkspur | 81,448 | 39,638 | 35,112 | 34,704 | 31,855 |
| Mill Valley | 0 | . 0 | - 0 | 0 | 0 |
| Napa | 173,924 | 186,874 | 121,406 | 106,652 | 101,274 |
| Novato | 2,432 | 411 | 0 | 0 | 0 |
| Petaluma | 268,293 | 236,959 | 230,722 | 228,055 | 251,499 |
| Rohnert Park | 0 | 0 | 0 | 0 | 0 |
| San Anselmo | 0 | 0 | 0 | 0 | 0 |
| San Rafael | 110,663 | 110,345 | 94,162 | 83,416 | 89,963 |
| Santa Rosa | 217,172 | 224,966 | 189,840 | 196,385 | 191,257 |
| Ukiah | 9,796 | 5,115 | 4,406 | 3,546 | 3,437 |
| Vallejo | 1,826,736 | 1,522,681 | 1,100,298 | 1,082,943 | 1,018,330 |
| Sacramento | | | | | |
| Davis | 625,628 | 439,409 | 437,489 | 358,297 | 657,238 |
| Fairfield | 171,240 | 274,710 | 777,160 | 694,465 | 883,408 |
| Sacramento | 4,027,022 | 3,670,964 191 | 2,809,305 | 2,950,086 | 3,096,270 |

Total Industrial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|---------------------|-----------|-----------|-----------|-----------|-----------|
| Community | | | | | |
| Sacramento (cont.) | | | | | |
| Vacaville | 839,748 | 566,637 | 186,910 | 157,128 | 169,052 |
| Woodland | 805,996 | 563,202 | 474,448 | 343,707 | 449,689 |
| San Francisco | | | | | |
| Daly City | 78,495 | 66,983 | 66,061 | 61,335 | 66,644 |
| Millbrae | 19,343 | 14,833 | 0 | 0 | 0 |
| Pacifica | 21,135 | 8,015 | 3,931 | 3,887 | 4,048 |
| San Bruno | 62,670 | 54,938 | 26,869 | 23,716 | 34,772 |
| San Francisco | 7,758,646 | 6,757,394 | 5,652,579 | 5,320,968 | 5,082,198 |
| South San Francisco | 2,073,819 | 1,808,766 | 1,113,011 | 809,816 | 871,668 |
| San Joaquin | | | | | |
| Atwater | 295,309 | 209,798 | 228,919 | 213,847 | 181,782 |
| Bakersfield | 401,403 | 330,780 | 265,715 | 241,521 | 271,742 |
| Clovis | 3,197 | 3,659 | 3,963 | 4,338 | 4,491 |
| Fresno | 1,513,308 | 1,599,443 | 1,224,541 | 1,153,839 | 1,189,746 |
| Los Banos | 219,361 | 249,069 | 74,326 | 44,159 | 88,713 |
| Madera | 19,014 | 32,287 | 12,685 | 12,157 | 18,207 |
| Merced | 619,527 | 371,231 | 469,406 | 323,739 | 426,801 |
| Ridgecrest | 134,535 | 137,344 | 94,324 | 28,077 | 111,125 |
| Sanger | 149,538 | 143,719 | 117,402 | 93,750 | 99,836 |
| San Jose | | | | | |
| Belmont | 21,005 | 16,846 | 14,655 | 9,973 | 0 |
| Burlingame | 98,219 | 94,219 | 86,103 | 84,782 | 79,356 |
| Campbell | 17,969 | 27,200 | 23,489 | 19,623 | 18,482 |
| Cupertino | 137,771 | 134,368 | 148,047 | 186,981 | 206,362 |
| Foster City | 0 | 0 | 0 | 0 | 0 |
| Gilroy | 1,791,294 | 1,273,942 | 378,926 | 380,756 | 491,703 |
| Los Altos | 2,170 | 1,943 | 1,693++ | 2,034 | 1,863 |
| Los Gatos | 69,972 | 73,160 | 75,284 | 70,759 | 64,532 |
| Menlo Park | 507,473 | 466,615 | 353,493 | 356,710 | 320,282 |
| Milpitas | 683,889 | 696,742 | 423,288 | 208,114 | 227,593 |

Total Industrial Gas Sales (MCF) - Communities of 10,000 Population or More - Pacific Gas and Electric Company (Continued)

| Year | 1975 | 1976 | 1977 | 1978 | 1979 |
|--|-------------|-------------|-------------|-------------|-------------|
| Community | | | | | |
| an Jose (cont.) | | | | | |
| lountain View | 696,060 | 631,795 | 588,324 | 557,619 | 528,715 |
| lorgan Hill | 0 | 0 | 18,864 | 7,662 | 11,411 |
| edwood City | 603,672 | 611,492 | 589,731 | 563,498 | 571,178 |
| an Carlos | 178,472 | 174,206 | 154,435 | 145,330 | 147,602 |
| an Jose | 6,703,139 | 6,232,538 | 6,121,920 | 5,869,984 | 6,387,930 |
| an Mateo | 271,349 | 261,626 | 192,935 | 167,718 | 171,579 |
| anta Clara | 3,584,935 | 3,874,429 | 2,798,244 | 2,968,605 | 3,073,899 |
| anta Cruz | 563,872 | 507,352 | 448,222 | 396,416 | 370,601 |
| aratoga | 37,919 | 35,363 | 36,996 | 33,155 | 35,279 |
| unnyvale | 2,342,758 | 1,966,390 | 1,447,797 | 1,435,468 | 1,465,715 |
| atsonville | 397,372 | 390,274 | 424,033 | 382,181 | 380,888 |
| hasta | | | | | |
| edding | 112,027 | 95,164 | 65,819 | 99,069 | 107,156 |
| tockton | | | | | |
| eres | 0 | 0 | 11,131 | 7,082 | 9,374 |
| odi | 762,854 | 592,335 | 577,101 | 523,856 | 665,763 |
| anteca | 15,166 | 20,580 | 21,746 | 32,186 | 38,478 |
| odesto | 1,635,502 | 1,502,391 | 1,490,099 | 1,340,666 | 1,468,287 |
| tockton | 4,679,168 | 4,426,444 | 3,508,288 | 2,490,341 | 3,530,308 |
| racy | 1,027,751 | 994,198 | 1,025,229 | 1,083,829 | 1,143,918 |
| urlock | 443,253 | 281,120 | 207,623 | 306,113 | 346,223 |
| otal Sales in Communities of 10,000 Popula- tion and Over | 113,702,068 | 109,740,587 | 65,048,222 | 57,617,977 | 69,813,186 |
| ales in Communi- ties of Less Than 10,000 Population | 202,097,356 | 74,520,540 | 97,480,930 | 81,357,214 | 116,351,750 |
| at a 1 | 315,799,424 | 284,261,131 | 162,529,152 | 138,975,191 | 186,164,93 |

++ : See Table C-2

Population, Acreage, Distribution Plant, and Main Mileage Pacific Gas and Electric Company

| · · · · · · · · · · · · · · · · · · · | | Total | Plant in Se | ervice (\$) | Miles of | Miles of Distribution | | |
|---------------------------------------|------------|---------|-------------------------|---------------------------------------|-------------------|-----------------------|--|--|
| Community | Population | Acreage | End of 1979 | 9 End of 1978 | End of 19 | 79 End of 19 | | |
| Coast Valleys | | | | · · · · · · · · · · · · · · · · · · · | | | | |
| Monterey | 26,302 | 5,056 | 4,175,024 | 4.020.736 | 104.05 | 104.10 | | |
| Pacific Grove | 13,505 | 1,728 | 2,202,007 | 2,200,086 | 55.38 | 55.38 | | |
| Salinas | 58,896 | 8,512 | 7,965,727 | 7,453,037 | 203.54 | 200.26 | | |
| Seaside | 35,935 | 5,760 | 2,460,646 | 2,367,341 | 71.26 | 71.26 | | |
| Colgate | | | | | | | | |
| Yuba City | 13,986 | 2,240 | 2,731,675 | 2,548,083 | 58.90 | 55.83 | | |
| De Sabla | | | | | | | | |
| Chico | 19,580 | 7,040 | 4,510,462 | 4,286,338 | 88.96 | 86.47 | | |
| Drum | | | | | | | | |
| Roseville | 17,895 | 17,856 | 4,202,708 | 3,658,965 | 95.65 | 89.72 | | |
| East Bay | | | | | | | | |
| Alameda | 70,968 | 6,400 | 5,671,343 | 5,218,047 | 141.34 | 140.88 | | |
| Albany | 14,674 | 1,088 | 1,423,307 | 1,372,221 | 36.63 | 36.64 | | |
| Antioch | 28,060 | 4,736 | 4,896,329 | 4,485,517 | 123.90 | 11/ 69 | | |
| Berkeley | 116,716 | 6,784 | 11,008,896 | 10,720,852 | 289.30 | 291.09 | | |
| Concord | 85,164 | 16,512 | 12.666.516 | 11 755 759 | 305 75 | 206 92 | | |
| El Cerrito | 25,190 | 2,944 | 3,089,513 | 2,934,513 | 75 01 | 75 92 | | |
| Fremont | 100,869 | 53,952 | 15.957.133 | 14 727 645 | 345 01 | 225 20 | | |
| Hayward | 93,058 | 24,256 | 10,891,569 | 10,074,061 | 071 70 | 323.89 | | |
| Lafayette | 20,484 | 7,936 | 4,526,570 | 4 307 325 | 2/1./0 | 264.60 | | |
| Livermore | 37,703 | 7,616 | 6 670 122 | 6 518 475 | 107.49 | 103.64 | | |
| Martinez | 16,506 | 4 544 | 3 / 69 922 | 0,510,475 | 100.22 | 159.98 | | |
| Moraga | 14,205 | 4.864 | 2,400,022 | 3,165,176 | 84.1/ | 77.37 | | |
| Newark | 27.153 | 5 276 | 4,000,700 | 4,404,939 | 58.61 | 58.01 | | |
| Oakland | 361,561 | 34,176 | 4,293,728 39,045,685 | 4,008,813 | 80.61 1,010.07 | 78.13 1,012.78 | | |

Population, Acreage, Distribution Plant, and Main Mileage Pacific Gas and Electric Company (Continued)

| | | Total | Plant in Se | ervice (\$) | Miles of 1 | Distribution Mai |
|------------------|---|---------|---|--|--|--|
| Community | Population | Acreage | End of 1979 | 9 End of 1978 | End of 19 | 75 End of 1978 |
| East Bay (cont.) | 2000)(()) (1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(| | 98-96-97-97-98-98-98-98-98-98-98-98-98-98-98-98-98- | an ta ang tang ang tang tang tang tang t | and a second | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |
| Piedmont | 10,917 | 1,280 | 1,761,128 | 1,721,702 | 47.01 | 47,26 |
| Pinhole | 15,850 | 2,240 | 1,882,259 | 1,811,980 | 41.50 | 41.30 |
| Pittsburg | 20,651 | 5,376 | 4,470,213 | 4,034,047 | 102.99 | 96.57 |
| Pleasant Hill | 24,610 | 3,712 | 3,989,000 | 3,724,217 | 94.59 | 91.27 |
| Pleasanton | 18,328 | 8,128 | 4,531,821 | 4,294,458 | 115.39 | 114.38 |
| Richmond | 79,043 | 20,544 | 9,747,754 | 9,304,423 | 235.25 | 233.02 |
| San Leandro | 68,698 | 8,128 | 8,330,034 | 7,938,850 | 199.57 | 197.22 |
| San Pablo | 21,461 | 1,600 | 1,872,542 | 1,797,660 | 42.94 | 42.91 |
| Union City | 14,724 | 9,408 | 4,430,570 | 4,233,963 | 97.93 | 95.13 |
| Walnut Creek | 39,844 | 9,408 | 7,337,723 | 6,735,424 | 148.32 | 140.31 |
| Humboldt | | | | | | |
| Arcata | 8,985 | 4,416 | 1,946,950 | 1,847,531 | 46.31 | 45.41 |
| Eureka | 24,337 | 5,248 | 4,851,487 | 4,659,587 | 120.03 | 120.27 |
| North Bay | | | | | | |
| Benicia | 8,783 | 1,984 | 2,459,054 | 2,217,344 | 56.08 | 54.18 |
| Larkspur | 10,487 | 2,112 | 1,450,364 | 1,376,825 | 36.12 | 35.00 |
| Mill Valley | 12,942 | 3,008 | 2,543,235 | 2,452,401 | 58.46 | 58.36 |
| Napa | 35,978 | 8,384 | 7,441,942 | 7,031,282 | 187.95 | 183.93 |
| Novato | 31,006 | 13,120 | 5,259,909 | 4,603,482 | 123.63 | 118.48 |
| Petaluma | 24,870 | 4,800 | 4,537,520 | 4,220,877 | 108.45 | 102.39 |
| Rohnert Park | 6,133 | 3,584 | 2,017,777 | 1,737,814 | 51.59 | 44.89 |
| San Anselmo | 13,031 | 1,728 | 1,871,746 | 1,803,798 | 46.45 | 46.55 |
| San Rafael | 38,977 | 9,152 | 6,727,702 | 6,454,637 | 173.64 | 173.45 |
| Santa Rosa | 50,006 | 12,736 | 12,675,490 | 11,646,974 | 300.09 | 279.77 |
| Ukiah | 10,095 | 2,432 | 1,850,657 | 1,747,789 | 42.49 | 39.83 |
| Vallejo | 66,733 | 9,728 | 9,392,154 | 8,465,803 | 228.85 | 217.46 |
| Sacramento | | | | | | |
| Davis | 23,488 | 3,648 | 4,025,947 | 3,726,489 | 96.04 | 94.07 |
| Fairfield | 44,146 | 9,856 | 6,866,836 | 6,172,392 | 154.00 | 144.18 |
| Sacramento | 254,413 | 60,032 | 42,988,841 195 | 39,249,469 | 969.70 | 935.07 |

Population, Acreage, Distribution Plant, and Main Mileage Pacific Gas and Electric Company (Continued)

| Community. | D. 1.4. | Total | Plant in Se | rvice (\$) | Miles of | Distribution M |
|--------------------|------------|---------|------------------|------------|----------|----------------|
| Community | Population | Acreage | End OI 1973 | | | |
| Sacramento (cont.) |) | | | | | |
| Vacaville | 21,690 | 5,824 | 5,831,640 | 5,256,693 | 134.73 | 126.35 |
| Woodland | 20,677 | 3,200 | 4,109,167 | 3,926,003 | 99.31 | 98.48 |
| San Francisco | | | | | | |
| Daly City | 66,922 | 4,416 | 6,038,305 | 5,775,647 | 141.39 | 141.14 |
| Millbrae | 20,781 | 2,112 | 2,182,510 | 2,067,347 | 53.61 | 53.61 |
| Pacifica | 36,020 | 8,064 | 3,690,571 | 3,553,121 | 95.29 | 95.26 |
| San Bruno | 36,254 | 3,584 | 3,916,183 | 3,768,717 | 98.19 | 98.16 |
| San Francisco | 715,674 | 29,056 | 47,871,683 | 46,203,472 | 1,194.53 | 1,194.00 |
| South San Francis | co 46,646 | 6,080 | 5,078,464 | 4,841,329 | 120.57 | 120.01 |
| San Joaquin | | | | | | |
| Atwater | 11,640 | 2,048 | 2,192,541 | 1,983,689 | 46.81 | 42.93 |
| Bakersfield | 69,515 | 16,576 | 14,185,068 | 12,645,350 | 331.09 | 313.56 |
| Clovis | 13,856 | 2,240 | 4,513,111 | 3,602,693 | 88.73 | 77.55 |
| Fresno | 165,972 | 26,752 | 32,377,960 | 29,261,120 | 787.42 | 762.62 |
| Los Banos | 9,188 | 3,456 | 1,721,398 | 1,580,220 | 41.32 | 39.27 |
| Madera | 16,044 | 4,160 | 3,269,199 | 2,999,765 | 77.47 | 72.87 |
| Merced | 22,670 | 4,800 | 5,109,513 | 4,802,144 | 123.25 | 120.87 |
| Ridgecrest | 7,629 | 4,992 | 2,457,771 | 2,304,763 | 63.87 | 59.76 |
| Sanger | 10,008 | 1,664 | 1,770,763 | 1,580,025 | 38.86 | 37.99 |
| San Jose | | | | | | |
| Belmont | 23,667 | 2,944 | 2,855,766 | 2,746,744 | 68.68 | 67.85 |
| Burlingame | 27,320 | 2,944 | 3,134,299 | 3,020,651 | 79.72 | 79.62 |
| Campbell | 24,770 | 2,176 | 3,064,268 | 2,911,702 | 73.18 | 72.13 |
| Cupertino | 18,216 | 4,864 | 4,451,365 | 3,142,260 | 77.37 | 75.11 |
| Foster City | 9,327 | 2,368 | 2,313,421 | 2,243,814 | 57.08 | 56.08 |
| Gilroy | 12,665 | 3,136 | 3,004,437 | 2,732,420 | 62.12 | 56.93 |
| Los Altos | 24,956 | 3,648 | 4,637,585 | 4,449,265 | 112.03 | 111.47 |
| Los Gatos | 23,735 | 5,632 | 4,478,477 | 3,909,206 | 101.19 | 99.82 |
| Menlo Park | 26,734 | 7,744 | 3,788,853 | 3,523,198 | 74.91 | 74.84 |
| Milpitas | 27,149 | 5,952 | 3,618,948 196 | 3,241,235 | 87.08 | 83.42 |

| Community Populatio | | Total Acreage | Plant in Ser End of 1979 | vice (\$) End of 1978 | Miles of Distribution Mar End of 1979 End of 1978 | | |
|---------------------|---------|------------------|-----------------------------|--------------------------|--|----------|--|
| San Jose (cont.) | | | | | | | |
| Mountain View | 51,092 | 6,976 | 2,942,741 | 2,764,623 | 160.87 | 157.92 | |
| Morgan Hill | 6,485 | 5,120 | 6,283,321 | 5,943,479 | 64.50 | 62.95 | |
| Redwood City | 55,686 | 13,120 | 6,118,190 | 5,867,521 | 147.83 | 145.84 | |
| San Carlos | 25,924 | 2,944 | 3,685,398 | 3,522,431 | 84.10 | 83.54 | |
| San Jose | 445,779 | 87,168 | 67,035,089 | 64,021,151 | 1,677.72 | 1,642.77 | |
| San Mateo | 78.991 | 7,232 | 9,023,296 | 8,676,796 | 223.99 | 223.79 | |
| Santa Clara | 87,717 | 10,560 | 10,105,598 | 9,529,975 | 250.71 | 248.02 | |
| Santa Cruz | 32,076 | 7,808 | 5,762,667 | 5,484,102 | 160.14 | 158.15 | |
| Saratoga | 27,110 | 7,808 | 5,733,416 | 5,497,325 | 133.59 | 132.59 | |
| Sunnyvale | 95,408 | 13,696 | 11,031,126 | 10,546,577 | 275.51 | 273.58 | |
| Watsonville | 14,569 | 2,624 | 2,669,546 | 2,505,061 | 64.58 | 63.14 | |
| Shasta | | | | | | | |
| Redding | 16,659 | 9,728 | 6,964,861 | 6,600,799 | 171,59 | 166.58 | |
| Stockton | | | | | • | | |
| Ceres | 6,029 | 1,920 | 2,401,358 | 2,033,588 | 50.49 | 43.39 | |
| Lodi | 28,691 | 4,544 | 5,461,647 | 5,203,469 | 141.05 | 135.05 | |
| Manteca | 13,845 | 1,920 | 3,301,411 | 2,985,790 | 80.06 | 74.14 | |
| Modesta | 61,712 | 6,080 | 16,265,111 | 15,222,248 | 370.49 | 355.43 | |
| Stockton | 107,644 | 19,136 | 19,271,810 | 18,007,242 | 478.80 | 458.38 | |
| Tracy | 14,724 | 3,712 | 2,626,873 | 2,413,790 | 65.84 | 63.58 | |
| Turlock | 13,992 | 2,944 | 4,315,208 | 4,009,390 | 95.61 | 90.25 | |

Population, Acreage, Distribution Plant, and Main Mileage Pacific Gas and Electric Company (Continued)

Source: Pacific Gas and Electric Company.

Table C-10

Average Degree-Days for the Period 1941-1970 - Meteorological Stations in the Pacific Gas and Electric Company's Service Area

| | | | | | | oompan | IJ 0 001. | 100 140 | | | | | | | |
|-------------|------|-------|------|----------|------|--------|-------------|----------|----------|------|-------|------|----------|---------|--------|
| Meter. | Jan. | Feb. | Mar. | Apr. | May. | June | July | Aug. | Sept. | Oct. | Nov. | Dec. | Total | Peak | Load |
| Station | · | | | | | C | OAST VAL | LEYS DT | VISTON | | | | Allituar | rioiren | Factor |
| Salinas | 465 | 364 | 372 | 296 | 214 | 139 | 102 | 96 | 72 | 136 | 275 | 42.8 | 2959 | 465 | 0.53 |
| Santa Maria | 450 | 364 | 378 | 303 | 245 | 167 | 112 | 102 | 94 | 159 | 270 | 409 | 3053 | 450 | 0.56 |
| Average | 457 | 364 | 375 | 300 | 229 | 153 | 107 | 99 | 83 | 148 | 272 | 419 | 3006 | 457 | 0.55 |
| | | | | | | | COLGAT | E DIVIS | LON | | | | | | 0.33 |
| Marysville | 605 | 406 | 335 | 186 | 58 | 9 | 0 | 0 | 0 | 76 | 333 | 577 | 2585 | 605 | 0.35 |
| | | | | | | | DE SAB | LA DIVI | SION | | | | | | |
| Red Bluff | 614 | 420 | 366 | 218 | 64 | 8 | 0 | 0 | 0 | 82 | 339 | 577 | 2688 | 614 | 0.36 |
| Marysville | 605 | 406 | 335 | 186 | 58 | 9 | 0 | 0 | 0 | 76 | 333 | 577 | 2585 | 605 | 0.35 |
| Average | 609 | 413 | 351 | 202 | 61 | 8 | 0 | 0 | 0 | 79 | 336 | 577 | 2636 | 609 | 0.36 |
| | | | | | | | DRUM | DIVISĮ | ON | | | | | | |
| Sacramento | 617 | 426 | 372 | 227 | 120 | 20 | 0 | 0 | . 5 | 101 | 360 | 595 | 2843 | 617 | 0.38 |
| | | | | | | | EAST B. | AY DIVI | SION | | | | | | |
| Oakland | 508 | 367 | 350 | 270 | 193 | 114 | 80 | 74 | 59 | 135 | 291 | 468 | 2909 | 508 | 0.48 |
| | | | | | | | HUMBOL | ΟΤ ΟΙΥΙ | SION | | | | | | |
| Eureka | 549 | 465 | 518 | 459 | 388 | 294 | 270 | 248 | 252 | 329 | 399 | 508 | 4679 | 549 | 0.71 |
| | | | | | | | NORTH B. | AY DIVI | SION | | | | | | |
| Santa Rosa | 586 | 420 | 406 | 289 | 171 | 78 | 20 | 22 | 33 | 134 | 354 | 552 | 3065 | 586 | 0.43 |
| Ukiah | 589 | 426 | 412 | 285 | 141 | 47 | 0 | 7 | 12 | 131 | 369 | 558 | 2977 | 589 | 0.42 |
| Average | 587 | 423 | 409 | 287 | 156 | 63 | 10 | 14 | 23 | 132 | 362 | 555 | 3021 | 587 | 0.43 |
| | | | | | | | SACRAME | NTO DIV | ISION | | | | | | |
| Sacramento | 617 | 426 | 372 | 227 | 120 | 20 | 0 | 0 | 5 | 101 | 360 | 595 | 2843 | 617 | 0.38 |
| | | | | | | S. | AN FRANC | ISCO DI | VISION | | | | | | |
| S/F City | 437 | 325 | 332 | 291 | 257 | 194 | 202 | 177 | 102 | 127 | 233 | 403 | 3080 | 437 | 0.59 |
| S/F Airport | 518 | 386 | 372 | 291 | 210 | 120 | 93 | 84 | 66 | | 291 | 4/4 | 3042 | 518 | 0.49 |
| Average | 4// | 356 | 352 | 291 | 233 | 157 | 148 | 130 | 84 | 132 | 262 | 439 | 3061 | 477 | 0.53 |
| | (1) | (0.0 | 211 | 100 | | | SAN JOAQI | JIN DIV | ISTON | | 015 | FOF | 0(50 | (11 | 0.26 |
| Fresno | 611 | 423 | 344 | 182 | 51 | 9 | 0 | 0 | 0 | 90 | 345 | 595 | 2650 | 611 | 0.36 |
| Bakersfield | | 353 | 266 | 140 | | | 0 | 0 | 0 | | 276 | 530 | 2185 | | 0.33 |
| Average | 577 | 388 | 305 | 161 | 36 | 5 | 0 | 0 | 0 | 12 | 311 | 202 | 2417 | 5// | 0.35 |
| | (01 | | | | 100 | 50 | SAN JU | SE DIVI | SION | | 0.7.6 | 150 | 0/16 | / 01 | 0. / 0 |
| San Jose | 481 | 350 | 322 | 228 | 123 | 50 | 12 | 15 | 13 | 90 | 276 | 426 | 2410 | 481 | 0.42 |
| Ded Dluff | 611 | 120 | 267 | 210 | 61 | 0 | SHAST | A DIVIS | | 0.2 | 220 | 577 | 2600 | 61/ | 0.26 |
| Ked biuli | 014 | 420 | 300 | 218 | 64 | 8 | U STOCKT | U | STON | 02 | 222 | J// | 2000 | 014 | 0.30 |
| Ctoolttop | 622 | 1.1.5 | 201 | 214 | 67 | 15 | STUCKI | | DIUN | 00 | 262 | 601 | 2806 | 622 | 0.37 |
| SLOCKEON | 032 | 443 | 701 | <u> </u> | 0/ | 10 | <u> </u> | <u> </u> | <u> </u> | 00 | 202 | 001 | 2000 | 0.52 | 0.37 |

Source: Monthly Normals of Temperature, Precipitation, and Heating and Cooling Degree-Days 1941-1970 Climatography of the United States No. 81-California. U.S. Department of Commerce, National Oceanic and Atmospheric Administration, Environmental Data Service.

APPENDIX D

NATIONAL FUEL GAS DISTRIBUTION CORPORATION DATA

The purpose of this appendix is to present the community-level data used in the econometric analysis of the National Fuel Gas Distribution Corporation, as reported in chapter 3. These data correspond to the year 1979 and cover 173 communities. The residential, commercial, industrial, and public authorities gas sales and numbers of customers are presented in table D-1. The distribution plant in service, disaggregated into various categories, the total plant in service, and the population and acreage are indicated in table D-2.

| , | Commund the | | Gas S | ales (MCF) | Average Number of Customers | | | | | | |
|-----|-------------------|-------------|------------|------------|-----------------------------|------------|-------------|------------|------------|-----------------------|------------|
| | · | Residential | Commercial | Industrial | Public Authoritie | Total s | Residential | Commercial | Industrial | Public Authorities | Total 3 |
| 1. | Alfred, T. | 43995 | 12337 | 5028 | 4591 | 65951 | 287 | 27 | 8 | 5 | 327 |
| 2. | Alfred, V. | 63749 | 113694 | 0 | 169089 | 346532 | 381 | 44 | 0 | 20 | 445 |
| 3. | Alma, T. | 30309 | 747 | 7314 | 3820 | 42190 | 242 | 7 | • 13 | 7 | 269 |
| 4. | Almond T. | 27511 | 780 | 0 | 21399 | 49690 | 218 | 8 | 0 | 4 | 230 |
| 5. | Almond, V. | 30932 | 4443 | 0 | 386 | 35761 | 200 | 12 | 0 | 2 | 214 |
| 6. | Amity, T. | 17863 | 11422 | 0 | 380 | 29665 | 151 | 6 | 0 | 2 | 159 |
| 7. | Andover, T. | 90410 | 6947 | 954 | 12902 | 111213 | 45 | 33 | 1 | 9 | 637 |
| 8. | Angelica, T. | 5684 | 639 | 0 | 0 | 6323 | 327 | 5 | 0 | 0 | 50 |
| 9. | Angellica, V. | 47658 | 8258 | 0 | 6882 | 62798 | 324 | 23 | 0 | 8 | 358 |
| 10. | Belfast, T. | 49353 | 6485 | 0 | 8579 | 64417 | 381 | 31 | 0 | 7 | 362 |
| 11. | Belmont, V. | 61250 | 11531 | 0 | 21203 | 93984 | 803 | 45 | 0 | 16 | 442 |
| 12. | Bolivar, T. | 124874 | 16215 | 4815 | 9744 | 155648 | 256 | 62 | 9 | 11 | 885 |
| 13. | Caneadea, T. | 54720 | 33577 | 0 | 2960 | 91257 | 51 | 12 | 0 | 5 | 273 |
| 14. | Centerville, T. | 7307 | 440 | 0 | 113 | 8860 | 227 | 3 | 0 | 2 | 56 |
| 15. | Cura, T. | 33292 | 4193 | 192814 | 228 | 230527 | 623 | 6 | 2 | 2 | 237 |
| 16. | Cuba, V. | 103752 | 16760 | 57269 | 29803 | 207584 | 190 | 53 | 7 | 10 | 693 |
| 17. | Friendship, T. | 25985 | 12364 | 0 | 147 | 38496 | 386 | 5 | 0 | 1 | 196 |
| 18. | Friendship, V. | 67478 | 6323 | 50739 | 14834 | 139374 | 261 | 23 | 1 | 15 | 425 |
| 19. | Genesee, T. | 32749 | 4778 | 1967 | 815 | 40309 | 276 | 11 | 4 | 6 | 282 |
| 20. | Independence, T. | 34998 | 2553 | 2305 | 5372 | 45228 | 532 | 17 | 3 | 7 | 303 |
| 21. | Scio, T. | 77930 | 4738 | 5528 | 8410 | 96606 | 982 | 26 | 6 | 5 | 569 |
| 22. | Wellsville, T. | 134871 | 33822 | 154279 | 16632 | 339604 | 1913 | 60 | 6 | 6 | 1054 |
| 23. | Wellsville, V. | 321432 | 79066 | 25824 | 44886 | 471208 | 427 | 180 | 7 | 18 | 2118 |
| 24. | Willing, T. | 56095 | 1920 | · 1 | 681 | 58697 | 241 | 14 | 2 | - 3 | 446 |
| 25. | Wirt, T. | 38420 | 464 | 300 | 7427 | 46611 | 0 | 3 | 21 | 8 | 273 |
| 26. | Carrollton, T. | 0 | 430 | 0 | 0 | 430 | 4 | 2 | 0 | 0 | 2 |
| 27. | Dayton, T. | 486 | 0 | 0 | 0 | 486 | 405 | 0 | 0 | 0 | 4 |
| 28. | Delevan, V. | 60852 | 6199 | 0 | 14458 | 81509 | 141 | 20 | 0 | 9 | 434 |
| 29. | East Otto, T. | 20010 | 1255 | . 0 | 1426 | 22691 | 642 | 5 | 0 | 4 | 150 |
| 30. | Ellicottville, R. | 95368 | 24326 | 31808 | 11022 | 162524 | 13 | 53 | 4 | 8 | , 707 |
| 31. | Farmersville, T. | 1842 | 1392 | 20 | 1095 | 4349 | 211 | 2 | 0 | 1 | 16 |
| 22 | Franksville, T. | 30711 | 966 | 95 | 52 | 31824 | 160 | 5 | 1 | 1 | · 218 |

Table D-1: Gas Sales and Average Numbers of Customers - 1979 - National Fuel Gas Distribution Corporation

| C | mmun i tv | | Gas S | ales (MCF) | | | Average Number of Customers | | | | |
|-------------|-------------------|-------------|------------|------------|-----------------------|---------|-----------------------------|------------|------------|------------------------------|-------|
| | | Residential | Commercial | Industrial | Public Authorities | Total | Residential | Commercial | Industrial | Public Authorities | Total |
| 22 | Freedon T | 22310 | 325 | 0 | 481 | 23116 | 160 | 1 | -20 | 1 | 142 |
| 34 | Covanda V | 120373 | 42091 | 19906 | 11623 | 193993 | 804 | 78 | 4 | 8 | 894 |
| 35 | Great Valley, T. | 54672 | 3535 | 23557 | 9263 | 91027 | 420 | 13 | 3 | 4 | 440 |
| 36 | Little Valley, T. | 92208 | 12715 | 20796 | 22060 | 147779 | 593 | 37, | 4 | 24 | 658 |
| 37. | Machias T. | 72181 | 7756 | . 0 | 14048 | 93985 | 643 | 22 | 0 | 6 | 671 |
| 38 | Mansfield, T. | 3061 | 0 | Ō | 848 | 3909 | 18 | 0 | 0 | 1 | 19 |
| 39 | New Albion, T. | 92360 | 9937 | 35793 | 14152 | 152242 | 588 | 36 | 3 | 12 | 639 |
| 40. | Olean, C. | 266434 | 82588 | 209372 | 14590 | 572984 | 1653 | 74 | 2 | 15 | 1744 |
| 41 | Olean, T. | 38325 | 16829 | 24505 | 7157 | 86816 | 263 | 27 | 1 | 1 | 292 |
| 42. | Otto, T. | 21134 | 2326 | 2562 | 1013 | 27035 | 126 | 9 | 2 | 2 | 139 |
| 43. | Perrysburg, V. | 21486 | 797 | 0 | 51291 | 73574 | 143 | 4 . | 0 | 4 | 151 |
| 44. | Perrysburg, T. | 42441 | 1308 | 0 | 14380 | 58129 | 288 | 6 | 0 | 2 | 296 |
| 45. | Persia, T. | 11125 | 777 | 0 | 968 | 12870 | 83 | 3 | 0 | 2 | 88 |
| 46. | Portville, T. | 91457 | 37172 | 0 | 22894 | 151523 | 622 | 43 | 0 | 3 | 668 |
| .47 | Portville V. | 63364 | 16247 | 55672 | 3610 | 138893 | 369 | 35 | 1 | 9 | 414 |
| 48. | Salam Anca. C. | 397434 | 88971 | 38416 | 46761 | 571582 | 2412 | 145 | 6 | 15 | 2578 |
| 49 | Salam Anca, T. | 27025 | 8 | 0 | 6430 | 33463 | 192 | 0 | 0 | 2 | 194 |
| 50. | Yorkshire, T. | 53674 | 12910 | 265 | 290 | 67139 | 418 | 28 | 1 | 3 | 450 |
| 51. | Dunkirk, T. | 83583 | 27750 | 310583 | 28307 | 450223 | 511 | 49 | . 5 | 5 | 570 |
| 52. | Arkwright, T. | 8980 | 92 | 0 | 755 | 9827 | 64 | 1 | 0 | 2 | 67 |
| 53. | Brocton, V. | 80817 | 17424 | 42061 | 25631 | 165933 | 512 | 32 | 1 | 5 | 550 |
| 54. | Cassadaga, V. | 56627 | 8750 | 1511 | 3747 | 70635 | 361 | 26 | 1 | 5 | 393 |
| 55. | Chautaliquat, T. | 88766 | 9534 | 0 | 20376 | 118676 | 764 | 26 | 0 | 1 | 791 |
| 56. | Dunkirk, C. | 881268 | 193756 | 1710336 | 87354 | 2872714 | 5454 | 342 | 16 | 39 | 5851 |
| 57. | Forestville, V. | 48445 | 5654 | 953 | 14760 | 69812 | 305 | 22 | 2 | 4 | 333 |
| 58. | Fredonia, V. | 473722 | 80751 | 156423 | 272723 | 983619 | 2905 | 163 | 2 | 16 | 3086 |
| 59. | Hanover, T. | 156842 | 36219 | -3600 | 19023 | 208484 | 1263 | 54 | 0 | 9 | 1326 |
| 60. | Mayville, V. | 96243 | 24811 | 9623 | 35849 | 166526 | 583 | 52 | 1 | 11 | 647 |
| 61. | Pomfret, T. | 139309 | 44496 | 572 | 34637 | 219014 | 939 | 51 | . 1 | / | 998 |
| 62. | Portland, T. | 87544 | 3981 | 0 | 1332 | 92857 | 690 | 12 | 0 | 4 | 700 |
| 53. | Ripley, T. | 86257 | 14434 | 11495 | 10166 | 122352 | 522 | 26 | 2 | 5 | 222 |
| β4 . | Sheridan, T. | 127164 | 21948 | 0 | 1697 | 150809 | 794 | 49 | 0 | 4 | 847 |
| 65. | Sherman, T. | 46970 | 10286 | 0 | 10954 | 68210 | 250 | 29 | 0 | 7 | 280 |
| \$6. | Silvercreek, V. | 182368 | 49520 | 23246 | 7438 | 262572 | 1150 | 109 | 2 | 8 | 1209 |
| \$7. | Stockton, T. | 23280 | 622 | 0 | 3183 | 27085 | 145 | 6 | 0 | 5 | 120 |
| 68 . | Villenova, T. | 194 | 0 | 0 | 0 | 194 | 3 | 0 | 0 | 0 | 303 |
| 69. | Westfield, T. | 35849 | 9441 | 676 | 0 | 45966 | 281 | 21 | | | |

Table D-1: Gas Sales and Average Numbers of Customers - 1979 - National Fuel Gas Distribution Corporation (Continued)

| Decidential Commencial Teductulal Dublia Total Decidential Commencial Tedu | strial Publ | ic Total |
|--|-------------|----------|
| Authorities | AULIDI | ities |
| 70. Westfield, V. 220369 45465 191087 17009 473930 1278 101 | 6 12 | 1397 |
| 71. Akron, V. 165129 26326 23119 9216 223790 1099 70 | 5 14 | 1188 |
| 72. Alden, T. 254511 17162 2974 11655 286302 1630 34 | -1 11 | 1674 |
| 73. Alden, V. 114602 15283 2181 1835 133901 751 40 | 1 2 | 794 |
| 74. Amherst, T. 4938904 903203 15334 267305 6124746 31132 974 | 7 64 | 32177 |
| 75. Angola, V. 127335 13178 35133 4015 179661 814 47 | 1 9 | 871 |
| 76. Aurore, T. 386342 30733 0 28509 445584 2198 51 | 0 13 | 2262 |
| 77. Blasdell, V. 162121 39162 213405 16860 431548 1126 49 | 2 8 | 1185 |
| 78. Boston, T. 328513 21532 3149 5929 359123 2122 57 | 1 7 | 2187 |
| 79. Brant, T. 89331 10109 0 4591 104031 596 28 | 0.10 | 634 |
| 80. Buffalo, C. 22747232 6125422 6273705 1895977 37042336 125398 5273 1 | 79 367 | 131217 |
| 81. Cheektomiaga,T. 4444263 1358704 495461 236198 6534626 30772 1040 | 3 68 | 31923 |
| 82. Clarence, T. 886749 265538 63593 63134 1279014 5146 365 | 3 25 | 5539 |
| 83. Colden, T. 118232 16571200 5818 140421 779 32 | 0 9 | 820 |
| 84. Collins, T. 155050 13450 97825 16321 282646 993 40 | 2 12 | 1047 |
| 85. Concord, T. 92473 10828 0 7348 110649 624 36 | 0 5 | 665 |
| 86. Depew, V. 923469 124012 705217 43938 1796636 6347 192 | 21 22 | 6582 |
| 87. E. Aurdra, V. 391445 106427 27893 46162 571927 2261 210 | 2 22 | 2495 |
| 88. Eden, T. 297467 60856 1924 25181 385428 1996 100 | 4 8 | 2108 |
| 89. Elma, T. 566754 47978 14115 30775 659622 3211 100 | 3 16 | 3330 |
| 90. Evans, T. 655329 60798 0 46114 762241 4779 143 | 0 21 | 4943 |
| 91. Farmham.V. 25724 1287 0 5043 32054 144 1 | 3 6 | 154 |
| 92. Gowanda, V. 55167 15011 3718 4796 78692 363 33 | 1 5 | 402 |
| 93. GrandIsland.T. 555679 130879 73065 44074 803697 4004 99 | 4 13 | 4120 |
| 94. Hamburg, T. 1693620 383674 440151 148114 2665559 11244 448 | 7 52 | 11751 |
| 95. Hamburg, V. 457529 97598 535 35681 591343 2743 216 | 1 20 | 2980 |
| 96. Holland, T. 142348 11980 30409 18153 202890 910 33 | 3 7 | 953 |
| 97. Kenmore, V. 859107 124724 0 15769 999600 6546 265 | 0 19 | 6830 |
| 98. Lackawanna, V. 1146913 276269 9059655 59411 10542248 8098 251 | 7 26 | 8382 |
| 99. Lancaster, V. 673546 75472 77491 51084 877593 4330 115 | 19 | 4474 |
| 100. Lancaster, T. 516654 63185 73178 45100 698117 3239 83 | 3 15 | 3340 |
| 101. Marilla, T. 166418 2589 0 5172 174179 1145 10 | 0 5 | 1160 |
| 102. Newstead, T. 59444 15501 48576 0 123521 368 38 | 3 0 | 409 |
| 103. Northcollins.T. 53730 6370 -2140 7223 65183 333 23 | 0 3 | 359 |
| 104. Northcollins, V. 76623 19079 6224 2180 104106 490 39 | 1 5 | 535 |
| 105. OrchardPark, V. 226065 38694 0 28178 292937 1198 78 | 0 9 | 1285 |
| 106. OrchardPark, T. 923661 118403 0 62355 1104419 5529 173 | 0 27 | 5729 |

Table D-1: Gas Sales and Average Numbers of Customers - 1979 - National Fuel Gas Distribution Corporation (Continued)
| Community | | Gas S | ales (MCF) | | | Average Number of Customers | | | | |
|--|---------------|------------|------------|---------------------------|--------------|-----------------------------|------------|------------|-----------------------|-------|
| annaharmana attinumena data data data data data data data da | Residential | Commercial | Industrial | Public Authorit | Total ics | Residential | Commercial | Industrial | Public Authorities | Total |
| 107. Sardinia, T. | 71351 | 9950 | 102313 | 3835 | 187449 | 441 | 30 | 7 | 7 | 480 |
| 108. Sldan, V. | 257722 | 85687 | 50 | 9981 | 353440 | 1759 | 36 | 0 | 5 | 1800 |
| 109. Springville, V. | 219925 | 70782 | 32572 | 32926 | 356205 | 1351 | 119 | 4 | 16 | 1490 |
| 110. Tonawanda, C. | 768577 | 188784 | 902775 | 54336 | 1914472 | 5709 | 222 | 27 | 29 | 5987 |
| 111. Tonawanda, T. | 3015905 | 726010 | 3245150 | 152891 | 7139956 | 23166 | 845 | 49 | 64 | 24124 |
| 112. Wales, T. | 73274 | 8684 | 0 | 5889 | 87847 | 527 | 19 | 0 | 6 | 552 |
| 113. West Seneca, T. | 2361327 | 318729 | 89856 | 146178 | 2916090 | 15156 | 465 | 4 | 57 | 15682 |
| 114. Williamsville, V. | 333025 | 87513 | 0 | 6160 | 426698 | 2206 | 126 | 0 | 6 | 2338 |
| 115. Alexander, T. | 52068 | 5746 | 11093 | 24183 | 93090 | 321 | 18 | 1 | 6 | 346 |
| 116. Batavia, C. | 871612 | 255245 | 415142 | 142997 | 1.684996 | 5191 | 311 | 20 | 38 | 5560 |
| 117. Batavia, T. | 167624 | 67119 | 0 | 6821 | 241564 | 1121 | 102 | 10 | 5 | 1238 |
| 118. Bethany, T. | 23449 | 1726 | -60 | 0 | 25115 | 145 | 4 | 0 | 0 | 149 |
| 119. Corfu, V. | 30472 | 11825 | 0 | 345 | 42642 | 177 | 25 | 0 | 2 | 204 |
| 120. Darien, T. | 50021 | 7865 | 0 | 2098 | 59984 | 320 | 21 | 0 | 3 | 344 |
| 121. Elba, T. | 32453 | 5538 | 0 | 9853 | 47844 | 231 | 23 | 0 | 6 | 260 |
| 122. Pavilion, T. | 17104 | 910 | 0 | 0 | 18014 | 110 | . 5 | 0 | 0 | 115 |
| 123. Pembroke, T. | 85360 | 9613 | 19173 | 6719 | 120865 | 576 | 28 | 1 | 7 | 612 |
| 124. Oakfield, T. | 118110 | 23789 | 0 | 3345 | 145244 | 743 | 43 | 20 | 5 | 811 |
| 125. Stafford, T. | 3136 | 809 | 0 | 0 | 3945 | 25 | 3 | 0 | 0 | 28 |
| 126. Avon, T. | 141 | 0 | 0 | 0 | 141 | 1 | 0 | 0 | 0 | 1 |
| 127. Lima, T. | 98759 | 40771 | 138290 | 6142 | 283962 | 556 | 64 | 2 | 9 | 631 |
| 128. Honeoye Falls,V. | 121344 | 39911 | 20466 | 3659 | 185380 | 725 | 74 | 4 | 7 | 810 |
| 129. Mendon, T. | 1686 | 0 | 0 | 0 | 1686 | 9 | 0 | Ó | 0 | 9 |
| 130. Lewiston, T. | 394315 | 215732 | 31256 | 97706 | 739009 | 2548 | 59 | 2 | 11 | 2620 |
| 131. Cambria, T. | 36761 | 4744 | 0 | 39545 | 81050 | 271 | 8 | ō | 1 | 280 |
| 132. Lewiston, V. | ⊈ 4762 | 49857 | 0 | 6049 | 150668 | 768 | 93 | 0 | 3 | 864 |
| 133. NiagraFalls,V. | 2714015 | 830291 | 4184084 | 129100 | 7857490 | 19343 | 1177 | 32 | 80 | 20632 |
| 134. Niagra, T. | 303398 | 131113 | 866670 | 125063 | 1426244 | 2227 | 249 | 5 | 9 | 2490 |
| 135. North TonawandaC. | . 1410784 | 192793 | 493464 | 100402 | 2197443 | 10004 | 348 | 42 | 42 | 10436 |
| 136. Dorter, T. | 80489 | 6016 | 0 | 18082 | 104587 | 535 | 16 | 0 | 12 | 563 |
| 137. Wilson, T. | 16730 | 1391 | 7134 | 13206 | 38461 | 1/13 | 4 | 1 | 2 | 150 |
| 138. Wilson, V. | 34407 | 8021 | 0 | 8577 | 51005 | 270 | 24 | ō | 6 | 300 |
| 139. Youngstown,V. | 67668 | 11716 | 0 | 1202 | 80586 | 552 | 26 | 0 0 | 2 | 580 |
| 140. Wheatfield,T. | 324198 | 62217 | 158818 | 10456 | 555689 | 2453 | 120 | ŭ | 7 | 2584 |
| 141. Bristol, T. | 27773 | 2395 | 0 | 962 | 31130 | 170 | 6 | 0 | 4 | 180 |

Table D-1: Gas Sales and Average Numbers of Customers - 1979 - National Fuel Gas Distribution Corporation (Continued)

.

.

| Comm | nunity | | Gas S | ales (MCF) | | | Average Number of Customers | | | | |
|----------------------------|--------------------|-------------|------------|------------|-----------------------|---------|-----------------------------|------------|------------|-----------------------|-----------|
| Contraction of Contraction | | Residential | Commercial | Industrial | Public Authorities | Total | Residential | Commercial | Industrial | Public Authorities | Total |
| 142. E | EastBloomfield,T. | 60846 | 6775 | 7837 | 12251 | 87709 | 299 | 17 | | , | 111 |
| 143. E | CastBloomfield, V. | 36690 | 12231 | 0 | 7643 | 56564 | 212 | 30 | 1 | 6 | 323 |
| 144. W | VestBloomfield, T. | 63147 | 3811 | 619704 | 1898 | 688560 | 4450 | 180 | 0 | 4 . | 240 |
| 145. R | tichmond,T. | 2986 | 32525 | 7717 | 4288 | 47516 | 20 | 100 | 0 | 30 | 4000 |
| 146. A | Arcade,V. | 71651 | 31322 | 146828 | 2614 | 252415 | 491 | 61 | 2 | 2 | 20 543 |
| 147. A | rcade,T. | 60606 | 4886 | 21535 | 1284 | 88311 | 446 | 7 | 0 | 5 | 205 |
| 148. A | ttica,V. | 152933 | 31240 | 0 | 9243 | 193416 | 853 | 65 | 4 | 4 | 401 |
| 149. A | ttica,T. | 13735 | 11467 | 61491 | 39688 | 126381 | 78 | 3 | 0 | 1 | 927 |
| 150. B | ennington,T. | 56342 | 2121 | 1189 | 1177 | 60829 | 339 | 2 | 5 | 6 | 250 |
| 151. G | astile,V. | 73232 | 13399 | 0 | 5615 | 92246 | 401 | 33 | 1 | 2 | 661 |
| 152. C | astile,T. | 5863 | 0 | 0 | 953 | 6816 | 30 | 1 | 0 | / | 32 |
| 153. E | agle,T. | 25823 | 2637 | 0 | 3444 | 31904 | 157 | 0 | 1 | 0 | 170 |
| 154. Ga | ainsville,T. | 27747 | 2108 | 440 | 19554 | 49849 | 157 | 9 | 0 | 4 | 170 |
| 155. G | eneseeFalls,T. | 476 | 0 | 0 | 0 | 476 | 201 | 0 | 0 | 4 | 270 |
| 156. Ja | ava,T. | 48246 | 6507 | 0 | 1000 | 55753 | 302 | 28 | 0 | Ŭ, | 22/ |
| 157. M: | iddlebury,T. | 13106 | 2768 | 0 | 800 | 16674 | 145 | 20 | 0 | 4 | 159 |
| 158. 01 | rangeville,T. | 10351 | 126 | 0 | 0 | 10477 | 80 | 1 | 1 | 4 | . 80 |
| 159. P: | ike,T. | 6829 | 93 | 404 | 0 | 7326 | 40 | 1 | 0 | 0 | 41 |
| 160. SH | heldon,T. | 794910 | 109080 | 4140 | 29590 | 937720 | 710 | 10 | 0 | 10 | 730 |
| 161. S: | ilverSprings,V. | 46430 | 4402 | 47807 | 3405 | 102044 | 498 | 35 | 0 | 10 | 5/3 |
| 162. Wy | yoming,V. | 29661 | 5288 | 0 | 3944 | 38893 | 278 | 15 | 1 | 4 | 200 |
| 163. Co | ovington,T. | 5863 | 0 | 0 | 0 | 5863 | 36 | 0 | 2 | 4 | 299 |
| 164. Ca | anisted,T. | 15092 | 843 | 0 | 0 | 15935 | 107 | 3 | 0 | 0 | 110 |
| 165. Ca | anisted,V. | 149641 | 16080 | 2161 | 16279 | 184161 | 921 | 70 | 1 | 12 | 1004 |
| 166. Fi | remont,T. | 11857 | 130 | 0 | 1142 | 13129 | 79 | 70 | 1 | 2 | 83 |
| 167. Gi | reenwood,T. | 29207 | 1736 | 0 | 6316 | 37259 | 200 | 16 | 1 | 6 | 223 |
| 168. Ho | ornell,C. | 655899 | 184993 | 121860 | 61915 | 1024667 | 3716 | 250 | 10 | 19 | 4004 |
| 169. Ho | ornellsville,T. | 134897 | 47447 | 6252 | 17101 | 205697 | 942 | 233 | 10 | 12 | 1050 |
| 170. Ho | oliard,T. | 11063 | 702 | -80 | 419 | 12104 | 62 | 5 | 2 | 1 | 68 |
| 171. No | orthHornell,V. | 52610 | 20215 | 0 | 2202 | 75027 | 309 | 8 | 0 | 5 | 322 |
| 172. We | estUnion,T. | 6265 | 305 | 0 | 426 | 6996 | 39 | 2 | U In | 2 | 43 |
| 173. Cl | larksville,T. | 25502 | 745 | Ó | 1066 | 27313 | 183 | 1 | 0 | 2 | 186 |

Table D-1: Gas Sales and Average Numbers of Customers - 1979 - National Fuel Gas Distribution Corporation (Continued)

Source: National Fuel Gas Distribution Corporation.

| · | | | | | | | | | | |
|--|-----------------|-------------|--------------|--------|------------|----------|--------|-----------------|------------|-----------|
| | Community | Land & | Structures | | Measuring | | | Total Plant | Population | Land Area |
| | Sommerrey | Land Rights | and | | Regulating | | | in Service (\$) | | (acres) |
| and the second | | | Improvements | Mains | Stations | Services | Total | | | (40100) |
| 1. | Alfred, T. | 0 | 0 | 125451 | 332 | 37279 | 163062 | 634800 | | |
| 2. | Alfred, V. | 0 | 256 | 255811 | 5727 | 58414 | 320208 | 389534 | 3804 | 768 |
| 3. | Alma, T. | 0 | 13 | 106677 | 26 | 24175 | 130891 | 216610 | | |
| 4. | Almond T. | 0 | 0 | 110147 | 0 | 16211 | 126358 | 295753 | | |
| 5. | Almond, V. | 100 | 755 | 38233 | 1652 | 12909 | 53650 | 69795 | | |
| б. | Amity, T. | 168 | 3583 | 89864 | 4018 | 23541 | 121174 | 142507 | | |
| 7. | Andover, T. | 0 | 0 | 172937 | 889 | 24414 | 198240 | 577491 | | |
| 8. | Angelica, T. | 0 | 374 | 140902 | 197 | 8076 | 149549 | 162227 | | |
| 9. | Angellica, V. | 202 | 1201 | 137624 | 3055 | 42466 | 184548 | 222885 | | |
| 10. | Belfast, T. | 654 | 631 | 84103 | 8140 | 30523 | 124051 | 132139 | | |
| 11. | Belmont, V. | 1095 | 3526 | 143068 | 11952 | 52324 | 211964 | 219062 | | |
| 12. | Bolivar, T. | 0 | 0 | 175006 | 183 | 52585 | 227774 | 338130 | | |
| 13. | Caneadea, T. | 0 | 0 | 176769 | 0 | 23549 | 200319 | 243517 | | |
| 14. | Centerville, T. | 0 | 0 | 25850 | 0 | 6095 | 31945 | 35216 | | |
| 15. | Cura, T, | 0 | 0 | 44106 | 0 | 24753 | 68859 | 152875 | | |
| 16. | Cuba, V. | 1032 | 1041 | 119309 | 6847 | 76154 | 204383 | 290405 | | |
| 17. | Friendship, T. | 464 | 3489 | 400474 | 11883 | 64339 | 480649 | 507747 | | |
| 18. | Friendship, V. | 0 | 0 | 0 | 0 . | 64339 | 43443 | 69031 | | |
| 19. | Genesee, T. | 0 | 512 | 237381 | 8518 | 3865 | 250277 | 362856 | | |
| 20. | Independence T | . 0 | 299 | 220514 | 0 | 51722 | 272535 | 364952 | | |
| 21. | Scio, T. | 0 | 359 | 365278 | 4708 | 76163 | 446508 | 616002 | | |
| 22. | Wellsville T | 0 | 1144 | 641997 | 4168 | 128982 | 776290 | 1503070 | | <u></u> |
| 23. | Wellsville, V. | 505 | 1899 | 499675 | 20693 | 202296 | 725068 | 1022977 | | 1408 |
| 24. | Willing, T. | 0 | 0 | 264224 | 0 | 49235 | 313459 | 372061 | 5815 | |
| 25. | Wirt, T. | 0 | 27 | 188246 | 8339 | 70642 | 267253 | 407134 | | |
| 26. | Carrollton T | 0 | 0 | 0 | 0 | 96 | 96 | 311 | | |
| 27. | Davton, T | 0 | 0 | 0 | 0 | 76 | 76 | 360 | | |
| 28. | Delevan V | 307 | 887 | 99025 | 6035 | 19281 | 125535 | 168588 | | |
| 29. | East Otto T | 244 | 563 | 99104 | 1093 | 10626 | 111630 | 116333 | | |
| 30. | Ellicottuille B | 386 | 782 | 186026 | 4825 | 25464 | 217484 | 317936 | | |
| 31. | Farmereville T | • 0 | 0 | 16432 | 2313 | 3436 | 22181 | 34050 | | |
| 32, | Franksville, T. | 0 | 0 | 91902 | 1399 | 19207 | 112508 | 199634 | | |

Table D-2: Distribution Plant in Service, Total Plant in Service, Population and Land Area - 1979 - National Fuel Gas Distribution Corporation

-

| Community | | Land & Land Rights | Structures and Improvements | Measuring and Mains Regulating Stations | | Services | Total | Total Plant in Service (\$) | Population | Land Area (acres) |
|-------------|-------------------|-----------------------|-----------------------------------|--|-------|----------|--------|--------------------------------|------------|----------------------|
| 33. | Freedon, T. | 0 | 0 | 53514 | 0 | 13182 | 66696 | 77499 | | |
| 34. | Govanda, V. | 0 | 0 | 116507 | 3661 | 28688 | 148856 | 205375 | 3110 | 1024 |
| 35. | Great Valley, T. | 432 | 865 | 228364 | 3362 | 35767 | 268790 | 289428 | | |
| 36. | Little Valley, T. | 0 | 0 | 125655 | 0 | 11515 | 137170 | 169323 | | |
| 37. | Machias T. | 321 | 876 | 191853 | 21084 | 53352 | 267487 | 367747 | | |
| 38. | Mansfield, T. | 0 | 0 | 17129 | 0 | 2503 | 19632 | 49239 | | |
| 39. | New Albion, T. | 0 | 709 | 133044 | 1980 | 12946 | 148679 | 161386 | | |
| 40. | Olean, C. | 1623 | 14777 | 441010 | 39560 | 136617 | 633586 | 870345 | 19169 | 3968 |
| 41. | Olean, T. | 0 | 0 | 228843 | 9288 | 43562 | 281692 | 301023 | | |
| 42. | Otto, T. | 395 | 1399 | 32684 | 1455 | 9158 | 45090 | 93330 | | |
| 43. | Perrysburg, V. | 111 | 1376 | 48451 | 6068 | 9633 | 65638 | 413989 | | |
| 44. | Perrysburg, T. | 963 | 2711 | 225642 | 8494 | 29286 | 267097 | 79870 | | |
| 45. | Persia, T. | 0 | 0 | 28138 | 0 | 6090 | 34228 | 38418 | | |
| 46. | Portville, T. | 2836 | 1660 | 410544 | 28977 | 108981 | 552998 | 596324 | | |
| 47. | Portville, V. | 648 | 1330 | 86310 | 9634 | 38280 | 136200 | 145323 | | |
| 48. | Salam Anca, C. | 1844 | 8605 | 773674 | 22381 | 102130 | 908633 | 1110103 | 7877 | 1920 |
| 49. | Salam Anca, T. | 273 | 0 | 100551 | 0 | 19512 | 120336 | 139316 | | |
| 50. | Yorkshire, T. | 347 | 808 | 222206 | 7608 | 29035 | 260005 | 383295 | | |
| 51. | Dunkirk, T. | 350 | 1292 | 419630 | 8968 | 124390 | 554631 | 918404 | | |
| 52. | Arkwright, T. | 0 | 0 | 58627 | 0 | 5638 | 64265 | 2002692 | | |
| 53. | Brocton, V. | 789 | 3305 | 128540 | 9839 | 57055 | 199527 | 226821 | | |
| 54. | Cassadaga, V. | 0 | 0 | 141790 | 2934 | 42312 | 187037 | 201078 | | |
| 55. | Chautaliquat T. | 0 | 103 | 464383 | 8934 | 135905 | 609325 | 1886405 | | |
| 56. | Dunkirk, C. | 4242 | 27003 | 863577 | 46259 | 331435 | 272516 | 1727435 | 16855 | 2944 |
| 57. | Forestville V | 203 | 0 | 75346 | 2559 | 30643 | 108751 | 120182 | | |
| 58. | Fredonia. V. | 3741 | 7051 | 702778 | 31004 | 206294 | 950867 | 1238412 | 10326 | 3584 |
| 59 | Hanover T | 830 | 125 | 504137 | 21980 | 173010 | 700081 | 1992613 | | |
| 60. | Mayville, V. | 787 | 77 | 221903 | 5366 | 41022 | 269154 | 302361 | | |
| 61 | Pomfret T | 0 | 76 | 705147 | 8979 | 125933 | 840136 | 2173149 | | |
| 62 | Portland T | 0 | 14955 | 364760 | 38155 | 81332 | 499203 | 1580543 | | · |
| 53 | Ripley T | 0 | 0 | 356890 | 6457 | 55685 | 419032 | 1524724 | | |
| 54 | Sheridan T | 0 | 0 | 514148 | 5392 | 87668 | 607208 | 1043885 | | |
| 65 | Sherman T | 0 | 0 | 21532 | 0 | 2475 | 24007 | 434923 | | |
| 66 | Silvercreek V | 2265 | 2811 | 211330 | 10615 | 76436 | 303457 | 346782 | 3182 | 768 |
| 67 | Stocktop T | 0 | 0 | 50716 | 2020 | 11379 | 64116 | 1496115 | | |
| 68. | Villenova T | - 0 | 0 | 2524 | 0 | 0 | 2524 | 2662 | | |
| <u>69</u> . | Westfield, T. | 0 | 14663 | 248421 | 33562 | 33844 | 330450 | 659590 | ` | |

Table D-2: Distribution Plant in Service, Total Plant in Service, Population and Land Area - 1979 - National Fuel Gas Distribution Corporation (Continued)

| | Community | Land & Land Rights | Structures and | Mains | Measuring and Regulating | Services | Total | Total Plant in Service (\$) | Population | Land Area (acres) |
|--------------|------------------|-----------------------|-------------------|----------|--------------------------------|----------|---------|--------------------------------|-------------------|----------------------|
| 70. | Westfield, V. | | | | Stations | 71000 | | | | |
| 71. | Akron, V. | 5046 | 2862 | 258637 | 18946 | 74262 | 359/53 | 423313 | 3651 | 2432 |
| 72. | Alden, T. | 2123 | 5052 | 361374 | 12960 | 19231 | 460/61 | 497071 | 2863 | 1152 |
| 73. | Alden, V. | 1292 | 0 | //0129 | 3469 | 20170 | 962017 | 1507019 | | |
| 74. | Amherst, T. | 22109 | 1/1 | 283941 | 5888 | 2121251 | 323228 | 350445 | 2651 | 1728 |
| 75. | Angola V | 34108 | 61964 | 12884804 | 219758 | 2131351 | 38/50/ | 17631152 | | |
| 76. | Aurore, T. | 0 | 340 | 261019 | 2982 | 55520 | 319861 | 401149 | 2676 | 768 |
| 77. | Blasdell V | 876 | 3995 | 1363211 | 20117 | 210764 | 598963 | 1927217 | | |
| 78. | Boston T | 559 | 1950 | 279417 | 5586 | 69889 | 357401 | 494340 | 3910 | 704 |
| 79. | Bront T | 1191 | 2351 | 1161453 | 20158 | 214850 | 403197 | 1682971 | | |
| 80. | Buffelo C | 50 | 0 | 360113 | 5588 | 59714 | 425466 | 563524 | | |
| 81 | Checktoniana T | 142009 | 337823 | 19703168 | 625097 | 4224131 | 032256 | 29599312 | 462768 | 26432 |
| 82. | Clarence T | 33889 | 54602 | 10299659 | 155315 | 1687741 | 253925 | 13714349 | | |
| 83 | Colden T | 3103 | 2574 | 2555144 | 26435 | 572942 | 161992 | 3799657 | · | |
| 84 | Colling T | 4754 | 160 | 603550 | 14987 | 87148 | 710598 | 959340 | | |
| 85 | Concord T | 0 | 435 | 537193 | 13426 | 75202 | 626256 | 1723857 | | |
| 86 | Depoir V | 355 | 5546 | 396140 | 23184 | 62232 | 319832 | 704487 | | ** |
| 87 | E August . | 7320 | 10826 | 2465058 | 48981 | 405589 | 937774 | 3196719 | 22158 | 3264 |
| 88 | E. Aurora, V. | 2187 | 5591 | 697982 | 18679 | 114674 | 839114 | 1061313 | 7033 | 1536 |
| 80. | ruen, I. | 270 | 0 | 1046712 | 27970 | 232420 | 307371 | 2364314 | 2962 | 2688 |
| 07. 00 | Erna, I. | 3044 | 10484 | 1863065 | 42604 | 395202 | 314398 | 2572219 | | |
| 01 | Evans, T. | 7451 | 9890 | 2367283 | 47204 | 609516 | 041345 | 3668229 | | |
| 71. | rarmnam,v. | 50 | 741 | 88764 | 3308 | 16354 | 109216 | 120917 | | |
| 92. 02 | Gowanda, V. | 718 | 3337 | 120995 | 5301 | 22374 | 152725 | 169516 | | |
| 93. 04 | GrandIsland, T. | 15819 | 0 | 2472358 | 11994 | 556340 | 3060288 | 3586423 | | |
| 94. | Hamburg, T. | 11582 | 38751 | 4844954 | 105923 | 1110081 | 6111292 | 7340337 | | |
| 9 5 . | Mamburg, V. | 2423 | 12610 | 895612 | 27265 | 176947. | 1115962 | 1374238 | 10215 | 1280 |
| Уð. 07 | Holland, T. | 328 | 4696 | 454871 | 39809 | 127378 | 627082 | 841170 | | |
| 97. | Kenmore, V. | 6059 | 5651 | 906602 | 24225 | 375055 | 1317592 | 1484836 | 20980 | 896 |
| 98. | Lackawanna, V. | 6912 | 15891 | 1564953 | 31562 | 376797 | 1996115 | 2569408 | 28657 | 3904 |
| 99. | Lancaster, V. | 6360 | 11468 | 1287297 | 24326 | 266464 | 1795735 | 1733517 | 13365 | 1728 |
| 100. | Lancaster, T. | 8938 | 5817 | 1957656 | 29969 | 483555 | 2485935 | 2809664 | | |
| 101. | Marilla, T. | 305 | 823 | 872604 | 3580 | 154080 | 1022392 | 1320275 | | |
| 102. | Newstead, T. | 0 | 0 | 512459 | 3792 | 51910 | 602160 | 735655 | | |
| 103. | Northcollins, T. | 50 | 777 | 268883 | 13412 | 49585 | 332707 | 893568 | | |
| 104. | Northcollins, V. | 1296 | 3608 | 177007 | 5343 | 27352 | 214605 | 241953 | | |
| 105. | OrchardPark, V. | 0 | 2331 | 467635 | 6944 | 62176 | 539915 | 607811 | 3732 | 1216 |
| 106. | OrchardPark, T. | 4462 | 8997 | 3474523 | 59045 | 628963 | 4175691 | 4798128 | | |

Table D-2: Distribution Plant in Service, Total Plant in Service, Population and Land Area - 1979 - National Fuel Gas Distribution Corporation (Continued)

| Community | Land & . Land Rights | Structures and Improvements | Mains | Measuring and Regulating Stations | Services | Total | Total Plant in Service (\$) | Population | Land Area (acres) |
|----------------------|-------------------------|-----------------------------------|----------------|--|----------|---------|--------------------------------|---------------------------------------|--|
| 107. Sardinia, T. | 317 | 2712 | 233901 | 31/0 | 2570/ | | | ੶ੑ੶ੑ੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶ | annan an a |
| 108. Sldan, V. | 265 | 2516 | 228052 | 9/57 | 55704 | 275783 | 438377 | | |
| 109. Springville, V | 501 | 4171 | 448519 | 10102 | 55346 | 294437 | 317630 | 5216 | 512 |
| 110. Tonawanda, C. | 5660 | 17750 | 2005867 | 26288 | . 80993 | 549287 | 617591/ | 4350 | 1856 |
| 111. Tonawanda, T. | 48341 | 45065 | 6895106 | 146521 | 280832 | 2736698 | 3139413 | 21898 | 2368 |
| 112. Wales, T. | 330 | 2123 | 261295 | 2600 | 2265571 | 9402196 | 14369034 | | |
| 113. West Seneca, T. | 18957 | 3/719 | 5812266 | 116370 | 30732 | 297172 | 399675 | | |
| 114. Williamsville. | 701 | 2201 | 361/3/ | 12551 | 1214/80 | 7215106 | 10585303 | | |
| 115. Alexander, T. | 101 | 2201 | 260792 | 12551 | 305481 | 682369 | 612711 | 6835 | 704 |
| 116. Batavia, C. | 79/1 | 10597 | 1007000 | 0 | 27847 | 389056 | 414206 | | |
| 117. Batavia, T. | 7041 | 19301 | 100/023 | 55140 | 397005 | 2367397 | 2667457 | 17338 | 3648 |
| 118. Berbany T | 0 | 0 | /59111 | 3736 | 114653 | 877575 | 1966051 | | |
| 119. Corfu V | 0 | 0 | 76638 | 511 | 72373 | 89552 | 294560 | ' | |
| 120. Darien T | 0 | U | 133705 | 8857 | 31538 | 174100 | 330743 | | |
| 120. barren, 1. | 0 | 0 | 257517 | 0 | 35986 | 293504 | 357172 | | |
| 122 Pavilion T | 3404 | 0 | 154748 | 0 | 4913 | 163065 | 181029 | | |
| 122. ravinon, 1. | 0 | 0 | 153052 | 0 | 15757 | 168810 | 261398 | | |
| 125. rembroke, 1. | 0 | 0 | 322124 | 1075 | 60083 | 383281 | 754212 | | |
| 124. Oakrield, T. | 0 | 0 | 78697 | 0 | 17321 | 96018 | 110995 | | |
| 125. Starford, T. | 0 | 0 | 10421 | 0 | 2679 | 13101 | 16056 | | |
| 126. Avon, T. | 0 | 0 | 0 | 0 | 0 | 0 | 321812 | | |
| 127. Lima, T. | 0 | 0 | 252642 | 1987 | 15385 | 270014 | 356731 | | |
| 128. Honeoye Falls,V | 3710 | 3710 | 257891 | 14173 | 49225 | 328710 | 362445 | | |
| 129. Mendon, T. | 0 | 0 | 19940 | 0 | 861 | 20801 | 22967 | | |
| 130. Lewiston, T. | 6298 | 0 | 2234155 | 26377 | 410050 | 2676881 | 3303210 | | |
| 131. Cambria, T. | 0 | 0 | 3 30593 | 0 | 28456 | 3590/9 | 436497 | | |
| 132. Lewiston, V. | 490 | 3272 | 498563 | 9196 | 81059 | 592580 | 646864 | 2202 | |
| 133. NiagraFalls, V. | 27337 | 48374 | 7368586 | 120606 | 292513 | 7857/17 | 10520194 | 3292 | 640 |
| 134. Niagra, T. | 9804 | 3771 | 1600396 | 50114 | 293426 | 1057512 | 2644928 | 82012 | 85/6 |
| 135. North Tonawanda | 20857 | 37766 | 4143348 | 76042 | 1160071 | 5/20006 | 6076841 | | |
| 136. Dorter, T. | 0 | 0 | 415833 | 0 | 57864 | J430000 | 585544 | 36012 | 6400 |
| 137. Wilson, T. | 3727 | 0 | 249597 | Ő | 13440 | 4/4020 | 316500 | | |
| 138. Wilson, V. | 0 | 0 | 190282 | 0 | 12013 | 200/05 | 282130 | | · |
| 139. Youngstown V. | 0 | 0 | 287302 | 758 | 72/65 | 202296 | 462210 | · | |
| 140. Wheatfield T | 54854 | Ő | 1939863 | 28/37 | 257520 | 360525 | 402319 | | - |
| 141. Bristol. T. | 0 | 0 | 192794 | 7/72 | 14000 | 1264675 | 2001721 | | |
| | | | 1)2/94 | 14/3 | 10305 | 217169 | 232278 | | |

Table D-2: Distribution Plant in Service, Total Plant in Service, Population and Land Area – 1979 – National Fuel Gas Distribution Corporation (Continued)

| | | | | | | | | · . | |
|-------------------------|-----------------------|-----------------------------------|--------|--|----------|---------|--------------------------------|------------|----------------------|
| | Land & Land Rights | Structures and Improvements | Mains | Measuring and Regulating Stations | Services | Total | Total Plant in Service (\$) | Population | Land Area (acres) |
| 142. EastBloomfield.T. | 10 | <u>,</u> | | | | | | | |
| 143. East Bloomfiled.V. | 10 | 0 | 214555 | 0 | 14161 | 228725 | 246643 | | |
| 144. West Bloomfield.T. | 40 | 0 | 130456 | 3128 | 30488 | 164112 | 176104 | | |
| 145. Richmond T | 0 | 0 | 543572 | 4654 | 44221 | 592447 | 645106 | | |
| 146. Arcade V | 0 | 0 | 89118 | 3779 | 1482 | 94379 | 109444 | | |
| 147. Arcade T | 4977 | 1969 | 177860 | 18884 | 32024 | 235714 | 250205 | | |
| 148 Attion V | 0 | 0 | 167522 | 8210 | 20961 | 106603 | 339293 | | |
| 140. Attion T | 1235 | 4141 | 366323 | 14125 | 103172 | /88005 | 404678 | | |
| 150 Romadastas m | 0 | 0 | 31864 | 0 | 7097 | 280(1 | 543475 | 2911 | 896 |
| 151 Cootdie V | 0 | 0 | 317304 | 2445 | 23372 | 26202 | 43667 | | ~ |
| 152 Creekle W | 342 | 2131 | 126634 | 4181 | 23812 | 343120 | 404137 | | |
| 152. Castile,T. | 0 | 0 | 33767 | 4101 | 23013 | 15/100 | 164856 | | |
| 155. Eagle, T. | 319 | 809 | 32970 | 5789 | 4/4/ | 38514 | 46643 | | |
| 154. Gainsville,T. | 0 | 0 | 42957 | 5788 | 9119 | 49005 | 106552 | | |
| 155. GeneseeFalls,T. | 0 | õ | 0 | 0 | 8245 | 51202 | 102492 | | |
| 156. Java,T. | 0 | õ | 220355 | 0 | 219 | 219 | 47068 | | |
| 157. Middlebury,T. | 0 | ő | 02000 | 0 | 26163 | 255518 | 307390 | | |
| 158. Orangeville,T. | 0 0 | · 0 | 93099 | 0 | 8977 | 102075 | 163965 | | |
| 159. Pike,T. | Ő | 0 | 1/(0/ | 0 | 3035 | 54362 | 62081 | | |
| 160. Sheldon,T. | ő | 0 | 14694 | 2397 | 4839 | 21930 | 93397 | | |
| 161. SilverSprings, V. | 134 | 12(7 | 399372 | 0 | 36201 | 435573 | 518172 | | |
| 162. Wyoming, V. | 151 | 1267 | 66248 | 1201 | 12277 | 81426 | 116917 | | |
| 163. Covington, T. | 101 | 4009 | 45504 | 4704 | 11175 | 65543 | 73337 | | |
| 164. Canisted.T. | 205 | 0. | 13727 | 0 | 3088 | 16814 | 19168 | | |
| 165. Canisted V. | 285 | 563 | 58197 | 2575 | 8366 | 69986 | 102102 | | |
| 166. Fremont T | 1700 | 1772 | 69178 | 5023 | 8366 | 238200 | 310255 | 2770 | (10 |
| 167. Greenwood T | 0 | 0 | 68051 | 0 | 3468 | 71520 | 70659 | 2112 | 640 |
| 168 Hornoll C | 0 | 351 | 102278 | 273 | 24864 | 127766 | 79058 | | |
| 160 Hornellerdile T | 666 | 4257 | 890836 | 13902 | 321130 | 1220700 | 248400 | | |
| 170 Holdend m | 0 | 530 | 290749 | 2565 | 632/6 | 257000 | 1824/12 | 12144 | 1664 |
| 170. HOLLARD, T. | 0 | 0 | 38210 | 2303 | 1638 | 200/0 | /172/3 | | |
| 171. NorthHornell,V. | 0 | 0 | 56492 | 0 | 1099/ | 39848 | 66/16 | <u></u> | |
| 1/2 WestUnion,T. | 0 | 0 | 29656 | 0 | 17004 | /63/6 | 102543 | | |
| 1/3. Clarksville,T. | 0 | 42 | 64985 | 1252 | /920 | 37582 | 40176 | | |
| | | | | 14.72 | 25/13 | 91992 | 204110 | | |

Table D-2: Distribution Plant in Service, Total Plant in Service, Population and Land Area - 1979 - National Fuel Gas Distribution Corporation (Continued)

Source: National Fuel Gas Distribution Corporation.

APPENDIX E

COMPUTER PROGRAM OF THE GUMCP MODEL

The purpose of this appendix is to present the listing of the GUMCP model developed in chapter 4. This listing depicts the MAIN program and the following subroutines: MARCOS, EVAL1, DIST, REVREQ, EVAL1, and EVAL2. The listing of the linear programming code LPCODE used in MARCOS is not presented here.

| FORTRAN | IV GI | RELEASE | 2.0 | MAIN | DATE = 802 | 67 17/05/08 |
|--|-------|---|--|---|--|---|
| 0001 0002 | | C | MAIN PROG IMPLICIT COMMON /M 1 TOP(200 2 X(200), | RAM REAL*8(A-H,O-Z) AIN1/ A(150,200),B),B(150),C(200),B0 SLACK(150),TOL(8),J | INV(150,150),TAB(UND(200),ROW(200) DUAL(150),BIG,SMA | 150,200),SOL(150), ,COL(150),S(150), LL,DETERM,OBJ |
| 0003 | | c | COMMON /M NINTO,N L NINTO,N L NINTO, I | AIN2/ LABCOL(200), 1 OUTOF, ISTATE, MNOW, 1 TERS, ITRMAX, IPRINT | LABROW(150), LABTH M, NCOL, N, NUMEQU, H , ISBND, IRMAX, IRCN | CM(200), INFEAS, SFEAS, ISDEGN, T, MAXM, MAXN |
| 0004 0005 0006 0007 0008 0009 0010 0011 0012 0013 0014 | | c cccccccccccccccccccccccccccccccccccc | DIMENSION DIMENSION DIMENSION DIMENSION DIMENSION REAL KMIN REAL NEWP NM=12 ICASE=1 ICASE=0 PAVGM=169 BASIC DAT ********** RMR, RMC, F ELR, ELC, E BL, SL=BAS DDM(IM) = SUPT1=MAX SUP2T=MAX SUP4T=MAX SUP4T=MAX SUP4T=MAX SUP4T=MAX SUP4T=MAX CC = COMP DC = DEMA WRC = WIN KMIN=TAKE CWH=COST CFL=COST DPROM=MAX COMP=CAS CIP=CAS F SHP=MARKE DSTCM=MAX CS = STOF CIST=STOF | DDM(12), PR(12), PC DCMR(12), DCMC(12) PRNCV(12, 50), PCMC DGMRV(12, 50), DCMC PRO(12), PCO(12), P IS, NPT2, NPD 9.665 A ********************************** | (12), PI(12), ELR(1 , DGMI(12), DGMT(12) V(12,50), PIMCV(12 V(12,50), DGMIV(12) IO(12) ************************************ | 2), ELC(12), ELI(12) 2) 2) 50) 50) 50) 50) 50) 51 51 51 51 52 51 51 52 51 51 52 51 51 52 51 51 52 51 51 52 51 51 52 51 51 52 51 52 52 52 52 52 52 52 52 52 52 |
| 0015 | | C C C | RMR=0.5 | | | |
| | | | | | | |

| FORTRAN | IV G1 | RELEASE | 2.0 | MAIN | DATE = 80267 | 17/05/08 |
|-------------------|-------|---------|----------------------------------|------------------|-----------------------|--------------|
| 0016 | | | RMC=0.5 | | | |
| 0017 | | | RMI=0.5 | | | |
| 0018 | | C | DA 9 IM=1 WW | | | |
| 0010 | | | FLR(1M) = -0.5 | | | |
| 0020 | | | ELC(IM) = -0.5 | | | |
| 0021 | | | ELI(IM) = -0.5 | | | |
| 0022 | | 2 | CONTINUE | | | |
| 0023 | | | BLR=3203.742 | | | |
| 0024 | | | SLR=23.912 | | | |
| 0020 | | | BLC= 1516.625 | | | |
| 0020 | | | BLU- 7. 104 BLI= 10170 264 | | | |
| 0028 | | | SLI=3.567 | | | |
| | | C | | | | |
| 0029 | | | DDM(1) = 506.6 | | | |
| 0030 | | | DDM(2) = 248.2 | | | |
| 0031 | | | DDM(3) = 50.5 | | | |
| 6633 | | | DDM(5) = 18.0 | | | |
| 0034 | | | DDN(6) = 120.5 | | | |
| 0035 | | | DDM(7) = 371.6 | | | |
| 0036 | | | DDM(8) = 712.6 | | | |
| 0037 | | | DDM(9) = 1071.6 | | | |
| 0038 | | | DDM(10) = 1207.7 | | | |
| 0039 AA4A | | | DDM(12) = 202 5 | | | |
| 0010 | | C | DDIN 14/- 074.0 | | | |
| 0041 | | - | SUP1T=200000. | | | |
| 0042 | | | SUP2T=500000. | | | |
| 0043 | | | SUPWHT=2000. | | | |
| 0044 0045 | | | SUPFL1=2500. | | | |
| 0040 | | | 1202.4 | | | |
| 0047 | | | WRC = 8.075 | | | |
| 0048 | | | CC2 = 1009.2 | | | |
| 0049 | | | DC2 = 1860.0 | | | |
| 0050 | | | KMIN = 0.75 | | | |
| 0051 | | | CWH = 787. | | | |
| 0012 | | С | UFL-1401. | | | |
| 0053 | | ч | DPROM= 3000. | | | |
| 0054 | | | COMP=921.12903 | | | |
| 0055 | | | CIP=14398.11 | | | |
| 0056 | | 6 | SHP=0.1 | | | |
| 0057 | | G | DOTOM- 100000 | | | |
| 0058 | | | CS=33 23 | | | |
| 0059 | | | CIST=50. | | | |
| | | С | | | | |
| 0060 | | | CIPT1=232.0397 | | | |
| 0061 | | | COM2=209.48495 | | | |
| 0002 | | c | ALLKOK=0.1206 | | | |
| | | C | | | | |
| 0063 | | 4 | WRITE(6.4) | | | |
| $006\overline{4}$ | | 4 | FORMAT(1H1,40X, 1100(1H*)///) | SUMMARY OF BASIC | DATA ASSUMPTIONS'/5X, | 100(1H*)/5X, |

| FORTRAN | IV G1 | RELEASE | 2.0 | main | DATE = 80267 | 17/05/08 |
|---------|-------|---------|--|--|---|--|
| 0065 | | | WRITE(6.3) RMR. | RMC.RMI | | |
| 0066 | | 3 | FORMAT(////10X, 120X, COMMERCIAL | 'RATES OF MARKET G | ROWTH' / 20X, ' RES IDF | ential=', F6.2/ |
| 0067 | | | WRITE(6.5) (ELR | (IM), IM=1, NM), (ELC | (IM), $IM=1$, NM), (ELI) | (IM), IM=1, NM) |
| 0068 | | 5 | FORMAT(///10X, ' | MONTHLY DEMAND ELA | STICITIES'//20X, 'F | ESIDENTIAL', 3X, |
| 0069 | | | WRITE(6.6) BLR. | SLR. RIC. SLC. RI.I. SI. | | y was a set of ear for f |
| ña7á | | 6 | FORMAT(/// 16% | BASE- AND SPACE-WE | ATING LOAD COFFFIC | LIENTS' //20X. 'BE |
| 0010 | | | ISIDENTIAL', 3X, 2 2.3X.2F12.3///) | F12.3/20X, 'COMMERC | IAL ', 3X, 2F12.3/20 | X, 'INDUSTRIAL ' |
| 0071 | | | WRITE(6.41) | | | |
| 0072 | | 41 | FORMATC////INX. | ' MEAN DECREE DAYS | DATA'//) | |
| 0073 | | | NDT=0. | | | |
| 0074 | | | DO 35 IM=1. NM | | | |
| 0075 | | | nor=nor+now/ IM | | | |
| 0076 | | 28 | CONTINUE | | | |
| 6677 | | 00 | WOITE/ A GAL DOT | (TATABLE TATAL TATAL NAME) | | |
| 0011 | | . 06 | COMATIEV 1 AVA | ANDYSAT TWEET AT THE CONTACT THE | TT TAVES TO 4 / PS | STATISTICS STATES |
| 0000 | | 96 | 1E DAYS=', 12F8.1 | ANNUAL IVIAL DEGN) | LE DAIG- ,F7.1//JA | N, TRIVINLE DEGRE |
| 0013 | | | ICFL | T, SUP2T, SUPWAT, SUP | FLT, CCI, IRI, WRG, CC | 12, INC2, KMIN, CWH, |
| 06800 | | 7 | FORMAT(////10X, UP2T=',F12.0/20 22.3/20X,'DC1=', UF12.3/20X.'KMIN | 'SUPPLY CHARACTERI X, 'SUPWHT=', F12.0/ F12.3/20X, 'WRC=', F =', F12.3/20X, 'CWH= | STICS'//20X,'SUP17 20X,'SUPFLT=',F12. '12.3/20X,'CC2=',F1 ',F12.3/20X,'CFL=' | F=',F12.0/20X,'S .0/20X,'CC1=',F1 12.3/20X,'DC2=', '.F12.3///) |
| 0081 | | | WRITE(6.8) DPR | OM. COMP. CIP. SHP | | , _ , _ , _ , , , , , , , , , , , , |
| 0082 | | 8 | FORMAT(////10X, 1X.'COMP='.F12.3 | 'PRODUCTION CHARAC | TERISTICS'//20X,'1 20X,'SHP='.F12.3// | DPROM=', F12.0/20 |
| 0083 | | | WRITE(6.9) DSTC | M.CS.CIST | ····· | - |
| 0084 | | 9 | FORMAT(////10X, 1'CS='.F12.3/20X | 'STORAGE CHARACTER | ISTICS'//20X, 'DST | CM=',F12.0/20X, |
| 0085 | | | WRITE(6,10) CIP | TI.ALLROR | | |
| 0086 | | 10 | FORMAT(///10X,' 1 OF RETURN ALLR | TRANSMISSION INVES OR=',F12.3////) | TMENT CIPT1=',F12. | . 3//// 10X, ' RATE |
| | | C | | | | |
| | | Ğ | AVERACE COST AN | ALYSIS | | |
| | | Ĉ | | | | |
| 0087 | | | WHITE(6,300) | | | |
| 0088 | | 300 | FORMAT(///40X,' 1/5X.100(1H*)/// | BASE AND AVERAGE () | OST PRICING ANALYS | SIS'/5X, 100(1H*) |
| 0089 | | | PAVG=1. | | | |
| 0090 | | | DO 1 IM=1.NM | | | |
| 0091 | | | PR(IM) = 1. | | | |
| 0092 | | | PC(IM) = 1 | | | |
| 0093 | | | PI(IM) = 1 | | | |
| AAQA | | 1 | CONTINUE | | | |
| AAAK | | £. | CALL | IGAD/RED DEC DEF | GID GIC GIT DMD DI | W BMI NDW DAUG |
| 0070 | | | 100 pr pi fid fi | C SY I INCRED INCREASED | MI NORT DIATI | TAN & LEALE & LALLES E HACES |
| 0096 | | | CALL MARC 1T, CIPTI, SUPIT, S | OSC CC1, CC2, DC1, DC2 UP2T, SUPWHT, SUPFLT | , KMIN, WRC, CWH, CFL , DPROM, DSTCM, SHP, I | , COMP, CIP, CS, CIS DGMT, DDGT, OMC1, |
| 0.0.0 m | | | andwr is, DGI, PR, P | U, TI/ | 1980 AL 10 000 000 000 000 000 000 000 000 000 | |
| 0097 | | | UALL DIST | (DGMT, 1MP, PEAK, CMP | "12, NP12, CMPD, NPD) | |
| 0098 | | | NEWPIS=NEWPIS+N | PTZ+NPD | | |
| 0099 | | | CALL REVR | eq(Allror, NEWP IS, D | GT, OMC1, X, PAVG, NE | TPIS) |
| 0100 | | | CALL EVAL 1DGT, X, NETPIS, AL | li(IMP, PEAK, DGMT, EL LROR, CRS, CCS, CIS, C | .R, ELC, ELI, DGMR, DCI IRST, CCST, CIST, CST | MC, DCMI, PAVG, , PS, TS) |

| FORTRAN | IV G1 | RELEASE | 2.0 | MAIN | DATE = 80267 | 17/05/08 |
|--------------|-------|------------|--------------|--|------------------------------------|-----------------------|
| | | C | | | | |
| | | č | START OF | THE ITERATIVE EQUIL | IBRATION PROCEDURE | |
| | | Ē | | | | |
| 0101 | | | WRITE(6,1 | 1) | | |
| 0102 | | 11 | FORMAT(1H | 11,40X,'ITERATIVE EG | UILIBRIUM PROCEDURE'/S | 5X, 100(1H*)/5X, |
| | | 0 | 1100(1H*)/ | (//) | | |
| A100 | | U | IFLICASE | ED 1) DAVO-DAVON | | |
| 0100 | | | NTWAY=50 | LG.I) LAAC-LAACU | | |
| 0107 | | | NTMAY= 10 | | | |
| 0100 0106 | | | NTMAX=5 | | | |
| 0107 | | | WRITE(6.2 | 03) NTMAX | | |
| 0108 | | 203 | FORMAT(// | 30X, 'MAXIMAL NUMBER | R OF ITERATIONS NTMAX= | , 15/30X, 34(1H=)/ |
| | | | 1/) | - | | |
| 0109 | | | IT=1 | | | |
| 0110 | | | DO 50 IM= | 1, NM | | |
| 0111 | | | PK=0. | 7 BAD \ 73 72- 4 | | |
| 0112 | | | DDMCW/IM | ITP = PR = I. | MACMONIACOMO | |
| 0110 0114 | | | PCMCV(IM | II = PC(IM) + PK+(CMP) | 24 CHE DJ + COHA POLCMDB) LCOMO | |
| 0115 | | | PIMCV(IM | IT = PI(IM) + PK + (CMP) | 2+CHFD)+COH2 P9+CMPD)+COM9 | |
| 0116 | | 50 | CONTINUE | | | |
| 0117 | | ••• | DO 100 IT | '= 1.NTMAX | | |
| 0118 | | | WRITE(6,1 | 08) IT | | |
| 0119 | | 108 | FORMAT(1H | <pre>[1,20X,'ITERATION NU</pre> | JMBER', I5/21X, 16(1H*)// | () |
| 0120 | | | WRITE(6,1 | 2) | | |
| 0121 | | 12 | FORMAT(5X | , 'MONTHLY MARGINAL | COSTS'//) | |
| 0122 | | | DO 101 IM | I= 1, NM | | |
| 0123 | | | PRO(IM) = P | RMCV(IM, IT) | | |
| 0124 | | | PUU(IM) = P | CHGV(IM, IT) | | |
| 0120 | | | TIUCIND-F | ON THE DECKIM | | |
| 0120 | | 19 | FORMAT(9) | S IN, FROUTH) / MONTH= / 14 9V /CO | 10T= 1 F 10 9) | |
| 0128 | | 101 | CONTINUE | , 1001121- , 1- 2 , 011, 00 | 191~ , F La. 09 | |
| 0129 | | | CALL | LOAD(BLR, BLC | C. BLI. SLR. SLC. SLI. RMR. I | RMC. RMI. DDM. PAVG. |
| ,- | | | 1PRO, PCO, P | 10, ELR, ELC, ELI, DGM | R, DGMC, DGMI, DGMT, DDGT) | ,,,,,,,, |
| 0130 | | | DO 102 IM | 1= 1, NM | | |
| 0131 | | | DGMRV(IM, | IT) = DGMR(IM) | | |
| 0132 | | | DGMCV(IM, | IT) = DGMC(IM) | | |
| 0133 | | 100 | DGMIV(IM, | IT) = DGMI(IM) | | |
| 0134 | | 102 | CONTINUE | 1) 00 50 100 | | |
| 0130 | | C | TEST OF D | I) GU IU IVO Nemand_ciddiv foiiti i | | |
| 0136 | | u . | WRITE(6 1 | (4) | i biti on | |
| 0137 | | 14 | FORMATC | 15X. TEST OF DEMANI | -SUPPLY FOULLIBRUM'Z | |
| 0138 | | | ID=0 | , only allow of selection | | , |
| 0139 | | | DO 104 IM | f= 1, NM | | |
| 0140 | | | DR=DGMRV(| IM, IT) - DGMRV(IM, IT- | -1) | |
| 0141 | | | DC = DGMCV(| IM, IT) - DGMCV(IM, IT- | -1) | |
| 0142 | | | DI=DGMIV(| IM, IT) -DGMIV(IM, IT- | -1) | |
| 0143 | | | DRA=DABS(| DRO DG) | | |
| VI 199 | | | DUA-DABS(| LPG J TVT V | | |
| 0120 A126 | | | WRITE(A 1 | 5) IM DB DC DI | | |
| Q147 | | 15 | FORMATICS | (.'MONTH='. 14.3Y'N) | R='. F12 3.3X. 'DC=' F12 | 3.3X.'DI=' F19 9 |
| | | 10 | 1) | a contraction of the other of the | e treestate the stree | erintei me. Ji teinop |
| 0148 | | | EPS=10. | | | |
| 0149 | | | IF((DRA.G | T. EPS). OR. (DCA. GT. I | EPS).OR.(DIA.GT.EPS)) | I D= 1 |

| FORTRAN | IV | G1 | RELEASE | 2.0 | MAIN | DATE - | 89267 | 17/05/08 |
|--------------|----|----|---------|--|--|--|--|----------------------------|
| 0150 0151 | | | 104 | CONTINUE IF(ID.EQ.1) | CO_TO 106 | | | |
| 0152 | | | 105 | WRITE(6,105) FORMAT(///5X IN', I5///) | ,'EQUILIBRIUM O | F SUPPLY AND | DEMAND REACHED | AT ITERATIO |
| 0154 | | | | GO TO 200 | | | | |
| 0155 | | | 106 | CALL M 1T, CIPT1, SUP1 2NEWPIS, DGT, P | ARCOS(CC1,CC2,D T,SUP2T,SUPWHT, R.PC.PI) | C1, DC2, KMIN, V SUPFLT, DPROM | WRC, CWH, CFL, COMI , DSTCM, SHP, DGMT, | P,CIP,CS,CIS DDGT,OMC1, |
| 0156 | | | _ | CALL D | ISTODCMT. IMP. PE | AK. CMPT2. NPT | 2. CMPD. NPD) | |
| 0157 | | | | NEWP IS=NEWP I | S+NPT2+NPD | integ with his yith he | | |
| 0158 | | | | CALL R | EVREQ(ALL.ROR. NE | WPIS.DCT.OMC | I. XE. PAVGE. NETP | IS) |
| 0159 | | | | CALL E | VAL2(IMP. PEAK. B | LR. BLC. BLL. SI | R. SLC. SLI. DDM. 1 | DGMR. DGMC. |
| | | |) 6 | DGMI, DGMT, DG 2CCS.CIS.CRST | T, PRO, PCO, PIO, E . CCST. CIST. CST. | LR, ELC, ELI, PARE, ELI, PARE, ELC, ELI, PARE, ELC, ELI, PARE, PARE, ELC, ELI, PARE, ELI | AVGE, XÉ, NÉTPIS, A MR. RMC, RMI) | ALLROR, CRS, |
| 0160 | | | | IT1= IT+1 | , , , | | ······ | |
| 0161 | | | | DO 107 IM=1.] | NM | | | |
| 0162 | | | | PK=0. | | | | |
| 0163 | | | | IF(IM.EQ.IMP) |) PK=1. | | | |
| 0164 | | | | PRMCV(IM, IT1) | = PR(IM) + (CMPT2) | +CMPD) *PK+COI | 12 | |
| 0165 | | | | PCMCV(IM, IT1) |) = PC(IM) + (CMPT2) | +CMPD) *PK+COI | 12 | |
| 0166 | | | | PIMCV(IM, IT1) | = PI(IM) + (CMPT2) | +CMPD) *PK+CO | 12 | |
| 0167 | | | 107 | CONTINUE | | | | |
| 0168 | | | 100 | CONTINUE | | | | |
| 0169 | | | 200 | STOP | | | | |
| 0170 | | | | END | | | | |
| | | | | | | | | |

| FORTRAN IV C1 | RELEASE 2.0 | MARCOS | DATE = 80267 | 17/05/08 | |
|--|--|---|---|--|------------------------------|
| 0001 | SUBROUTIN 1T, CIPTI, SU 2NEWPIS, DG C MAIN PROGRAI C LINEAR PROG C | E MARCOS(CC1,CC2,D UP1T,SUP2T,SUPWHT, F,PR,PC,PI) M LINPRO RAMMING CODE | C1, DC2, KMIN, WRC, CWH, CFL SUPFLT, DPROM, DSTCM, SHP, | ,COMP,CIP,CS,CIS DGMT,DDGT,UMC1, | 1576 1577 1582 |
| 0002 0003 | IMPLICIT I COMMON ∕M 1 TOP(290) 2 X(200),5 | REAL*8(A-H, O-Z) AIN1/ A(150, 200), B), B(150), C(200), BO SLACK(150), TOL(8), | INV(150,150),TAB(150,20 UND(200),ROW(200),COL(1 DUAL(150),BIG,SMALL,DET | 9),SOL(150), 50),S(150), ERM,OBJ | 0008 |
| 0004 | C COMMON /M 1 NINTO, N 2 ITYPE, I | AIN2/ LABCOL(200), DUTOF, ISTATE, MNOW, FERS, ITRMAX, IPRINT | LABROW(156), LABTEM(206) M, NCOL, N, NUMEQU, ISFEAS, , ISBND, IRMAX, IRCNT, MAXM | , INFEAS, ISDEGN, , MAXN | 0012 0014 0015 0016 |
| 0005 0006 0007 0008 0009 0010 0011 0012 0013 0014 | C REAL KMIN REAL NEWP DIMENSION 1GINST(12) DIMENSION DIMENSION DIMENSION DIMENSION DIMENSION DIMENSION DIMENSION DIMENSION DIMENSION | IS DCMT(12), PR(12), P , COUST(12), SUP1(12) CSTOR(12), FSN(12) COMAX(12), GIMAX(1 VCINS(12), VCOUS V1X(12), V2X(12), V VPRO(12), VTRAN(12 RHS(139) | C(12), PI(12),), SUP2(12), SUPV(12), PRO 2), RSTOR(12) (12), VSMAX(12), VSMIN(12 VV(12), VSUV(12), VDGMT(1) | D(12)) 2) | 0010 |
| | C ALL CAS FI C PROC=EXIS' C PT10=EXIS' C A10,B10 = C A20,B20=UI C RMIN AND C C STC0=EXIS' C | LOWS ARE EXPRESSED FING PRODUCTION CA FING TRANSMISSION UNIT SLOPE AND IN NIT SLOPE AND INTE RMAX = MINIMAL AND FING CERTIFIED STO | IN MMCF PACITY (MMCF) CAPACITY (MMCF) TERCEPT OF MAX. DELIVER RCEPT OF MAX. WITHDRAWA MAXIMAL SATURATION RAT RAGE CAPACITY (MMCF) | IES TO STORACE LS FROM STORACE ES | |
| 0015 0016 0017 0018 0019 0020 0021 0022 0023 0023 | WTOT = 12 PROC=947. PT10=55000 A10=-0.07 A20=0.152 B10=0.140 B20=-0.060 RMAX=1.18 RMIN=0.77 STC0=1475 | .*WRC 66667 0. 766852 44512 43129 65677 94.1 | | | |
| 0025 0026 0027 | C WRITE(6,4 42 FORMAT(1H 1'EXISTING 2MONTHLY T 52 FORMAT(// 13X,'B10=' C START OF C | 2) PROC, PT10 1,40X,'OUTPUT FROM MONTHLY PRODUCTIO RANSMISSION CAPACI 10X,'EXISTING STOR ,F10.5,3X,'A20=',F LP MODEL SET-UP | SUBROUTINE MARCOS'/40X N CAPACITY PROCE', F12.3 TY PT10=', F12.3///) AGE CHARACTERISTICS'//5 10.5,3X,'B2O=', F10.5/5X | 7,39(1H=)////10X, //10X,'EXISTING X,'A10=',F10.5, (,'STCO=',F10.1/) | |
| | G STANT MAR. | D OF OF C APPING | | | |

| FORTRAN | IV | G1 | RELEASE | 2.0 | | MARCOS | | DATE : | 80267 | 17/05/08 |
|-------------|----|----|---|------------------------|----------|-----------|------------|--------|-------------|----------|
| 0028 | | | | N=6*NM+7 | | | | | | |
| 0029 | | | | $M = 11 \times NM + 7$ | 2 | | | | | |
| 0030 | | | | DO 54 1=1 | . NM | | | | | |
| 0031 | | | | II=I+NM | | | | | | |
| 0032 | | | | 12= 1+2×N1 | A | | | | | |
| 0033 | | | | 13=1+3×N | ā | | | | | |
| 0034 | | | | 14=1+4×N | 4 | | | | | |
| 0035 | | | | C(1) = -CS | | | | | | |
| 0036 | | | | C(11) = -CS | 2 | | | | | |
| 0037 | | | | C(12) = -C(12) | 1 | | | | | |
| 0038 | | | | IF(I.GE. |)) C(12) | =-CC1-WT | m | | | |
| 0039 | | | | C(13) = 0. | | | | | | |
| 0040 | | | | C(I4) = -C(| 12 | | | | | |
| 0041 | | | 54 | CONTINUE | | | | | | |
| 0042 | | | | 11=5*NM+1 | 1.1.1 | | | | | |
| 0043 | | | | 12=11+1 | | | | | | |
| 0044 | | | | 13=12+1 | | | | | | |
| 0045 | | | | 14=13+1 | | | | | | |
| 0046 | | | | C(13)=-C) | /H×12. | | | | | |
| 0047 | | | | C(I4)=-CI | L*12. | | | | | |
| 0048 | | | | C(I1) = -0. | 4*DC1 | | | | | |
| 0049 | | | | C(12) = -0 | 4×DC2 | | | | | |
| 0050 | | | | DO 57 IM: | :1,NM | | | | | |
| 0051 | | | | I=64+IM | | | | | | |
| 0052 | | | | C(I) = -CO! | P | | | • | | |
| 0053 | | | 57 | CONTINUE | | | | | | |
| 0054 | | | | I 1=77 | | | | | | |
| 0055 | | - | | 12=78 | | | | | | |
| 0056 | | | | 13=79 | | | | | | |
| 0057 | | | 1. S. | C(II) = -CI | P | | | | | |
| 0058 | | | | C(12) = -C | IST | | | | | |
| 0059 | | | | C(I3) = -C | PTI | | | | | |
| | | | ç | | | | | | | |
| | | | C · | | | • | | | | |
| 0010 | | | C | INITIALIZ | le a s f | s to be a | LL ZEROES | | | |
| 0000 | | | | 10 5 1=1, | M | | | | | |
| 0001 | | | | B(I)=0. | - | | | | | |
| 0002 | | | | DU D J=1, | 11 | | | | | |
| 0003 | | | | $A(1, J) = \emptyset$ | | | | 1 | | |
| 0004 004 | | | 152 | BUUND(J)= | -1. | | | | | |
| 0000 | | | C C | CONTINUE | | | | | | |
| | | | G | WRITING (| of const | TRAINTS (| 1) AND (2) | G | INST, COUST | |
| 0066 | | | V4 | F1=A10±R | 1114010 | | | | | |
| 0067 | | | | F2=A20xP | AIN+ROA | | | | | |
| 0068 | | | | $N_0 7 I = 1$ | NM | | | | | |
| 0069 | | | | 11=1 | 1412 | | | | | |
| 0070 | | | | 12=NM+I | | | | | | |
| 0071 | | | | B(11)=F13 | STCA | | | | | |
| 0072 | | | | B(12)=F2 | STCA | | | | | |
| 0073 | | | | S(11)=1 | | | | | | |
| 0074 | | | | S(12) = 1 | | | | | | |
| 0075 | | | | J1=I | | | | | | |
| 0076 | | | | J2 = I + NM | | | | | | |
| 0077 | | | | J3=78 | | | | | | |
| 0078 | | | | A(I1.J1): | : 1 . | | | | | |
| 0079 | | | | A(11, J3): | -F1 | | | • | | |

| FORTRAN | IV | G1 | RELEASE | 2.0 | MARCOS | DATE | L = 80267 | 17/05/08 |
|--|----|----|----------|---|--------------------------------|----------------|--------------------|----------|
| 0080 0081 0082 0083 0084 0085 0086 0087 0088 0088 0089 0099 0099 0099 | | | 8 7 | A(12,J2)=1. A(12,J3)=-F2 NM1=I-1 IF(NM1.EQ.0) DO 8 J=1,NM1 J1=J J2=J+NM A(11,J1)=-A10 A(12,J1)=-A20 A(12,J2)=A20 CONTINUE CONTINUE | GO TO 7 | | | |
| | | | Č | WRITING OF CO | NSTRAINTS (3) | AND (4) | GINST, COUST | |
| 0093 0094 0095 0096 0097 0098 0099 0100 0101 0102 0104 0103 0104 0105 0106 0107 0108 0108 0108 0108 0109 0111 | | | 10 56 | DO 56 I=1, NM I1=2*NM+I I2=3*NM+I B(I2)=0. S(I1)=1. S(I2)=1. J3=78 A(I1,J3)=-(RM IMAX=I DO 10 J=1, IMAI J1=J J2=J+NM A(I1,J1)=1. A(I2,J2)=1. A(I2,J1)=-1. CONTINUE CONTINUE | MIN) *STCO AX-RMIN) X | | | |
| 0112 0113 0114 0115 0115 0116 0117 0120 0122 0123 0123 0124 | | | CC | CONTRACT DEMA WRITING OF CO DO 55 I=1,NM I1=4*NM+I 12=5*NM+I S(I1)=1. S(I2)=1. J1=2*NM+I J2=5*NM+1 J3=3*NM+1 J3=3*NM+2 A(I1,J1)=1.0 A(I2,J3)=1.0 A(I2,J4)=-1.0 | ND CONSTRAINT NSTRAINTS (5) | S AND (6) E | OP1, SUPMX1, SUP2, | SUPMX2 |
| 0126 0127 0128 0128 | | | C C | WRITING OF CO TAKE-OR-PAY C DO 6 I=1,NM I1=6*NM+I I2=7*NM+I S(I1)=1. | NSTRAINTS (7) ONSTRAINTS (| ?) AND (8) | | |

| FORTRAN | IV | G1 | RELEASE | 2.0 | MARCOS | DATE = 80267 | 17/05/08 |
|--|-----|----|---------------|--|--------------------------|------------------------|----------|
| 0130 0131 0132 0133 0134 0135 0136 0136 0137 0138 | | | 6 | S(I2)=1. J1=4*NM+I J2=3*NM+I J3=5*NM+2 A(I1,J1)=-1.0 A(I1,J3)=KMIN A(I2,J2)=1.0 A(I2,J1)=-1.0 CONTINUE | | | |
| 0139 0140 0141 0142 0143 0144 0145 0144 0145 0146 0147 0148 0149 0150 | · . | | C C 30 | $\begin{array}{c} \text{MAXIMUM ANNUAL} \\ \text{CONSTRAINTS} \\ \text{I1=8*NM+1} \\ \text{I2=8*NM+2} \\ \text{B(I1)=SUP1T} \\ \text{B(I2)=SUP2T} \\ \text{S(I1)=1.} \\ \text{S(I2)=1.} \\ \text{DO 30 I=1,NM} \\ \text{J1=2*NM+I} \\ \text{J2=3*NM+I} \\ \text{A(I1,J1)=1.0} \\ \text{A(I2,J2)=1.0} \\ \text{CONTINUE} \end{array}$ | SUPPLIES (9) AND (10) | | |
| 0151 0152 0153 0154 0155 0156 0157 0158 0159 0159 0160 | | | C | CONSTRAINTS ON 11=99 12=100 J1=63 J2=64 B(11)=SUPWHT B(12)=SUPFLT S(11)=1. S(12)=1. A(11, J1)=1. A(12, J2)=1. | WELL-HEAD AN | D FIELD-LINE PURCHASES | |
| 0161 0162 0163 0165 0166 0167 0168 0167 0169 0170 0171 0172 0173 0174 0175 0176 0177 0178 0179 0180 0181 | | | C 58 59 | CONSTRAINTS ON DO 58 IM=1, NM I=100 + IM B(I)=PROC S(I)=1. J1=64 + IM J2=77 A(I,J1)=1. A(I,J2)=-1. CONTINUE I=113 J=77 B(I)=DPROM S(I)=1. A(I,J)=1. I=114 B(I)=-DDGT $*$ SHP S(I)=1. DO 59 IM=1, NM J=64 + IM A(I,J)=-1. CONTINUE | PRODUCTION | | |

| FORTRAN | IV | G1 | RELEASE | 2.0 | MARC | 08 | DATE | = 802 | 267 | 17/05/08 |
|---------|----|----|---------|------------------------|------------|------------|---|-------|--------|----------|
| | | | C | CONTRAINT | ON STORACE | EXPANSION | (| | | |
| 0182 | | | | I = 115 | | | | | | |
| 0183 | | | | J=78 | | | | | | |
| 0184 | | | | B(I)=DSTCM | Ĺ | | | | | |
| 0185 | | | | S(I) = 1. | | | | | | |
| 0186 | | | • | A(I, J) = 1. | | | | | | |
| | | | C | CONTRAINT | ON TRANSMI | SSION EXPA | ANS I ON | | | |
| 0187 | | • | | DO 60 IM=1 | , NM | | | | | |
| 0188 | | | | I = 115 + IM | I . | | | | | |
| 0189 | | | | B(I) = PT10 | | | | | | |
| 0190 | | | | S(D) = 1. | | | | | | |
| 0191 | | | | J1=24 + IM | | | | | | |
| 0192 | | | | J2=36 + IM | ſ | | | | | |
| 0193 | | | | J3=63 | | | | | | |
| 0194 | | | | J4=64 | _ | | | | | |
| 0195 | | | | J5=64 + IM | [| | | | | |
| 0196 | | | | J6=79 | | | | | | |
| 0197 | | | | A(I, JI) = 1. | | | | | | |
| 0198 | | | | A(I, J2) = 1. | | | | | | |
| 0199 | | | | A(I, J3) = 1. | | | | | | |
| 0200 | | | | A(1, J4) = 1. | | | | | | |
| 0201 | | | | A(I, J5) = 1. | | | | | | |
| 0202 | | | | A(1, J6) = -1 | • | | | | | |
| 0203 | | | . 60 | CONTINUE | | | | | | |
| | | | C | DEMAND REC | UIREMENTS | | | | | |
| 0204 | | | | DO 62 IM=1 | , NM | | | | | |
| 0205 | | | | 1 = 127 + 1M | | | | | | |
| 0206 | | | | B(1) = DGMT(| IMD | | | | | |
| 0207 | | | | S(1) = 0. | | | | | | |
| 0208 | | | | JI=IM | | | | | | |
| 0209 | | | | J2=NM + IM | [| | | | | |
| 0210 | | | | J3=2*NM + | IM | | | | | |
| 0211 | | | | J4=3*NM + | IM | | | | | |
| 0212 | | | | 12=03 | | | | | | |
| 0213 | | | | J6=64 | | | | | | |
| 0214 | | | | J7=64 + 11 | | | | | | |
| 0215 | | | | A(1, J1) = -1 | • | | | | | |
| 0216 | | | | A(1, J2) = 1. | | | | | | |
| 0217 | | | | A(1, J3) = 1. | | | | | | |
| 0218 | | | | $A(1, J_4) = 1.$ | | | | | | |
| 0219 | | | | A(1, J5) = 1. | | | 1. A. | | | |
| 0220 | | | | A(1, J6) = 1. | | | | | | |
| 0221 | | | | A(1, J7) = 1. | | | | | | |
| 0222 | | | 62 | CONTINUE | | | | | | |
| | | | u . | ***** | ******** | ******** | <u> </u> | ***** | ****** | ***** |
| 0000 | | | C | NTO N | | | | | | |
| 0223 | | | | NEQ=M | | | | | | |
| 0224 | | | | FIAXET 150 | | | | | | |
| 0220 | | | | HAXN=200 | , | | | | | |
| V220 | | | | TIAANUU= 220 |) | | | | | |
| 0220 | | | | HAXSUK= 8 | | | | | | |
| V220 | | | | $I \Gamma L A D I = I$ | | | | | | |
| 0229 | | | | IKUUND=3 | | | | | | |
| 0230 | | | | ILUGIC=1 | | | | | | |
| 0231 | | | | IGOMRY= 1 | | | | | | |
| 0232 | | | | ISBND=1 | | | | | | |
| 0233 | | | | 1 THMAX= 300 |) | | | | | |
| 0234 | | | | IPRINT=0 | | | | | | |

| FORTRAN | IV | G1 | RELEASE | 2.0 | MARCOS | DATE = 8020 | 57 | 17/05/08 |
|--------------|----|----|---------|---|---|--------------------|-------------|---|
| 0235 | | | | I RMAX= 1 | | | | |
| 0236 | | | | NUMEQU=0 | | | | |
| 0237 | | | | DO 135 I=1,M | | | | |
| 0238 | | | 101 | IF(S(I).EQ.0) | NUMEQU= NUMEQU+ | •1 | | |
| 0239 | | | 135 | CONTINUE | | | | |
| 0240 | | | | ISTATE=0 | | 4 | | |
| 9241 0040 | | | | ISTEAS-V DIC-1 OF11 | | | | |
| 0242 0243 | | | | SMATI = 1 AF-Q | | | | |
| 0244 | | | | TOL(1) = 1.0E - 6 | 6 | | | |
| 0245 | | | | TOL(2) = 1.0E - 5 | 5 | | | |
| 0246 | | | | TOL(3) = 1.0E-5 | 5 | | | |
| 0247 | | | | TOL(4) = 1.0E - 6 | 5 | | | |
| 0248 | | | | TOL(5) = 1.0E - 5 | 5 | | | |
| 0249 | | | | TOL(6) = 1.0E - 5 | 5 | | | |
| 0250 | | | | TOL(7) = 1.0E - 5 | 5 | | | |
| 0251 | | | | TOL(8) = 1.0E-3 | 3 | | | |
| 0252 | | | | DETERME 1.0 | | | | |
| 0203 0054 | | | | CALL FIRSTS | | | | |
| 0209 0255 | | | | IF(MNOW NF | M CO TO O | | | |
| 0256 | | | | IF(ISTATE N | | | | |
| 0257 | | | | DO 1 I=1.M | | | | |
| 0258 | | | | IF(S(I)) EQ. | 1.0) GO TO 1 | | | |
| 0259 | | | | GO TO 2 | | | | |
| 0260 | | | 1 | CONTINUE | | | | |
| 0261 | | | | GO TO 9 | | | | |
| 0262 | | | 2 | CALL INVERT | | | | |
| 0263 | | | | CALL FULTAB | | | | |
| 0264 | | | | DO 4 1=1,M | | | | |
| V265 | | | | HOLD=0.0 | | | | |
| 0200 | | | 0 | UU 3 K=1, M | (P) +DINU(P I) | | | |
| 0201 0268 | | | 3 4 | HOLD = | C(K) +DINA(K, I) | | | |
| 0269 | | | 75 | ISTATE=4 | | | | |
| 0270 | | | . 9 | CALL OUTPUT | | | | |
| | | | C**** | ***** | ****** | ******* | ***** | ***** |
| | | | C**** | ***** | ***** | ***** | * | |
| 0271 | | | | WRITE(6,17) | | | | |
| 0272 | | | 17 | FORMAT(//5X, | CONSTRAINTS VAL | LUES AND DUAL PRI- | Ces'///) | |
| 0273 | | | | DO 14 I=1, NEC | D | | | |
| 0274 | | | | $\operatorname{RHS}(1) = 0.$ | | | | |
| 0270 | | | | DU = 10 J = 1, N | \ | | | |
| 9279 0977 | | | 15 | CONTINUE | TACI, JJ *A(J) | | | |
| 0211 | | | 10 | WRITE(6 16) | I RHQ(I) B(I) DI | TATICES | | |
| 0279 | | | 16 | FORMAT(2X. ' I | = 14.3X.'BHS=' | F20.5.3X.'B='.F2 | 0.5.3X.'DH | AL='.F20.5) |
| 0280 | | | 14 | CONTINUE | , | | 010,011, 20 | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |
| 0281 | | | | DO 43 IM=1,NI | M | | | |
| 0282 | | | | 12=NM+IM | | | | |
| 0283 | | | | 13=2*NM+ IM | | | | |
| 0284 | | | | 14=3*NM+IM | | | | |
| 0285 | | | | $15 = 4 \times NM + IM$ | | | | |
| 0280 | | | | | | | | |
| 0201 | | | | 17-07-NM+1M | | | • | |
| V200 8980 | | | | 10-64NTT1M 10=10011M | | | | |
| 0207 020A | | | | VPRO(IM)=DIM | ((10) | | | |
| UMIV | | | | VI AUX III - DUAL | 64 L 7 J | | | |

| FURTHAN | IV GI | RELEASE | 2.0 | MARCOS | DAIE | - 04201 | 16/ | 991490 |
|--------------|-------|----------------|-------------------|-----------------------------|--------------------|--------------|--------------|------------|
| 0291 | | | I 10=115+IM | | | | | |
| 0292 | | | VTRAN(IM) = DUAL | L(I10) | | | | |
| 0293 | | | I I I = 127+ IM | | | | | |
| 0294 | | | VDGMT(IM) = DUAI | L(I11) | | | | |
| 0295 | | | VGINS(IM) = DUAL | LCIMD | | | | |
| 0296 | | | VGOUS(IM) = DUAL | L(12) | | | | |
| 0297 | | | VSMAX(IM) = DUAL | L(I3) | | | | |
| 0298 | | | VSMIN(IM) = DUAI | L(I4) | | | | |
| 0299 | | | V1X(IM) = DUAI | L(15) | | | | |
| 0300 | | | V2X(IM) = DUAI | L(16) | | | | |
| 0301 | | | VVV(IM) = DUAI | L(17) | | | | |
| 0302 | | | VSUV(IM) = DUAL | L(18) | | | | |
| 0303 | | | PR(IM) = -VDGMT(| (IM) | | | | |
| 0304 | | | PC(IM) = -VDGMT(| (IM) | | | | |
| 0305 | | | PI(IM) = -VDGMT(| (IM) | | | | |
| 0306 | | 43 | CONTINUE | | | | | |
| 0307 | | | I 1=8*NM+1 | | | | | |
| 0308 | | | 12=8*NM+2 | | | | | |
| 0309 | | | VS1T=DUAL(I1) | | | | | |
| 0310 | | | VS2T=DUAL(12) | | | | | |
| 0311 | | | 13=99 | | | | | |
| 0312 | | | I4=100 | | | | | |
| 0313 | | | VSWH=DUAL(13) | | | | | |
| 0314 | | | VSFL=DUAL(14) | | | | | |
| 0315 | | | 15=113 | | | | | |
| 0316 | | | I6=114 | | | | | |
| 0317 | | | VPRMX=DUAL(15) |) | | | | |
| 0318 | | | VPRMN=DUAL(I6) |) | | | | |
| 0319 | | | 17=115 | | | | | |
| 0320 | | | VSTC=DUAL(17) | | | | | |
| 0321 | | | WRITE(6, 20) | | | | | |
| 0322 | | 20 | FORMAT(1H1,40) | X, 'OPTIMAL SO | LUTION CHARA | CTERISTICS | '/40X, 32() | (H*)///) |
| 0323 | | | DO 18 IM=1, NM | | | | | |
| 0324 | | | II = IM | | | | | |
| 0325 | | | 12 = NM + 1M | | | | | |
| 0326 | | | 13=2*NM+IM | | | | | |
| 0327 | | | 14=3*NM+1M | | | | | |
| 0328 | | | GINST(IM) = X(I) | 1) | | | | |
| 0329 | | • | GOUST(IM) = X(I) | 2) | | | | |
| 0330 | 1 J | | FSN(IM)=GOUST | (IM)-GINST(IM |) | | | |
| 0331 | | | SUPICIMI = X(13) |) | | | | |
| 0332 | | | SUP2(IM) = X(I4) |) | | • | | |
| 0333 | | 18 | CONTINUE | | | | | |
| 0334 | | | $10=0\times NM+1$ | | | | | |
| 0335 | | | 10= 3*NM+2 | | | | | |
| 0330 | | | SUPMAI=X(15) | | | | | |
| 0337 | | | SUPMA2=A(10) | V1 /00 | | | | |
| 0338 | | | DAIMAI - SUPPL | A1/30. | | | | |
| 0339 | | | DAIFIAZ = SUPM | A4/ JV. | | | | |
| 0340 | | | WRITE(0,0999) | ITO MAU I IN A UMPUPA | | | | |
| 8041 0010 | | | WRITE(0,21) S | urnai, yaimai | | | | |
| V042 A949 | | | WOITE(& GIV O | TOMUS TAUMUM | | | | |
| 0343 0944 | | 6000 | TODMAT(//) | CONSOL IDATE | | | | |
| V344 A94E | | · ወንንን ማልልል | FUTURI (/ / ' | DANUANDIE | /) / / · · · · | | | |
| 0040 0946 | | (UUU 0 1 | FORMAT(/ / EV) | CANDANULL' MAVIMAI MONTH | / / IV GINDDIV | 10 0/ FV | INATIMAT 1 | NATEV OTTO |
| 0%°CU | | 4 I | 1PLY=', F10.2// | DAAIDAL RUNIN) | LI BUTTLI=', | F 10.4/, 0A, | THAN I FIALS | UAILI SUP |
| 0347 | | | DO 63 IM=1.NM | | | | | |

| FORTRAN | IV | Gl | RELEASE | 2.0 | MARCOS | DATE = 80267 | 17/05/08 |
|---------|----|----|---------|---|--|---|--|
| 0348 | | | | [1=48 +]] | 1 | | |
| 0349 | | | | 12=64 + 11 | 1 | | |
| 0350 | | | | SUPV(IM)= | K(I1) | | |
| 0351 | | | | PROD(IM) = 2 | K(12) | | |
| 0352 | | | 63 | CONTINUE | | | |
| 0353 | | | | DO 19 IM= | I, NM | | |
| 0354 | | | | WRITE(6,6 | 5) IM, FSN(IM), SUP1(I | m, sup2(IM), supv(IM) | , PROD(IM) , DGMT(IM |
| 0355 | | | 65 | FORMAT(1X. 12X, 'SUPV= | ,'IM=',I3,2X,'FSN=', ',F9.2,2X,'PROD=',F8 | F9.2,2X,'SUP1=',F9.2 .2,3X,'DGMT=',F9.2) | 2,2X,'SUP2=',F9.2, |
| 0356 | | | 19 | CONTINUE | | | |
| 0357 | | | | I 1=63 | | | |
| 0358 | | | | 12=64 | | | |
| 0359 | | | | 13=77 | | | |
| 0360 | | | | 14=78 | | | |
| 0361 | | | | 15=79 | | | |
| 0362 | | | | SUPWH=X(I | 1) | | |
| 0363 | | | | SUPFL=X(I: | 2) | | |
| 0364 | | | | DPRO=X(13) |) | | |
| 0365 | | | | DSTC=X(I4) | | | |
| 0366 | | | | DPT1=X(15) | | | |
| 0367 | | | | WRITE(6,60 | 5) SUPWH, SUPFL, DPRO, | DSTC, DPT1 | · · · · · · · · · · · · · · · · · · · |
| 0368 | | | 66 | FORMAT(/// 1/10X, 'DSTV | //10X,'SUPWH=',F10.3 C=',F10.3/10X,'DPT1= | /10X,'SUPFL=',F10.3/ ',F10.3//) | '10X, 'DPRO=', F10.3 |
| 0369 | | | | DCHT = (DC) | 1×SUPMX1 + DC2×SUPMD | 2)*0.4 | |
| 0370 | | | | CCHT=0. | | | |
| 0371 | | | | CST=0. | | | |
| 0372 | | | | WCST=0. | | | |
| 0373 | | | | DO 22 1M= | 1, NM | | |
| 0374 | | | | CCHT=CCHT- | SUP1(IM) *CC1 + SUPV | (IM) *CC2 | |
| 0375 | | | | CST= CST+ | (GINST(IM)+GOUST(IM) |) *CS | |
| 0376 | | | | IF(IM.GE. | B) WCST=WCST+WTOT*SU | P1(IM) | |
| 0377 | | | 22 | CONTINUE | | | |
| 0378 | | | | DCHTS=-DC | HT/OBJ | | |
| 0379 | | | | CCHTS=-CC | HT/OBJ | | |
| 0380 | | | | CSTS=-CST | ∕0BJ | | |
| 0381 | | | | WCSTS=-WC | ST/OBJ | | |
| 0382 | | | | WRITE(6,2 | 3) DCHT, DCHTS, CCHT, C | CHTS, CST, CSTS, WCST, V | WCSTS |
| 0383 | | | 23 | FORMAT(// | 10X, 'TOTAL DEMAND CH | LARGE=', 3X, F15.2, 5X, ' | 'OR', F10.5, 3X, 'OF M |
| | · | | | 1MINIMUM C 2,'OF MINI 35,3X,'OF 2 4.F10.5.3X | DST'/10X,'TOTAL COMP MUM COST'/10X,'TOTAI MINIMUM COST'/10X,'T .'OF MINIMUM COST'// | ODITY CHARGE=',F15.2 STORAGE COST=',4X,H OTAL WINTER CHARGE=' | 2,5X,'OR',F10.5,3X,' OF F15.2,5X,'OR',F10. ',3X,F15.2,5X,'OR' |
| 0384 | | | | STCAP=STC | 0+DSTC | | |
| 0385 | | | | A1=A10*ST | CAP*RMIN | | |
| 0386 | | | | A2=A20*ST | CAP*RMIN | | |
| 0387 | | | | B1=B10*ST | CAP | | |
| 0388 | | | | B2=B20*ST | CAP | | |
| 0389 | | | | GSTOR(1) = 0 | 0. | | |
| 0390 | | | | DO 24 IM= | 2.NM | | |
| 0391 | | | | CSTOR(IM) | =GSTOR(IM-1)+GINST() | M-1)-COUST(IM-1) | |
| 0392 | | | | RSTOR(IM) | = (GSTOR(IM) / STCAP) + I | MIN | |
| 0393 | | | 24 | CONTINUE | | | |
| 0394 | | | | WRITE(6.2 | 5) | | |
| 0395 | | | 25 | FORMAT(// | 10X. SUMMARY OF MONT | THLY STORAGE GAS FLOW | WS AND STOCKS'//) |
| 0396 | | | | GINTT=0. | | | |
| 0397 | | | | COUTT=0. | | | |
| 0398 | | | | DO 26 JM= | 1.NM | | |
| | | | | | | | |

| FORTRAN | IV G1 | RELEASE | 2.0 | MARCOS | DATE = 80267 | 17/05/08 |
|---------------|-------|---------|-----------------------------|--|--|-----------------------|
| A399 | | | CINTT=CINTT | +CINST(IM) | | |
| 8469 | | | COUTT = COUTT | + COUST(IM) | | |
| 0401 | | | CIMAX(IM) = A | 10*CSTOR(IM) + A1 | + B1 | |
| 0402 | | | COMAX(IM) = A | $20 \times GSTOR(IM) + A2$ | + 82 | |
| 0403 | | | WRITE(6 27) | IM CSTOR(IM) BST | OR IN CINST IN CIMAN | IM COUST IM |
| 0.200 | | | 1COMAX(IM) | 111,0010101110,1001 | on no , on or , or how , or here | , 11D , 000D1 (11D , |
| 0404 | | 27 | FORMAT(1X, ' | IM=', I3, 2X, 'CSTOR CIMAX=' F10 2 2X | =', F10.2,2X, 'RSTOR=', F | 10.2,2X,'GINST=' |
| 0405 | | 26 | CONTINUE | onna, rio.a, an, | 60051-, 110. a, an, 6012 | ···· , · ··· / |
| 0406 0406 | | | WRITE(6 40) | CINTT COUTT | | |
| 0407 | | 49 | FORMAT(/10X 1 OF STORAGE | (,'YEARLY FLOW INT | 0 STORAGE=', F12.2/10X, | YEARLY FLOW OUT |
| 0408 | | | WRITE(6.44) | · · · · · · · · · · · · · · · · · · · | | |
| 0409 | | 44 | FORMAT(///4 | 0X. 'DUAL VALUES S | UMMARY'/40X,20(1H*)/// | |
| 0410 | | | DO 45 IM=1. | NM | | |
| 0411 | | | WRITE(6.46) | IM. VGINS(IM). VGO | US(IM), VSMAX(IM), VSMIN | |
| 0412 | | 46 | FORMAT(1X. ' | IM='. 12.2X. 'VGINS | =', F10.3.2X.'VGOUS=' | |
| | | | 1. F13. 5. 2X. | VSMAX=', F11.3.2X. | 'VSMIN=', F11.3) | |
| 0413 | | 45 | CONTINUE | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, | 1011111 (11110) | |
| 0414 | | | WRITE(6.48) | | | |
| 0415 | | 48 | FORMAT(//10 | X.80(1H*)//) | | |
| 0416 | | | DO 47 IM=1. | NM | | |
| 0417 | | | WRITE(6.51) | IM. VIX(IM), V2X(I | MO. VVV(IM), VSUV(IM), VD | CMTCIMO |
| 0418 | | 51 | FORMAT(1X. | IM=', 12.2X, 'V1X=' | .F10.3.2X. 'V2X='.F10.3 | 2X. 'VVV='. F10.3 |
| 6410 | | 47 | 1,2X, 'VSUV=' | ,F10.3,2X, 'VDGMT= | ',F10.3) | |
| 0717 | | -206 | WDITE(S SO) | VEIT VEOT | | |
| 0720 | | 50 | FORMAT(///1 | AV VOIT-V FIR E | 1AV IVEOTS FIE EZZI | |
| 0761 | 2 A. | | WDITE(6 4Q) | UA, VBII- , PIU-U/ | 10A, VS21- , 110.0/// | |
| 0744 0499 | | | DO 67 IM=1 | NM | | |
| 0723 Q494 | | | WRITE(6 68) | IM UPRO(IM) VTDA | NITM | |
| 0727 | | 68 | FORMAT(15V | 'IM=' 19 OV 'VDDA | - I FIR R RV IMPOAN- I F | 1 民 民) |
| 0740 | | 67 | CONTINUE. | 111-, 10, 0A, VI 10 | - , FIJ.J.J.J., VIILAN- , F | 10.07 |
| 0320 04.97 | | 0. | WRITE(6 49) | | | |
| 6429 | | | WRITE(6, 60) | VOWN VORI VDDWV | UDDAWN UCTY | |
| 0720 6490 | | 60 | FORMAT(///1 | V WOWN-' FIA 7 7 | VIULLY, VOIC | DMV- 1 E10 9 |
| 0400 | | U Z | 13X, 'VPRMN=' | ,F10.3,3X,'VSTC=' | , F10.3//) | un- , r 10.0, |
| U TOU 0491 | | | DIG~DDDA~01 | D . DOTOSOTOT . D | 1 W T I D T 1 | |
| 0701 | | | 0MC1001 | r T DOILALIDI T D | F11*G1F11 | |
| 94JA | | | VELDIC-DIC | | | |
| 01233 | | | NEWFIS-FIS/ | CRF | | |
| 040ª 040ª | | | DG1-07. | RI BA | | |
| 0400 | | | | , IVII | | |
| 0430 | | FAI | DGI=DGI+DGP | | | |
| 0437 | | 201 | UNIINUE UDITECCE | | The state of the s | |
| 0438 | | r on | WHILE(0,000 | PIS, NEWPIS, UNCI | , IU-I | |
| 0439 | | 900 | FURMAT(//80 | (IH*)//5X, IUIAL | INVESTMENT IN PRODUCTION | UN, STURAGE AND T |
| | | | INANSMISSION | GAPACITY'//15X,' | ANNUALIZED COST PIS=', | 10.2/10X, TUTAL |
| | | | 2 DISCUUNTED | UUSI NEWPIS=', FI | 0.2///5X, PURCHASES, P | RUDUCTION AND ST |
| | | | JURAGE UPERA | TING CUSIS OMCI=' | , FID. 2//DX, TUTAL ANNU | al gas demand (m |
| 0440 | | | ANUF DGI'=', | r13.2//80(1H*)/// | J | |
| 0440 | | | RETURN | | | |
| V~~1 | | | END | | | |

| FORTRAN | (IV | G1 | RELEASE | 2.0 | LOAD | DATE = 8 | 10267 | 17/05/08 |
|---------|------|----|----------|----------------------------------|--|---|-----------------------|--------------------|
| 0001 | | | | SUBROUTINE 1PR, PC, PI, EL | LOAD(BLR, BL R, ELC, ELI, DGMR, D | C, BLI, SLR, SLC, S CMC, DGMI, DGMT, D | LI, RMR, RMC, DCT) | RMI, DDM, PAVG, |
| 0002 | | | | IMPLICIT RE | AL*8(A-H, O-Z) | | | |
| 0003 | | | | DIMENSION D | DM(12), PR(12), PC | (12),PI(12),ELF | ((12), ELC(12 |),ELI(12) |
| 0004 | | | | CIMENSION D | GMRO(12), DGMCO(1 | 2),DGMIO(12),DC | MTO(12) | |
| 0005 | | | | DIMENSION D | GMR(12), DGMC(12) | , DGMI(12), DGMT(| 12) | |
| 0006 | | | | NM= 12 | | | | |
| 0007 | | | | WRITE(6, 1) | · · · · · · · · · · · · · · · · · · · | | | |
| 0008 | | | 1 | FORMAT(//20 11H=)//) | X, 'SUBROUTINE LO | ADCAS MARKET | CHARACTERIS | TICS'/20X,45(|
| 0009 | | | . | WRITE(6, 37) | | | | |
| 0010 | | | 37 | FURMAT(//// | IVX, CAS DEMAND | PATTERNS'//) | | |
| 0011 | | | 40 | HATTELO,40J | 17 1 73 A (17) 13 13 13 44 A 17 1 / | ******** | | |
| 0012 | | | 40 | FURMAT(///2 | X, BASE DEMAND (| rificity (77) | | |
| 0013 | | | | DCMCTO-0 | | | | |
| 0014 | | | | DGHGIU=0. | | | | |
| 0013 | | | | DOMITIO-0. | | | | |
| 0010 | | | | | NM | | | |
| 0010 | | | | DC 33 11-1, | DIDTCIDAUDW(IM)) | + ((DAVC / DD(THO) | **(-FIR(IM) |)) |
| 0010 | | | | DCMCO(IM) = (| BICTCICADDUC IU) | *((PAVC/PC(IM)) | **(-FIC(IM)) |)))) |
| 0019 | | | | PCMIO(IM) = (| RI [+SI [*DDM(IM)) | *((PAVC/PI(IM)) | **(-FII(IM)) |)) |
| 0020 | | | | $DCMTO(IM) = \Gamma$ | CMBO(IM) + DCMCO(I) | M + $DCMIO(IM)$ | | ,, |
| 0022 | | | | ECMBTO= DCME | TO+DCMRO(IM) | | | |
| 0023 | | | | DCMCTO=DGMC | TO+DGMCO(IM) | | | |
| 0024 | | | | DGMITO=DGMI | TO+DGMIO(IM) | | | |
| 0025 | | | | DCMTTO=DCMI | TO+DGMTO(IM) | | ÷ | |
| 0026 | | | | WRITE(6.38) | IM. DGMRO(IM). DG | MCO(IM), DGMIO(| (M). DGMTO(IM | D |
| 0027 | | | 38 | FORMAT(' | MONTH=', 14, ' E | GMRO=', F10.2,' | DCMCO= ', F | 10.2,' DGMI |
| 6628 | | | 44 | CONTINUE | A, Donio- , Fie.2 | , | | |
| 6620 | | | 00 | LANTE(A 30) | DOMBTO DOMOTO D | CMITO DOMITO | | |
| 0027 | | | 30 | FORMAT(//3) | TOTAL 7V | DOMRTO: FII 9 | ' DOMOTO= | , FQ 2 28 'DC |
| 0000 | | | | 1MITO=', F10. | 2, ' DCMTTO', F | 11.2//) | Dururo | yr 7 Ym ymri y 200 |
| 0032 | | | 34 | FORMATC | 22X FORFCASTED | DEMAND (MMCF) ' | 1/1 | |
| 0033 | | | | DCMBT=0 | | | | |
| 0034 | | | | DGMCT=0. | | | | |
| 0035 | | | | DGMIT=0. | | | | |
| 0036 | | | | DCMTT=0. | | | | |
| 0037 | | | | DO 64 IM=1. | NM | | | |
| 0038 | | | | DGMR(IM) = (I) | .+RMR) *DCMRO(IM) | | | |
| 0039 | | | | DCMC(IM) = (1 | .+RMC) *DGMCO(IM) | | | |
| 0040 | | | | DGMI(IM) = (1 | .+RMI) *DGMIO(IM) | | | |
| 0041 | | | | DGMT(IM) = D(| GMR(IM)+DGMC(IM)+ | DGMI(IM) | | |
| 0042 | | | | DCMRT=DCMR1 | F+DGMR(IM) | | | |
| 0043 | | | | DGMCT=DGMC1 | F+DGMC(IM) | | | |
| 0044 | | | | DGMIT=DGMI7 | C+DGMI(IM) | | | |
| 0045 | | | | DGMTT= DGMT | T+DGMT(IM) | | | |
| 0046 | | | | WRITE(6,32) | IM, DGMR(IM), DGN | IC(IM), DGMI(IM). | , DGMT(IM) | |
| 0047 | | | 32 | FORMAT(' $1 = ', F10.2, 4$ | MONTH=', 14, ' 1 X, 'DCMT =', F10.2 | GMR =', F10.2,' | DGMC = ', F | '10.2,' DGMI |
| 0048 | | | 64 | CONTINUE | | | | |
| 0049 | | | | WRITE(6,42) | DGMRT, DGMCT, DCM | IT, DCMTT | | |
| 0020 | | | 42 | FORMAT($//3$) 1MIT =', F10. | (, TOTAL', 7X,' 2,' DGMTT', F | DGMRT =', F11.2 '11.2//) | , ' DGMCT' = | ',F9.2,2X,'DG |
| 0051 | | | | DDGT=RMR*D(| GMRTO+RMC*DGMCTÓ+ | RMI*DGMITO | | |
| 0052 | | | | WRITE(6,31) | DDGT | | | |

| FORTRAN | IV | G1 | RELEASE | 2.0 | LOA | D | DAT | E = 802 | 67 | 17/05/08 |
|----------------------|----|----|---------|--------------------------------|---------|--------|-----------|----------|--------------------------|----------|
| 0053 0054 0055 | | | 31 | FORMAT(//10X, RETURN END | ' TOTAL | DEMAND | INCREMENT | (MMCF) ' | ,F _. 10.2///) | |

| FORTRAN | I٧ | G1 | RELEASE | 2.0 | DIST | DATE = 80267 | 17/05/08 |
|-------------|----|----|---------------|----------------------|---------------------------|---------------------------------------|--------------------|
| 0001 | | | | SUBROUTINE DIST | (DGMT. IMP. PEAK. CMP | T2.NPT2.CMPD.NPD) | |
| 0002 | | | | IMPLICIT REAL*8 | (A-H, O-Z) | , , , | |
| 0003 | | | | DIMENSION DCMT(| 12) | | |
| 0004 | | | | REAL NPT2, NPD | | | |
| 0005 | | | | WRITE(6,7) | | | |
| 0006 | | | 7 | FORMAT(1H1,40X, | OUTPUT OF SUBROUT | INE DIST'/40X,25(| 1日=)////) |
| 0007 | | | | CRF=0.1241 | | | |
| | | | C | DETERMINATION O | F THE PEAK MONTH (| IMP) | |
| 0008 | | | | PEAK=0. | | | |
| 0009 | | | | DO 1 IM=1,12 | | | |
| 0010 | | | | IF (PEAK. GT. DGMT | (IM)) CO TO 1 | | |
| 0011 | | | | PFAK= DCMT(I M) | | | |
| 0012 | | | | IMP = IM | | | |
| 0013 | | | 1 | CONTINUE | | | |
| | | | C | CALCULATION OF | FINAL LOAD-RELATED | TRANSMISSION PLA | NT PT2 |
| 0014 | | | | P120=88500. | | | |
| 0015 | | | | IF(PEAK.GT.PT20 |) GO TO 2 | | |
| 0016 | | | | CMPT2=0. | | | |
| 0017 | | | | NPT2=0. | | | |
| 0018 | | | - | GO TO 3 | | | |
| 0019 | | | 2 | CNPT2=216.30822 | | | |
| 0020 | | | - | NFT2=216.30822* | (PEAK-PT20)/CRF | · · · · · · · · · · · · · · · · · · · | |
| | | | C | CALCULATION OF | FINAL LOAD-RELATED | DISTRIBUTION PLA | NT PD |
| 0021 | | | 3 | PL/0=59081. | | | |
| 0022 | | | | IF(PEAK. GT. PDO) | CO TO 4 | | |
| 0023 | | | | CFIPD=0. | | | |
| 0024 | | | | NED=0. | | | |
| 0025 | | | | | | | |
| 0026 | | | 4 | CIPD=1954.964 | DAR DDAL (ODD | | |
| 0027 | | | HIP. | NPD=1954.964*(P | EAK-PDO)/CRF | | |
| 0028 | | | 9 | WALLE(6,6) IMP, | PEAK, CMP12, NP12, CM | IPD, NPD | |
| 0029 | | | ¹⁰ | FURMATUZIOX, PE | AK MUNTH=', | ANOMICOTON MADOIN | |
| | | | | 114, 3X, PEAK LUA | D^{-1} , F10.2//10X, 1H | ANSMISSION MARGIN | AL $CUS1=7,F12.3,$ |
| | | | | 201, NEW TRANSMI | SSIUN PLANTE', F15. | Z//IVX, DISTRIBUT | ION MARGINAL COS |
| 0000 | | | | JI- , FIZ. J, JX, 'N | LW DISTRIBUTION PL | ANI=', FID. 2/////) | |
| 0030 | | | | | | | |
| 0031 | | | | £-1.17 | | | |
| | | | | | | | |

| FORTRAN | IV | Gl | RELEASE | 2.0 | EVAL 1 | DATE = | 80267 | 17/05/08 |
|--------------|----|-------|---------|-------------------------|---|--------------------------------------|-----------------------------|--|
| 0001 | | | | SUBROUTI 1DCT, X. NE | NE EVAL1(IMP, PEAK, D TPIS, ALLROR, CRS, CCS | CMT, ELR, ELC, E , CIS, CRST, CCS | LI, DGMR, DG T.CIST, CS1 | MC,DGMI,PAVG, (,PS,TS) |
| 0002 | | | | IMPLICIT | ' REAL*8(A-H, 0-Z) | | | |
| 0003 | | | | DIMENSIO | N DGMR (12) , DGMC (12) | DGMI(12).DGM | IT(12).ELR(| 12), ELC(12), |
| | | | | 1ELI(12). | CRS(12), CCS(12), CIS | (12) | | • |
| 0004 | | | | REAL NET | PIS | | | |
| 0005 | | | | NM= 12 | | | | |
| 0006 | | | | FL=DGT/(| 12.*PEAK) | | | |
| 0007 | | | | WRITE(6. | 10) | | | |
| 0008 | | | 10 | FORMAT(1 | H1,40X,'OUTPUT OF S | UBROUTINE EVA | L1'/40X,26 | (1日=)////) |
| 0009 | | | | WRITE(6, | 1) IMP, PEAK, FL, DGT | | | |
| 0010 | | | 1 | FORMAT(/ | ///10X,'GAS CONSUMP | TION EVALUATI | ON CRITERI | A'/10X,35(1H*)// |
| | | | | 1/29X,'PE 20R-',F8. | AK MONTH=', 13/20X,' 4/20X,'TOTAL GAS CO | PEAK LOAD (MP NSUMPTION=',F | ICF)=',F10. 12.2////) | 2/20X, 'LOAD FACT |
| 0011 | | | | WRITE(6, | 2) PAVG, X | | | |
| 0012 | | | 2 | FORMAT(2 | ØX, 'AVERACE VOLUMET | RIC RATE=', F1 | 0.3/20X, 'A | CHIEVED GAS SALE |
| | | | | 1S REVENU | E=',F15.2///) | | | |
| 0013 | | | | WRITE(6, | 3) | | | |
| 0014 | | | 3 | FORMAT(/ | <pre>//10X,'EFFICIENCY C</pre> | RITERIA'/10X, | 19(1田*)/// | () |
| 0015 | | | | CRST=0. | | | | |
| 0016 | | | | CCST=0. | | | | |
| 0017 | | | | CIST=0. | | | | • |
| 0018 | | | | DO 5 IM= | 1, NM | | | |
| 0019 | | | | E1 = 1./(1 | + ELR(IM) | | | |
| 0020 | | | | E2=1./(1 | .+ELC(IM)) | | | |
| 0021 | | | | E3-1./(1 | ·+ELI(IM) | | | |
| 0022 | | | | F1=1./E1 | | | | |
| 0023 | | | | F2=1./E2 | | | | |
| 0024 | | | | F3=1./E3 | | TTO THE SALE | (10000 www | |
| 0020 | | | | URS(IM) = | DOMA(IM) *(PAVG**(-E | LR(IM)))*EI*(| (10000.** | $(\mathbf{PAVG} \times \mathbf{F} \mathbf{I})$ |
| 0020 | | | | CLOCIND = | DGMU(IM)*(PAVG**(-E | しし(1m)))本世名本(!(1w)))+での女(| (10000.** | 2)-(PAVG**F2)) |
| 0021 | | | | | DGHI(IH) A(FAVGAA(-E | LICIMITATOR | (10000.**1 | 3)~(FAVG**F3)) |
| 0020 | | | | COUT-COS | | | | |
| 0027 AA2A | | | | | TTCIS(IM) | | | |
| 6671 | | | | WRITE(6 | A) IN CRS(IN) COS(I | M) CIS(IM) | | |
| 0031 | | | A. | FORMAT(3 | X MONTH=' 13.3X 'R | FSIDENTIAL SI | TRPLUS=' FI | 5 6 3X COMMERCE |
| 0004 | | | -15 | 1AL SURPI | IIS = ' F 15 0 3Y ' INDI | STRIAL SURPLI | IS=' F15 (A) | done, on, dominitar |
| 0033 | | | 5 | CONTINUE | 100- ,110.0,0A, 1800 | DIMAL SOLULO | /0- ir 10.0/ | · · · · · · · · · · · · · · · · · · · |
| 0000 | | | | CST=CRST | +CCST+CIST | | | |
| 0035 | | | | $PS = \Delta LLBC$ | BENETPIS | | | |
| 0036 | | · · · | • | TS=CST+F | S | | | |
| 0037 | | | | WRITE(6. | 6) CRST.CCST.CIST.C | ST | | |
| 0038 | | | 6 | FORMAT | //3X. 'TOTAL RESIDEN | TIAL SURPLUS' | '.F15.0/3X. | 'TOTAL COMMERCIA |
| | | | Ū | 1L SURPLU 2SUMER SI | IS', F15.0/3X, 'TOTAL IBPLUS', F15.0//) | INDUSTRIAL SU | JRPLUS', F1 | 5.0/3X, 'TOTAL CON |
| 0039 | | | | WRITE(6 | 7) PS. TS | | | |
| 0040 | | | 7 | FORMAT | VISX. PRODUCER SURP | LUS'. F15.0//? | X. 'TOTAL | SURPLUS', F15,0//) |
| 0041 | | | u u | RETURN | | | | |
| 0042 | | | | END | | | | |

| FORTRAN | IV G1 | RELEASE | 2.0 | EVAL2 | DATE = | 80267 | 17/05/08 |
|--------------|-------|---------|------------------------------------|--|----------------------------------|------------------------------------|-------------------------------|
| 0001 | | | SUBROUTINE EV 1 DGMI, DGMT, DGT | AL2(IMP, PEAK, BLR C, PR, PC, PI, ELR, EL | , BLC, BLI, SLI C, ELI, PAVGE | R, SLC, SLI, DD , XE, NETPIS, A | M, DCMR, DGMC, LLROR, CRS, |
| 0000 | | • | IMPIICIT REAL | $x_{8}(A-H) = 7$ | , 18, I AVG, 100 | (, IURC, IUII) | |
| 0003 | | | DIFENSION DDA | f(12), DGMR(12), DG | MC(12), DGMI | (12), DGMT(12 |), PR(12), PC(12 |
| 0004 | | | DFAI NETPIQ | | 147,000(12) | , 448(12), 418 | |
| 0005 | | | NM= 12 | | | | |
| 0006 | | | FL=DGT/(12.*F | PEAK) | | | |
| 0007 | | | WRITE(6.10) | | | | |
| 0008 | | 10 | FOFMAT(1H1,40 | X, OUTPUT OF SUB | ROUTINE EVAL | L2'/40X,26(1 | H=)////) |
| 0009 | | | WRITE(6,1) IN | IP, PEAK, FL, DGT | | | |
| 0010 | | 1 | FOFEIAT(////10 | X, GAS CONSUMPTI | ON EVALUATIO | ON CRITERIA' | /10X,35(1H*)// |
| | | | 1/20X, 'PEAK MC | DNTH=', I3/20X, 'PE | AK LOAD (MM | CF)=',F10.2/ | 20X, 'LOAD FACT |
| | | | 20R=', F8.4/20) | K, TOTAL GAS CONS | UMPTION=', F | 12.2////) | |
| 0011 | - 1 | | WRITE $(6,2)$ PA | VGE, XE | | | |
| 0012 | | 2 | FORMATCZOX, | HEURETICAL EQUIL | IBRIUM VULU | METRIC RATES | = ', F10.3/ |
| 0010 | | | 120A, 'EQUILIBE | CIUM GAS SALES RE | VENUE REQUI | REMENT= ', FID | .2////> |
| 0013 | | 9 | FORMAT(///IGS | VIEFFICIENCY CDI | TEDIAY / IAV | 10(104)///> | |
| 0014 | | 3 | rurnai(/// iv/ | , crficienci chi | ILNIA / IVA, | 19(114)///) | |
| 0013 | | | CCST=0. | | | | |
| 0010 | | | CIST=0. | | | | |
| 0018 | | | DO 5 IM=1.NM | | | | |
| 0019 | | | $E_{1=1}/(1+E_{1})$ | R(IM)) | | | |
| 0020 | | | E2=1./(1.+ELC) | C(IM)) | | | |
| 0021 | | | E3=1./(1.+EL) | (IM)) | | | |
| 0022 | | | F1=1./E1 | | | | |
| 0023 | | | F2=1./E2 | | | | |
| 0024 | | | F3=1./E3 | | | | |
| 0025 | | | CRS(IM) = (BLR+ | SLR*DDM(IM))*(PA | VC**(-ELR(I | MD))*E1*((1€ | 000.**F1)- |
| | | | 1(PR(IM) * * F1)) | *(1.+RMR) | | | |
| 0026 | | | CCS(IM)=(BLC+ 1(PC(IM)**F2)) | +SLC*DDM(IM))*(PA)*(1.+RMC) | VG**(-ELC(I) | MD))*E2*((16 | 000.**F2)- |
| 0027 | | | CIS(IM)=(BLI+ 1(PI(IM)**F3)) | +SLI*DDM(IM))*(PA)*(1.+RMI) | VG**(-ELI(I | M)))*E3*((10 | /000.**F3)- |
| 0028 | | | CRST=CRST+CRS | S(IM) | | | |
| 0029 | | | CCST=CCST+CCS | S(IM) | | | |
| 0030 | | | CIST=CIST+CIS | S(IM) | | | |
| 0031 | | | WRITE $(6, 4)$ II | I, CRS(IM), CCS(IM) | ,CIS(IM) | | |
| 0032 | | 4 | FOPMAT(3X, 'MO 1AL SURPLUS=' | DNTH=', 13, 3X, 'RES , F15.0, 3X, 'INDUST | TDENTIAL SU RIAL SURPLU | RPLUS=', F15. S=', F15.0) | Ø, 3X, COMMERCI |
| 0033 | | 5 | CONTINUE | | | | |
| 0034 | | | CST=CRST+CCS | T+CIST | | | |
| | | C | CALCULATION (| DF ACTUAL GAS SAL | ES REVENUES | XA | |
| 0035 | | | XA=0. | | | | |
| 0030 | | | DU & IM=I, NM | | | MT/ 1M3 - DT/ 11 | |
| 0037 | | | CONTINUE | m)*PR(Im)+DGnG(Ir | J*PG(IM)*DG | | 1) |
| 00000 | | Q | DF=YA-YF | | | | |
| 00007 | | | WRITE(6 9) X | A DF | | | |
| 0040 | | 9 | FORMAT(/5X. ' | ACTUAL CAS SALES | REVENUES= ' | F15.2//5X.'(| AS SALES REVEN |
| 00 × 1 | | , | IUE SURPLUS (- | +) OR DEFICIT (-) | =',F15.2/// |) | The states and VEAL |
| 0074 AAA9 | | | TS=ALLRUMANE | 11 13TUF 40.0160 | | | |
| agaa | | | WRITE (6 6) CI | BOT COST CIST CON | • | | |
| 0045 | | 6 | FORMATI | TOTAL RESIDENT | AL SURPLUS | F15.0/5X." | MAL COMMERCIA |
| JUTU | | v | 1L SURPLUS', F | 15.0/5X, 'TOTAL IN | DUSTRIAL SU | RPLUS', F15. | >/5X, 'TOTAL CON |

FORTRAN IV C1 RELEASE 2.0

EVAL2

DATE = 80267

| 1 | 71 | 05 | /08 | |
|---|----|----|-----|--|
|---|----|----|-----|--|

2SUMER SURPLUS', F15.0//) WRITE(6,7) PS.TS 7 FORMAT(///5X, 'PRODUCER SURPLUS', F15.0//5X, 'TOTAL SURPLUS', F15.0//) REJURN END

| FORTRAN | IV | G1 | RELEASE | 2.0 | REVREQ | DATE = 84 | 0267 17/05/08 |
|--------------|----|----|------------|---|---------------------|----------------------|---|
| 0001 | | | | SUBROUTINE REV | REQ(ALLROR, | NEWPIS, DGT, OMC1, X | , PAVG, NETP IS) |
| 0002 | | | | IMPLICIT REAL* | 8(A-H, 0-Z) | | |
| 0003 | | | | REAL NEWPIS, NE | TPIS, INVIXG | | |
| 0004 | | | 9 | WRIIE(0,2) FORMAT(////AQV | | SUPPORTINE BEVRE | 01/408 97(1H=)////) |
| 0005 | | | 6 2 | ATP IS=0.03625 | , control of | SODIDOTTINE IESVIE | |
| 0000 | | | | PIS8EG=6173385 | 11. | | |
| 0008 | | | | DEPAVG=0.02939 | | | |
| 0009 | | | | TAP00=22469051 | 9. | | |
| 0010 | | | | APDF=0.82528 | | | |
| 0011 | | | | REPPIS=ATPIS*P | ISBEG | 10 | |
| 0012 | | | | TUTPIS=PISBEG+ | REPPIS+NEWP | 18 | |
| 0013 | | | | DEF CAF - DEFAVG + TAPD + APD + TAPD + TAPD + APD + | IVIEIS Frafferve | | |
| 0015 | | | | NETPIS=TOTPIS- | TAPD | | |
| 0016 | | | | WRITE(6.3) REP | PIS. TOTPIS. | DEPEXP, TAPD, NETPI | S |
| 0017 | | | 3 | FORMAT(//10X,' | REPPIS=', FI | 5.3/10X, 'TOTP IS=' | ,F15.3/10X,'DEPEXP=',F15 |
| | | | | 1.3/10X, 'TAPD=' | ,F15.3/10X, | 'NETPIS=', F15.3// | (Z) |
| 0018 | | | | REVIXR=0.04145 | 4 | | |
| 0019 | | | | A3=0.002288 | | | |
| 0020 | | | | A4=9.002975 | | | |
| 0021 | | | | CON2=209.48495 | | | |
| 0022 | | | | PRPTXR=0.021 | | | |
| 0023 | | | | FEDITE-0 44 | | | |
| 0024 | | | | reprin-0.40 | | | |
| 0020 | | | | AG=0.01750 | | | |
| 0020 | | | | A7 = 9.1 | | | |
| 0028 | | | | OOP SEV= A3*TOTP | IS | | |
| 0029 | | | | ONUINC=A4*TOTP | is | | |
| 0030 | | | | OMC 2= DGT*COM2 | | | |
| 0031 | | | | ACOPEX=OMC1+OM | C2+DEPEXP | | |
| 0032 | | | | PRP TAX= PRPTXR* | NETP IS | | |
| 0033 | | | | ΡΑΥΓΑΧ= ΡΑΥΤΧR * | OMC2 | | |
| 0034 | | | | INVTXC=A7*(NEW | PIS+REPPIS) | | TT THAT THE A ST T BY STATUTE |
| 0033 | | | 4 | FORMATIC CLOSE | MEV, UNUINC, | UMC2, ACUPEA, PRPTA | ELE OLION COMODEL FIE O |
| 6630 | | | eff | 1/16V 'ACOPEV-' | FIS 9/16V | PEDTAV- FIS 9/1 | , r_{10} , σ_{10} , σ_{10} , r_{10} , σ_{10} , $\sigma_$ |
| | | | | 2^{1} INVTXC=' F15 | 3//) | 110 IAA- , F10.0/1 | VA, IATIAN- , FIG. OF IVA, |
| 0037 | | | | $X0 = 15 \times 10^{-1}$ | 6*TOTPIS | | |
| 0038 | | | | X1=ALLBOR*NETP | IS | | |
| 0039 | | | | X2=FEDITR*X0+I | NVTXC | | |
| 0040 | | | | X3 = (X1 - X2) / (1) | -FEDITR) | | |
| 0041 | | | | X4= ACOPEX+PRP1 | ΆΧ+ΡΑΥΤΑΧ | | |
| 0042 | | | | X5 = (X3 + X4) / (1. | -REVTXR) | | |
| 0043 | | | | X6 = OOPREV + ONUI | NC | 2 mm | |
| 0044 | | | | WRITE $(6,5)$ X0, | XI, X2, X3, X4 | x, X5, X6 | |
| 0040 | | | 5 | FURMATCIOX, XO | - ', F 15.3/ 10 | X, XI= ', FID. 3/10X | 5, 'X2=', F10.37 F15 0/10V 1V6-1 F15 0//) |
| DDAS | | | | 110A, AJ- , FIJ. | 3/104, 44- | , r 10.3/10A, AD- , | r 10.3/10A, A0- , r 10.3//) |
| 00-20 | | | С | X LE THE CAS F | EVENUE REOL | IREMENT | |
| 0047 | | | V 2 | PAVG=X/DGT | | · A A VERA REAL Y A | |
| | | | C | PAVE IS THE AV | ERAGE VOLUM | ETRIC GAS RATE | |
| 0048 | | | - | WRITE(6,1) NEW | PIS, DGT, OMO | 1, X, PAVG | |
| 0049 | | | 1 | FORMAT(//10X,' | NEWPIS=', FI | 5.3/10X, 'DGT=', F1 | 15.3/10X, 'OMC1=', F15.3/10 |
| | | | | 1X, 'X=', F15.3/1 | 0X, 'PAVC=', | F10.3///) | |
| 0050 | | | | RETURN | | | |
| 005 l | | | | END | | | |

APPENDIX F

SAMPLE OUTPUT OF THE GUMCP MODEL

The purpose of this appendix is to present a sample output of the GUMCP model developed and applied in chapter 4. This output includes (1) the basic data assumptions, (2) the results of the average cost pricing policy, and (3) the results of the first iteration of the marginal cost pricing equilibrating procedure.

SUMMARY OF BASIC DATA ASSUMPTIONS

RATES OF MARKET GROWTH RESIDENTIAL= 0.50 COMMERCIAL= 0.50 INDUSTRIAL= 0.50

MONTHLY DEMAND ELASTICITIES

| RESIDENTIAL | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 |
|-------------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|
| LOMMERCIAL | -9.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 |
| INDUSTRIAL | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 | -0.50 |

BASE- AND SPACE-HEATING LOAD COEFFICIENTS

| RESIDENTIAL | 3203.742 | 23.912 |
|-------------|-----------|--------|
| COMMERCIAL | 1516.625 | 9.104 |
| INDUSTRIAL | 10179.266 | 3.567 |

MEAN DECREE DAYS DATA

AVC. ANNUAL TOTAL DEGREE DAYS= 6258.9

MONTHLY DEGREE DAYS= 506.6 248.2 50.5 11.0 18.9 120.5 371.6 712.6 1071.6 1207.7 1046.3 892.5

SUPPLY CHARACTERISTICS

| SUP1T= | 200000. |
|---------|----------|
| SUP2T= | 500000. |
| SUPWHT= | 2000. |
| SUPFLT= | 2500. |
| CC1= | 1202.400 |
| DC1= | 980.000 |
| WRC= | 8.075 |
| CC2= | 1009.200 |
| DC2= | 1860.000 |
| KMIN= | 0.750 |
| CWII= | 787.000 |
| CFL= | 1481.000 |

MONTH= MONTH= 15317.56 9138.70 D(;MCO= D(;MCO= 6128.71 3776.24 DCM10= 11986.31 DCM10= 11064.60 DGMTO= 33432.59 DGMTO= 23979.53 DGMRO= 1 DGMRO= 2

BASE DEMAND (MMCF)

GAS DEMAND PATTERNS

SUBROUTINE LOAD---GAS MARKET CHARACTERISTICS

BASE AND AVERAGE COST PRICING ANALYSIS

RATE OF RETURN ALLROR= 0.121

TRANSMISSION INVESTMENT CIPTI= 232.040

CIST= 50.000

| DSTCM= | 100000. |
|--------|---------|
| CS= | 33.230 |
| OIOT- | E0 000 |

STORACE CHARACTERISTICS

| DPROM= | 3000. |
|--------|-----------|
| COMP= | 921.129 |
| CIP= | 14398.109 |
| SHP= | 0.100 |

PRODUCTION CHARACTERISTICS

| TOTAL | | DGMRTO= | 188086.22 | DGMCTO= | 75172.34 | DCM1TO= | 144473.48 | DCMTTO | 407732.04 | |
|---------|----|---------|-----------|---------|----------|---------|-----------|---------|-----------|--|
| MONTH= | 12 | DGMR0= | 24545.20 | DGMCO= | 9641.95 | DCMI0= | 13362.81 | DGMT0= | 47549.96 | |
| MONTH= | 11 | DGMRO= | 28222.87 | DCMCO= | 11042.14 | DGNI0= | 13911.42 | DGIIIO= | 53176.43 | |
| MON'TH= | 10 | DGMR0= | 32082.27 | DCMCO= | 12511.53 | DCH10= | 14487.13 | DCMIO= | 59080.92 | |
| MONTH= | 9 | DCMRO= | 28827.85 | DGMCO= | 11272.47 | DGM10= | 14001.66 | DGMTO= | 54101.98 | |
| MONTH= | 8 | DGMRO= | 20243.44 | DGMCO= | 8004.14 | DCMI0= | 12721.11 | DGMTO= | 40968.68 | |
| MONTH= | 7 | DGMRO= | 12089.44 | DCMCO= | 4899.67 | DCM10= | 11504.76 | DGMTO= | 28493.88 | |
| MONTH= | 6 | DCMRO= | 6085.14 | DGMCO= | 2613.66 | DCM10= | 10609.09 | DGMTO= | 19307.88 | |
| MONTH= | 5 | DGMRO= | 3655.68 | DCMCO= | 1688.69 | DCM10= | 10246.68 | DGMTO= | 15591.05 | |
| MONTH= | 4 | DCMRO= | 3466.77 | DGMCO= | 1616.77 | DCMIO= | 10218.50 | DGMTO= | 15302.05 | |
| MONTH= | 3 | DCMR0= | 4411.30 | DN;MCO= | 1976.38 | DCMI0= | 10359.40 | DGMIU= | 16747.07 | |

FORECASTED DEMAND (MMCF)

| TOTAL. | | DCMRT = | 282129.34 | DGMCT | =112758.50 | DCMIT = | 216710.22 | DCMIT | 611598.06 |
|----------------|----|------------|-----------|----------|--------------------|---------|-----------|-----------|-----------|
| MONTH= | 12 | DGMR = | 36817.81 | DGMC = | 14462.92 | DGrif = | 20094.22 | Derii = | (1324,94 |
| MONTH= | 11 | DCMR = | 42334.31 | DRIFIC = | 16563.21 | DGrii = | 20007.10 | DOMP - | 71204 04 |
| MONTH= | 10 | DGPUR = | 48123.40 | DUFFIC = | 18707.29 | DGHI - | 21100.10 | DONT - | 70764 68 |
| HOR THE | | 17671116 = | 90471.((| | 10700.01 | DOMI - | 01700 70 | DOM'N'L - | 88621 99 |
| MONTELL- | 0 | DOMD - | 40041 77 | IM'MC - | 16048 71 | IMIMI = | 21002 50 | DCMT = | 81152.97 |
| MANTH | à | DCMR = | 30365 16 | DCMC = | 12006.20 | DGMT = | 19081.67 | DCMT = | 61453.03 |
| MONTH= | 7 | DGMR = | 18134.17 | DCMC = | 7349.51 | DCMI = | 17257.14 | DCMT = | 42740.82 |
| MONTH= | 6 | DGMR = | 9127.71 | DGMC = | 3920.49 | DGMI = | 15913.63 | DGMT = | 28961.83 |
| MONTH= | 5 | DGMR = | 5483.52 | DGMC = | 2533.04 | DGPUI = | 15370.02 | DGPLE = | 23386.38 |
| MONTH= | 4 | DGMR = | 5200.16 | DUMC = | 2420.10 | DOLL = | 10046.60 | 17(7)11 ~ | 00006 80 |
| rion 111= | 3 | penic = | 0010.90 | DUNC = | 2704.07 0405 15 | DONL - | 15007.10 | INCREP - | 00050 07 |
| PROPERTY AND A | 6 | DOMD - | 10100.00 | DUMO - | 0064 57 | DOME = | 15539 10 | DOM'T = | 25120 61 |
| MON'TH= | ō | DCMR = | 13708 05 | DCMC = | 5664.36 | DGML = | 16596.89 | DGMT = | 35969.30 |
| MONTH= | 1 | DGMR = | 22976.35 | DCMC = | 9193.07 | DGMI = | 17979.46 | DGMT = | 50148.88 |

TOTAL DEMAND INCREMENT (MMCF) 203866.02

OUTPUT FROM SUBROUTINE MARCOS

EXISTING MONTHLY PRODUCTION CAPACITY PROC= 947.667 EXISTING MONTHLY TRANSMISSION CAPACITY PTIO= 55000.000

OPTIMAL SOLUTION

| OBJ | -69 | 326310 | 0.7 I | STATE | 4 I | TERATIO | ons 9 | 6 DETERMINA | INT · | -7.85810 IN | IFEAS 0 | NINTO | Ø NOU | TOF 0 | |
|------|--------|----------|--------------|--------|--------------|------------|----------------|--------------------|----------------------------|-----------------------------|--------------|----------------------|--------------------|--------------------------------|---------------|
| N | 79 NC | 0L 67 | M 13 | 9 MNO | W 139 | ISFEA | 5 1 | IRCNT 1 | | | | | | | |
| v w | COTOR | | | | | | | | | | | | | | |
| V A1 | 56101 | 2617 | .74557 | 4255 | . 12929 | 15103 | .81867 | 18255.75102 | 16837.85335 | 15530.08174 | 13858.22 | 011 | 0.0 | 0.0 | 0.0 |
| | | 0 | . 0 | 0 | . 0 | 0 | . 0 | 9.0 | 0.0 | 0.0 | 0.0 | | 0.0 | 0.0 | 9053.41412 |
| | | 24381 | .54804 | 20664 | . 69960 | 17514 | .40663 | 14844.47135 | 12542.19379 | 9.0 | 0.0 | 98 211 9259 | 4.38942 | 0.0 | 4267.47838 |
| | | 36525 | .54611 | 36525 | .0 .54611 | 37343 | .01322 | 48709.72814 | 48700.72814 | 48700.72814 | 48700.72 | 814 4878 | 0.72814 | 36525.54611 | 36525.54611 |
| | | 36525 | .54611 | 36525. | 54611 | 36525 | .54611 | 36525.54611 | 37343.08263 | 48700.72814 | 48700.72 | 814 4870 | 0.72814 | 48700.72814 | 48700.72814 |
| | | 15557 | .07370 | 48700 | .72814 | :: 900 | . 00000 | 9.0 | 1698.88390 | 1698.88390 | 1698.88 | 390 169 | 8.88390 | 1698.88390 | 1698,88390 |
| | | 1698 | . 88390 | 1698 | . 88390 | 1698 | . 88390 | 1698.88396 | 1698.88390 | 1698.88390 | 751.21 | 71510000 | 0.00000 | 12956.68574 | |
| SLA | CK VEC | TOR | | | | | | | | | | | | | |
| | | 17344 | .90332 | 15504 | . 20310 | 4325 | . 02401 | 0.0 | 0.0 | 0.0 | 465.66 | 273 1324 | 7.53501 | 13950.70054 | 15844.37996 |
| | | 17449 | .37717 | 18809 | .70036 | 12581 | . 50388 | 12980.56646 | 13629.24023 | 15931.74394 | 18714.74 | 442 2128 | 1.59329 | 23649.07873 | 16708.28288 |
| | | 15655 | .07179 | 24168 | 48501 | 901.1P | . 0 63306 | 0.0 69154 73355 | 90090,92097 86660 90018 | 94040.79000 101513 67153 | 96.17 74 | 0991 0120 557 687 | 2 17485 | - 94443.37304 - 91976 69352 | 40232 44454 |
| | | 57070 | .29789 | 72600 | 37963 | 86458 | . 59974 | 77405.18562 | 53023.63758 | 32358.93798 | 14844.47 | 135 | 0.0 | 3014.87991 | 15557.07370 |
| | | 15557 | .07370 | 14572 | 68428 | 15557 | .07370 | 11289.59532 | 0.0 | 15557.07370 | 11185.26 | 048 | 0.0 | 5705.50671 | 11476.21292 |
| | | 12175 | . 18204 | 12175 | . 18204 | 12175 | . 18204 | 12175.18204 | 12175.18204 | 12175.18204 | 11357.64 | 552 | 0.0 | 0.0 | 0.0 |
| | | 19175 | . 9 | 12175 | . 0 19704 | 10175 | . 0 18364 | 12175 19204 | 19.19 1. 10 0 | 9.0 | 9.9 A A | | 4.4 A A | 817.53652 | 12175.10204 |
| | | 12110 | . 0 | 0. | . 0 | 9 | . 10202 . Ø | 9.0 | 0.0 | 0.0 | 132788.55 | AA 1 | 0.0 | 0.0 | 2300.00000 |
| | | ø | .0 | ø. | .0 | Ø. | .0 | 0.0 | 0.0 | 0.0 | 0.0 | | 0.0 | 0.0 | 0.0 |
| | | 0 | .0 | 0. | .0 | 2248 | . 78285 | 0.0 | 0.0 | 15190.06195 | 27732.25 | 574 2773 | 2.25574 | 26747.86632 | 27732.25574 |
| | | 23464 | .77735 | 11357. | .64552 | 15557 | . 07370 | 11185.26048 | 6 0.0 0 0 | 5706.50671 | 11476.21 | 292 | 19.19 A 4 | 9.9 6 6 | 0.0 |
| | | v | . 0 | 0. | | U . | | 0.0 | 0.0 | 0.0 | 0.0 | | 0.0 | 0.0 | |
| DUAI | . VECT | DR | | | | | | | | | | | | | |
| | | 6 | .0 | 0. | .0 | 0. | .0 | 0.0000 | 68.41705 | 5.76133 | 0.0 | | 0.0 | 0.0 | 0.0 |
| | | 99 29 | 11759 | 763 | . 0 83947 | 171 | 74241 | 0.0 202 63279 | 0.0 | 0.0 | U.U G G | | v.v 0 0 | 0.0 | 0.0 |
| | | . Ő | .0 | | .0 | | .0 | 0.0 | 0.0 | 0.0 | 0.0 | | 0.0 | 0.0 | 0.0 |
| | | 0 | . 0 | 0. | .0 | 0 | . 0 | 0.0 | 0.0 | 0.0 | 0.0 | 106 | 3.43711 | 0.0 | 0.0 |
| | | 0 | .0 | 0. | .0 | 0 | . 0 | 0.0 | 5.76133 | 0.0 | 0.0 | 38 | 6.23865 | 0.0 | 0.0 |
| | | 61 | . U 19047 | 01 | 12067 | . 0 | 76199 | 0.0 | 0.0 | 0.0 | 9.0 60.06 | 489 Û | 6.40999 5 76199 | 91.13867 | 911.31134 |
| | | 91 | . 0 | | . 0 | ő. | . 6 | 9.0 | 1003.43862 | 1003.43862 | 1003.43 | 862 100 | 3.43862 | 940.33543 | 1003.43862 |
| | | 1009 | . 19995 | 1009 | 19995 | 1009 | 19995 | 1009.19995 | 1009.19995 | 1009.19995 | 0.0 | 19 | 8.96128 | 5773.51692 | 0.0 |
| | | 1134 | .11594 | 1134. | 11594 | 1134 | . 11594 | 1134.11594 | 1071.01274 | 1134.11594 | 1139.87 | 727 120 | 6.33726 | 1231.01594 | 1617.25459 |
| | | 1231 | . 01594 | 1231. | 01594 | 9. | . 0 | 852.84519 | 15.29877 | 0.0 | 0.0 | - 120 | 9.9 9.90000 | -1202 20000 | -1202 39999 |
| | | -1202 | . 39990 | -1139 | 29670 | - 1202 | 39998 | -1208.16123 | -1274.62122 | -1299.29990 | -1917.57 | 825 - 129 | 9.29990 | -1299.29990 | 1202.07770 |
| | | | | | | | | | | | | | | | |
| C083 | r vect | or or | 00000 | | 00000 | | 00000 | 00 00000 | 00 00000 | 00 00000 | 00 00 | 000 0 | 0 00000 | -00 00000 | -00 00000 |
| | | -33 | .23009 | | 23000 | ~33. | 23000 | -33.23000 | -33.23000 | -33.23000 | -33,23 | 000 -J 444 -3 | 3.23000 3.23000 | -33 23000 | -33,23000 |
| | | -33 | .23000 | -33 | 23000 | -33 | 23000 | -33.23000 | -1202.39990 | -1202.39990 | -1202.39 | 990 -120 | 2.39990 | -1202.39990 | -1202.39990 |
| | | -1202 | .39990 | -1299 | 29990 | -1299 | 29990 | -1299.29996 | -1299.29990 | -1299.29990 | 0.0 | | 0.0 | 0.0 | 0.0 |
| | | 0 | .0 | 0 | .0 | 0 | .0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | -1009.19995 | -1009.19995 |
| | | -1009 | . 19995 | -1009. | . 19995 | -1009 | . 19995 | -1009.19995 | -1009.19990 | -1009.19990 | -1009.19 | 990 -100 015 -02 | 9.19990 | -921 12915 | -921, 12915 |
| | | -921 | 12915 | -921 | . 12915 | -921 | . 12915 | -921.12915 | -921.12915 | -921.12915 | -14398.10 | 937 -5 | 0.00000 | -232.03970 | , at 1 av 1 a |
| | | | | | | | | | | | | | _ | | |
| BOUI | ID VEC | I'OR | 00000 | | 00000 | | 00000 | -1 00000 | - 1 00000 | - 1 00000 | _1 00 | 000 - | 1 00000 | -1 00000 | -1.00000 |
| | | - 1 | . 00000 | -1 | | -1 | . 800000 | -1.00000 | -1.00000 | -1.00000 | -1.00 | 000 - 000 - | 1.00000 | -1.00000 | -1.00000 |
| | | -1 | .00000 | -1. | 00000 | -1 | . 00000 | -1.00000 | -1.00000 | -1.00000 | -1.00 | 000 - | 1.00000 | -1.00000 | -1.00000 |
| | | - 1 | .00000 | -1. | . 00000 | - 1 | .00000 | -1.00000 | -1.00000 | -1.00000 | -1.00 | 000 - | 1.00000 | -1.00000 | -1.00000 |
| | | -1 | . 40000 | -1. | . 00000 | -1 | . 00000 | -1.00000 | -1.00000 | -1.00000 | -1.00 | 000 - 000 - | 1 00000 | -1.00000 | -1.00000 |
| | | - 1 | | - 1 - | | -1. | | -1.00000 | -1.00000 | -1.00000 | -1.00 | | 1.000000 | 1.00000 | 1100000 |

| | | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1 -1.00000 -1 | . 00900 . 00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 |
|-------------------|---|--|---|---|---|---|--|--|--|--|--|
| B VF | CTOR | $11899.99841 \\11899.99841 \\7500.00048 \\60513.63911 \\0.0 \\0.0 \\0.0 \\0.0 \\9.7 \\0.0 \\9.7 \\0.0 \\9.7 \\0.0 \\9.7 \\0.0 \\0.0 \\0.0 \\0.0 \\0.0 \\0.0 \\0.0 \\0$ | $11899.99841 \\ 11899.99841 \\ 7500.00048 \\ 60513.63911 \\ 0.0$ | $\begin{array}{c} 11899.99841\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 60513.63911\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 947.66675\\ 5000.00000\\ 5000.00000\\ 20961.82665\end{array}$ | $\begin{array}{c} 11899.99841 \\ 7509.00048 \\ 7509.00048 \\ 60513.63911 \\ 0.0 \\ 0.0 \\ 0.0 \\ 0.0 \\ 0.0 \\ 0.0 \\ 947.66675 \\ -20305.6067510 \\ 55009.00000 \\ 542740.82011 \\ 6\end{array}$ | $\begin{array}{c} 1899.99841\\ 7500.00048\\ 0513.63911\\ 0513.63911\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 947.66675\\ 0000.00000\\ 1453.02617 \end{array}$ | $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | . 99841 . 00048 . 63911 . 0 . 0 . 0 . 0 . 0 . 000000 . 66675 . 00000 . 00000 . 00000 . 00000 . 38334 | $\begin{array}{c} 11899 \cdot 99841 \\ 7500 \cdot 00048 \\ 60513 \cdot 63911 \\ 0 \cdot 0 \\ $ | $\begin{array}{c} 11899.99841\\ 7300.00048\\ 60513.63911\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 9.0\\ 2000.00000\\ 947.66675\\ 55600.00000\\ 947.66675\\ 71324.94418 \end{array}$ | $\begin{array}{c} 11899.99841\\ 7300.00048\\ 60513.63911\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 2500.00000\\ 947.66675\\ 55000.00000\\ 25120.61133 \end{array}$ |
| 8 I C N | VECT | OR | 1 44444 | 1 44444 | 1 00000 | 1 00000 | 1.00000 1 | . 00000 | 1.00000 | 1.00000 | 1.00000 |
| | | 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 0.00000 0.00000 0.00000 | $\begin{array}{c} 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 1.60000\\ 0.6000\\ 0.6000\\ 0.6000\\ 0.6000\\ 0.6000\\ 0.600\\$ | 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 0.000000 0.000000 0.000000 0.000000 0.00000000 | 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 0.0000 | 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 0.0 | 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 1.00000 1 0.0000 0 | .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 .00000 | $\begin{array}{c} 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 1.00000\\ 0.0000\\ 0.0\\ 0.$ | $\begin{array}{c} 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 1,00000\\ 0,0000\\ 0,0\\ 0,$ | 1.00000 1.00000 1.00009 1.00009 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 0.0000 0.0 |
| | CONST | RAINTS VALUES | S AND DUAL PI | NICES | | | | | | | |
| H H H H H H H H H | 1 2 3 4 5 6 7 | NUS= NUS= NUS= NUS= NUS= NUS= | -5444.90490 -3604.20469 7574.9744 11899.9984 11899.9984 |) B= B= B= B= B= | 1 1899 . 9984 1 1 1899 . 9984 1 | DUAL= DUAL= DUAL= DUAL= DUAL= | 9.9 9.9 9.9 9.9 9.90000 68.41793 | | | | |
| I = | 8 | NIS= NIS= | 11434.33569 |) B=) B= | 11899.99841 | DUAL= DUAL= DUAL= | 5.76133 0.0 0.0 | | | | |
| 1= | 8 9 10 | NIS= NIS= NIS= NIS= | 11434.33569 -1347.53660 -2050.70213 -3944.38150 | B= B= B= B= B= B= B= B= | 1 1899, 9984 1 1 1899, 9984 1 1 1899, 9984 1 1 1899, 9984 1 1 1899, 9984 1 | DUAL= DUAL= DUAL= DUAL= DUAL= | 5.76133 0.0 0.0 0.0 0.0 | | | | |
| | 8 9 10 11 12 13 14 15 15 | NIS= NIS= NIS= NIS= NIS= NIS= NIS= NIS= | $\begin{array}{c} 11634.33569\\ -1347.53669\\ -2050.70211\\ -3944.38151\\ -5549.37487\\ -5981.59346\\ -5480.56594\\ -5480.56594\\ -6129.23977\\ -8431.7434\\ -756\\ -7$ | b- b B= b B= b B= b B= b B= b B= b B= b B= | $\begin{array}{c} 11637,27641\\ 11639,998441\\ 11899,998441\\ 11899,998441\\ 11899,998441\\ 11899,998441\\ 11899,998441\\ 7500,000448\\ 7500,000448\\ 7500,00048\\ 7500,000468\\ 7500,000468\\ 7500,00048\\ $ | DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= BUAL= BUAL= DUAL= DUAL= | 5.76133 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0 | | | | |
| | 8 9 10 11 12 13 14 15 16 17 18 | NIS= NIS= NIS= NIS= NIS= NIS= NIS= NIS= | $\begin{array}{c} 11634: 3056\\ -1347: 5066\\ -2050: 70212\\ -3944: 3015\\ -5549: 3707\\ -5909: 7019\\ -5909: 7019\\ -5480: 56599\\ -6129: 23975\\ -8431: 7434\\ -11214: 7439\\ -13781: 5928\\ -1610: 97292\\ -100: 97292\\ -100: 9729$ | b = b = c = b = c = | $\begin{array}{c} 11099.97844\\ 11099.998441\\ 11899.998441\\ 11899.998441\\ 11899.998441\\ 11899.998441\\ 7500.600448\\ 7500.600448\\ 7500.600448\\ 7500.600448\\ 7500.600448\\ 7500.600448\\ 7500.600048\\ 75000000000000000000000000000000000000$ | DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= DUAL= | 5.76133 9.9 9.9 9.9 9.9 9.9 9.9 9.9 9.9 9.9 9 | | | | |
| | 8 9 10 11 12 13 14 15 16 17 18 19 20 21 22 24 | NIS= NIS= NIS= NIS= NIS= NIS= NIS= NIS= | $\begin{array}{c} 11634, 33569\\ -1347, 53669\\ -2050, 70212\\ -3944, 38152\\ -3944, 38152\\ -5549, 37187\\ -6909, 70199\\ -5081, 59344\\ -5480, 56591\\ -6129, 23977\\ -8431, 7434\\ -11214, 7439\\ -13781, 5928\\ -16149, 07782\\ -9208, 28244\\ 7560, 00044\\ 7560, 00044\\ 7560, 00044\\ 7500, 00046\\ -560, 00044\\ -5600, 00046\\ -5600, 0004\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600, 0000\\ -5600,$ | b b b B b B b B c | $\begin{array}{c} 11839,93841\\ 11839,93841\\ 11899,93841\\ 11899,93841\\ 11899,93841\\ 11899,93841\\ 11899,93841\\ 7500,00048\\ 75$ | DUAL= | $\begin{array}{c} 5.76133\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0$ | | | | |
| | 8 9 10 11 12 13 14 16 17 18 20 21 22 24 22 24 22 24 22 26 7 | NUS= NUS= NUS= NUS= NUS= NUS= NUS= NUS= | $\begin{array}{c} 11634: 3056\\ -1347: 5066\\ -2050: 70212\\ -3944: 3015\\ -3944: 3015\\ -5394: 3707\\ -59081: 5034\\ -6129: 23975\\ -6129: 23975\\ -6129: 23975\\ -8431: 7434\\ -6129: 23975\\ -8431: 7434\\ -11214: 7439\\ -11214: 7439\\ -113781: 5928\\ -16149: 07825\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92061: 20266\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -92062: 2026\\ -9206\\ -9206\\ -9206\\ -9206\\$ | b- b B= | $\begin{array}{c} 11899.99841\\ 11899.99841\\ 11899.99841\\ 11899.99841\\ 11899.99841\\ 11899.99841\\ 11899.99841\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 7500.00048\\ 60513.63911\\ 60513.6391\end{array}$ | BUAL: BUAL: | $\begin{array}{c} 5.76133\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0$ | | | | |
| T | 00 | 111117- | 16070 06847 | D - | | 60510 60011 | TYLAT - | 0 0 |
|------------|----------------|---------|---------------|------------|---|---------------------|----------|-------------|
| 1- | 29 | nua- | 10070.20347 | D- | | 00010.00711 | DOAL- | 0.0 |
| 1= | 30 | RHS= | 31600.34720 | B= | | 60513.63911 | DUAL= | 0.0 |
| I = | 31 | RHS= | 45458.56732 | B= | | 60513.63911 | DUAL= | 0.0 |
| Î | 44 | 005= | 26405 15210 | n= | | 60519 63911 | DUAL = | 0.0 |
| 1~ | 24 | nuio- | 30400.19319 | D- | | 60710.00711 | DUMI- | 0.0 |
| 1= | 33 | 1015= | 12023.60315 | 13= | | 00010.00911 | DUAL= | 0.0 |
| I = | 34 | NIS= | -8641.09445 | B= | | 60513.63911 | DUAL= | 0.0 |
| 1= | 95 | 11119= | -26155 56107 | B= | | 60513 63911 | DI(A) = | 4.4 |
| | 00 | 10103- | 41000 00010 | - D- | | 60010.00011 | DUAL - | 0.0 |
| 1= | 36 | 1015= | -41000.03242 | в- | | 00213.03311 | DOAL- | 0.0 |
| I= | 37 | 10HS= | -2617.74557 | B= | | 0.0 | DUAL= | 0.0 |
| Ιz | 28 | HHS= | -6879 87485 | B= | | 0.0 | DUAL= | 0.0 |
| * - | 00 | 111167- | 01076 60050 | - 11- | | 0.0 | INIAL - | 0.0 |
| 1- | 39 | nuis- | -21970.09332 | D | | 0.0 | 100/11/- | 0.0 |
| 1= | 40 | 101S = | -40232.44454 | B= | | Q.Q | DUAL= | 0.0 |
| 1= | 41 | RHS= | -57070.29789 | B= | | 0.0 | DUAL= | 0.0 |
| 1 = | 4.9 | 005= | -79600 97963 | B. | | A A | DHAL = | 61.64 |
| 1 - T - | 10 | 10103- | 06450 50074 | D | | 0.0 | DUAL - | Å Å |
| 1= | 43 | 1018= | -86458.39974 | D- | | 0.0 | DUAL- | 0.0 |
| 1= | 44 | NHS= | -77405.18562 | B= | | (). () | DUAL= | Q.Q |
| 1 = | 45 | RHS= | -53023.63758 | R= | | 0.0 | DUAL= | 0.0 |
| i - | 44 | 0.05- | -22250 02700 | N = | | 0 0 | DIIAL = | 0.0 |
| 1~ | 40 | 101.7~ | -32330.73170 | D- | | 0.0 | | 0.0 |
| 1= | 47 | RHS= | -14844.47135 | 13= | | 0.0 | DUAL= | 0.0 |
| 1= | 48 | RHS= | 0.0000 | B= | | 0.0 | DUAL= | 1063.43711 |
| Î = | 40 | DUS= | -9014 87991 | 8= | | 6 6 | DHAL = | 0.0 |
| R | | 10167- | 15555 07070 | 17 | | 0.0 | INITAL - | 0 0 |
| 1= | 26 | 1012= | -19994.04940 | 25 = | | 0.0 | DOAL- | 0.0 |
| I = | 51 | RHS = | -15357.07370 | B= | | 0.0 | DUAL= | 0.0 |
| 1 = | 52 | 1018= | -14572.68428 | B= | | 6.6 | DUAL= | 0.0 |
| ÷ | TO | 0067- | 15557 07070 | ñ- | | A A | DUAL - | 0 0 |
| 1- | 00 | 1415- | -10007.07070 | D- | | W . W | 10/11 | 0.0 |
| 1= | 54 | RHS= | -11289.59532 | В= | | (). () | DUAL= | 17.0 |
| I = | 55 | RHS= | 0.0000 | B= | | 0.0 | DUAL= | 5.76133 |
| 1 = | 56 | BUS= | -15557 47170 | H= | | 0.0 | DUAL= | 6 6 |
| | 50 | Turk - | 10001.01010 | 5 | | 0.0 | INTIAL - | Å Å |
| 1 = | 37 | iuis= | -11180.20040 | B= | | 0.0 | 100/41 | 0.0 |
| = | 58 | RHS= | 0.00000 | B= | | 0.0 | DUAL= | 386.23865 |
| 1= | 59 | HHS= | -5706.59671 | B= | | 61.61 | DUAL= | 0.0 |
| ÷ | ćó | DUG- | 11476 21202 | D- | | 0 0 | DUAL - | 4) 4) |
| 1~ | 00 | 1013- | -11470.21292 | D- | | 0.0 | DUAL- | 0.0 |
| 1= | 61 | 1018= | -12175.18204 | B= | | 9.9 | DUAL= | 0.0 |
| 1= | 62 | RHS= | -12175.18204 | B= | | 0.0 | DUAL= | 0.0 |
| Î = | 63 | nus= | -12175 18264 | R= | | 6 6 | DHAL= | 0.0 |
| 1 | 200 | 1010- | 10175 10004 | D= | | 0.0 | INTAL - | <u> </u> |
| 1 = | 64 | rus= | ~12175.18204 | B- | | 47.47 | DUAL- | 0.0 |
| I = | 65 | RHS= | -12175.18204 | 8= | | 0.0 | DUAL= | v.v |
| 1= | 66 | BHS= | -12175.18204 | B= | | 0.0 | DUAL= | 0.0 |
| ī | 67 | 0.05- | -11957 64559 | B - | | 6 6 | DHAL = | 64 64 |
| ÷ – | 20 | 1010- | 1100110400 | D- | • | A A | DUAL - | 46 4E000 |
| 1 = | 013 | 1015= | 0.00000 | B= | | 0.0 | DUAL- | 00.43977 |
| 1= | 69 | IHS= | 0.0000 | B= | | 0.0 | DUAL= | 91.13867 |
| 1= | 70 | BBS= | 0.0 | B= | | 0.0 | DUAL= | 477.37732 |
| л — | | 1111-1- | | ñ- | | 0 0 | DIAL = | 01 13367 |
| | 4.1 | 1013~ | -0.00000 | D | | 0.0 | DITAL - | 01 10047 |
| 1= | 72 | nus= | 0.00000 | 8= | | 0.0 | DOAL- | 91.13007 |
| I = | 73 | NIS= | -0.00000 | в= | | 0.0 | DUAL= | 5.76133 |
| 1= | 74 | BHS= | -0 00000 | B= | | 0.0 | DUAL= | 5.76133 |
| | | Duc- | 0.00000 | D- | | 0 0 | DUAL - | 5 76199 |
| s | 10 | 11175= | -0.00000 | D- | | 0.0 | DUAL- | |
| 1= | 76 | IOIS= | -0.00000 | <u></u> ಟ= | | 0.0 | DUAL= | 9.40133 |
| 1= | 77 | RHS= | -0.00000 | B= | | 0.0 | DUAL= | 68.86452 |
| I = | 79 | RHS= | -0 00000 | B= | | 0.0 | DUAL= | 5.76133 |
| 1 | 70 | 101.7- | -017 50(50 | - D | | Å Å | DUAL - | A A |
| 1 = | 69 | 1015= | -017.03002 | D- | | v.v | 10/11- | U.U |
| 1 = | 80 | IUIS= | -12175.18204 | В≃ | | 0.0 | DUAL= | 0.0 |
| 1= | 81 | BHS= | -12175.18204 | B= | | 0.0 | DUAL= | 0.0 |
| Ĩ = | 89 | RUS- | -19175 10904 | B= | | 63 . 63 | DHAL= | 0.0 |
| 1 | 02 | 101.3* | - 14110.10407 | D- | | 0. V | DUAL - | A A |
| 1= | 83 | 1018= | -12175.18204 | <u>ы</u> = | | U . V | DUAL- | v .v |
| I = | 84 | NHS= | -12175.18204 | B= | | 0.0 | DOVT= | 0.0 |
| 1= | 85 | BHS= | ~0.00000 | B= | | 0.0 | DUAL= | 1003.43862 |
| Î= | 86 | nne- | A AAAAA | n- | | 4 4 | DUAL = | 1003 43369 |
| 17 | 00 | 1018- | 0.00000 | | | 0.0 | DUAL- | 1000.30000 |
| 1 = | 87 | RUS= | 0.00000 | B= | | 0.9 | DUAL= | 1003.43802 |
| I= | 88 | RHS= | -0.00000 | B= | | 0.0 | DUAL= | 1003.43862 |
| Ī= | 89 | RUS= | -0.00000 | B= | | 0.0 | DUAL= | 940.33543 |
| ř- | 00 | 000- | | p~ | | ã Ã | DUAL = | 1001 49869 |
| | 99 | 1015= | -0.00000 | D- | | 0.0 | DUAL- | 1000.10002 |
| 1= | 91 | iuis= | -0.00000 | в= | | 17.17 | DUAL= | 1004 14449 |
| I = | 92 | RHS= | 0.00000 | B= | | 0.0 | DUAL= | 1009.19995 |
| I= | 93 | BHS= | 0.0 | B= | | 0.0 | DUAL= | 1009.19995 |
| | 64 | D10- | | D- | | a a | DUAL = | 1009 10005 |
| 1 = | 9 1 | 1015= | -0.00000 | D- | | 0.0 | DUAL- | 1007,19970 |
| 1= | 95 | HHS= | v. v | в= | | Q . Q | DOAL= | 1003 13330 |
| I = | 96 | NIS= | 0.00000 | B= | | 0.0 | DUAL= | 1009.19995 |
| 1 | - | | | - | | | | |

| I = | 97 | BHS= | 67211.44999 | B= | 2999999.999999 | DUAL= | 0.0 |
|-----|--------------|-------|--------------|------------|----------------|---------|--------------|
| Î= | - <u>6</u> 8 | BUS= | 500000,00000 | Ĩa≊ | 500000.00000 | DUAL= | 198.96128 |
| î = | áğ | RHS= | 2000.00000 | B = | 2000.00000 | DUAL= | 5773.51692 |
| Î= | 100 | BHS= | 0.0 | B= | 2500.00000 | DUAL= | 0.0 |
| Î= | 101 | BUS= | 947.66675 | B= | 947.66675 | DUAL= | 1134.11594 |
| î= | 102 | BHS= | 947.66675 | B= | 947.66675 | DUAL= | 1134.11594 |
| î= | 103 | BHS= | 947.66675 | B= | 947.66675 | DUAL= | 1134.11594 |
| 1= | 104 | RUIS= | 947.66675 | B= | 947.66675 | DUAL= | 1134.11594 |
| Î= | 105 | IUIS= | 947.66675 | B= | 947.66675 | - DUAL= | 1071.01274 |
| I = | 106 | IUIS= | 947.66675 | B= | 947.66675 | DUAL= | 1134.11594 |
| Ī= | 107 | RHS= | 947.66675 | B= | 947.66675 | DUAL= | 1139.87727 |
| Ī= | 108 | IUIS= | 947.66675 | B= | 947.66675 | DUAL= | 1206.33726 |
| Ĩ= | 109 | RHS= | 947.66675 | B= | 947.66675 | DUAL= | 1231.01594 |
| 1= | 110 | BHS= | 947.66675 | B= | 947.66675 | DUAL= | 1617.25459 |
| I = | 111 | IUIS= | 947.66675 | В= | 947.66675 | DUAL= | 1231.01594 |
| Ī= | 112 | HHS= | 947.66675 | B= | 947.66675 | DUAL= | 1231.01594 |
| 1= | 113 | RHS= | 751.21715 | B= | 3000.00000 | DUAL= | 0.0 |
| Î= | 114 | RUIS= | -20386.60675 | B= | -20386.60675 | DUAL= | 852.84519 |
| 1= | 115 | RHS= | 100000.00000 | B= | 100000.00000 | DUAL= | 15.29877 |
| 1= | 116 | RHS= | 39809.93805 | B= | 55090.00000 | DUAL= | 0.0 |
| 1= | 117 | IUIS= | 27267.74426 | B≃ | 55000.00000 | DUAL= | 0.0 |
| Î = | 118 | RHS= | 27267.74426 | B= | 55000.00000 | DUAL= | 0.0 |
| 1= | 119 | RHS= | 28252.13368 | B≃ | 55000.00000 | DUAL= | 0.0 |
| 1= | 120 | NIS= | 27267.74426 | B= | 55000.00000 | DUAL= | 0.0 |
| 1 = | 121 | RUIS= | 31535.22265 | B= | 55000.00000 | DUAL= | 0.0 |
| 1= | 122 | RUS= | 43642.35448 | B= | 55999.00000 | DUAL= | 0.0 |
| 1= | 123 | RHS= | 39442.92630 | B= | 55000.00000 | DUAL= | 0.0 |
| I = | 124 | RHS= | 43814.73952 | B= | 55000.00000 | DUAL= | 0.0 |
| I = | 125 | RHS= | 55000.00000 | B= | 55000.00000 | DUAL= | 232.03970 |
|] = | 126 | RHS= | 49293.49329 | B= | 55000.00000 | DUAL= | 0.0 |
| I = | 127 | RHS= | 43523.78708 | B= | 55000.00000 | DUAL= | 0.0 |
| I = | 128 | NIS= | 50148.87822 | B= | 50148.87822 | DUAL= | -1202.39990 |
| I = | 129 | RHS= | 35969.30072 | B= | 35969.30072 | DUAL= | -1202.39990 |
| I = | 130 | IUIS= | 25120.61133 | B= | 25120.61133 | DUAL= | -1202.39990 |
| I = | 131 | IUIS= | 22953.06840 | B= | 22953.06840 | DUAL= | -1202.39990 |
| I = | 132 | RHS= | 23386.57665 | B= | 23386.57665 | DUAL= | -1139.29670 |
|] = | 133 | RUS= | 28961.82665 | B= | 28961.82665 | DUAL= | -1202.39990 |
| I = | 134 | IUIS= | 42740.82011 | B= | 42740.82011 | DUAL= | -1208.16123 |
| 1 = | 135 | RHS= | 61453.02617 | B= | 61453.02617 | DUAL= | -1274.62122 |
| I = | 136 | RHS= | 81152.97330 | B= | 81152.97330 | DUAL= | -1299.29990 |
| I = | 137 | iuis= | 88621.38534 | B= | 88621.38534 | DUAL= | -1917.57825 |
| I = | 138 | RHS= | 79764.64566 | B= | 79764.64566 | DUAL= | -1299.29990 |
| I = | 139 | IUIS= | 71324.94418 | B= | 71324.94418 | DUAL= | - 1299.29990 |

OPTIMAL SOLUTION CHARACTERISTICS

CONSOLIDATED

MAXIMAL MONTHLY SUPPLY= 15557.07 MAXIMAL DAILY SUPPLY= 518.57

PANHANDLE

MAXIMAL MONTHLY SUPPLY= 48700.73 MAXIMAL DAILY SUPPLY= 1623.36

| IM= | 1 | FSN= -2617.75 | SUP 1= | 12542.19 | SUP2= | 36525.55 | SUPV= | 36525.55 | PROD= | 1698.88 | DGFIT= | 50148.88 |
|------|----|---------------|--------|----------|-------|----------|-------|----------|-------|---------|--------|----------|
| I M= | 2 | FSN= -4255.13 | SUP1= | 0.0 | SUP2= | 36525.55 | SUPV= | 36525.55 | PROD= | 1698.88 | DCMT= | 35969.30 |
| I M= | 3 | FSN=-15103.82 | SUP1= | 0.0 | SUP2= | 36525.55 | SUPV= | 36525.55 | PROD= | 1698.88 | DCMT= | 25120.61 |
| I M= | 4 | FSN=-18255.75 | SUP1= | 984.39 | SUP2= | 36525.55 | SUPV= | 36525.55 | PROD= | 1698.88 | DGMT= | 22953.07 |
| I M= | 5 | FSN=-16837.85 | SUP1= | 0.0 | SUP2= | 36525.55 | SUPV= | 36525.55 | PROD= | 1698.88 | DGMT= | 23386.58 |
| I M= | 6 | FSN=-15530.08 | SUP1= | 4267.48 | SUP2= | 36525.55 | SUPV= | 36525.55 | PROD= | 1698.88 | DCMT= | 28961.83 |
| 1 M= | 7 | FSN=-13858.22 | SUP1= | 15557.07 | SUP2= | 37343.08 | SUPV= | 37343.08 | PROD= | 1698.88 | DGMT= | 42740.82 |
| IM= | 8 | FSN= 9053.41 | SUP1= | Ø.C | SUP2= | 48700.73 | SUPV= | 48700.73 | PROD= | 1698.88 | DGML= | 61453.03 |
| IM≃ | 9 | FSN= 24381.55 | SUP1= | 4371.84 | SUP2= | 48700.73 | SUPV= | 48700.73 | PROD= | 1698.88 | DGMT= | 81152.97 |
| I M= | 10 | FSN= 20664.70 | SUP1= | 15557.07 | SUP2= | 48700.73 | SUPV= | 48700.73 | PROD= | 1698.88 | DGMT= | 88621.39 |
| 1 M= | 11 | FSN= 17514.47 | SUP1= | 9850.57 | SUP2= | 48700.73 | SUPV= | 48700.73 | PROD= | 1698.88 | DGMT= | 79764.65 |
| IM= | 12 | FSN= 14844.47 | SUP1= | 4080.66 | SUP2= | 48700.73 | SUPV= | 48700.73 | PROD= | 1698.88 | DCMT= | 71324.94 |

 SUP WI=
 2000.000

 SUPFL=
 0.0

 DPR0=
 751.217

 DSTU=
 100000.000

 DPT1=
 12956.686

| TOTAL DEMAND CHARGE= | 42331712.11 | OR | 0.06106 | OF | MINIMUM | COST |
|-------------------------|--------------|----|---------|----|---------|------|
| TOTAL COMMODITY CHARGE= | 585415016.49 | or | 0.84443 | OF | MINIMUM | COST |
| TOTAL STORACE COST= | 5746037.80 | OR | 0.00829 | OF | MINIMUM | COST |
| TOTAL WINTER CHARCE= | 3281064.42 | OR | 0.00473 | OF | MINIMUM | COST |

SUMMARY OF MONTHLY STORACE CAS FLOWS AND STOCKS

| IN= 2 GSTOR= 2617.75 RSTOR= 0.78 GINST= 4255.13 GINAX= 19759.33 GOUST= 0.0 COMAX= | 12980.57 |
|---|----------|
| | |
| IN= 3 CSTUR= 6872.87 RSTOR= 0.80 CINST= 15103.82 CIMAX= 19428.84 COUST= 0.0 COMAX= | 13629.24 |
| IM= 4 CSTOR= 21976.69 RSTOR= 0.86 CINST= 18255.75 CIMAX= 18255.75 COUST= 0.0 COMAX= | 15931.74 |
| IN= 5 GSTOR= 40232.44 RSTOR= 0.93 GINST= 16837.85 GIMAX= 16837.85 COUST= 0.0 COMAX= | 18714.74 |
| IN= 6 CSTOR= 57070.30 RSTOR= 1.00 CINST= 15530.08 CIMAX= 15530.08 COUST= 0.0 COMAX= | 21281.59 |
| IM= 7 CSTOR= 72600.38 RSTOR= 1.06 GINST= 13858.22 GIMAX= 14323.88 COUST= 0.0 COMAX= | 23649.08 |
| IM= 8 CSTOR= 86458.60 RSTOR= 1.12 CINST= 0.0 CIMAX= 13247.54 COUST= 9053.41 COMAX= | 25761.70 |

| IM= 1 IM= 2 IM= 3 IM= 4 IM= 6 IM= 6 IM= 7 IM= 8 IM= 7 IM= 8 IM= 10 IM= 11 IM= 12 | V1X= V1X= V1X= V1X= V1X= V1X= V1X= V1X= | 0.0 0.0 0.0 0.0 0.0 5.761 0.0 306.239 0.0 0.0 | V2X= V2X= V2X= V2X= V2X= V2X= V2X= V2X= | $\begin{array}{c} 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 6.0\\ 6.460\\ 91.139\\ 477.377\\ 91.139\\ 91.139\\ 91.139 \end{array}$ | VVV= VVV= VVV= VVV= VVV= VVV= VVV= VVV | $\begin{array}{c} 5.761\\ 5.761\\ 5.761\\ 5.761\\ 68.865\\ 5.761\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.$ | VSUV= VSUV= VSUV= VSUV= VSUV= VSUV= VSUV= VSUV= VSUV= VSUV= VSUV= | $\begin{array}{c} 1003.439\\ 1003.439\\ 1003.439\\ 1003.439\\ 940.335\\ 1003.439\\ 1009.200\\ 1009.200\\ 1009.200\\ 1009.200\\ 1009.200\\ 1009.200\\ 1009.200\\ 1009.200\\ 1009.200\end{array}$ |
|--|--|--|--|--|---|--|---|---|
| | VS 17 | r <u>-</u> | A .A | | | | | |

| IM= 1 | VGINS= | 0.0 | VGOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
|---------|--------|--------|--------|-----------|--------|-----|--------|----------|
| IM= 2 | VCINS= | 0.0 | VCOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 3 | VGINS= | 0.0 | VCOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 4 | VGINS= | 0.000 | VGOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 5 | VGINS= | 68.417 | VCOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 6 | VGINS= | 5.761 | VGOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 7 | VGINS= | 0.0 | VCOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 8 | VGINS= | 0.0 | VCOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 9 | VGINS= | 0.0 | VCOUS= | 29.11750 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| I M= 10 | VGINS= | 0.0 | VCOUS= | 763.83947 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 1 1 | VCINS= | 0.0 | VGOUS= | 171.74241 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 12 | VCINS= | 0.0 | VCOUS= | 292.63279 | VSMAX= | 0.0 | VSMIN= | 1063.437 |

DUAL VALUES SUMMARY *****

1.08 GINST=

| IM= 10 | CSTOR= | 53023.64 | RSTOR= | 0.98 | GINST= | 0.0 | CIMAX= | 15844.38 | COUST= | 20664.70 | GOMAX= | 20664.70 |
|--------|------------------|-----------|-------------------------|--------------|--------|-----|--------|----------|--------|----------|--------|----------|
| IM= 11 | CSTOR= | 32358.94 | RSTOR= | 0.90 | GINST= | 0.0 | CIMAX= | 17449.38 | COUST= | 17514.47 | COMAX= | 17514.47 |
| IM= 12 | CSTOR= | 14844.47 | RSTOR= | 0.83 | CINST= | 0.0 | CIMAX= | 18809.70 | COUST= | 14844.47 | COMAX= | 14844.47 |
| | YEARLY YEARLY | FLOW INTO | STORACE= OF STORACE= | 86458 864 | 58.60 | | | | | | | |

0.0

CIMAX=

13950.70 COUST= 24381.55 COMAX=

VDGMT= -1202.400 VINCINT= -1202.400

VDCMT= -1202.400

VDGNT= -1202.400 VDCMT= -1139.297

VDGMT= -1202.400

VDGNT= -1208.161

VDGNT= -1274.621 VDCMT= -1299.300 VDCMT= -1917.578

VDGNT'= -1299.300

VDCMT= -1299.300

24381.55

VS2T=

CSTOR=

77405.19 RSTOR=

IM= 9

198.96128

| IM- | | VPDO- | 1194 11804 | V/TDAN= | 0 0 |
|------|-----|---------|------------|------------|-----------|
| 111- | | 41 NO- | 1104,11079 | A 1 LOWL | 0.0 |
| I M= | 2 | VPRO= | 1134.11594 | VTRAN= | 0.0 |
| I M= | 3 | VPRO= | 1134.11594 | VTRAN= | 0.0 |
| I M= | 4 | VPRO= | 1134.11594 | VTRAN= | 0.0 |
| IM= | 5 | VPRO= | 1071.01274 | VTRAN= | 0.0 |
| I M= | 6 | VPRO= | 1134.11594 | VTRAN= | 0.0 |
| IM= | 7 | VPBO= | 1139.87727 | VTBAN= | 0.0 |
| TM= | à | VPRO= | 1206 33726 | VTRAN= | 0 0 |
| 111- | | VI 110- | 1200.00120 | V I I CHIN | 0.0 |
| I M= | - 9 | VPRO= | 1231.01594 | VTRAN= | 0.0 |
| I M= | 10 | VPRO= | 1617.25459 | VTRAN= | 232.03970 |
| I M= | 11 | VPRO= | 1231.01594 | VTRAN= | 0.0 |

IM= 12 VPRO= 1231.01594 VTRAN= 0.0

VSWH= 5773.517 VSFL= 0.0 VPRMX= 0.0 VPRMN= 852.845 VSTC= 15.299

TOTAL INVESTMENT IN PRODUCTION, STORAGE AND TRANSMISSION CAPACITY

ANNUALIZED COST PIS= 18822572.18 TOTAL DISCOUNTED COST NEWPIS= 151672576.

PURCHASES, PRODUCTION AND STORAGE OPERATING COSTS ONC1= 674440528.57 TOTAL ANNUAL GAS DEMAND (MNCF) DCT= 611598.06

OUTPUT OF SUBROUTINE DIST

PEAK MONTH=10PEAK LOAD=88621.39TRANSMISSIONMARCINAL COST=216.308DISTRIBUTIONMARGINAL COST=1954.964

NEW TRANSMISSION PLANT= NEW DISTRIBUTION PLANT= 211576.44

465353472.

OUTPUT OF SUBROUTINE REVREQ

 REPPIS=
 22378524.476

 TWTPIS=
 1256954652.476

 DEPEXP=
 36941896.994

 TAPD=
 255177921.146

 NETPIS=
 1001776640.

| OOPREV: | 2875912.194 |
|---------|-----------------------------------|
| ONUINC: | 3739440.177 |
| onc2= | 128120591.292 |
| ACOPEX | 839503016.853 |
| PRPTAX: | = 21037311.022 |
| PAYTAX: | = 3843617.892 |
| INVTXC | 63961616.0 |
| | |
| | |
| X0= : | 33192403.416 |

| 110- | 001747001710 |
|------|---------------|
| X1= | 120814247.880 |
| X2= | 79230120.859 |
| X3= | 77007639.572 |
| X4= | 864383945 767 |
| X3 = | 982103711.142 |
| X6= | 6615352.371 |
| | |

NEWPIS= 617237504. DGT= 611598.057 OMC1= 674440328.567 X= 975480358.771 PAVC= 1594.983

OUTPUT OF SUBROUTINE EVAL1

GAS CONSUMPTION EVALUATION CRITERIA

PEAK MONTH= 10 PEAK LOAD (MMCF)= 88621.39 LOAD FACTOR= 0.5751 TVTAL CAS CONSUMPTION= 611598.06

AVERAGE VOLUMETRIC RATE= 1594.983 ACHIEVED CAS SALES REVENUE= 975488358.77

EFFICIENCY CRITERIA

| MONTH= | 1 | RESIDENTIAL SURPLUS= | 110228601. | COMMERCIAL SURPLUS= | 44103576. | INDUSTRIAL SURPLUS= | 86256140. |
|---------|-----|----------------------|------------|---------------------|-----------|---------------------|------------|
| MONTH= | 2 | RESIDENTIAL SURPLUS= | 65764123. | COMMERCIAL SURPLUS= | 27174647. | INDUSTRIAL SURPLUS= | 79623286. |
| MONTH= | 3 | RESIDENTIAL SURPLUS= | 31744682. | COMMERCIAL SURPLUS= | 14222448. | INDUSTRIAL SURPLUS= | 74348539. |
| MONTH= | 4 . | RESIDENTIAL SURPLUS= | 24947676. | COMMERCIAL SURPLUS= | 11634629. | INDUSTRIAL SURPLUS= | 73534616. |
| MON'IH= | 3 | RESIDENTIAL SURPLUS= | 26307076. | COMMERCIAL SURPLUS= | 12152193. | INDUSTRIAL SURPLUS= | 73737401. |
| MONTH= | 6 | RESIDENTIAL SURPLUS= | 43790008. | COMMERCIAL SURPLUS= | 18808457. | INDUSTRIAL SURPLUS= | 76345364. |
| MONTH≈ | 7 | RESIDENTIAL SURPLUS= | 86998330. | COMMERCIAL SURPLUS= | 35259131. | INDUSTRIAL SURPLUS= | 82790834. |
| MONTH= | 8 | RESIDENTIAL SURPLUS= | 145676275. | COMMERCIAL SURPLUS= | 57599545. | INDUSTRIAL SURPLUS= | 91543939. |
| MONTH= | 9 | RESIDENTIAL SURPLUS= | 207451590. | COMMERCIAL SURPLUS= | 81119217. | INDUSTRIAL SURPLUS= | 100759084. |
| MONTH= | 10 | RESIDENTIAL SURPLUS= | 230871120. | COMMERCIAL SURPLUS= | 90035719. | INDUSTRIAL SURPLUS= | 104252621. |
| MONTH= | 11 | RESIDENTIAL SURPLUS= | 203098056. | COMMERCIAL SURPLUS= | 79461699. | INDUSTRIAL SURPLUS= | 100109659. |
| MONTH= | 12 | RESIDENTIAL SURPLUS= | 176632746. | COMMERCIAL SURPLUS= | 69385580. | INDUSTRIAL SURPLUS= | 96161776. |

| TOTAL | RESIDENTIAL SURPLUS | 1353510283. | |
|-------|---------------------|-------------|--|
| TOTAL | COMMERCIAL SURPLUS | 540975841. | |
| TOTAL | INDUSTRIAL SURPLUS | 1039663259. | |
| TOTAL | CONSUMER SURPLUS | 2934130333. | |

PRODUCER SURPLUS 120814248.

TOTAL SURPLUS 3054944631.

ITERATIVE DQUILIBRIUM PROCEDURE

ITERATION NUMBER

MONTHLY MARCINAL COSTS

| MONTH= | 1 | COST= | 1411.885 |
|---------|----|-------|----------|
| MONTH= | 2 | COST= | 1411.885 |
| MONTH= | 3 | COST= | 1411.885 |
| MONTH= | 4 | COST= | 1411.885 |
| MONTH= | 5 | COST= | 1348.782 |
| MONTH= | 6 | COST= | 1411.885 |
| MONTH= | 7 | COST= | 1417.646 |
| MONTH= | 8 | COST= | 1484.106 |
| MONTH= | -9 | COST= | 1508.785 |
| MON LH= | 10 | COST= | 4298.336 |
| MONTH= | 11 | COST= | 1508.785 |
| MONTH= | 12 | COST= | 1508.785 |

SUBROUTINE LOAD---CAS MARKET CHARACTERISTICS

1

GAS DEMAND PATTERNS

BASE DEMAND (MMCF)

| MONTHI MONTH MONTH HONTH HONTH MONTH MONTH MONTH MONTH MONTH MONTH MONTH | 1 2 3 4 5 6 7 8 9 10 11 12 | DCMRO= DCMRO= DCMRO= DCMRO= DCMRO= DCMRO= DCMRO= DCMRO= DCMRO= | $\begin{array}{c} 16280.51\\ 9713.21\\ 4688.62\\ 3684.71\\ 3975.35\\ 6467.68\\ 12823.32\\ 20986.00\\ 29639.89\\ 19543.07\\ 29017.87\\ 29017.87\\ 25236.61 \end{array}$ | DGMC0= DGHC0= DGHC0= DGHC0= DGHC0= DGHC0= DGHC0= DGHC0= DGHC0= DGHC0= DGHC0= DGHC0= | $\begin{array}{c} 6514.00\\ 4013.63\\ 2100.62\\ 1718.41\\ 1836.36\\ 2777.97\\ 5197.10\\ 8297.74\\ 11590.00\\ 7621.46\\ 11353.18\\ 9913.55\\ \end{array}$ | DCMIO= DCMIO= DCMIO= DCMIO= DCMIO= DCMIO= DCMIO= DCMIO= DCMIO= DCMIO= DCMIO= | 12739.84 11760.18 11010.65 10860.90 11142.70 11276.04 12203.14 13187.74 14396.07 88124.91 14303.28 13739.23 | DGMT0= DGMT0= DGMT0= DGMT0= DGMT0= DGMT0= DGMT0= DGMT0= DGMT0= DGMT0= | $\begin{array}{c} 35534.35\\ 25487.02\\ 17799.89\\ 16264.02\\ 16954.41\\ 20521.69\\ 30223.56\\ 42471.49\\ 55625.96\\ 35989.44\\ 54674.34\\ 46889.38\\ \end{array}$ |
|---|---|--|--|--|--|--|--|--|--|
| TOTAL | | DCMRTO= | 182056.85 | DGMCTO= | 72934.02 | DGM1TO= | 145444.67 | DGMITO | 400435.54 |

FORECASTED DEMAND (MPRCF)

| MONTH= | 1 | DGMR = | 24429.77 | DGMC = | 9771.00 | DGMI = | 19109.75 | DCMT = | 53301.52 |
|---------|---|--------|----------|---------|---------|--------|----------|--------|----------|
| MON'TH= | 2 | DGMR = | 14569.82 | DGMC = | 6020.45 | DGMI = | 17640.27 | DGMT = | 38230.53 |
| MONTH= | 3 | DCMR = | 7032.93 | DGrIC = | 3150.93 | DGMI = | 16515.97 | DGMT = | 26699.83 |
| MONTH= | 4 | DCMR = | 5527.07 | DGriC = | 2577.61 | DCMI = | 16291.34 | DCMT = | 24396.03 |
| MON'TH= | 5 | DCMR = | 5963.02 | DGMC = | 2754.54 | DCMI = | 16714.05 | DGMT = | 25431.61 |

| MONTH= | 6 | DGMR = | 9701.53 | DCMC = | 4166.95 | DGMI = | 16914.95 | DCMT = | 30782.53 |
|-----------|----|----------|-----------|-----------|--------------|----------|-----------|------------|-----------|
| MONTH= | 7 | DCMR = | 19234.98 | DGriC ≖ | 7795.65 | DGMI = | 18304.72 | DCMT = | 45335.34 |
| MONTH= | 8 | DGMR = | 31479.01 | DGMC = | 12446.61 | DCMI = | 19781.62 | DGMT = | 63707.24 |
| MONTH= | 9 | DGMR = | 44459.83 | DGMC = | 17385.00 | DCMI = | 21594.11 | DCMT = | 83438.94 |
| MONTH= | 10 | DCMR = | 29314.61 | DGHC = | 11432.19 | DCMI = | 13237.36 | DGMT = | 53984.16 |
| MONTH= | 11 | DGMR = | 43526.81 | DGrlC = | 17029.77 | DGMI = | 21454.93 | DCMT = | 82011.50 |
| MONTII= | 12 | DCMR = | 37854.91 | DGrIC = | 14870.32 | DGMI = | 20698.84 | DCML = | 73334.07 |
| (DO (DA) | | DOMD/D - | 070007 00 | 141306300 | - 100 404 00 | DOM IT - | 0.0.0 | TA AND THE | (00/50 01 |
| TUTAL | | DGPIRT = | 273009.20 | IN-ITCI | * 109401.03 | DGMTT = | 210167.01 | DGHIT | 600033.31 |

TOTAL DEMAND INCREMENT (MMCF) 200217.77

OUTPUT FROM SUBROUTINE MARCOS

EXISTING MONTHLY PRODUCTION CAPACITY PROC: 947.667 EXISTING MONTHLY TRANSMISSION CAPACITY PTIO: 55000.000

OPTIMAL SOLUTION

OBJ -682006594.8 ISTATE 4 ITERATIONS 94 DETERMINANT -7.03470 INFEAS 0 NINTO **0** NOUTOF 0 N 79 NCOL 67 M 139 MNOW 139 ISFEAS 1 IRCNT 1 X VECTOR

| SLACK | VECTOR | | | | | | | | | |
|--------|-------------|-------------------|--------------------|--------------|--------------|-------------------|----------------|-------------------|-------------|----------------|
| | 11899.99841 | 4590.76714 | 9.9 | 0.0 | 0.0 | 0.0 | 0.0 | 7564.02920 | 8807.54076 | 9861.48304 |
| | 10401.80398 | 11212.71058 | 7500.00048 | 7500.00048 | 8614.25725 | 10341.81143 | 11935.18900 | 13404.81124 | 14760.29006 | 0.0 |
| | 0.0 | 4544.39623 | 0.0 | 0.0 | 60313.63911 | 53204.40783 | 41872.10679 | 31419.96912 | 21779.63380 | 12888.04932 |
| | 4687.06129 | 20697.55210 | 34267.32144 | 41224.04919 | 51664.65386 | 60513.63911 | 0.0 | 7309.23128 | 18641.53231 | 29093.66999 |
| | 38734.00531 | 47625.58979 | 551-26.57781 | 39816.08700 | 26246.31767 | 19289.58992 | 8848.98525 | 0.0 | 18269.38164 | 26031.13571 |
| | 26031.13571 | 26031.13571 | 26031.13571 | 26031.13571 | 18034.56722 | 23874.15151 | 1701.72977 | 24543.46441 | 0.0 | 7085.81669 |
| | 0.0 | 0.0 | 7507.62915 | 10691.59936 | 10467.82059 | 5865.65002 | 0.0 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 0.0 | 10467.82059 | 10467.82059 | 2960.19144 | 0.0 | 0.0 | 4602.17056 | 10467.82059 | 10467.82059 |
| | 10467.82059 | 10467.82059 | 10467.82059 | 10467.82059 | 0.0 | 0.0 | 0.0 | 223.77877 | 0.0 | 0.0 |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 111291.16125 | 32077.31083 | 0.0 | 2300.00000 |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 0.0 | 2279.18493 | 0.0 | 100000.00000 | 18269.38164 | 26031.13571 | 33538.76486 | 36722.73507 | 36498.95630 |
| | 31896.78574 | 18034.56722 | 23674.15151 | 1701.72977 | 24543.46441 | 0.0 | 7085.81669 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | |
| | | | | | | | | | | |
| DUAL V | VECTOR | | 00 (1007 | 1110 70100 | 1000 0000 | 001 07870 | 00 44001 | | 0 0 | 0.0 |
| | 0.0 | 0.0 | 23.01937 | 1119.(9193 | 1203.00233 | 221.2(0(0 | 30.44001 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 9.9 | 0.00 | 100 40117 | | 0.0 | 0.0 | 0.0 | 0.0 | 0.00000 A A |
| | -0.00000 | 0.0 | 736.28234 | 132.43115 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 6.6 | 0.0 | 0.0 | 0.0 | 0.0 | 104.90007 | 1100 (0076 | 0.0 | 0.0 |
| | 6.6 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 1133.03010 | 201 00008 | 0.0 |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 100 10008 | 300 0005 | 371,77770 | 200 00005 |
| | 193.19995 | 21.70909 | 0.0 | v.v | 0.0 | 9.0 1000 10005 | 193.19990 | 470.07770 A A | 6 G | A A |
| | 682.09993 | 290.09993 | 6.9 | 0.0 | 1000 10008 | 1009.19990 | 1000 1000R | 0.0 A A | 0 49716 | 1000 10005 |
| | U.U. | 0.0 1000 1000F | 0.00 1000 10007 | 1000 10005 | 1009.19990 | 1007.17770 | 1007.17770 | 40.00 60.60 | 7000110 | 0 0 |
| | 1009.19993 | 1009.19990 | 1009.19990 | 1009.19990 | 170 00901 | 1007.17770 | 1979 86478 | 1460 70475 | 1460 70475 | 1460 70475 |
| | 1372.09470 | 1440 70478 | 1119.09%00 | 1001 49400 | 6 6 | 4 6 | 1012.07910 | 1407.17410 A A | A A | A 4 |
| | 1001.79473 | 1907.79970 | 0.0 | 1091.04400 | 0.0 | 010 00070 | 0.0 | -1202 20000 | -1030 08484 | -1000 10005 |
| | 9.9 | 0.0 | -16.00 10005 | - 1000 0000 | -1000 00000 | -1000 00000 | -1200 20000 | -1099 99059 | -1200 20000 | 1007.17770 |
| | 0.0 | -9.90/10 | -1009.19990 | -1202.07770 | -1299.27990 | -1299.29990 | - 14.77.4.7790 | -1720.00700 | 1 | |
| COST 1 | VECTOR | | | | | · · · · · · | | | | |
| - | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 |
| | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -33.23000 |
| | -33.23000 | -33.23000 | -33.23000 | -33.23000 | -1202.39990 | -1202.39990 | -1202.39990 | -1202.39990 | -1202.39990 | -1202.39990 |
| | -1202.39990 | -1299.29990 | -1299.29990 | -1299.29990 | -1299.29990 | -1299.29990 | 0.0 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | -1009.19995 | -1009.19995 |
| | -1009.19995 | -1009.19995 | -1609.19995 | -1009.19995 | -1009.19995 | -1009.19995 | -1009.19995 | -1009.19995 | -1009.19995 | -1009.19995 |
| | -391,99998 | -743.99996 | -9444.00000 | -17772.00000 | -921.12915 | -921.12915 | -921.12915 | -921.12915 | -921.12915 | -921.12915 |
| | -921.12915 | -921.12915 | -921.12915 | -921.12915 | -921.12915 | -921.12915 | -14398.10937 | ***** | -232.03970 | |
| | | | | | | | | | | |
| BOUND | VECTUR | - 1 00000 | -1 00000 | -1 00000 | - 1 00000 | - 1 00000 | -1 00000 | -1 00000 | - 1 00000 | - 1 00000 |
| | -1.00000 | -1.00000 | -1.00000 | _1 00000 | -1.00000 | -1.00000 | -1 00000 | -1 00000 | -1 00000 | -1 00000 |
| | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1,00000 | -1.00000 | -1.00000 | -1.00000 | -1 00000 | -1 00000 |
| | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1 00000 |
| | -1 00000 | -1.00000 | -1.00000 | -1.00000 | -1.00000 | -1 00000 | -1.00000 | -1 00000 | -1.00000 | -1.00000 |
| | -1 00000 | -1 00000 | -1.00000 | -1 00000 | -1 00000 | -1.00000 | -1.00000 | -1 66666 | -1.00000 | -1.00000 |
| | 1.00000 | ******** | ******** | | 1.00000 | | | | | - 100000 |

| | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 -1.00000 | -1.00000 |
|-----------|----------------------|----------------------|----------------------|----------------------|----------------------|--------------|----------------------|----------------------|----------------------|-------------|
| B VECTOR | | | | | | | | | | |
| | 11899.99841 | 11899.99841 | 11899.99841 | 11899.99841 | 11899.99841 | 11899.99841 | 11899.99841 | 11899.99841 | 11899.99841 | 11899.99841 |
| | 11899.99841 | 11899.99841 | 7500.00048 | 7500.00048 | 7500.00048 | 7500.00048 | 7500.00048 | 7500.00048 | 7500.00048 | 7500.00048 |
| | 7500.00048 | 7500.00048 | 7500.00048 | 7500.00048 | 60513.63911 | 69513.63911 | 60513.63911 | 60513.63911 | 60313.63911 | 00013.03911 |
| | 69513.63911 | 60513.63911 | 60513.63911 | 60513.63911 | 60513.63911 | 60513.63911 | 0.0 | 0.0 | 0.0 | |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | Q7. Q7 | 0.0 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 6 6 |
| | 0.0 0 0 | 0.0 | 0.0 6 6 | 0.0 | 6 6 | 0.0 | 0.0 0 0 | 0.0 | 0.0 | 0.0 |
| | 0.0 0.0 | 6 6 | 0.0 | 0.0 | 4.0 | 6.6 | 0.0 | 0.0 | 0.0 | ŏ.ŏ |
| | ŏ.ŏ | 0.0 | ě.ě | 0.0 | <u>.</u> | 0.0 2 | 200000.00000 | 300000.00000 | 2000.00000 | 2500.00000 |
| | 947.66675 | 947.66675 | 947.66675 | 947.66675 | 947.66675 | 947.66675 | 947.66675 | 947.66675 | 947.66675 | 947.66675 |
| | 947.66675 | 947.66675 | 3600.00000- | 20021.781761 | 00000.00000 | 55000.00000 | 55000.00000 | 55000.00000 | 55000.00000 | 55000.00000 |
| | 35000.00000 | 55000.00000 | 55000.00000 | 55000.00000 | 55000.00000 | 55000.00000 | 55000.00000 | 53301.51825 | 38230.53289 | 26699.83399 |
| | 24396.02714 | 25431.60826 | 30782.52967 | 45335.34464 | 63707.23919 | 83438, 93945 | 53984.16323 | 82011.50455 | 73334.06844 | |
| SIGN VECT | TOR | | | | | | - | | | |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1 00000 | 1.00000 | 1.00000 | 1 00000 | 1 44666 | 1 00000 | 1 00000 |
| | 1 00000 | 1 00000 | 1.00000 | 1.00000 | 1 88888 | 1 00000 | 1 88688 | 1 00000 | 1 00000 | 1 00000 |
| | 1 66666 | 1 00000 | 1 00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 |
| | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 1.00000 | 0.0 | 0.0 | 0.0 |
| | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | |
| | | | | | | | | | | |
| CONST | TRAINTS VALUES | 5 AND DUAL PH | RICES | | | | | | | |
| | | | | | | | | | | |
| I= t | RHS= | 9.9 | B= | 11899,9984 | 1 DUAL= | Q. | . 0 | | | |
| · | 101G_ | | n ñ- | 11000 0004 | 1 DUAL | | ā. | | | |

| | A . | 1010- | 0.0 | D- | | DOHL- | 0.0 |
|-----|-----|-------|--------------|-----|-------------|-------|------------|
| I = | 2 | RHS= | 7309.23128 | B= | 11899.99841 | DUAL= | 6.0 |
| I = | 3 | RHS≈ | 11899.99841 | B= | 11899.99841 | DUAL= | 23.61937 |
| I = | 4 | RHS= | 11899.99841 | B= | 11899.99841 | DUAL= | 1119.79193 |
| 1= | 5 | RUIS= | 11899,99841 | B= | 11899.99841 | DUAL= | 1203.80235 |
| 1= | 6 | NIS= | 11899.99841 | B= | 11899.99841 | DUAL= | 221.27573 |
| I = | 7 | NIS= | 11899.99841 | B= | 11899.99841 | DUAL= | 30.44001 |
| Ī= | 8 | BHS= | 4335,96922 | B= | 11899,99841 | DUAL= | 0.0 |
| Ī= | ÿ | RUS= | 3092.45765 | B= | 11899.99841 | DUAL= | 0.0 |
| Ĩ= | 10 | RHS= | 2038.51337 | B≃ | 11899.99841 | DUAL= | 0.0 |
| Ī= | 11 | HHS= | 1498, 19443 | B= | 11899.99841 | DUAL= | 0.0 |
| Ì= | 12 | IUIS= | 687.28783 | Ĩ8= | 11899.99841 | DUAL= | 0.0 |
| Ĩ= | 13 | BUS= | 0.0 | B= | 7500.00048 | DUAL= | 0.0 |
| Ĩ= | 14 | BHS= | 0.0 | B= | 7500.00048 | DUAL= | 0.0 |
| Î= | is | BHS= | -1114.25677 | B= | 7500.00048 | DUAL= | 0.0 |
| Î= | 16 | BHS= | -2841.81096 | B= | 7500.00048 | DUAL= | 0.0 |
| ĩ = | 17 | RHS= | -4435, 18852 | B= | 7500.00048 | DUAL= | 0.0 |
| Î= | 18 | BHS= | -5904.81077 | B= | 7500.00048 | DUAL= | 0.0 |
| Î= | 19 | HUS= | -7260.28958 | B= | 7500.00048 | DUAL= | 0.0 |
| Î= | 20 | BUS= | 7500,00048 | B= | 7500.00048 | DUAL= | 0.00000 |
| Î= | 21 | IUIS= | 7500.00048 | B= | 7500.00048 | DUAL= | -0.00000 |
| Ī= | 22 | BHS= | 2955.60425 | B= | 7500.00048 | DUAL= | 0.0 |
| Î= | 23 | BHS= | 7500.00048 | B= | 7500.00048 | DUAL= | 736.28234 |
| Ï= | 24 | BIIS= | 7500.00048 | B= | 7500.00048 | DUAL= | 132.43115 |
| 1= | 25 | BHS= | 0.0 | B= | 60513.63911 | DUAL= | 0.0 |
| I = | 26 | HIS= | 7309.23128 | B= | 60513.63911 | DUAL= | 0.0 |
| 1= | 27 | JUIS= | 18641.53231 | B= | 60513.63911 | DUAL= | 0.0 |
| I = | 28 | RUIS= | 29093.66999 | B= | 60513.63911 | DUAL= | 0.0 |
| | | | | | | | |

| Į= | 29 | RAS= | 38734.00531 | B= | 60513.63911 | DUAL≏ | 0.0 |
|----------|-------------|--------|----------------------|------------|--------------|---------|------------------|
| 1= | 30 | RHS= | 47625.58979 | B= | 60513.63911 | DUAL= | 0.0 |
| 1= | 31 | BHS= | 55826 57781 | B= | 60513.63911 | DUAL= | 8.8 |
| Î= | 30 | nus= | 39816 68760 | ñ= | 60513 63911 | DHAL = | 4 4 |
| 1- | 22 | 1013- | 26246 21767 | n= | 60513 63911 | DUAL = | <u> </u> |
| 12 | 0.4 | 0167- | 10000 50000 | D- | 60510 60011 | DUAL- | 0.0 |
| 1- | - 34k OF | 1018- | 19209.00992 | D- | | DUAL- | 0.0 |
| 1= | 35 | RHS= | 8848.98525 | H= | 60313.63911 | DUAL= | 9.9 |
| 1= | 36 | Rus= | 0.00000 | 8= | 60513.63911 | DUAL= | 0.0 |
| I = | 37 | RHS= | 0.0 | B= | 0.0 | DUAL= | 104.95507 |
| I = | 38 | RHS= | -7309.23128 | B= | 0.0 | DUAL= | 0.0 |
| I = | 39 | RHS= | -18641.53231 | B= | 0.0 | DUAL= | 0.0 |
| 1= | 40 | BHS= | -29093.66999 | B= | 0.0 | DUAL= | 0.0 |
| Î = | 41 | RHS= | -38734 00531 | R= | 0 0 | DUAL= | 63 . 63 |
| Î. | 42 | BUS= | -47625 58070 | ñ= | 8 6 | DUAL = | 6 6 |
| 1- | 40 | 1010- | | n- | 0.0 | DUAL~ | 0.0 |
| 1- | 4.0 | 1015- | -33620.37761 | D- D- | 0.0 | DUAL- | 0.0 |
| 1= | 44 | TOIS= | -39816.08700 | 8= | 0.0 | DUAL= | |
| 1 = | 40 | Rus= | -26246.31767 | B≃ | 0.0 | DUAL= | 0.0 |
| I = | 46 | RHS= | -19289.58992 | B= | 0.0 | DUAL= | 0.0 |
| 1= | 47 | RHS= | -8848.98525 | B= | 0.0 | DUAL= | 0.0 |
| I = | 48 | RHS= | -0.90000 | B= | 0.0 | DUAL= | 1133.63876 |
| 1= | 49 | RHS= | -18269.38164 | H= | 0.0 | DUAL= | 0.0 |
| Î= | 50 | RUS= | -26031 13571 | ñ= | 0.0 | DUAL = | 6.6 |
| i- | 51 | 010- | -26021 12571 | - D- | <u> </u> | DUAL - | 4 9 |
| 1 - | 80 | 000- | -96091 19571 | D- | 0.0 | DUAL - | 43 43 |
| 1- | 24 | 1013- | -20031.13371 | D- D- | V.V | DUAL- | 0.0 |
| 1= | 03 | 1015= | -26031.13571 | B= | 0.0 | DUAL- | 0.0 |
| 1= | 54 | Rus= | -26031.13571 | B= | 69.69 | DUAL= | 0.0 |
| 1= | 55 | RHS= | -18034.56722 | B= | 0.0 | DUAL= | 0.0 |
| 1= | 56 | nus= | -23874.15151 | B= | 0.0 | DUAL= | 0.0 |
| 1= | 57 | RIIS= | -1701.72977 | B≓ | 0.0 | DUAL= | 0.0 |
| 1= | 58 | RHS= | -24543.46441 | B= | 0.0 | DUAL= | 0.0 |
| 1= | 59 | RHS= | -0.0000 | B= | 0.0 | DUAL= | 391.99998 |
| î= | 66 | BUS= | -7085 81669 | B= | 0.0 | DUAL= | 0.0 |
| î | 61 | 1010- | 4 44444 | Ř= | 6 6 | DUAL = | 102 10005 |
| 1. | 49 | 1013- | 0.00000 | D- D- | 0.0 (A (A | INIAL - | 91 79490 |
| <u>.</u> | 04 | 1018- | | D- D- | 0.0 | ' DUAL- | A A |
| 1= | 63 | Iuis= | -7307.62913 | B= | 9.9 | DUAL- | 97.97 () () |
| 1= | 64 | IUIS= | -10691.59936 | B= | 0.0 | DUAL= | 17.17 |
| I = | 65 | RHS= | -10467.82059 | B= | 9.9 | DUAL.= | 63.69 |
| I = | 66 | iuis= | -5865.65002 | B= | 0.0 | DUAL= | 0.0 |
| I = | 67 | RHS= | 0.00000 | B= | 0.0 | DUAL= | 193.19995 |
| 1= | 68 | IUIS= | 0.00000 | B= | 0.0 | DUAL= | 290.09995 |
| 1= | 69 | BHS= | 0.0 | B= | 0.0 | DUAL= | 290.09995 |
| 1= | 70 | BHS= | 0.0000 | B≈ | 0.0 | DUAL= | 290.09995 |
| ī= | 71 | BHS= | A A | B= | 0.0 | DUAL= | 682.09993 |
| î- | | 000- | <u>ě ěnena</u> | ñ- | <u> </u> | DIAL = | 294 09995 |
| | 70 | BUG- | -10467 00050 | D- D- | 0.0 | DUAL = | Q 4 |
| 1- | 63 | BHO- | -10407.02037 | D- | 0.0 | DUAL- | 0.0 |
| i= | 74 | IIIS= | -10467.82039 | B= | v. v | DUAL- | 0.0 |
| 1= | 70 | 1015= | -2960.19144 | B= | 4.4 | DUAL- | 7000 1000 |
| 1 = | 76 | RHS= | -0.00000 | B= | 6.6 | DUAL= | 1003 . 13332 |
| I = | 77 | iuis= | -0.00000 | B= | 0.0 | DUAL= | 999.71279 |
| I = | 78 | RHS= | -4602.17056 | B= | 0.0 | DUAL= | 0.0 |
| I = | 79 | IUIS= | -10467.82059 | B= | 0.0 | DUAL= | 0.0 |
| I = | 80 | RHS= | -10467.82059 | B= | 0.0 | DUAL= | 0.0 |
| î= | 81 | RUS= | -10467 82059 | ñ= | A A | DUAL= | 0.0 |
| 1- | 00 | 1010- | -10467 99050 | B- | å å | DUAL = | Å Å |
| 1- | 02 | 1115- | - 10407 00050 | D- D- | 0.0 | DUAL- | 0.0 |
| 1- | 0.0 | 10187- | -10407.02009 | D- | 0.0 | DUAL- | 0.0 |
| i = | 04 | IUIS= | -10407.02039 | B- | 0.0 | DUAL- | |
| 1= | 85 | RHS= | 0.0000 | B= | 0.0 | DUAL= | 1009.19993 |
| 1 = | 86 | RHS= | 0.0000 | R= | 0.0 | DUAL= | 1009.19995 |
| I = | 87 | RHS= | 0.00000 | B= | 0.0 | DUAL= | 1009.19995 |
| I = | 88 | RHS= | -223.77877 | B= | 0.0 | DUAL= | 0.0 |
| I = | 89 | RHS= | 0.0000 | B= | 0.0 | DUAL= | 9.48716 |
| 1= | 90 | RHS= | 0.00000 | B= | 0.0 | DUAL= | 1009.19995 |
| Ĭ= | 91 | BHS= | -0.00000 | B = | 0.0 | DUAL= | 1009.19995 |
| Ī= | 92 | BHS= | -0.00000 | B= | 0.0 | DUAL= | 1009.19995 |
| î - | 67 | HHG= | -0 00000 | ñ= | ă ă | DIIAL = | 1000 10005 |
| ÷- | 70 | 101.7- | - 0.00000 A AAAAA | B= | 0.V | DUAL = | 1000 10005 |
| ÷- | 77 | 010- | 0.0000 A A | D- D- | U.U A A | DUAL- | 1007.17770 |
| 1- | 90 | 000- | v.v A A | D- D- | U.U 0 0 | DUAL- | 1007.17770 |
| 1 = | 70 | nuis= | 0.0 | D = | v.v | DUAL- | 1963.12230 |
| | | | | | | | |

| I = | 97 | BHS= | 88708.83875 | B= | 200000 | 00000 | DUAL= | .0.0 |
|-----|-----|--------|--------------|----|---------|-------|--------|----------------|
| Ť= | ú8 | RUS= | 467922.68917 | ละ | 500000. | 00000 | DUAL= | 0.0 |
| Î= | úú | HHS= | 2000.00000 | ñ= | 2000. | 00000 | DUAL= | 2908.17118 |
| Î= | 100 | BHS= | 0.0 | B= | 2500. | 00000 | DUAL= | 0.0 |
| î = | 101 | IIIIS= | 947.66675 | B≍ | 947. | 66675 | DUAL= | 1372.89475 |
| Î= | 102 | BHS= | 947.60675 | B= | 947. | 66675 | DUAL= | 1201.47969 |
| î = | 103 | BHS= | 947.60675 | B= | 947. | 66675 | DUAL= | 1179.69480 |
| Î= | 104 | BHS= | 947.66675 | B= | 947. | 66675 | DUAL= | 170.49485 |
| Ĩ= | 105 | IIIIS= | 947.66675 | B= | 947. | 66675 | DUAL= | 179.98201 |
| Î= | 106 | BHS= | 947.66675 | B= | 947. | 66675 | DUAL= | 1179.69480 |
| Î = | 107 | RUS= | 947.66675 | B= | 947. | 66675 | DUAL= | 1372.89475 |
| Î= | 108 | RHS= | 947.66675 | B= | 947. | 66675 | DUAL= | 1469.79475 |
| 1 = | 109 | RIIS= | 947.66675 | B= | 947. | 66675 | DUAL= | 1469.79475 |
| Î = | 110 | RHS= | 947.60675 | B= | 947. | 66675 | DUAL= | 1469.79475 |
| 1= | 111 | RHS= | 947.66675 | B≃ | 947. | 66675 | DUAL= | 1861.79473 |
| I = | 112 | RHS= | 947.66675 | в= | 947. | 66675 | DUAL= | 1469.79475 |
| I = | 113 | RHS= | 720.81507 | B= | 3009. | 00000 | DUAL= | 0.0 |
| 1 = | 114 | IUIS= | -20021.78176 | B= | -20021. | 78176 | DUAL= | 1091.62400 |
| I = | 115 | RHS= | 0.0 | B= | 100000. | 00000 | DUAL= | 0.0 |
| I = | 116 | IUIS= | 36730.61836 | B= | 55000. | 00000 | DUAL= | 0.0 |
| 1 = | 117 | NIS= | 28968.86429 | B= | 55000. | 00000 | DUAL= | 0.0 |
| I = | 118 | RHS= | 21461.23514 | B= | 55000. | 00000 | DUAL= | 0.0 |
| 1= | 119 | nus= | 18277.26493 | H= | 55000. | 00000 | DUAL.= | 0.0 |
| I = | 120 | RHS= | 18501.04370 | B= | 55000. | 00000 | DUAL= | 0.0 |
| I = | 121 | RHS= | 23103.21426 | B= | 55000. | 00000 | DUAL= | 0.0 |
| 1= | 122 | IUIS= | 36965.43278 | B= | 55000. | 00000 | DUAL= | 0.0 |
| 1 = | 123 | RHS= | 31125.84849 | B= | 55000. | 00000 | DUAL= | 0.0 |
| I = | 124 | RHS= | 53298.27023 | 8= | 55000. | 00000 | DUAL= | 0.0 |
| 1 = | 125 | RHS= | 30456.53559 | B= | 55000. | 00000 | DUAL= | 0.0 |
| I = | 126 | NIS= | 55000.00000 | B= | 55000. | 00000 | DUAL= | 232.03970 |
| I = | 127 | RHS= | 47914.18331 | B= | 55000. | 00000 | DUAL= | 0.0 |
| I = | 128 | iuis= | 53301.51825 | B= | 53301. | 51825 | DUAL= | -1202.39990 |
| 1 = | 129 | iuis= | 38230.53289 | B= | 38230. | 53289 | DUAL= | -1030.98484 |
| I = | 130 | IUIS= | 26699.83399 | B= | 26699. | 83399 | DUAL= | -1009.19995 |
| I = | 131 | NHS= | 24396.02714 | B= | 24396. | 02714 | DUAL.= | 0.0 |
| I = | 132 | 101S = | 25431.60826 | B= | 25431. | 69826 | DUAL= | -9.48716 |
| I = | 133 | RHS= | 30782.52967 | B= | 30782. | 52967 | DUAL= | -1009.19995 |
| I = | 134 | NIS= | 45335.34464 | B= | 45335. | 34464 | DUAL= | -1202.39990 |
| I = | 135 | 1048= | 63707.23919 | B= | 63707. | 23919 | DUAL= | ~ 1299.29990 |
| I = | 136 | 1015= | 83438.93945 | B= | 83438. | 93945 | BUAL= | - 1299.29990 |
| I = | 137 | 0 S = | 53984.16323 | B≈ | 53984. | 16323 | DUAL= | -1299.29990 |
| I = | 138 | NIS= | 82011.50455 | 8= | 82011. | 59455 | DUAL= | - 1923 . 33958 |
| I = | 139 | NIS= | 73334.06844 | B≠ | 73334. | 06844 | DUAL≃ | -1299.29990 |

OPTIMAL SOLUTION CHARACTERISTICS

CONSOLIDATED

MAXIMAL MONTHLY SUPPLY= 26031.14 MAXIMAL DAILY SUPPLY= 867.70

PANHANDLE

MAXIMAL MONTHLY SUPPLY= 41871.28 MAXIMAL DAILY SUPPLY= 1395.71

SUP2= 41871.28 SUPV= 41871.28 PROD= 1668.48 DCMT= 53301.52 IM= FSN= 0.0 SUP 1 = 7761.75 1 SUPV= 41871.28 DGMT= 38230.53 1 M= $\frac{2}{3}$ FSN= -7309.23 SUP 1= 0.0 SUP2= 41871.28 PROD= 1668.48 1 M= FSN=-11332.30 SUP1= 0.0 SUP2= 34363.65 SUPV= 34363.65 PROD= 1668.48 DCMT= 26699.83 SUPV= 31403.46 SUPV= 31403.46 SUPV= 36005.63 FSN=-10452.14 SUP1= SUP2= 31179.68 PROD= 1668.48 DGMT= 24396.03 IM= 4 0.0 DCMT= 25431.61 DCMT= 30782.53 5 FSN= -9640.34 FSN= -8891.58 SUP 1= SUP2= 31403.46 PROD= 1668.48 PROD= 1668.48 IM= 0.0 SUP1= SUP2= 36005.63 IM= 6 0.0 FSN= -8200.99 SUPV= 41871.28 IM= 7 SUP1= 7996.57 SUP2= 41871.28 PROD= 1668.48 DGMT= 45335.34 IM≃ Ō FSN= 16010.49 FSN= 13569.77 SUP1= 2156.98 SUP1= 24329.41 SUP2= 41871.28 SUP2= 41871.28 SUPV= 41871.28 SUPV= 41871.28 PROD= 1668.48 DGMT= 63707.24 DGMT= 83438.94 PROD= 1668.48 IM= Q. FSN= 6956.73 SUPV= 41871.28 PROD= 1668.48 DGMT= 53984.16 IM= 10 SUP1= 1487.67 SUP2= 41871.28 IM= FSN= 10440.60 SUP1= 26031.14 SUP1= 18945.32 SUP2= 41871.28 SUPV= 41871.28 PROD= 1668.48 DCMT= 82011.50 11 FSN= 8848.99 SUP2= 41871.28 SUPV= 41871.28 DGMT= 73334.07 IM= 12 PROD= 1668.48

SUPWH= 2000.000 SUPFL= 0.0

SUPFL= 0.0 DPRO= 720.815 DSTC= 0.0 DPT1= 16570.900

| TOTAL | DEMAND CHARCE= | 41356436.81 | OR | 0.06057 | OF | MINIMUM | COST |
|--------|-------------------|--------------|----|---------|----|---------|------|
| TOTAL | COMMODITY CHARGE= | 579116891.64 | or | 0.84814 | OF | MINIMUM | COST |
| INTAL. | STORACE COST= | 3710233.88 | on | 0.00543 | OF | MINIMUM | COST |
| DATO | WINTER CHARCE= | 7068904.85 | on | 0.01035 | OF | MINIMUM | COST |

SUMMARY OF MONTHLY STORAGE CAS FLOWS AND STOCKS

| IM= | 1 | CSTOR= | 0.0 | RSTOR= | 0.0 | GINST= | 0.0 | GIMAX= | 11900.00 | COUST= | 0.0 | Comax= | 7560.00 |
|------|---|--------|----------|--------|------|--------|----------|--------|----------|--------|----------|--------|----------|
| IM= | 2 | GSTOR= | 0.0 | RSTOR= | 0.77 | GINST= | 7309.23 | GIMAX= | 11900.00 | COUST= | 0.0 | GOMAX= | 7500.00 |
| 1 M= | 3 | CSTOR= | 7309.23 | RSTOR= | 0.82 | GINST= | 11332.30 | GIMAX= | 11332.30 | COUST= | 0.0 | GOMAX= | 8614.26 |
| I M= | 4 | CSTOR= | 18641.53 | RSTOR= | 0.90 | GINST= | 10452.14 | GIMAX= | 10452.14 | COUST= | 0.0 | GOMAX= | 10341.81 |
| I M= | 5 | CSTOR= | 29093.67 | RSTOR= | 0.97 | CINST= | 9640.34 | CIMAX= | 9640.34 | COUST= | 0.0 | COMAX= | 11935.19 |
| IM= | 6 | CSTOR= | 38734.01 | RSTOR= | 1.03 | CINST= | 8891.58 | GIMAX= | 8891.58 | GOUST= | 0.0 | GOMAX= | 13404.81 |
| IM= | 7 | CSTOR= | 47625.59 | RSTOR= | 1.09 | CINST= | 8200.99 | CIMAX= | 8200.99 | COUST= | 0.0 | COMAX= | 14760.29 |
| IM= | 8 | CSTOR= | 55826.58 | RSTOR= | 1.15 | CINST= | 0.0 | GIMAX= | 7564.03 | COUST= | 16010.49 | COMAX= | 16010.49 |

| IM= 9 | CSTOR= | 39816.09 | RSTOR= | 1.04 | CINST= | 0.0 | GIMAX= | 8807.54 | GOUST= | 13569.77 | GOMAX= | 13569.77 |
|---------|--------|----------|--------|------|--------|-----|--------|----------|--------|----------|--------|----------|
| IM= 10 | GSTOR= | 26246.32 | ESTOR= | 0.95 | CINST= | 0.0 | GINAX= | 9861.49 | COUST= | 6956.73 | COMAX= | 11201.12 |
| IM= 11 | GSTOR= | 19289.59 | RSTOR= | 0.90 | CINST= | 0.0 | CIMAX= | 10101.80 | GOUST= | 10440.60 | GOHAX= | 10449.60 |
| IM = 12 | GSTOR= | 8848.99 | RSTOR= | 0.83 | CINST= | 0.0 | GIMAX= | 11212.71 | COUST= | 8848.99 | COMAX= | 8848.99 |

| YEABLY | FLOW | INTO STORACE= | 55826.58 |
|--------|-------------|-----------------|----------|
| YEARLY | FLOW | OUT OF STORAGE= | 55826.58 |

DUAL VALUES SUMMARY

| IM= 1 | VGINS= | 0.0 | VGOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 104.955 |
|--------|--------|----------|--------|-------------|--------|-----|--------|----------|
| IM= 2 | VGINS= | 0.0 | VGOUS= | Э.О | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 3 | VGINS= | 23.619 | VGOUS= | 0.0 | VSMAX= | 0.0 | VSNIN= | 0.0 |
| 111= 4 | VG1HS= | 1119.792 | VGOUS= | 3.0 | VSNAX= | 0.0 | VSMIN= | 0.0 |
| IM= 5 | VGINS= | 1203.802 | VCOUS= | 3.0 | VSMAX= | 0.0 | VSM1N= | 0.0 |
| 111= 6 | VGINS= | 221.276 | VCOUS= | 0. C | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IN= 7 | VCINS= | 30.440 | VCOUS= | э.ө | VSMAX= | 0.0 | VSMIN= | 0.0 |
| 1M= 8 | VGINS= | 0.0 | VGOUS= | 3.03000 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IN= 9 | VG1NS= | 0.0 | VCOUS= | -3.00000 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM= 10 | VGINS= | 0.0 | VGOUS= | 0.0 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| 1M=11 | VGINS= | 0.0 | VCOUS= | 735.28234 | VSMAX= | 0.0 | VSMIN= | 0.0 |
| IM=12 | VCINS= | 0.0 | VCOUS= | 132.43115 | VSNAX= | 0.0 | VSMIN= | 1133.639 |

| IM= 1 | V1X= | 0.0 | V2X= | 193.200 | VVV= | 0.0 | VSUV= | 1009.200 | VDCMT= | -1202.400 |
|--------|------|---------|-------|---------|------|----------|--------|----------|---------|-----------|
| IN= 2 | V1X= | 0.0 | V2X= | 21.785 | VVV= | 0.0 | VSUV= | 1009.200 | VDGNT= | -1030.985 |
| IN= 3 | V1X= | 0.0 | V:2X= | 0.0 | VVV= | 0.0 | VSUV= | 1009.200 | VDGHT= | -1009.200 |
| IH= 4 | V1X= | 0.0 | V2X= | 0.0 | VVV= | 1009.200 | VSUV= | 0.0 | VDCMT= | 0.0 |
| III= 5 | VIX= | 0.0 | V:2X= | 0.0 | VVV= | 999.713 | VSUV= | 9.487 | VDGnT= | -9.487 |
| IN= 6 | V1X= | 0.0 | V2X= | 0.0 | VVV= | 0.0 | VS:UV= | 1009.200 | VDCrff= | -1009.200 |
| IM= 7 | V1X= | 0.0 | V:2X= | 193.200 | VVV= | 0.0 | VSUV= | 1009.200 | VDGMT= | -1202.400 |
| IM= 8 | V1X= | 0.0 | V:2X= | 290.100 | VVV= | 0.0 | vsuv= | 1009.200 | VDCMI'= | -1299.300 |
| III= 9 | V1X= | 0.0 | V2X= | 290.100 | vvv= | 0.0 | vsuv= | 1009.200 | VDGMI'= | -1299.300 |
| IN= 10 | V1X= | 0.0 | V2X= | 290.109 | VVV= | 0.0 | VSUV= | 1009.200 | VDGrff= | -1299.300 |
| IM= 11 | V1X= | 392.000 | V2X= | 682.100 | VVV= | 0.0 | VSUV= | 1009.200 | VDCrff= | -1923.340 |
| IM= 12 | V1X= | 0.0 | V2X= | 290.100 | vvv= | 0.0 | VSUV= | 1009.200 | VDGMT= | -1299.300 |

VS1T= VS2T= $0.0 \\ 0.0$

| I M= | 1 | VPRO= | 1372.89475 | VTRAN= | 0.0 |
|-------|----|-------|------------|----------|-----------|
| I 11= | 2 | VPR0= | 1201.47969 | VTRAN= | 0.0 |
| I M= | 3 | VPR0= | 1179.69480 | VTRAN= | 0.0 |
| 114= | -1 | VPR0= | 170,49485 | VTRAN= | 0.0 |
| I M= | 5 | VPRO= | 179.98201 | V'IIIAN= | 0.0 |
| 1 M= | 6 | VPR0= | 1179.69480 | VIRAN= | 0.0 |
| I M= | 7 | VPRO= | 1372.89475 | VTRAN= | 0.0 |
| IH= | 8 | VPn0= | 1469.79475 | VTRAN= | 0.0 |
| =11 I | 9 | VPRO= | 1469.79475 | VTIMI= | 0.0 |
| 111= | 10 | VPR0= | 1469.79475 | VTRAN= | 0.0 |
| 1 M= | 11 | VPRO= | 1861.79473 | VTRAN= | 232.03970 |

OUTPUT OF SUBROUTINE DIST

| PEAK MONTH= | 9 PE | ak load= | 83438.94 | · | |
|--------------|----------|----------|----------|-------------------------|------------|
| TRANSMISSION | MARCINAL | L COST= | 0.0 | NEW TRANSMISSION PLANT= | 0.0 |
| DISTRIBUTION | MARCINAL | L COST= | 1954.964 | NEW DISTRIBUTION PLANT= | 383713792. |

OUTPUT OF SUBROUTINE REVREQ

REPPIS= 22378524.476 TUTPIS= 1138043932.476 DEPEXP= 33447110.956 TAPD= 252293744.067 RETPIS= 865750016.

| OOPRE | .V= | 2603 | 844.472 |
|-------|-------|-------|----------|
| ONUIN | IC= | 3385 | 689.777 |
| OMC2= | 125 | 82783 | 11.450 |
| ACOPE | :X= 8 | 27858 | 1056.409 |
| PRPTA | .X≃ | 18600 | 751.735 |
| PAYTA | X= | 3774 | 835.094 |
| INVTX | IC= | 52078 | 528.0 |
| | | | |
| X0= | 3995 | 2327. | 852 |
| X1= | 10682 | 1438. | 752 |

| X2= | 65894598.167 |
|------|---------------|
| ХЗ≖ | 75790442.516 |
| X4= | 850233643.237 |
| X3 = | 966071617.156 |
| X6 = | 5989525.249 |

NEWPIS= 498326784. DCT= 600653.310 OMC1= 668583114.003 X= 960082091.908 PAVG= 1598.396

OUTPUT OF SUBROUTINE EVAL2

CAS CONSUMPTION EVALUATION CRITERIA

PEAK MONTH= 9 PEAK LOAD (MMCF)= 83438.94 LOAD FACTOR= 0.5999 TOTAL CAS CONSUMPTION= 600653.31

THEORETICAL EQUILIBRIUM VOLUMETRIC RATES= 1598.396 EQUILIBRIUM GAS SALES REVENUE REQUIREMENT= 960082091.91

EFFICIENCY CRITERIA

MONTH= RESIDENTIAL SURPLUS= 114563728. COMMERCIAL SURPLUS= 45838104. INDUSTRIAL SURPLUS= 89648465. 1 RESIDENTIAL SURPLUS= RESIDENTIAL SURPLUS= COMMERCIAL SURPLUS= COMMERCIAL SURPLUS= 82754751. 77489422. 76426623. MONTH= 2 68350528. 28243385. INDUSTRIAL SURPLUS= MONTH= 3 32993153. 14781796. INDUSTRIAL SURPLUS= INDUSTRIAL SURPLUS= MONTH= 4 RESIDENTIAL SURPLUS= 25928831. COMMERCIAL SURPLUS= 12092202. INDUSTRIAL SURPLUS= INDUSTRIAL SURPLUS= INDUSTRIAL SURPLUS= RESIDENTIAL SURPLUS= COMMERCIAL SURPLUS= COMMERCIAL SURPLUS= MONTH= 5 27713679. 12801953. 77680037. 19548165. 79347913. 83941308. MONTH= 45512203. 6 MONTH= 7 RESIDENTIAL SURPLUS= 90308912. COMMERCIAL SURPLUS= 36600861. RESIDENTIAL SURPLUS= RESIDENTIAL SURPLUS= RESIDENTIAL SURPLUS= MONTH= 8 149103698. COMMERCIAL SURPLUS= 58954727. INDUSTRIAL SURPLUS= 93697754. 211230708. COMMERCIAL SURPLUS= COMMERCIAL SURPLUS= 82596955. INDUSTRIAL SURPLUS= INDUSTRIAL SURPLUS= 102394599. MONTH= 9 59775534. MONTH= 10 132375035. 51623960. INDUSTRIAL SURPLUS= 101933343. MONTH= 11 **RESIDENTIAL SURPLUS=** 206797867. COMMERCIAL SURPLUS= 80909243. MONTH= 12 RESIDENTIAL SURPLUS= 179856441. COMMERCIAL SURPLUS= 70649568. INDUSTRIAL SURPLUS= 97913543.

ACTUAL GAS SALES REVENUES= 1030271552.99

| CAR | SALES | REVENUE | SURPLUS | (+) | OR | DEFICIT | (-)= | 76189461.0 | 68 |
|-----|-------|---------|---------|-----|----|---------|------|------------|----|
|-----|-------|---------|---------|-----|----|---------|------|------------|----|

| TOTAL | RESIDENTIAL SURPLUS | 1284728783. |
|-------|---------------------|-------------|
| TOTAL | COMMERCIAL SURPLUS | 514640920. |
| TOTAL | INDUSTRIAL SURPLUS | 1025194291. |
| TOTAL | CONSUMER SURPLUS | 2824563995. |

PRODUCER SURPLUS 143151504.

TOTAL SURPLUS 2967715498.



APPENDIX G

COMMENTS OF FIRST DRAFT REVIEWERS

This appendix contains the comments of reviewers of an early draft of this report: Walter J. Cavagnaro, California Public Utilities Commission; Stephen P. Reynolds and other staff, Pacific Gas and Electric Company; and John R. Yurtchuk, National Fuel Gas Distribution Corporation.



ADDRESS ALL COMMUNICATIONS TO THE COMMISSION CALIFORNIA STATE BUILDING SAN FRANCISCO, CALIFORNIA 94102 TELEPHONE: (415) 557-0507

Public Atilities Commission

STATE OF CALIFORNIA

January 7, 1981

FILE NO.

Dr. Jean-Michel Guldman Senior Faculty Associate The National Regulatory Research Institute The Ohio State University 2130 Neil Avenue Columbus, Ohio 43210

Dear Dr. Guldman

Thank you for the opportunity to comment on the draft of your Gas Capacity Cost Study. NRRI is to be commended for initiating studies in this area and it is hoped that such studies will continue. I would like to stress that California's interest mainly focuses on Gas Supply Cost including storage and transmission facilities. In California, we have experienced rapidly escalating Canadian gas prices which together with the phased deregulation of domestic gas is presenting us with marginal supply cost substantially in excess of average cost. I am enclosing for your information, a copy of a paper presented by Irwin M. Stelzer at a seminar on August 6, 1980. His view on the marginal cost of gas (Page 6) is quite interesting.

Through my association with other state commissions and NARUC, as well as our experience in California, I feel there is a need to develop a simplified marginal cost methodology and recommendations for reconciliation between marginal cost and the revenue requirement in meeting the PURPA goals of conservation, efficiency and equity. I hope that NRRI will provide the states with such a report as soon as possible. I would also encourage you to develop the link between your model and the utilities resource planning models. I am sure that PG&E will continue to cooperate with you in your further studies.

Very truly yours

Walter Cavaynaro

Walter J. Cavagnaro Energy Policy Staff Policy and Planning Division

WJC:asa cc Steve Reynolds, PG&E att PACIFIC GAS AND ELECTRIC COMPANY

IP G and E

77 BEALE STREET • SAN FRANCISCO, CALIFORNIA 94106 • (415) 781-4211 • TWX 910-372-6587

S. P. REYNOLDS MANAGER RATE DEPARTMENT

December 19, 1980

Dr. J. M. Guldmann Senior Faculty Associate The National Regulatory Research Institute The Ohio State University 2130 Neil Avenue Columbus, Ohio 43210

Dear Dr. Guldmann:

Thank you for the opportunity to comment on Chapter 3 of the final draft of your gas capacity cost study. Although Pacific Gas and Electric Company is not generally supportive of either an econometric or a historical approach to estimating marginal costs, we read your study with interest. The draft has been circulated within PGandE, and many of our staff have had a chance to review it. Attached please find a summary of their comments. Should you require additional information, please do not hesitate to contact either Mr. T. C. Long (Ext. 4743) or Ms. L. G. Baldwin (Ext. 2998).

You may also have our approval to release the study to Mr. Walter Cavagnero of the California Public Utilities Commission. We might suggest, however, that you send him a copy of PGandE's comments along with the report.

Sincerely,

Steph P. Reynolds

Attachment

PGandE Comments on Chapter 3 -Gas Capacity Cost Study

PGandE agrees with you that research in the area of gas distribution costs has been limited, and, thus, applauds the objectives of your study. You cite several weaknesses with the analysis performed by previous researchers (the Real Estate Research Corporation): designation of prototype neighborhoods too general to be of much use; failure to reflect costs due to differences in terrain, topography, and climate; no investigation of costs for commercial and industrial customers; and, neglect of the situations that may cause different types of investment, such as reinforcement, pressurizing, or extension. Your approach makes some good progress towards addressing these shortcomings in its recognition of the importance of localized conditions in evaluating gas distribution costs.

PGandE would, however, like to offer comments on your study along two veins. The first section of our comments deals with the conceptual economic basis for evaluating marginal costs. This is followed by a discussion of more specific topics: the econometric model specification, the data supporting the analysis, the interpretation of results, and areas for further work.

I. Conceptual Basis for Marginal Costing

Your stated objective (p. 15) is to perform an econometric analysis of distribution plant costs, and to use the resulting distribution plant cost functions to predict future costs and marginal costs. Your use of crosssectional regression analysis and of historic accounting cost data to support that analysis, however, make us skeptical that your model has the capability of predicting future costs (if what you mean is next year's costs as opposed to the costs of a 95th PGandE community). Verification of the predictive ability of your model would be a desirable addition to the analysis. We also have reservations about the applicability of your model to gas marginal costing. It would be helpful to the reader for you to define what you mean by marginal costs early on in Chapter 3.

PGandE defines the marginal cost of gas service as "the change in the total cost of supplying gas as a result of a change in the quantity supplied." Gas service involves the process of hooking up customers, acquiring gas supplies, and then providing a gas system that delivers the supplies to the customers. Accordingly, we view the marginal cost of gas service as having three components: a marginal customer cost; a marginal commodity cost; and a marginal capacity cost. The marginal customer cost is associated with providing service to an additional customer. The marginal commodity cost is the variable cost of providing the last unit of gas supplied. The marginal capacity cost is related to constructing and maintaining a system with sufficient capacity to meet the last unit of peak day gas demand. Therefore, your designation of marginal distribution costs overlaps with two of our identified marginal cost components - the marginal customer cost and the distribution portion of the marginal capacity cost - one of which varies with a change in customers, and the other with a change in demand. Your specifications, however, attempt to explain all distribution

investment with either customers or sales and do not attempt to break down the cost of distribution plant investment by cost causation.

PGandE is suspicious of the use of historic accounting data to calculate marginal costs. Marginal costs, by their nature, are prospective, forward-looking costs, not historic costs. Only those costs which result from investing in resources to supply and deliver additional increments of gas or to hook-up additional customers, should be counted as marginal costs. Sunk costs, or costs which are presently on the books, are not considered costs in an economic sense. Thus, the use of historic accounting costs (which in PGandE's case include investments made back as early as 1910), coupled with the limitations of the data described below and your assumptions concerning vintaging, make us skeptical that your approach will produce marginal costs that are grounded in economic theory.

II. Discussion of Specific Topics

The Appropriate Use of Cross-Section Regression Analysis

In principal, cross-section regression is appropriate for analysis of long-run cost determinants but is not suitable for analysis of short-run adjustment to changes in cost determinants. This distinction follows from recognition that the information isolated by cross-section analysis is consistent only with the economic concept of the long-run, i.e. the period of time in which all factors are fully variable.

Ideally, cross-section data represent a wide and independent variation of the factors that determine the cost of distribution capacity. Moreover, each cross-section is held to be in full adjustment to the local determinants of cost. Regression analysis on cross-section data therefore focuses on the relationship that independently varying cost determinants have to total plant cost under conditions of full adjustment, in particular, adjustment to long-run equilibrium.

Data Limitations

You state on page 20 of your study that "most gas distribution utilities keep track of their capital investments at the community level." As you know, PGandE does not maintain statistics on distribution plant for individual communities. Therefore, as we agreed, the PGandE data on distribution cost for service by communities was developed as the product of the miles of distribution gas mains in each community and the <u>system</u> historical unit cost per mile of distribution main.

This approach to allocation of <u>system</u> historical costs between individual communities obscures many of the factors that cause real variation of plant costs between communities. For example, the local differences in distribution costs due to differences in technology, pre-existing land uses, and local terrain can not be discerned. Furthermore, it may be that the mileage of gas mains is not a suitable basis for allocation of total system costs between communities with varying proportions of residential, commercial and industrial customers. PGandE acknowledges that your model makes some intuitive sense: the cost of distribution capacity is a function of weather, density, and composition of customer population. However, because of data limitations, the interpretation of the model results are less clear. For example, since the unit cost per mile of distribution main does not vary by community, isn't the dependent variable really the miles of gas main per community?

An effect of the method by which historical data was generated for community level distribution plant is to impose the assumption that the vintage composition of local plant is constant across all communities. PGandE <u>does</u> <u>not</u> verify the assumption of similar vintages of plant across communities as you claim (page 22). Indeed, we are certain that the vintage of distribution plant varies substantially across PGandE service communities.

As noted (page 21) one obvious problem with your approach is "related to the use of the original cost balance for measuring the value of plant in service, instead of its replacement costs, which should be the correct reference for measuring total and marginal costs." The fact that the vintage of distribution plant varies substantially between PGandE communities compounds this problem. As a result, the estimated model cannot be used to predict historical plant costs for any given community.

Dynamic Analysis

PGandE has two comments concerning your dynamic analysis of distribution costs. First, analysis of the change in capacity cost should be adjusted by the initial conditions of local capacity utilization: for instance, is the current situation one of overcapacity or undercapacity? Second, the historical cost method of plant accounting may misrepresent the cost for addition of incremental capacity.

First, the dynamic analysis must be qualified with respect to the level of PGandE's capacity utilization in 1979. Under normal conditions investments in transmission and major distribution facilities have lead times and life times longer than one year, so that the planner typically prebuilds for anticipated growth. However, the significant dislocations in the energy market over the last decade have caused the system to diverge from normal levels of capacity utilization because of a substantially lower average use per customer than estimated earlier for planning purposes. Because of this reason, the use of a single period analysis under the recent conditions of excess capacity may have yielded costs that are lower than will be required on average in the future.

Second, the dynamic analysis focuses on the change in the distribution plant cost for one year and relates this change in cost to the change in customers. A problem exists with this approach in that the procedures of historical cost accounting may introduce an upward bias on the cost of additional plant. Load growth may be by upgrading an existing pipeline with a larger diameter pipeline. In the instance of pipeline replacement the book investment in the larger pipeline is based on current cost. Meanwhile, the smaller pipeline that is replaced, is retired from plant based on its historical cost. As a result, historical cost accounting will tend to overstate the year to year increase in plant.

Interpretation of Regression Estimates

PGandE has several comments that relate to the general results of the regression analysis:

<u>Plausibility of Cost Estimates</u>: PGandE finds your "short-run" residential marginal cost to be low, while the "long-run" residential marginal cost are difficult to judge. The Gas System Planning Department provided the following estimate of the total cost for a typical sub-division customer:

| Service: | \$340.00 | | |
|-----------------------|----------|----------------|--|
| Meter & regulator: | 65.00 | | |
| Total (without main): | 405.00 | (1980 dollars) | |
| Local main: | 182.28 | | |
| Total (with main): | 587.28 | (1980 dollars) | |

This cost estimate includes an allowance of \$182.28 for a local distribution main. This cost added to the "customer costs" would be \$587.28 which is higher than your derived costs of \$359.357 per residential customer. Your estimate of "long-run" residential cost of \$326.75 on page 54 would need to be translated to current cost. However, the replacement cost multiple of 2.79 (which, if applied to \$326.75 would yield \$911.64), is not applicable to residential cost estimates because the multiple pertains only to the historical system technology and customer composition. Consequently the estimate of long-run residential costs is difficult to put into perspective.

The Specification of Heating-Degree-Day's Variable: PGandE designs the distribution system to have the capacity necessary to meet demand on an abnormal peak day. Therefore, the finding that the peak-month heating-degree-day measure (DDM) was clearly more significant than the annual heating degree-day measure (DDT) is consistent with planning criteria for capacity of the gas distribution system. However, since peak day is the critical influence, an even better explanatory variable would be peak day demand.

The Test for a Separable Cost Structure: By use of the regression analysis the hypothesis that the distribution system is characterized by joint, non-separable costs is tested. This hypothesis is tested by determining whether the additive (separable and linear) or multiplicative (non-separable and log-linear) form of the regression estimator achieves a significantly better fit. The results indicate that distribution costs are not separable between customer classes. PGandE finds the result that distribution costs are non-separable plausible.

The results for residential and commercial/industrial customers further indicate that total distribution costs exhibit economies of scale. This seems realistic. However, the implied marginal cost function for residential customers indicates that costs for additional residential customers are positively influenced by the presence of commercial/industrial customers, which seems less realistic. Historically, major extensions of the gas system were more desirable when there was a simultaneous hook-up of residential and commercial/industrial customers. PGandE has typically made an analysis in order to certify that the costs of the prospective additions to gas distribution plant would be recovered from commercial/industrial customers. Thus it seems the residential sector benefited from an externality (technological and pecuniary) related to the presence of commercial and industrial customers.

<u>Problems In Interpretation</u>: You note that the ratio between replacement and historical costs for PGandE's gas distribution plant is 2.79 and proceed to use this value at various points. PGandE thinks that the meaning of the system ratio of 2.79 must be clarified because at numerous points certain inferences are based on questionable application of this ratio.

The ratio of replacement to historical costs represents a system average and is specific to the historical technology and equipment composition of system-wide distribution facilities. One problem with your analysis occurs on page 48, line 9. Obviously, one would expect the vintage of plant being retired to differ significantly from the system average vintage. Therefore the estimate for "truly new distribution plant" should be substantially less than \$59,944,556, because the factor 2.79 is too low to be appropriate for retired capital.

Another instance of questionable inference occurs on page 58, line 3. At this point you have estimated a 'dynamic' cost of \$466.46 per customer. It must be noted that this value is supposed to reflect the costs for plant added in 1979 and implies use of current technology. You then compare this 1979 investment against the 'static' estimate of historical distribution costs, adjusted by the 2.79 ratio of replacement cost to historical cost. The problem is that the numbers are not comparable. Technology is certain to have changed, so that the difference in your estimates could be due to technological change or other influences, and not specifically to the disparity of short-run and long-run costs.

<u>Areas For Further Work</u>: It would be instructive for you to more carefully align your definitions of short-run and long-run costs with economic theory. The economic definition of long-run is the period over which all factors of production are variable, while the short-run simply refers to any lesser time. Your distribution costs seem to be comprised of two parts: (1) the cost of hooking up new customers, such as the cost of meter, regulator and service; and (2) the distribution costs incurred to serve additional volumes of gas demand. It appears that you implicitly designate a one year period as the short-run, categorize the former costs as short-run costs, and use the dynamic model to estimate them. The latter remaining costs therefore, fall into the long-run category and are (in addition to the former) analyzed with your static model. This would be a very neat approach to estimating the two components in which PGandE is interested - marginal customer costs and the distribution portion of marginal capacity costs - if your construct can be verified.

If as is suggested (page 15) the results are to be the basis for projection of future costs, there is a problem of translating the model's predictions based on historical cost into values relevant today or in the future. PGandE suggests that if the model is to be practically useful for cost forecasting, further research must address the translation of historical costs into replacement costs with allowance for technological change. Also, to the extent that costs of additional plant vary between locations within the PGandE gas distribution network, use of the present model for forecasting may misrepresent additional costs. For example, communities may differ by terrain, state of development, and the type of investment required to meet growth.

PGandE requests that any future work be accompanied by more information related to sample design, correlation analysis of independent variables, and error analysis. Furthermore, experimentation with plausible alternative formulations of the model and introduction of other explanatory variables such as terrain, zoning, income, etc. would be interesting.

Corrections to Tables

PGandE would like to bring to your attention several figures that need to be changed in Tables 3.11 and 3.12. The suggested corrections are shown on the attached tables. Also, the total gas plant in service that you quote in Table 3.12 is exclusive of production and intangible plant; this should be noted.²¹

 21 Author's note: These corrections have been made.



January 7, 1981

Dr. J. M. Guldman The National Regulatory Research Institute The Ohio State University 2130 Neil Avenue Columbus, Ohio 43210

Dear Dr. Guldman:

I have enclosed my comments regarding your marginal cost study. Your direct approach in estimating total cost functions is theoretically appealing, yet I feel requires some fine-tuning with respect to its econometrics. Since you plan to address these problems, that would certainly alleviate any of the concern I might have in utilizing the results.

I look forward to receiving any further work you may undertake on this project, and should you require any additional information, please feel free to contact me.

Sincerely,

John R. Yurtchuk Economist

JRY: ms Enc. Comments of John R. Yurtchuk, Economist - National Fuel Gas Distribution

The study attempts to identify various total cost functions in aggregate and disaggregate form so that estimates of marginal costs and scale economics can be made. The three types of independent variables used were: number of customers, MCF sales, and a density variable. The performance of the estimating equations utilizing the stated regressors either jointly or separately generally superior when a multiplicative-type function was used.

The methodology exhibited in this study rests upon sound microeconomic principles and offers a tractable approach to identifying certain characteristics of a utility's operations. However, in reviewing the empirical component it appears that certain econometric difficulties may exist.

The first problem lies in the values for the coefficients of determination. Certainly some of the R-squares are "acceptable" in that a significant portion of the variation in the dependent variable is being explained. However, there do exist a number of equations whose R-square value is simply too low, indicating that the regression equation lacks significant explanatory power. Specifically, the following equations have R-square values below .50 with some even less than .40: 3.91, 3.92, 3.93, 3.99, 3.100, 3.101, 3.105, 3.106, 3.113, 3.114, 3.129, 3.130, 3.131, 3.137, 3.138, and 3.139. Statistical theory would suggest that something is missing from these relations. Since results from these equations are discussed and economically interpreted, it is assumed that the R-square levels are regarded as acceptable in this stage of the study.

In the following equation, TMCF and CIPMCF are the two included independent variables.

(Equation 3.82) PS = $9.5978 \times \text{TMCF}^{0.8028} \times \text{CIPMCF}^{0.0965}$

where PS - Total Distribution Plant

These two right hand side variables are linearly related to one another in the following way:

TMCF = RMCF + CIPMCF

where TMCF - Total Gas Sales (MCF) RMCF - Residential Gas Sales (MCF) CIRMCF - Total Non-Residential Gas Sales (MCF) Clearly, since CIPMCF is a component of TMCF, their respective effects on the dependent variable, PS, are inseparable. Thus the problem of multicollinearity with a consequencial loss of precision becomes a distinct possibility.²²

Lastly, it appears highly probable that simultaneous equation bias exists in many of the regressions in light of the accounting relationships that are present among the independent variables. Further work is thus called for to account for these identities.

²²Author's note: Thanks are due to Mr. Yurtchuk for pointing out a typographical error in equation 3.82 in the first draft (TMCF must be replaced by RMCF). This error has been corrected.